UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

(Mark One)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2020

OR

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission file number: 001-35878



(Exact name of registrant as specified in its charter)

Grand Duchy of Luxembourg (State or Other Jurisdiction of Incorporation or Organization) 4. rue Albert Borschette L-1246

4, rue Albert Borschette L-1246 (Address of principal executive offices, including zip code) Luxembourg

98-1009418 (I.R.S. Employer Identification No.)

+352 27 84 1600 (Registrant's telephone number, including area code)

Securities registered pursuant to Section 12(b) of the Securities Exchange Act of 1934: None

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes 🗵 No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer	\boxtimes	Accelerated filer	
Non-accelerated filer		Smaller reporting company	
		Emerging growth company	

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes 🗆 No 🗵

As of November 3, 2020, 142,145,054 common shares of the registrant were outstanding, with a nominal value of \$0.01 per share.

TABLE OF CONTENTS

PART I. FINANCIAL INFORMATION

Item 1.	Financial Statements:	
	Condensed Consolidated Balance Sheets as of December 31, 2019 and September 30, 2020 (Unaudited)	<u>6</u>
	Unaudited Condensed Consolidated Statements of Operations for the Three and Nine Months Ended September 30, 2019	
	and 2020	<u>7</u>
	Unaudited Condensed Consolidated Statements of Comprehensive Loss for the Three and Nine Months Ended September 30, 2019 and 2020	<u>8</u>
	Unaudited Condensed Consolidated Statements of Changes in Shareholders' Deficit for the Three and Nine Months Ended September 30, 2019 and 2020	<u>9</u>
	Unaudited Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2019 and 2020	<u>11</u>
	Notes to the Condensed Consolidated Financial Statements (Unaudited)	<u>13</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>39</u>
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	<u>53</u>
Item 4.	Controls and Procedures	<u>53</u>
PART II.	OTHER INFORMATION	
Item 1.	Legal Proceedings	<u>55</u>
Item 1A.	Risk Factors	<u>55</u>
Item 6.	Exhibits	<u>56</u>
<u>SIGNATI</u>	URES	<u>57</u>

INTRODUCTION

In this Quarterly Report on Form 10-Q, or Quarterly Report, unless otherwise indicated or the context otherwise requires, (1) the terms "we," "us," "our," "the Company" and "Intelsat" refer to Intelsat S.A., and its subsidiaries on a consolidated basis, (2) the term "Intelsat Holdings" refers to our indirect wholly-owned subsidiary, Intelsat Holdings S.A., (3) the term "Intelsat Investments" refers to Intelsat Investments S.A., Intelsat Holdings' direct wholly-owned subsidiary, (4) the term "Intelsat Luxembourg" refers to Intelsat (Luxembourg) S.A., Intelsat Investments' direct wholly-owned subsidiary, (5) the term "Intelsat Envision" refers to Intelsat Envision Holdings LLC, Intelsat Luxembourg's direct wholly-owned subsidiary, (6) the terms "Intelsat Connect" and "ICF" refer to Intelsat Connect Finance S.A., Intelsat Envision's direct wholly-owned subsidiary, and (7) the term "Intelsat Jackson" refers to Intelsat Jackson Holdings S.A., Intelsat Connect's direct wholly-owned subsidiary. In this Quarterly Report, unless the context otherwise requires, all references to transponder capacity or demand in the C-band and Ku-band frequencies only.

FINANCIAL AND OTHER INFORMATION

Unless otherwise indicated, all references to "dollars" and "\$" in this Quarterly Report are to, and all monetary amounts in this Quarterly Report are presented in, U.S. dollars. Unless otherwise indicated, the financial information contained in this Quarterly Report has been prepared in accordance with United States generally accepted accounting principles ("U.S. GAAP").

Certain monetary amounts, percentages and other figures included in this Quarterly Report have been subject to rounding adjustments. Accordingly, figures shown as totals in certain tables may not be the arithmetic aggregation of the figures that precede them, and figures expressed as percentages in the text may not total 100% or, as applicable, when aggregated may not be the arithmetic aggregation of the percentages that precede them.

In this Quarterly Report, we refer to and rely on publicly available information regarding our industry and our competitors. Although we believe the information is reliable, we cannot guarantee the accuracy and completeness of the information and have not independently verified it.

FORWARD-LOOKING STATEMENTS

Some of the statements in this Quarterly Report and oral statements made from time to time by our representatives constitute forward-looking statements that do not directly or exclusively relate to historical facts. The Private Securities Litigation Reform Act of 1995 provides a "safe harbor" for certain forward-looking statements as long as they are identified as forward-looking and are accompanied by meaningful cautionary statements identifying important factors that could cause actual results to differ materially from the expectations expressed or implied in the forward-looking statements.

When used in this Quarterly Report, the words "may," "will," " might," "should," "expect," "plan," "anticipate," "project," "believe," "estimate," "predict," "intend," "potential," "outlook" and "continue," and the negative of these terms, and other similar expressions are intended to identify forwardlooking statements and information. Examples of these forward-looking statements include, but are not limited to, statements regarding the following: our belief that the growing worldwide demand for reliable broadband connectivity everywhere at all times, together with our leadership position in our attractive sector, global scale, efficient operating and financial profile, diversified customer sets and sizeable contracted backlog, provide us with a platform for long-term success; our ability to confirm and consummate a plan of reorganization in the Chapter 11 Cases (as defined below); the effects of the Chapter 11 Cases on our liquidity or results of operations or business prospects; other risks related to the Chapter 11 Cases as described further below; our belief that the new and differentiated capacity of our next generation Intelsat Epic satellites will provide inventory to help offset recent trends of pricing pressure, new capacity from other satellite operators, and improved access to fiber links in our network services business; our outlook that the increased volume of services provided by our Intelsat Epic fleet is expected to stabilize the level of business activity in the network services sector; our expectation that over time incremental demand for capacity to support the new 4K format, also known as ultra-high definition, could offset some of the reductions in demand related to use of new compression technologies in our media business; our expectation that our new services and technologies will open new sectors that are much larger and faster growing than those we support today; our belief that selectively investing, employing a disciplined yield management approach, and emphasizing the development of strong distribution channels for our four primary customer sets will drive stability in our core business; our expectation that developing and scaling our differentiated managed service offerings in targeted verticals, leveraging the global footprint, higher performance and better economics of our Intelsat Epic fleet, in addition to the flexibility of our innovative terrestrial network, will drive revenue growth; our belief that completing targeted investments and partnerships in differentiated space and ground infrastructure will provide a seamless interface with the broader telecommunications ecosystem; our ability to incorporate new technologies into our network that could change the types of applications we can serve and increase our share of the global demand for broadband connectivity; our projection that our government business will benefit from the increasing demands for mobility services from the U.S. government for aeronautical and ground mobile

requirements; our intention to maximize the value of our spectrum rights; our expectations as to our ability to comply with the final U.S. Federal Communications Commission ("FCC") order regarding clearing C-band spectrum in North America, including the availability of adequate resources and funds required to comply and the receipt of accelerated clearing payments set forth in the FCC order; our belief that developing differentiated services and investing in related software- and standards-based technology will allow us to unlock opportunities that are essential to providing global broadband connectivity; the trends that we believe will impact our revenue and operating expenses in the future; our assessments regarding how long satellites that have experienced anomalies in the past should be able to provide service on their transponders; our belief as to the likelihood of the cause of the failure of Intelsat 29e occurring on our other satellites; our assessment of the risks of future anomalies occurring on our satellites; our plans for satellite launches in the near-term; our expected capital expenditures in 2020 and during the next several years; our belief that the diversity of our revenue allows us to benefit from changing market conditions and lowers our risk from revenue fluctuations in our service applications and geographic regions; our belief that the scale of our fleet can reduce the financial impact of any satellite anomalies or launch failures and protect against service interruptions; and the impact on our financial position or results of operations of pending legal proceedings.

Forward-looking statements reflect our intentions, plans, expectations, anticipations, projections, estimations, predictions, outlook, assumptions and beliefs about future events. These forward-looking statements speak only as of their dates and are not guarantees of future performance or results and are subject to risks, uncertainties and other factors, many of which are outside of our control. These factors could cause actual results or developments to differ materially from the expectations expressed or implied in the forward-looking statements and include known and unknown risks. Known risks include, among others, the risks discussed in Part I—Item 1A—Risk Factors in our Annual Report on Form 10-K for the year ended December 31, 2019 (the "2019 Annual Report"), as supplemented by the risks discussed in Part II—Item 1A—Risk Factors in our Quarterly Report on Form 10-Q for the quarter ended March 31, 2020, and Part II—Item 1A—Risk Factors in our Quarterly Report on Form 10-Q for the quarter ended March 31, 2020, the political, economic, regulatory and legal conditions in the markets we are targeting for communications services or in which we operate and other risks and uncertainties inherent in the telecommunications business in general and the satellite communications business in particular.

Other factors that may cause results or developments to differ materially from historical results or developments or the forward-looking statements made in this Quarterly Report include, but are not limited to:

- risks associated with operating our in-orbit satellites;
- satellite launch failures, satellite launch and construction delays and in-orbit failures or reduced satellite performance;
- potential changes in the number of companies offering commercial satellite launch services and the number of commercial satellite launch opportunities available in any given time period that could impact our ability to timely schedule future launches and the prices we pay for such launches;
- our ability to obtain new satellite insurance policies with financially viable insurance carriers on commercially reasonable terms or at all, as well as the ability of our insurance carriers to fulfill their obligations;
- · possible future losses on satellites that are not adequately covered by insurance;
- U.S. and other government regulation;
- · changes in our contracted backlog or expected contracted backlog for future services;
- · pricing pressure and overcapacity in the markets in which we compete;
- our ability to access capital markets for debt or equity;
- the competitive environment in which we operate;
- customer defaults on their obligations to us;
- our international operations and other uncertainties associated with doing business internationally;
- the impact of the novel coronavirus ("COVID-19") pandemic on our business, the economic environment and our expected financial results;
- our ability to consummate the Company's transaction through which we expect to purchase the equity of Gogo Inc.'s ("Gogo") commercial aviation business (the "Gogo Transaction"), including without limitation the receipt of certain regulatory approvals and potential further approvals by the Bankruptcy Court (as defined below);
- our expectations as to the timing, benefits, and impact on our future financial performance associated with the Gogo Transaction;
- our ability to successfully integrate Gogo's commercial aviation business (including without limitation, the achievement of synergies and cost reductions);
- litigation; and
- other risks discussed in our 2019 Annual Report or this Quarterly Report.

Further, many of the risks and uncertainties that we face are currently amplified by, and will continue to be amplified by, the risks and uncertainties regarding the Company and certain of its subsidiaries' voluntary commencement of cases under Chapter 11 (the "Chapter 11 Cases") of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the Eastern District of Virginia (the "Bankruptcy Court"), including but not limited to:

- our ability to improve our liquidity and long-term capital structure and to address our debt service obligations through the restructuring;
- our ability to obtain timely approval by the Bankruptcy Court with respect to the motions that we have filed or will file in the Chapter 11 Cases;
 objections to the Company's restructuring process or other pleadings filed that could protract the Chapter 11 Cases or interfere with the
- objections to the Company's restructuring process or other pleadings filed that could protract the Chapter 11 Cases or interfere with the Company's ability to consummate the restructuring;
- the length of time that the Company will operate under Chapter 11 protection and the continued availability of operating capital during the pendency of the Chapter 11 Cases;
- our substantial level of indebtedness and related debt service obligations and restrictions, including those expected to be imposed by covenants in any exit financing, that may limit our operational and financial flexibility;
- the conditions to which our debtor-in-possession (DIP) financing is subject and the risk that these conditions may not be satisfied for various reasons, including for reasons outside of our control;
- our ability to develop and execute our business plan during the pendency of the Chapter 11 Cases;
- increased administrative and legal costs related to the Chapter 11 process;
- potential delays in the Chapter 11 process due to the effects of the COVID-19 pandemic; and
- our ability to continue as a going concern and our ability to maintain relationships with regulators, suppliers, customers, employees and other third parties as a result of such going concern, during the restructuring and the pendency of the Chapter 11 Cases.

Although we believe that the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee our future results, level of activity, performance or achievements. Because actual results could differ materially from our intentions, plans, expectations, anticipations, projections, estimations, predictions, outlook, assumptions and beliefs about the future, you are urged not to rely on forward-looking statements in this Quarterly Report and to view all forward-looking statements made in this Quarterly Report with caution. We do not undertake any obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise.

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

INTELSAT S.A. (DEBTOR-IN-POSSESSION)

CONDENSED CONSOLIDATED BALANCE SHEETS

(in thousands, except per share amounts)

	Dee	cember 31, 2019	September 30, 2020 (unaudited)			
ASSETS				(
Current assets:						
Cash and cash equivalents	\$	810,626	\$	1,001,000		
Restricted cash		20,238		19,350		
Receivables, net of allowances of \$40,028 in 2019 and \$29,838 in 2020		255,722		215,725		
Contract assets		47,721		40,479		
Prepaid expenses and other current assets		39,230		104,838		
Total current assets		1,173,537		1,381,392		
Satellites and other property and equipment, net		4,702,063		4,696,123		
Goodwill		2,620,627		2,620,627		
Non-amortizable intangible assets		2,452,900		2,440,700		
Amortizable intangible assets, net		276,752		253,425		
Contract assets, net of current portion		74,109		57,715		
Other assets		504,394		564,434		
Total assets	\$	11,804,382	\$	12,014,416		
LIABILITIES AND SHAREHOLDERS' DEFICIT						
Current liabilities:						
Accounts payable and accrued liabilities	\$	88,107	\$	172,271		
Taxes payable		6,402		8,223		
Employee related liabilities		44,648		35,371		
Accrued interest payable		308,657		13,837		
Current portion of long-term debt		,		5,400,953		
Contract liabilities		137,706		127,879		
Deferred satellite performance incentives		42,835		42,337		
Other current liabilities		62,446		58,416		
Total current liabilities		690,801		5,859,287		
Long-term debt		14,465,483				
Contract liabilities, net of current portion		1,113,450		1,060,732		
Deferred satellite performance incentives, net of current portion		175,837		146,559		
Deferred income taxes		55,171		63,922		
Accrued retirement benefits, net of current portion		125,511		113,258		
Other long-term liabilities		166,977		224,728		
Liabilities subject to compromise				10,170,344		
Shareholders' deficit:				, ,		
Common shares, nominal value \$0.01 per share		1,411		1,421		
Paid-in capital		2,565,696		2,572,324		
Accumulated deficit		(7,503,830)		(8,144,803)		
Accumulated other comprehensive loss		(63,135)		(61,191)		
Total Intelsat S.A. shareholders' deficit		(4,999,858)		(5,632,249)		
Noncontrolling interest		11,010		7,835		
Total liabilities and shareholders' deficit	\$	11,804,382	\$	12,014,416		
Town naonnies and shareholders deficit	φ	11,004,582	Ψ	12,014,410		

See accompanying notes to unaudited condensed consolidated financial statements.

UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (in thousands, except per share amounts)

(in thousands, except per share amo	ounts)
-------------------------------------	-------	---

	Three Months Ended September 30, 2019	Three Months Ended September 30, 2020	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Revenue	\$ 506,658	\$ 489,449	\$ 1,544,514	\$ 1,430,303
Operating expenses:				
Direct costs of revenue (excluding depreciation and amortization)	104,743	120,267	305,595	331,771
Selling, general and administrative	60,750	69,215	168,263	214,586
Depreciation and amortization	161,536	162,573	496,438	488,235
Satellite impairment loss	—	—	381,565	—
Impairment of non-amortizable intangible and other assets				46,243
Total operating expenses	327,029	352,055	1,351,861	1,080,835
Income from operations	179,629	137,394	192,653	349,468
Interest expense, net	315,964	138,075	953,246	678,937
Other income (expense), net	(5,115)	3,067	(32,374)	8,564
Reorganization items	—	(36,367)	—	(335,059)
Loss before income taxes	(141,450)	(33,981)	(792,967)	(655,964)
Provision for (benefit from) income taxes	6,248	(18,650)	3,884	(17,691)
Net loss	(147,698)	(15,331)	(796,851)	(638,273)
Net income attributable to noncontrolling interest	(594)	(600)	(1,785)	(1,784)
Net loss attributable to Intelsat S.A.	\$ (148,292)	\$ (15,931)	\$ (798,636)	\$ (640,057)
Net loss per common share attributable to Intelsat S.A.:				
Basic	\$ (1.05)	\$ (0.11)	\$ (5.70)	\$ (4.51)
Diluted	\$ (1.05)	\$ (0.11)	\$ (5.70)	\$ (4.51)

See accompanying notes to unaudited condensed consolidated financial statements.

UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE LOSS

(in thousands)

		Three Months Ended September 30, 2019	Three Months Ended September 30, 2020	Nine Months Ended September 30, 2019		Nine Months Ended September 30, 2020
Net loss	\$	(147,698)	\$ (15,331)	\$ (796,851)	S	\$ (638,273)
Other comprehensive income (loss), net of tax:						
Defined benefit retirement plans:						
Reclassification adjustment for amortization of unrecognized prior service credits, net of tax included in other income (expense), net	l	(625)	(626)	(1,877)		(1,878)
Reclassification adjustment for amortization of unrecognized actuarial loss, net of tax included in other income (expense), net		736	1,274	2,207		3,822
Adoption of ASU 2018-02 (see Note 13—Income Taxes)		_	_	(16,191)		_
Other comprehensive income (loss)		111	 648	(15,861)	_	1,944
Comprehensive loss		(147,587)	 (14,683)	(812,712)	_	(636,329)
Comprehensive income attributable to noncontrolling interest		(594)	(600)	(1,785)	_	(1,784)
Comprehensive loss attributable to Intelsat S.A.	\$	(148,181)	\$ (15,283)	\$ (814,497)	Ş	\$ (638,113)

See accompanying notes to unaudited condensed consolidated financial statements.

UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN SHAREHOLDERS' DEFICIT (in thousands, except where otherwise noted)

	Commo	n Sha	res						
	Number of Shares (in millions)	A	mount	Paid-in Capital	Accumulated Deficit	Accumulated Other Comprehensive Loss	tal Intelsat S.A. Shareholders' Deficit	ľ	Noncontrolling Interest
Balance at December 31, 2018	138.0	\$	1,380	\$ 2,551,471	\$ (6,606,426)	\$ (43,430)	\$ (4,097,005)	\$	14,396
Net income (loss)	_		_	—	(120,622)	—	(120,622)		580
Dividends paid to noncontrolling interests	_		_	_	_	_	_		(1,925)
Share-based compensation	2.6		26	2,913	_	_	2,939		
Postretirement/pension liability adjustment, net of tax	_		_	_	_	110	110		
Adoption of ASU 2018-02 (see Note 13—Income Taxes)					16,191	(16,191)			
Balance at March 31, 2019	140.6	\$	1,406	\$ 2,554,384	\$ (6,710,857)	\$ (59,511)	\$ (4,214,578)	\$	13,051
Net income (loss)	_		_		(529,722)		 (529,722)		610
Dividends paid to noncontrolling interests	_		_	_	_	_	_		(1,469)
Share-based compensation	0.1		1	3,586	_	—	3,587		_
Postretirement/pension liability adjustment, net of tax	_		_	_	_	109	109		_
Balance at June 30, 2019	140.7	\$	1,407	\$ 2,557,970	\$ (7,240,579)	\$ (59,402)	\$ (4,740,604)	\$	12,192
Net income (loss)				 _	 (148,292)		 (148,292)		594
Dividends paid to noncontrolling interests	_					_	_		(1,436)
Share-based compensation	0.3		3	4,676	—	—	4,679		—
Postretirement/pension liability adjustment, net of tax	_		_	_		111	111		_
Balance at September 30, 2019	141.0	\$	1,410	\$ 2,562,646	\$ (7,388,871)	\$ (59,291)	\$ (4,884,106)	\$	11,350

See accompanying notes to unaudited condensed consolidated financial statements.

UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN SHAREHOLDERS' DEFICIT (in thousands, except where otherwise noted)

	Commo	n Sh	ares						
	Number of Shares (in millions)		Amount	Paid-in Capital	Accumulated Deficit	ccumulated Other Comprehensive Loss	tal Intelsat S.A. Shareholders' Deficit		Noncontrolling Interest
Balance at December 31, 2019	141.1	\$	1,411	\$ 2,565,696	\$ (7,503,830)	\$ (63,135)	\$ (4,999,858)	\$	11,010
Net income (loss)	—		_	—	(218,771)	—	(218,771)		556
Dividends paid to noncontrolling interests	_		_	_	_	_	_		(1,879)
Share-based compensation	1.0		10	971			981		
Postretirement/pension liability adjustment, net of tax	_		_	_	_	648	648		_
Adoption of ASU 2016-13 (see Note 1—General)			_	—	(916)	_	(916)		
Balance at March 31, 2020	142.1	\$	1,421	\$ 2,566,667	\$ (7,723,517)	\$ (62,487)	\$ (5,217,916)	\$	9,687
Net income (loss)			_		(405,355)	 	 (405,355)		628
Share-based compensation				2,737			2,737		
Postretirement/pension liability adjustment, net of tax			_			648	648		
Balance at June 30, 2020	142.1	\$	1,421	\$ 2,569,404	\$ (8,128,872)	\$ (61,839)	\$ (5,619,886)	\$	10,315
Net income (loss)	_				 (15,931)		(15,931)	_	600
Dividends paid to noncontrolling interests	_		_	_	_	_	_		(3,080)
Share-based compensation	_			2,920			2,920		
Postretirement/pension liability adjustment, net of tax			_	_		648	648		_
Balance at September 30, 2020	142.1	\$	1,421	\$ 2,572,324	\$ (8,144,803)	\$ (61,191)	\$ (5,632,249)	\$	7,835

See accompanying notes to unaudited condensed consolidated financial statements.

UNAUDITED CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands)

		e Months Ended tember 30, 2019	Nine Months Ended September 30, 2020
Cash flows from operating activities:	¢	(70(951)	¢ ((20.272)
Net loss	\$	(796,851)	\$ (638,273
Adjustments to reconcile net loss to net cash provided by operating activities: Depreciation and amortization		406 429	488,235
Provision for doubtful accounts		496,438 13,429	36,360
Foreign currency transaction loss Loss on disposal of assets		5,119 171	7,330
Satellite impairment loss		381,565	
Impairment of non-amortizable intangible and other assets		581,505	46,243
Share-based compensation		10,215	9,399
Deferred income taxes		(1,029)	4,720
Amortization of discount, premium, issuance costs and related costs		31,011	19,689
Non-cash reorganization items		51,011	19,085
Debtor-in-possession financing fees			52,182
Amortization of actuarial loss and prior service credits for retirement benefits		336	1,970
Unrealized losses on derivative financial instruments		21,104	372
Unrealized losses on investments and loans held-for-investment		36,482	72
Sales-type lease		7,064	12.
Other non-cash items		(82)	
Changes in operating assets and liabilities:		(82)	
Receivables		(4,015)	2,769
Prepaid expenses, contract and other assets		(7,556)	(67,253
Accounts payable and accrued liabilities		2,933	60,624
Accrued interest payable		(1,076)	48,713
Contract liabilities		(17,336)	(62,737
Accrued retirement benefits		(9,616)	(12,253
Other long-term liabilities		(4,877)	(1,062
Net cash provided by operating activities		163,429	194,729
ash flows from investing activities:		105,427	174,72
Payments for satellites and other property and equipment (including capitalized interest)		(202,265)	(419,952
Origination of loans held-for-investment		(19,424)	(11),932
Proceeds from loans held-for-investment		(1),424)	973
Capital contribution to unconsolidated affiliate (including capitalized interest)		(338)	(2,692
Other proceeds from satellites		7,500	5,62
Net cash used in investing activities		(214,527)	(418,346
Cash flows from financing activities:		(214,527)	(+10,5+0
Proceeds from issuance of long-term debt		400,000	
Debt issuance costs		(4,650)	_
Proceeds from debtor-in-possession financing		(4,050)	500,000
Debtor-in-possession financing fees			(52,182
Principal payments on deferred satellite performance incentives		(20,991)	(25,428
Dividends paid to noncontrolling interest		(4,830)	(4,959
Proceeds from exercise of employee stock options		989	(1,95)
Other financing activities		298	_
Net cash provided by financing activities		370,816	417,431
Effect of exchange rate changes on cash, cash equivalents and restricted cash			
let change in cash, cash equivalents and restricted cash		(2,241)	(4,328
		317,477	189,486
Cash, cash equivalents, and restricted cash, beginning of period	¢	507,157	\$30,864
Cash, cash equivalents, and restricted cash, end of period	\$	824,634	\$ 1,020,350

	Months Ended mber 30, 2019	e Months Ended otember 30, 2020
Supplemental cash flow information:		
Cash paid for reorganization items included in cash flows from operating activities	\$ —	\$ 53,983
Interest paid, net of amounts capitalized	840,180	532,234
Income taxes paid, net of refunds	9,284	4,717
Supplemental disclosure of non-cash investing activities:		
Accrued capital expenditures	\$ 3,825	\$ 51,221
Conversion of loans held-for-investment to equity securities		4,802

See accompanying notes to unaudited condensed consolidated financial statements.

NOTES TO THE CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

September 30, 2020

Note 1 General

Basis of Presentation

The accompanying condensed consolidated financial statements of Intelsat S.A. and its subsidiaries ("Intelsat S.A.," "we," "us," "our" or the "Company") have not been audited, but are prepared in accordance with United States generally accepted accounting principles ("U.S. GAAP") for interim financial information. Accordingly, they do not include all of the information and footnotes required by U.S. GAAP for complete financial statements. References to U.S. GAAP issued by the Financial Accounting Standards Board ("FASB") in these footnotes are to the FASB Accounting Standards Codification ("ASC"). The unaudited condensed consolidated financial statements include all adjustments (consisting only of normal and recurring adjustments) that are, in the opinion of management, necessary for a fair presentation of these financial statements. The results of operations for the periods presented are not necessarily indicative of operating results for the full year or for any future period. These unaudited condensed consolidated financial statements and notes thereto included in our Annual Report on Form 10-K for the year ended December 31, 2019, on file with the U.S. Securities and Exchange Commission ("SEC").

Use of Estimates

The preparation of these condensed consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities as of the date of these condensed consolidated financial statements, the reported amounts of revenues and expenses during the reporting periods, and the disclosures of contingent liabilities. Accordingly, ultimate results could differ from those estimates.

Bankruptcy Accounting

Our condensed consolidated financial statements included herein have been prepared as if we are a going concern and reflect the application of ASC 852, *Reorganizations* ("ASC 852"). ASC 852 requires the financial statements, for periods subsequent to the commencement of the Chapter 11 proceedings, to distinguish transactions and events that are directly associated with the reorganization from the ongoing operations of the business. Accordingly, we classify liabilities and obligations whose treatment and satisfaction are dependent on the outcome of the reorganization under the Chapter 11 proceedings as liabilities subject to compromise on our condensed consolidated balance sheets. In addition, we classify all income, expenses, gains or losses that are incurred or realized as a result of the Chapter 11 proceedings as reorganization items in our condensed consolidated statements of operations. See Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters.

Impact of COVID-19 on the Company

As a result of the novel coronavirus ("COVID-19") pandemic, in the first, second and third quarters of 2020, in an effort to safeguard public health, governments around the world, including United States ("U.S.") federal, state and local governments, implemented a number of orders and restrictions on travel and businesses, among other things. Some of these measures remain in effect and have negatively impacted the U.S. and other economics around the world in the short-term, while the long-term economic impact of COVID-19 remains unknown.

The COVID-19 pandemic has had an adverse impact on our business, results of operations and financial condition, a trend we expect to continue. Among the impacts of the COVID-19 pandemic were a reduction of revenue and a decreased likelihood of collection from certain mobility customers. We continue to closely monitor the ongoing impact on our employees, customers, business and results of operations.

Gogo Inc. Transaction

On August 31, 2020, following approval from the U.S. Bankruptcy Court for the Eastern District of Virginia (the "Bankruptcy Court"), Intelsat Jackson and Gogo Inc., a Delaware corporation ("Gogo"), entered into a purchase and sale agreement (the "Purchase and Sale Agreement") with respect to Gogo's commercial aviation business for \$400.0 million in cash, subject to customary adjustments (the "Purchase Price"). The transaction further propels the Company's efforts in the growing commercial in-flight connectivity market, pairing our high-capacity global satellite and ground network with Gogo's installed base of more than 3,000 commercial aircraft to redefine the connectivity experience.

We expect to fund the Purchase Price with proceeds from our existing DIP Facility (as defined below) and cash on hand. In connection therewith, on August 24, 2020, the DIP Debtors and DIP Lenders (each as defined below) entered into an amendment ("DIP Amendment No. 1") to the DIP Credit Agreement (as defined below), in connection with the transactions contemplated by the Purchase and Sale Agreement (the "Gogo Transaction"). DIP Amendment No. 1 was approved by the Bankruptcy Court on August 31, 2020. The Gogo Transaction is expected to close before the end of the first quarter of 2021, subject to the receipt of certain regulatory approvals and the satisfaction or waiver of certain other customary closing conditions.

Cash and Cash Equivalents and Restricted Cash

Cash and cash equivalents consist of cash on hand and highly liquid investments with original maturities of three months or less, which are generally time deposits with banks and money market funds. The carrying amount of these investments approximates fair value. Restricted cash represents legally restricted amounts being held as a compensating balance for certain outstanding letters of credit.

The following table provides a reconciliation of cash, cash equivalents and restricted cash reported within our condensed consolidated balance sheets to the total sum of these amounts reported in our condensed consolidated statements of cash flows (in thousands):

	As of December 31, 2019	As of September 30, 2020
Cash and cash equivalents	\$ 810,626	\$ 1,001,000
Restricted cash	20,238	 19,350
Cash, cash equivalents and restricted cash	\$ 830,864	\$ 1,020,350

Recently Adopted Accounting Pronouncements

In June 2016, the FASB issued ASU 2016-13, *Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments* ("ASU 2016-13"), which changes how companies measure and recognize credit impairment for any financial assets. The standard requires companies to immediately recognize an estimate of credit losses expected to occur over the remaining life of the financial assets that are within the scope of the standard. We adopted ASU 2016-13 and its amendments in the first quarter of 2020, on a modified retrospective basis. The adoption of ASU 2016-13 and its amendments increased our reserve for credit losses by \$0.9 million as of January 1, 2020.

In January 2017, the FASB issued ASU 2017-04, *Intangibles—Goodwill and Other (Topic 350): Simplifying the Test for Goodwill Impairment* ("ASU 2017-04"), which is intended to simplify the subsequent measurement of goodwill. The amendments in ASU 2017-04 modify the concept of impairment from the condition that exists when the carrying amount of goodwill exceeds its fair value to the condition that exists when the carrying amount of a reporting unit exceeds its fair value. An entity will no longer determine goodwill impairment by calculating the implied fair value of goodwill by assigning the fair value of a reporting unit to all of its assets and liabilities, as if that reporting unit had been acquired in a business combination. We adopted ASU 2017-04 in the first quarter of 2020, on a prospective basis. As a result, we will measure impairment using the difference between the carrying amount and the fair value of the reporting unit, if required. In the first quarter of 2020, we conducted a goodwill impairment analysis and determined the fair value of our reporting unit to be greater than its carrying value, resulting in no impairment of goodwill. See Note 10—Goodwill and Other Intangible Assets for further information.

In August 2018, the FASB issued ASU 2018-13, *Fair Value Measurement (Topic 820)—Disclosure Framework—Changes to the Disclosure Requirements for Fair Value Measurement* ("ASU 2018-13"), as part of its disclosure framework project to improve the effectiveness of disclosures in the notes to financial statements. Changes in unrealized gains and losses, the range and weighted average of significant unobservable inputs used to develop Level 3 fair value measurements, and the narrative description of measurement uncertainty were applied prospectively for only the most recent interim period presented. All other amendments were applied retrospectively for all periods presented. ASU 2018-13 and its amendments were adopted by the Company in the first quarter of 2020.

Recently Issued Accounting Pronouncements

In August 2018, the FASB issued ASU 2018-14, *Compensation—Retirement Benefits—Defined Benefit Plans—General (Subtopic 715-20)— Disclosure Framework—Changes to the Disclosure Requirements for Defined Benefit Plans* ("ASU 2018-14"), as part of its disclosure framework project to improve the effectiveness of disclosures in the notes to financial statements. ASU 2018-14 modifies and clarifies disclosure requirements for employers that sponsor defined benefit pension or other postretirement plans. The amendments remove certain disclosure requirements and require additional disclosures. ASU 2018-14 will be effective for the Company for annual periods in fiscal years ending after December 15, 2020, on a retrospective basis to all periods presented. We are in the process of evaluating the impact that ASU 2018-14 will have on our condensed consolidated financial statements and associated disclosures.

In December 2019, the FASB issued ASU 2019-12, *Income Taxes (Topic 740): Simplifying the Accounting For Income Taxes* ("ASU 2019-12"). The standard removes certain exceptions for recognizing deferred taxes for investments, performing intra-period allocation and calculating income taxes in interim periods. It also adds guidance to reduce complexity in certain areas, including recognizing deferred taxes for tax goodwill and allocating taxes to members of a consolidated group. ASU 2019-12 will be effective for the Company for annual periods in fiscal years beginning after December 15, 2020. We are in the process of evaluating the impact that ASU 2019-12 will have on our condensed consolidated financial statements and associated disclosures.

Note 2 Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters

Voluntary Reorganization under Chapter 11

On May 13, 2020, Intelsat S.A. and certain of its subsidiaries (each, a "Debtor" and collectively, the "Debtors") commenced voluntary cases (the "Chapter 11 Cases") under title 11 of the United States Code (the "Bankruptcy Code") in the Bankruptcy Court. Primary factors causing us to file for Chapter 11 protection included the Company's intention to participate in the accelerated clearing process of C-band spectrum set forth in the U.S. Federal Communications Commission's ("FCC") March 3, 2020 final order (the "FCC Final Order"), requiring the Company to incur significant costs related to clearing activities well in advance of receiving reimbursement for such costs and the need for additional financing to fund the C-band clearing process, service our current debt obligations, and meet our operating requirements, as well as the economic slowdown impacting the Company and several of its end markets due to the COVID-19 pandemic. On August 14, 2020, the Company filed its final C-band spectrum transition plan with the FCC. On September 17, 2020, the Company announced it finalized all of its required contracts with satellite manufacturers and launch-vehicle providers to move forward and meet the accelerated C-band spectrum clearing timelines established by the FCC.

The Chapter 11 process can be unpredictable and involves significant risks and uncertainties. Pursuant to various orders from the Bankruptcy Court, the Debtors have received approval from the Bankruptcy Court to generally maintain their ordinary operations and uphold certain commitments to their stakeholders, including employees, customers, and vendors, during the restructuring process, subject to the jurisdiction of the Bankruptcy Court and in accordance with the applicable provisions of the Bankruptcy Code. Our ability to fund operating expenses may be subject to obtaining further approvals from the Bankruptcy Court in connection with the Chapter 11 Cases.

On June 9, 2020, Intelsat Jackson received approval from the Bankruptcy Court (the "DIP Order") to enter into a non-amortizing multiple draw superpriority secured debtor-in-possession term loan facility (the "DIP Facility"), in an aggregate principal amount of \$1.0 billion on the terms and conditions as set forth in the DIP Facility credit agreement (the "DIP Credit Agreement") with certain of the Debtors' prepetition secured parties (the "DIP Lenders"), and on June 17, 2020, Intelsat Jackson and certain of its subsidiaries as guarantors (together with Intelsat Jackson, the "DIP Debtors") entered into the DIP Credit Agreement with the DIP Lenders, as amended by DIP Amendment No. 1 (see Note 1—General—*Gogo Inc. Transaction*), which was approved by the Bankruptcy Court on August 31, 2020. For additional information regarding the DIP Facility, DIP Credit Agreement and DIP Amendment No. 1, see Note 11—Debt.

On July 11, 2020, the Debtors filed with the Bankruptcy Court schedules and statements setting forth, among other things, the assets and liabilities of each of the Debtors, subject to the assumptions filed in connection therewith. These schedules and statements may be subject to further amendment or modification after filing.

The filing of the Chapter 11 Cases constituted an event of default that accelerated substantially all of our obligations under the documents governing the prepetition existing indebtedness of Intelsat S.A., Intelsat Luxembourg, Intelsat Connect and Intelsat Jackson. For additional discussion regarding the impact of the Chapter 11 Cases on our debt obligations, see Note 11—Debt.

While the Chapter 11 Cases are pending, the Debtors do not anticipate making interest payments due under their respective unsecured debt instruments; however, the Debtors expect to make monthly interest payments on their senior secured debt instruments pursuant to the adequate protection requirements under the DIP Order. For the three and nine months ended September 30, 2020, the contractual interest expense pursuant to our unsecured debt instruments that was not recognized in our condensed consolidated statements of operations was \$196.4 million and \$298.9 million, respectively.

In order for the Debtors to emerge successfully from Chapter 11, the Debtors must obtain the required votes of creditors accepting a plan of reorganization as well as the Bankruptcy Court's confirmation of such plan. A reorganization plan determines the rights and satisfaction of claims of various creditors and security holders and is subject to the ultimate outcome of negotiations and Bankruptcy Court decisions ongoing through the date on which it is confirmed.

Delisting of Intelsat S.A. Common Shares

On May 20, 2020, the New York Stock Exchange ("NYSE") filed a Form 25 with the SEC to delist the Company's common shares, \$0.01 par value from the NYSE. The delisting became effective 10 days after the Form 25 was filed. The deregistration of the

common shares under Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act") became effective 90 days after the filing date of the Form 25. The common shares remain registered under Section 12(g) of the Exchange Act. The Company's common shares began trading on the OTC Pink Marketplace on May 19, 2020 under the symbol "INTEQ."

Liabilities Subject to Compromise

Prepetition unsecured liabilities of the Debtors subject to compromise under the Chapter 11 proceedings have been distinguished from secured liabilities that are not expected to be compromised and post-petition liabilities in our condensed consolidated balance sheets. Liabilities subject to compromise have been recorded at the amounts expected to be allowed by the Bankruptcy Court. The ultimate settlement amounts of these liabilities remain at the discretion of the Bankruptcy Court and may vary from the expected allowed amounts.

Liabilities subject to compromise consisted of the following (in thousands):

	S	As of September 30, 2020
Accounts payable	\$	9,874
Debt subject to comprise		9,782,161
Accrued interest on debt subject to compromise		341,676
Other long-term liabilities subject to compromise		36,633
Total liabilities subject to compromise	\$	10,170,344

Reorganization Items

The expenses, gains and losses directly and incrementally resulting from the Chapter 11 Cases are separately reported as reorganization items in our condensed consolidated statements of operations.

Reorganization items consisted of the following (in thousands):

	Months Ended mber 30, 2020	e Months Ended tember 30, 2020
Adjustment of debt discount, premium and issuance costs	\$ _	\$ 196,974
Debtor-in-possession financing fees	—	52,182
Professional fees	36,262	85,233
Other reorganization costs	105	670
Total reorganization items	\$ 36,367	\$ 335,059

Going Concern

Our condensed consolidated financial statements have been prepared assuming that we will continue as a going concern, which contemplates continuity of operations, realization of assets, and satisfaction of liabilities in the normal course of business. In connection with the preparation of our condensed consolidated financial statements, we conducted an evaluation as to whether there were conditions and events, considered in the aggregate, that raised substantial doubt as to the Company's ability to continue as a going concern. As reflected in our condensed consolidated financial statements, the Company had cash and cash equivalents of \$1.0 billion and an accumulated deficit of \$8.1 billion as of September 30, 2020. The Company generated income from operations of \$349.5 million and a net loss of \$638.3 million for the nine months ended September 30, 2020.

In light of the Company's Chapter 11 proceedings, our ability to continue as a going concern is contingent upon, among other things, our ability to, subject to the Bankruptcy Court's approval, implement a business plan of reorganization, emerge from the Chapter 11 proceedings and generate sufficient liquidity following the reorganization to meet our contractual obligations and operating needs. As a result of risks and uncertainties related to, among other things, (i) the Company's ability to obtain requisite support for the business plan of reorganization from various stakeholders, and (ii) the disruptive effects of the Chapter 11 proceedings on our business making it potentially more difficult to maintain business, financing and operational relationships, substantial doubt exists regarding our ability to continue as a going concern.

The filing of the Chapter 11 Cases constituted an event of default that accelerated substantially all of our obligations under the documents governing the prepetition existing indebtedness of Intelsat S.A., Intelsat Luxembourg, Intelsat Connect and Intelsat Jackson. As such, we have reclassified all such debt obligations, other than debt subject to compromise, to current portion of long-term debt on our condensed consolidated balance sheet as of September 30, 2020. For additional discussion regarding the impact of the Chapter 11 Cases on our debt obligations, see Note 11—Debt.



Our condensed consolidated financial statements do not include any adjustments related to the recoverability and classification of recorded asset amounts or the amounts and classification of liabilities that might be necessary should we be unable to continue as a going concern.

Note 3 Share Capital

Under our Articles of Incorporation, we have an authorized share capital of \$10.0 million, represented by 1.0 billion shares of any class with a nominal value of \$0.01 per share. At September 30, 2020, there were approximately 142.1 million common shares issued and outstanding.

Note 4 Revenue

(a) Revenue Recognition

We earn revenue primarily by providing services to our customers using our satellite transponder capacity. Our customers generally obtain satellite capacity from us by placing an order pursuant to one of several master customer service agreements. On-network services are comprised primarily of services delivered on our owned network infrastructure, as well as commitments for third-party capacity, generally long-term in nature, that we integrate and market as part of our owned infrastructure. In the case of third-party services in support of government applications, the commitments for third-party capacity are shorter and matched to the government contracting period, and thus remain classified as off-network services. Off-network services can include transponder services and other satellite-based transmission services, such as mobile satellite services ("MSS"), which are sourced from other operators, often in frequencies not available on our network. Under the category Off-Network and Other Revenues, we also include revenues from consulting and other services.

For each service type, the price per unit in our contracts is generally fixed for each defined time period. While the number of units or price per unit in our multi-year contracts may be different by year or another time period, the number of units and price per unit are fixed for each defined time period and the total contract price is fixed. To determine the proper revenue recognition method for contracts, we evaluate whether two or more services should be combined and accounted for as a single performance obligation. Our specific revenue recognition policies are as follows:

Satellite Utilization Charges

The Company's contracts for satellite utilization services often contain multiple service orders for the provision of capacity on or over different beams, satellites, frequencies, geographies or time periods. Under each separate service order, the Company's satellite services, comprised of transponder services, managed services, channel services, and occasional use managed services, are delivered in a series of time periods that are distinct from each other and have the same pattern of transfer to the customer. In each period, the Company's obligation is to make those services available to the customer. Throughout each service period, the Company provides services that are able to be used continuously, and the customer simultaneously receives and consumes the benefits provided by the Company. We believe that, given that our services are stand-ready obligations that are available continuously, the passage of time most faithfully reflects our satisfaction of the performance obligation. We also have certain obligations, including providing spare or substitute capacity if available, in the event of satellite service failure under certain long-term agreements. While we are generally not obligated to refund satellite utilization payments previously made, credits may be granted for sustained service outages in certain limited circumstances.

Similar to satellite utilization charges, we have determined that the customer simultaneously receives and consumes benefits provided by the Company for satellite related consulting and technical services, tracking, telemetry and commanding services ("TT&C") and in-orbit backup services, as detailed below. Therefore, similar to satellite utilization charges, we believe that the passage of time most faithfully reflects our satisfaction of the performance obligation for these services:

Satellite-Related Consulting and Technical Services

We recognize revenue from the provision of consulting services as those services are performed. We recognize revenue for consulting services with specific performance obligations, such as transfer orbit support services or training programs over the service period.

TT&C

We earn TT&C services revenue from providing operational services to other satellite owners and from certain customers on our satellites. TT&C agreements entered into in connection with our satellite utilization contracts are typically for the period of the related service agreement. We recognize this revenue over the term of the service agreement.

In-Orbit Backup Services

We provide back-up transponder capacity that is held on reserve for certain customers on agreed-upon terms. We recognize revenues for in-orbit backup services over the term of the related agreement.

Revenue Share Arrangements

We recognize revenues under revenue share agreements for satellite-related services either on a gross or net basis in accordance with principal versus agent considerations.

We occasionally sell products or services individually or in some combination to our customers. When products or services are sold together, we allocate revenue for each performance obligation based on each obligation's relative selling price. In these arrangements, revenue for products is recognized when the transfer of control passes to the customer, while service revenue is recognized over the service term.

Contract Assets

Contract assets include unbilled amounts typically resulting from sales under our long-term contracts when the total contract value is recognized on a straight-line basis and the revenue recognized exceeds the amount billed to the customer.

Contract Liabilities

Contract liabilities consist of advance payments and collections in excess of revenue recognized and deferred revenue. Our contracts at times contain prepayment terms that range from one month to one year in advance of providing the service. As a practical expedient, we do not need to adjust the promised amount of consideration for the effects of a significant financing component if we expect, at contract inception, that the period of time between when the Company transfers a promised good or service to a customer and when the customer pays for that good or service will be one year or less. For a small subset of contracts with advance payments that contain prepayment terms greater than one year and up to fifteen years, we assess whether a significant financing component exists by considering the difference between the amount of promised consideration and the cash selling price of the promised services. The prepayment amount, resulting in a difference between the amount of promised consideration and the cash selling price of the promised services. The Company considers the timing difference between payment and the promised transfer of services, combined with the Company's incremental borrowing rates, to determine whether a significant financing component exists. When a significant financing component exists, the amount of revenue recognized exceeds the amount of cash received from the customer. After receiving cash from the customer but prior to the Company is effectively receiving from the customer. Once the Company begins providing services, additional interest expense is recorded each period using the effective interest method, as well as corresponding additional revenue, which is recognized ratably over the service period.

For the three months ended September 30, 2019 and 2020, we recognized revenue of \$54.8 million and \$47.8 million, respectively, and for the nine months ended September 30, 2019 and 2020, we recognized revenue of \$200.3 million and \$185.5 million, respectively, that were included in the contract liability balances as of January 1, 2019 and 2020, respectively. In addition, the total amount of consideration included in contract assets as of January 1, 2019 and 2020 that became unconditional for the nine months ended September 30, 2019 and 2020 was \$9.1 million and \$15.1 million, respectively.

Assets Recognized from the Costs to Obtain a Customer Contract

We recognize an asset for the incremental costs of obtaining a contract with a customer if we expect the benefit of those costs to be longer than one year. We have determined that our sales incentive program meets the requirements to be capitalized due to the incremental nature of the costs and the expectation that the Company will recover such costs. The assets recognized from the costs to obtain a customer contract are amortized over a period that is consistent with the transfer to the customer of the services to which the asset relates. We capitalized \$1.7 million and \$1.5 million for our sales incentive program and amortized \$1.2 million for the three months ended September 30, 2019 and 2020, respectively, and we capitalized \$5.7 million and \$3.9 million for our sales incentive program and amortized \$4.7 million and \$3.9 million for the nine months ended September 30, 2019 and 2020, respectively. As of December 31, 2019 and September 30, 2020, capitalized costs relating to our sales incentive program were \$9.4 million and \$9.5 million, respectively, which were included within other assets in our condensed consolidated balance sheets.

Contract Modifications

Contracts are often modified to account for changes in contract specifications or requirements. We consider contract modifications to exist when the modification either creates new rights or obligations or changes the existing enforceable rights and



obligations of either party. Most of our contract modifications are for goods and services that are distinct from the existing contract, as they consist of additional months of service priced at the Company's standalone selling prices of the additional services and are therefore treated as separate contracts. For contract modifications that do not result in additional distinct goods or services, the effect of a contract modification on the transaction price and our measure of progress for the performance obligation to which it relates, is recognized as an adjustment to revenue.

Significant Judgments

We occasionally enter into certain contracts in which the customer makes payments in advance of services to be delivered, which may be years in the future. The reasons for the prepayments in these contracts vary, but generally can be either for the customer's benefit or for the Company's benefit (such as the ability to use the cash received from the customer to pay for the construction of a satellite asset). The determination of whether contracts with a prepayment provision contain a significant financing component requires judgment. The Company makes this determination based on various factors, including the differences between the amount of promised consideration and cash selling prices, the length of time between payment and the transfer of services and prevailing interest rates in the market.

While most satellite utilization contracts contain multiple performance obligations for each transponder service on different satellites, the service period for the different satellite utilization performance obligations is generally the same time period. In the event that the time period for multiple performance obligations is not the same, we allocate the total transaction price to each performance obligation in an amount based on the estimated relative standalone selling price of the promised good or service underlying such performance obligation. Judgment is required to determine the standalone selling price for each distinct performance obligation. In order to estimate standalone selling prices, we use an adjusted market assessment approach which involves an evaluation of the market and an estimate of the price that our customers are willing to pay, or an expected cost plus a margin approach.

When more than one party is involved in providing goods or services to a customer, we generally recognize the transaction on a gross basis due to the level of control that we have prior to the transfer of the good or service. These arrangements include instances where we procure equipment from vendors and sell to third-party customers, when we enter into revenue sharing arrangements with other parties and when we purchase capacity for voice, data and video services provided by third-party commercial satellite operators for which the desired frequency type or geographic coverage is not available on our network. Our third-party capacity arrangements (off-network) are more significant and, in determining whether we are the principal or the agent in these arrangements, we consider whether or not we control the service before it is transferred to the customer. In this determination, we consider the definition of control as set forth in ASC 606, *Revenue from Contracts with Customers* ("ASC 606"), in ASC 606-10-25-25. When we purchase satellite transponder capacity from a third party, we have the ability to direct the use of and obtain substantially all of the remaining benefits from the purchased capacity. We obtain the right to the service to be performed by the third party, which gives the Company the ability to direct that party to provide the service to the customer on the Company's behalf. No other third party can direct the use of or obtain any benefits from the capacity.

We also considered the factors in ASC 606-10-55-39 in the Company's determination of control. In the vast majority of cases, when we resell capacity to third party customers, we are primarily responsible for the fulfillment of the services and acceptability of the service. Additionally, the Company has full discretion in establishing the pricing for transponder services with the customer and assumes the credit risk associated with capacity purchased from the third party. In the event the service is not acceptable to the customer, we are required to identify an alternative solution. Based on these considerations, we have concluded that we are the principal in the transaction for these arrangements. When these factors are not met, the Company recognizes revenue for third-party capacity arrangements on a net basis.

Judgment is required in determining whether we are the principal or the agent in transactions involving third parties.

Remaining Performance Obligations

Our remaining performance obligation is our expected future revenue under our existing customer contracts and includes both cancelable and noncancelable contracts. Our remaining performance obligation was approximately \$6.1 billion as of September 30, 2020, approximately 92% of which related to contracts that were non-cancelable and approximately 8% of which related to contracts that were cancelable subject to substantial termination fees. We assess the contract term of our cancelable contracts as the full stated term of the contract assuming each contract is not canceled since the termination penalty upon cancellation is substantive. As of September 30, 2020, the weighted average remaining customer contract life was approximately 4.0 years. Approximately 29%, 27%, and 44% of our total remaining performance obligation as of September 30, 2020 is expected to be recognized as revenue during 2020 and 2021, 2022 and 2023, and 2024 and thereafter, respectively. The amount included in the remaining performance obligation represents the full-service charge for the duration of the contract and does not include termination fees. The amount of the termination fees, which is not included in the remaining performance obligation amount, is generally calculated as a percentage of the remaining performance obligation associated with the contract. In certain cases of breach for non-payment or customer financial distress or bankruptcy, we may not be able to recover the full value of certain contracts or termination fees. Our remaining performance obligation includes 100% of the remaining performance obligation of our consolidated ownership interests, which is consistent with the accounting for our ownership interest in these entities.

(b) Business and Geographic Segment Information

We operate in a single industry segment in which we provide satellite services to our communications customers around the world. Our revenues are disaggregated by billing region, service type and customer set. Revenue by region is based on the locations of customers to which services are billed. Our satellites are in geosynchronous orbit, and consequently are not attributable to any geographic location. Of our remaining assets, substantially all are located in the U.S.

The following table disaggregates revenue by billing region (in thousands, except percentages):

	Three Months September 30		Three Months September 30		Nine Months En September 30, 20		Nine Months E September 30,		
North America	\$ 271,001	53 %	\$ 266,516	54 %	\$ 804,014	52 %	\$ 758,999	53 %	
Europe	60,708	12 %	54,082	11 %	186,497	12 %	160,866	11 %	
Latin America and Caribbean	56,702	11 %	52,474	11 %	183,045	12 %	157,956	11 %	
Africa and Middle East	61,080	12 %	60,349	12 %	185,186	12 %	181,151	13 %	
Asia-Pacific	57,167	11 %	56,028	11 %	185,772	12 %	171,331	12 %	
Total	\$ 506,658		\$ 489,449		\$ 1,544,514		\$ 1,430,303		

The following table disaggregates revenue by type of service (in thousands, except percentages):

		onths Ended Der 30, 2019		Ionths Ended Iber 30, 2020	Nine Month September 3		Nine Months Ended September 30, 2020			
On-Network Revenues										
Transponder services	\$ 364,3	12 72 %	\$ 353,	758 72 %	\$ 1,111,182	72 %	\$ 1,028,692	72 %		
Managed services	93,0	80 18 %	69,	138 14 %	277,616	18 %	226,749	16 %		
Channel	6	01 —%		280 — %	1,877	<u> </u>	1,096	%		
Total on-network revenues	457,9	93 90 %	423,	176	1,390,675	90 %	1,256,537	88 %		
Off-Network and Other Revenues										
Transponder, MSS and other off- network services	39,1	29 8%	54,	478 11 %	126,704	8 %	141,783	10 %		
Satellite-related services	9,5	36 2 %	11,	495 2 %	27,135	2 %	31,983	2 %		
Total off-network and other revenues	48,6	65 10 %	65,	973 13 %	153,839	10 %	173,766	12 %		
Total	\$ 506,6	58	\$ 489,	149	\$ 1,544,514	=	\$ 1,430,303	-		

By customer application, our revenues from network services, media, government and satellite-related services were \$180.8 million, \$222.9 million, \$95.7 million and \$7.3 million, respectively, for the three months ended September 30, 2019, as compared to \$169.6 million, \$203.5 million, \$108.0 million and \$8.3 million, respectively, for the three months ended September 30, 2020. Revenues from network services, media, government and satellite-related services were \$570.2 million, \$672.3 million, \$282.3 million and \$19.7 million, respectively, for the nine months ended September 30, 2019, as compared to \$495.7 million, \$611.9 million, \$299.9 million and \$22.8 million, respectively, for the nine months ended September 30, 2020.

Our largest customer accounted for approximately 15% and 14% of our revenue during the three months ended September 30, 2019 and 2020, respectively, and 15% and 14% during the nine months ended September 30, 2019 and 2020, respectively. Our ten largest customers accounted for approximately 42% and 43% of our revenue during the three months ended September 30, 2019 and 2020, respectively, and approximately 40% and 41% during the nine months ended September 30, 2019 and 2020, respectively, and approximately 40% and 41% during the nine months ended September 30, 2019 and 2020, respectively.

Note 5 Net Loss per Share

Basic net loss per common share attributable to Intelsat S.A. ("EPS") is computed by dividing net loss attributable to Intelsat S.A.'s common shareholders by the weighted average number of common shares outstanding during the periods. Diluted EPS assumes the issuance of common shares pursuant to share-based compensation plans and conversion of the Intelsat S.A. 4.5% Convertible Senior Notes due 2025 (the "2025 Convertible Notes"), unless the effect of such issuances would be anti-dilutive.

The following table sets forth the computation of basic and diluted EPS (in thousands, except per share data or where otherwise noted):

	hree Months Ended September 30, 2019		Three Months Ended September 30, 2020	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Numerator:	 	_			
Net loss attributable to Intelsat S.A.	\$ (148,292)	\$	(15,931)	\$ (798,636)	\$ (640,057)
Denominator:					
Basic weighted average shares outstanding (in millions)	140.9		142.1	140.1	141.9
Diluted weighted average shares outstanding (in millions):	140.9		142.1	140.1	141.9
Basic EPS	\$ (1.05)	\$	(0.11)	\$ (5.70)	\$ (4.51)
Diluted EPS	\$ (1.05)	\$	(0.11)	\$ (5.70)	\$ (4.51)

In June 2018, Intelsat S.A. completed an offering of \$402.5 million aggregate principal amount of its 2025 Convertible Notes. We do not expect to settle the principal amount of the 2025 Convertible Notes in cash, and therefore use the if-converted method for calculating any potential dilutive effect of the conversion on diluted EPS, if applicable. Under the indenture governing the 2025 Convertible Notes (the "2025 Indenture"), the 2025 Convertible Notes are eligible for conversion depending upon the trading price of our common shares and under other conditions set forth in the indenture until December 15, 2024, and thereafter without regard to any conditions. The commencement of the Chapter 11 Cases constituted an event of default under the 2025 Indenture. See Note 11—Debt for additional information on the impact of the Chapter 11 Cases on our debt obligations.

Due to a net loss for each of the three and nine months ended September 30, 2019 and 2020, there were no dilutive securities, and therefore, basic and diluted EPS were the same. The weighted average number of common shares that could potentially dilute basic EPS in the future was 26.5 million and 22.3 million for the three months ended September 30, 2019 and 2020, respectively, and 26.8 million and 22.3 million for the nine months ended September 30, 2019 and 2020, respectively, and 26.8 million and 22.3 million for the nine months ended September 30, 2019 and 2020, respectively, and 26.8 million and 22.3 million for the nine months ended September 30, 2019 and 2020, respectively.

Note 6 Fair Value Measurements

ASC 820, *Fair Value Measurements and Disclosure* ("ASC 820") defines fair value, establishes a market-based framework or hierarchy for measuring fair value and provides for certain required disclosures about fair value measurements. The guidance is applicable whenever another accounting pronouncement requires or permits assets and liabilities to be measured at fair value but does not require any new fair value measurements.

The fair value hierarchy prioritizes the inputs used in valuation techniques into three levels as follows:

- Level 1—unadjusted quoted prices for identical assets or liabilities in active markets;
- Level 2—quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets or liabilities in markets that are not active, and inputs other than quoted market prices that are observable or that can be corroborated by observable market data by correlation; and
- Level 3—unobservable inputs based upon the reporting entity's internally developed assumptions which market participants would use in
 pricing the asset or liability.

Recurring Fair Value Measurements

The tables below present assets measured and recorded at fair value in our condensed consolidated balance sheets on a recurring basis and their corresponding level within the fair value hierarchy (in thousands). No transfers between Level 1, Level 2 and Level 3 fair value measurements occurred for the nine months ended September 30, 2020.



				Fair Value	e Measu	rements as of Decemb	er 31, 2	.019
Description	As of De	ecember 31, 2019		Level 1		Level 2		Level 3
Assets								
Marketable securities ⁽¹⁾	\$	5,145	\$	5,145	\$	—	\$	—
Undesignated interest rate cap contracts ⁽²⁾		372		—		372		
Common stock warrant ⁽³⁾		3,239		—		—		3,239
m + 1 +	¢	8,756	\$	5,145	\$	372	\$	3,239
Total assets	\$	8,/30	ψ	5,145	Ψ	312	Ψ	- 3
lotal assets	<u>></u>	8,730	ф 	,	• Measu	rements as of Septeml	ç Der 30, 2	
Description	<u>ج</u> As of Se	8,730	ф	,			¢	
	As of Seg		φ	Fair Value	Measu	rements as of Septeml	oer 30, 2	2020
Description	As of Sep \$			Fair Value Level 1	weasu S	rements as of Septeml	©er 30, 2	2020
<u>Description</u> <u>Assets</u>		ptember 30, 2020		Fair Value Level 1		rements as of Septeml	©er 30, 2	2020
Description Assets Marketable securities ⁽¹⁾		ptember 30, 2020 4,823		Fair Value Level 1		rements as of Septeml	©er 30, 2 	2020 Level 3

- (1) The valuation measurement inputs of these marketable securities represent unadjusted quoted prices in active markets and, accordingly, we have classified such investments within Level 1 of the fair value hierarchy. The cost basis of our marketable securities was \$4.3 million and \$3.9 million as of December 31, 2019 and September 30, 2020, respectively. We sold marketable securities with a cost basis of \$0.1 million resulting in a nominal gain during each of the three months ended September 30, 2019 and 2020. We sold marketable securities with a cost basis of \$0.6 million and \$0.5 million resulting in a nominal gain during each of the nine months ended September 30, 2019 and 2020, respectively. These amounts are included in other income (expense), net in our condensed consolidated statements of operations.
- (2) The valuation of our interest rate derivative instruments reflects the fair value of premiums paid, taking into account observable inputs including current interest rates, the market expectation for future interest rate volatility and current creditworthiness of the counterparties. As a result, we have determined that the valuation in its entirety is classified as Level 2 within the fair value hierarchy.
- (3) We valued the common stock warrant using a valuation technique that reflects the risk-free interest rate, time to maturity and volatility of comparable companies. We identified the inputs used to calculate the fair value as Level 3 inputs and concluded that the valuation in its entirety is classified as Level 3 within the fair value hierarchy.

The following table presents a reconciliation of the preferred and common stock warrants which are measured and recorded at fair value on a recurring basis using Level 3 inputs (in thousands):

	Three Months Ended September 30, 2019		Three Months Ended September 30, 2020	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Balance as of beginning of period	\$ 247	\$	3,239	\$ 4,100	\$ 3,239
Unrealized loss included in other income (expense), net	(247))	_	(4,100)	_
Balance as of end of period	\$ —	\$	3,239	\$ _	\$ 3,239

Nonrecurring Fair Value Measurements

The carrying values of certain assets may be adjusted to fair value in subsequent periods on a nonrecurring basis if an event occurs or circumstances change that indicate that the asset is impaired or, for investments in equity securities without readily determinable fair values, observable transactions for identical or similar investments of the same issuer support a change in the investment fair value. During the first quarter of 2020, as a result of our interim impairment assessments, we recognized an impairment of non-amortizable intangible assets of \$12.2 million. This fair value measurement is classified as Level 3 within the fair value hierarchy due to the use of significant unobservable inputs. See Note 10—Goodwill and Other Intangible Assets for additional information.

Other Fair Value Disclosures

See Note 9—Investments, Note 10—Goodwill and Other Intangible Assets and Note 11—Debt for fair value disclosures related to our loan receivables, impairment analysis and debt, respectively. The carrying amounts of the Company's other financial instruments are reasonable estimates of their related fair values due to their short-term nature.

Note 7 Retirement Plans and Other Retiree Benefits

(a) Pension and Other Postretirement Benefits

We maintain a noncontributory defined benefit retirement plan covering substantially all of our employees hired prior to July 19, 2001. The cost of providing benefits to eligible participants under the defined benefit retirement plan is calculated using the plan's benefit formulas, which take into account the participants' remuneration, dates of hire, years of eligible service and certain actuarial assumptions. In addition, as part of the overall medical plan, we provide postretirement medical benefits to certain current retirees who meet the criteria under the medical plan for postretirement benefit eligibility. In 2015, we amended the defined benefit retirement plan to end the accrual of additional benefits for the remaining active participants. We have received authorization from the Bankruptcy Court to continue making contributions in the ordinary course during our Chapter 11 Cases.

The defined benefit retirement plan is subject to the provisions of the Employee Retirement Income Security Act of 1974, as amended. We expect that our future contributions to the defined benefit retirement plan will be based on the minimum funding requirements of the Internal Revenue Code and on the plan's funded status. Any significant decline in the fair value of our defined benefit retirement plan assets or other adverse changes to the significant assumptions used to determine the plan's funded status would negatively impact its funded status and could result in increased funding in future years. The impact on the funded status is determined based upon market conditions in effect when we completed our annual valuation. For the nine months ended September 30, 2020, we made cash contributions to the defined benefit retirement plan of \$3.1 million. We anticipate that our remaining contributions to the defined benefits paid. We anticipate that our contributions to fund postretirement medical benefits in 2020 will be approximately \$0.9 million.

Included in accumulated other comprehensive loss at September 30, 2020 was \$94.4 million (\$63.0 million, net of tax) that has not yet been recognized in net periodic benefit cost.

The tables below show the components of net periodic benefit cost (income) for the three and nine months ended September 30, 2019 and 2020 (in thousands). These amounts are recognized in other income (expense), net in the condensed consolidated statements of operations.

	Pension Benefits											
		Months Ended mber 30, 2019		Months Ended nber 30, 2020		ne Months Ended ptember 30, 2019	Nine Months Ended September 30, 2020					
Interest cost	\$	3,848	\$	2,962	\$	11,543	\$	8,888				
Expected return on plan assets		(5,873)		(5,810)		(17,618)		(17,431)				
Amortization of unrecognized net loss		1,055		1,600		3,166		4,799				
Net periodic benefit income	\$	(970)	\$	(1,248)	\$	(2,909)	\$	(3,744)				

		Other Postretiren	nent Bo	enefits		
	onths Ended ber 30, 2019	Three Months Ended September 30, 2020		e Months Ended tember 30, 2019	Nine Months Ended September 30, 2020	
Interest cost	\$ 383	\$ 266	\$	1,149	\$	798
Amortization of unrecognized prior service credits	(636)	(636)		(1,908)		(1,908)
Amortization of unrecognized net gain	(307)	(305)		(922)		(915)
Net periodic benefit income	\$ (560)	\$ (675)	\$	(1,681)	\$	(2,025)

(b) Other Retirement Plans

We maintain a defined contribution retirement plan qualified under the provisions of Section 401(k) of the Internal Revenue Code for our employees in the United States. We have received authorization from the Bankruptcy Court to continue making our contributions in the ordinary course during the Chapter 11 Cases. We recognized compensation expense for this plan of \$2.1 million and \$2.2 million for the three months ended September 30, 2019 and 2020, respectively, and \$6.4 million and \$6.7 million for the nine months ended September 30, 2019 and 2020, respectively. We also maintain other defined contribution retirement plans in several non-U.S. jurisdictions, but such plans are not material to our financial position or results of operations.

Note 8 Satellites and Other Property and Equipment

(a) Satellites and Other Property and Equipment, net

Satellites and other property and equipment, net were comprised of the following (in thousands):

	December 31, 2019	September 30, 2020
Satellites and launch vehicles	\$ 10,407,690	\$ 10,361,059
Information systems and ground segment	968,482	1,020,132
Buildings and other	 280,109	288,484
Total cost	11,656,281	11,669,675
Less: accumulated depreciation	(6,954,218)	 (6,973,552)
Total	\$ 4,702,063	\$ 4,696,123

As of

As of

Satellites and other property and equipment are stated at historical cost, except for satellites that have been impaired. Satellites and other property and equipment acquired as part of an acquisition are stated based on their fair value at the date of acquisition.

During the first quarter of 2020, the price of our common shares and trading values of our debt securities experienced sustained reductions. We also witnessed certain declines in financial performance as compared to previously prepared internal budget and forecast projections. Among the impacts of the COVID-19 pandemic were a reduction of revenue and a decreased likelihood of collection from certain mobility customers. Based on our examination of these and other qualitative factors, we concluded that further testing of satellites and other property and equipment was required.

The Company evaluated the assets for potential impairment using internal projections of undiscounted cash flows expected to result from the use and eventual disposal of the assets. If the carrying amount of the assets exceeds the undiscounted cash flows expected to result from its use, an impairment expense is recognized for the amount by which the carrying amount of the asset group exceeds its fair value. The impairment expense cannot exceed the carrying amount of the long-lived assets (unless the carrying amount is not being reduced below fair value for any individual long-lived asset that is determinable without undue cost and effort).

In estimating the undiscounted cash flows, we primarily used our internally prepared budgets and forecast information. The key assumptions included in our model were projected growth rates, cost of capital, effective tax rates, and industry and economic trends. A change in estimated future cash flows or other assumptions could change our estimated fair values and result in future impairments. The conclusion of our analysis was that the undiscounted cash flows of the asset group were greater than its carrying value, resulting in no impairment.

Satellites and other property and equipment, net of accumulated depreciation, included construction-in-progress of \$191.5 million and \$606.5 million as of December 31, 2019 and September 30, 2020, respectively. These amounts relate primarily to satellites under construction and related launch services. As of September 30, 2020, we incurred C-band clearing related costs and expenses of \$312.1 million, of which \$311.6 million is capitalized and expected to be reimbursable under the FCC Final Order. Of this capitalized amount, \$288.2 million, \$0.4 million, and \$22.9 million is capitalized as satellites and other property and equipment, net of accumulated depreciation, other current assets, and other assets, respectively, in the condensed consolidated balance sheets.

Interest costs of \$9.6 million and \$10.5 million were capitalized for the three months ended September 30, 2019 and 2020, respectively, and \$26.7 million and \$20.7 million were capitalized for the nine months ended September 30, 2019 and 2020, respectively. Additionally, depreciation expense was \$152.7 million and \$154.4 million for the three months ended September 30, 2019 and 2020, respectively, and \$470.4 million and \$463.3 million for the nine months ended September 30, 2019 and 2020, respectively.

We have entered into launch contracts for the launch of both specified and unspecified future satellites. Each of these launch contracts may be terminated at our option, subject to payment of a termination fee that increases as the applicable launch date approaches. In addition, in the event of a failure of any launch, we may exercise our right to obtain a replacement launch within a specified period following our request for re-launch.

(b) Satellite Launches

Galaxy 30, the first satellite in Intelsat's Galaxy fleet refresh plan, was successfully launched on August 15, 2020. Galaxy 30 will replace Galaxy 14 at 125°W serving media customers in the North America region. Galaxy 30 is the first four-frequency Intelsat satellite with C-, Ku-, Ka- and L-band capabilities. In addition, Galaxy 30 will offer broadband, mobility and network services to mobile network, enterprise and government customers in the North America region. The satellite will also play an important role in the Company's U.S. C-band spectrum transition plan, and is expected to enter into service in early 2021.

Intelsat 39 was successfully launched on August 6, 2019. Intelsat 39 replaced Intelsat 902 at the 62°E location and delivers connectivity services in both the C- and Ku-bands to mobile network operators, enterprises and government customers, as well as aeronautical and maritime mobility service providers operating in the Europe, Africa, Middle East and Asia-Pacific regions. Intelsat 39 entered into service in October 2019.



(c) Significant Anomalies

In April 2019, the Intelsat 29e satellite (in service since 2016) experienced an anomaly that resulted in a total loss of the satellite. In accordance with our existing satellite anomaly contingency plans, we restored service for most Intelsat 29e customers on other satellites in our network, as well as on third-party satellites. We recorded a non-cash impairment charge of \$381.6 million in the second quarter of 2019, of which \$377.9 million related to the write off of the carrying value of the satellite and associated deferred performance incentive obligations and \$3.7 million related to prepaid regulatory fees.

A Failure Review Board comprised of the satellite's manufacturer, Boeing Satellite Systems, Inc., the Company and external independent experts was convened to complete a comprehensive analysis of the cause of the anomaly. The board concluded that the anomaly was caused by either a harness flaw in conjunction with an electrostatic discharge event related to solar weather activity or the impact of a micrometeoroid.

During the second quarter of 2020, the Company determined it unlikely that it will be able to utilize certain satellite and launch vehicle deposits prior to their respective expiration dates. As a result, the Company recorded a non-cash impairment charge of \$34.0 million related to the impairment of the carrying values of the deposits, which is included within impairment of non-amortizable intangible and other assets in the condensed consolidated statements of operations.

Note 9 Investments

We have an ownership interest in two entities that meet the criteria of a variable interest entity ("VIE"): Horizons Satellite Holdings LLC ("Horizons Holdings") and Horizons-3 Satellite LLC ("Horizons 3"), which are discussed in further detail below, including our analyses of the primary beneficiary determination as required under ASC 810, *Consolidation* ("ASC 810"). We also own noncontrolling investments in equity securities and loan receivables as discussed further below.

(a) Horizons Holdings

Horizons Holdings is a joint venture with JSAT International, Inc. ("JSAT") that consists of two investments: Horizons-1 Satellite LLC and Horizons-2 Satellite LLC. Horizons Holdings borrowed from JSAT a portion of the funds necessary to finance the construction of the Horizons 2 satellite pursuant to a loan agreement. The borrowing was subsequently repaid. We provide certain services to the joint venture and in return utilize capacity from the joint venture.

We have determined that this joint venture meets the criteria of a VIE under ASC 810, and we have concluded that we are the primary beneficiary because decisions relating to any future relocation of the Horizons 2 satellite, the most significant asset of the joint venture, are effectively controlled by us. In accordance with ASC 810, as the primary beneficiary, we consolidate Horizons Holdings within our condensed consolidated financial statements. Total assets of Horizons Holdings were \$22.2 million and \$16.8 million as of December 31, 2019 and September 30, 2020, respectively. Total liabilities were nominal as of December 31, 2019 and \$1.1 million as of September 30, 2020.

We have a revenue sharing agreement with JSAT related to services sold on the Horizons 1 and Horizons 2 satellites. We are responsible for billing and collection for such services, and we remit 50% of the revenue, less applicable fees and commissions, to JSAT. Amounts payable to JSAT related to the revenue sharing agreement, net of applicable fees and commissions, from the Horizons 1 and Horizons 2 satellites were \$1.6 million and \$2.1 million as of December 31, 2019 and September 30, 2020, respectively.

(b) Horizons-3 Satellite LLC

On November 4, 2015, we entered into an additional joint venture agreement with JSAT. The joint venture, Horizons 3, was formed for the purpose of developing, launching, managing, operating and owning a high-performance satellite located at the 169°E orbital location.

Horizons 3, which is 50% owned by each of Intelsat and JSAT, was set up with joint sharing of management authority and equal rights to profits and revenues from the joint venture. Similar to Horizons Holdings, we have a revenue sharing agreement with JSAT related to services sold on the Horizons 3e satellite. In addition, we are responsible for billing and collection for such services, and we remit 50% of the revenue, less applicable fees and commissions, to JSAT. Amounts payable to JSAT related to the revenue sharing agreement, net of applicable fees and commissions, from the Horizons 3e satellite were \$3.3 million and \$4.5 million as of December 31, 2019 and September 30, 2020, respectively.

We have determined that this joint venture meets the criteria of a VIE under ASC 810; however, we have concluded that we are not the primary beneficiary and therefore do not consolidate Horizons 3. The assessment considered both quantitative and qualitative factors, including an analysis of voting power and other means of control of the joint venture, as well as each owner's exposure to risk of loss or gain. Because we and JSAT equally share control over the operations of the joint venture and also equally share exposure to risk of losses or gains, we concluded that we are not the primary beneficiary of Horizons 3. Our investment, included within other assets in our condensed consolidated balance sheets, is accounted for using the equity method of accounting. The investment balance,

which is equivalent to our maximum exposure to loss, was \$110.2 million and \$103.8 million as of December 31, 2019 and September 30, 2020, respectively. The investment balance exceeded our equity in the net assets of Horizons 3 by \$11.6 million and \$11.1 million as of December 31, 2019 and September 30, 2020, respectively. This basis difference represents the capitalized interest that we incurred in relation to financing our investment and we recognize it as a reduction of our equity in earnings of Horizons 3 on a straight-line basis over the life of the satellite. We recognized a nominal amount of equity in earnings of Horizons 3 in other income (expense), net for each of the three and nine months ended September 30, 2019 and 2020.

In connection with our investment in Horizons 3, we entered into a capital contribution and subscription agreement, which requires us to fund our 50% share of amounts due thereunder in order to maintain our 50% interest in the joint venture. Pursuant to this agreement, we made contributions of \$5.0 million for the year ended December 31, 2019 and \$2.7 million during the nine months ended September 30, 2020. We received distributions of \$5.0 million for the year ended December 31, 2019 and \$9.0 million for the nine months ended September 30, 2020. The Company utilizes the cumulative earnings approach to determine whether distributions received from equity method investees are returns on investment or returns of investment. In addition, our indirect subsidiary that holds our investment in Horizons 3 has entered into a security and pledge agreement with Horizons 3, pursuant to which it has granted a security interest in its membership interest in Horizons 3. Further, our indirect subsidiary has granted a security interest to Horizons 3 in its customer capacity contracts and its ownership interest in its wholly-owned subsidiary that holds the FCC license required for the joint venture's operations.

The Horizons 3e satellite entered into service in January 2019. The Company purchases satellite capacity and related services from the Horizons 3 joint venture, and then sells that capacity to its customers. We incurred direct costs of revenue related to these purchases of \$5.6 million and \$4.9 million during the three months ended September 30, 2019 and 2020, respectively, and \$14.8 million and \$14.9 million during the nine months ended September 30, 2019 and 2020, respectively, and \$14.8 million and \$14.9 million during the nine months ended September 30, 2019 and 2020, respectively. The Company also sells managed ground network services to the Horizons 3 joint venture and provides program management services for a fee. We recorded an offset to direct costs of revenue related to the provision of these services of \$1.8 million and \$1.7 million during the three months ended September 30, 2019 and 2020, respectively, and \$3.9 million and \$5.2 million during the nine months ended September 30, 2019 and 2020, respectively, and \$3.9 million and \$5.2 million during the nine months ended September 30, 2019 and 2020, respectively, and \$3.9 million and \$5.2 million during the nine months ended September 30, 2019 and 2020, respectively, and \$3.9 million and \$5.2 million during the nine months ended September 30, 2019 and 2020, respectively, and \$3.9 million and \$5.2 million during the nine months ended September 30, 2019 and 2020, respectively, of amounts due to Horizons 3 were included in accounts payable and accrued liabilities.

(c) Investments in Equity Securities

The Company holds noncontrolling equity investments in seven separate privately held companies, including investments in equity securities without readily determinable fair values and common stock warrants.

In accordance with ASC 321, *Investments—Equity Securities*, we use the measurement alternative to measure the fair value of our investments in equity securities without readily determinable fair values. Accordingly, these investments are measured at cost, less any impairment, and are adjusted for changes in fair value resulting from observable transactions for identical or similar investments of the same issuer. We recorded impairment charges of \$5.0 million and \$34.1 million during the three and nine months ended September 30, 2019, respectively, and \$0.1 million during the nine months ended September 30, 2020, with no amounts being recognized during the three months ended September 30, 2020. We recognized an increase in fair value relating to investments of \$1.7 million during the three and nine months ended September 30, 2019, with no comparative amounts in 2020. These investments are recorded in other assets in our condensed consolidated balance sheets and had a total carrying value of \$27.2 million and \$31.9 million as of December 31, 2019 and September 30, 2020, respectively.

We measure our stock warrants at fair value (see Note 6—Fair Value Measurements and Note 12—Derivative Instruments and Hedging Activities for additional information). The warrants are recorded in other assets in our condensed consolidated balance sheets and had a cumulative fair value of \$3.2 million as of both December 31, 2019 and September 30, 2020.

(d) Loan Receivables

The Company has loan receivables from three privately held companies that it is holding for long-term investment. These loan receivables are reported at amortized cost, net of the allowance for credit losses. Amortized cost is the outstanding principal, adjusted for unamortized discounts and deferred transaction costs. The Company recognizes interest income on loan receivables using the effective-interest method applied on a loan-by-loan basis. Direct costs associated with originating loans are offset against any related fees received and the balance, along with any premium or discount, is deferred and amortized as an adjustment to interest income over the term of the related loan receivable using the effective interest method.

Loan receivables are recorded in other assets in our condensed consolidated balance sheets at an amortized cost basis, net of allowance for credit losses, of \$70.4 million and \$68.8 million as of December 31, 2019 and September 30, 2020, respectively. These amounts were net of an allowance for credit losses of \$4.6 million, unamortized discount of \$3.0 million and \$1.3 million, and unamortized deferred transaction costs of \$1.0 million and \$0.4 million as of the respective periods. As of December 31, 2019 and September 30, 2020, respectively, \$1.5 million and \$2.9 million of accrued interest related to our loan receivables was recorded in

prepaid expenses and other current assets in our condensed consolidated balance sheets. We recognized interest income related to our loan receivables of \$1.0 million and \$2.9 million for the three and nine months ended September 30, 2020, respectively, with no comparative amounts for the three and nine months ended September 30, 2019.

A loan is determined to be impaired and placed on non-accrual status when, in management's judgment based on current information and events, it is probable that the Company will be unable to collect all amounts due under the contractual terms of the applicable loan agreement. We recognized impairment losses of \$0.6 million related to loan receivables during the nine months ended September 30, 2020, with no comparable amounts for the three months ended September 30, 2020 and three and nine months ended September 30, 2019.

The fair value of loan receivables is evaluated on a loan-by-loan basis, and is determined based on assessments of discounted cash flows that are considered probable of collection. We consider these inputs to be Level 3 within the fair value hierarchy under ASC 820. The cumulative fair value of our loan receivables as of December 31, 2019 and September 30, 2020 was \$69.3 million and \$71.3 million, respectively.

Note 10 Goodwill and Other Intangible Assets

We account for goodwill and other non-amortizable intangible assets in accordance with ASC 350, *Intangibles—Goodwill and Other*, and have deemed these assets to have indefinite lives. Therefore, these assets are not amortized but are tested on an annual basis for impairment during the fourth quarter, or whenever events or changes in circumstances indicate that the carrying amount may not be fully recoverable.

During the first quarter of 2020, the price of our common shares and trading values of our debt securities experienced sustained reductions. We also witnessed certain declines in financial performance as compared to previously prepared internal budget and forecast projections. Among the impacts of the COVID-19 pandemic were a reduction of revenue and a decreased likelihood of collection from certain mobility customers. Based on our examination of these and other qualitative factors, we concluded that further testing of goodwill and other non-amortizable and amortizable intangible assets was required.

Determining the fair value of a reporting unit and other intangible assets often involves the use of estimates and assumptions that require significant judgment, and that could have a substantial impact on whether or not an impairment charge is recognized and the magnitude of any such charge. Estimates of fair value are primarily determined using discounted cash flows and market transactions. These estimates involve making significant estimates and assumptions, including projected future cash flows (including timing), discount rates reflecting the risks inherent in future cash flows, perpetual growth rates, and the determination of appropriate market comparisons.

(a) Goodwill

For the analysis of goodwill in the first quarter of 2020, we applied ASU 2017-04, which is further described in Note 1—General. If the carrying amount of a reporting unit exceeds its fair value, an impairment loss shall be recognized in an amount equal to that excess, limited to the total amount of goodwill allocated to that reporting unit. After recognizing the impairment loss, the loss establishes a new corresponding basis in the goodwill. Subsequent reversals of goodwill impairment losses are not permitted under applicable accounting standards.

Intelsat has only one reporting unit for purposes of the analysis of goodwill, and accordingly, the analysis is undertaken at the enterprise level. We determined the estimated fair value of our reporting unit using a discounted cash flow analysis, along with independent source data related to comparative market multiples and, when available, recent transactions, each of which is considered a Level 3 input within the fair value hierarchy under ASC 820. The discounted cash flows were derived from a five-year projection of cash flows plus a residual value, with the resulting projected cash flows discounted at an appropriate weighted average cost of capital.

In estimating the undiscounted cash flows, we primarily used our internally prepared budgets and forecast information. The key assumptions included in our model were projected growth rates, cost of capital, effective tax rates, and industry and economic trends, along with the C-band accelerated clearing incentive payments expected to be received subject to the achievement of certain milestones and the discount rate applied to those cash flows. A change in estimated future cash flows or other assumptions could change our estimated fair values and result in future impairments. The conclusion of our analysis was that the fair value of our reporting unit was greater than its carrying value, resulting in no impairment of goodwill.

The carrying amounts of goodwill consisted of the following (in thousands):

	As of December 31, 2019	As of September 30, 2020
Goodwill	\$ 6,780,827	\$ 6,780,827
Accumulated impairment losses	(4,160,200)	(4,160,200)
Net carrying amount	\$ 2,620,627	\$ 2,620,627



(b) Orbital Locations, Trade Name and Other Intangible Assets

Orbital Locations. Intelsat is authorized by governments to operate satellites at certain orbital locations—i.e., longitudinal coordinates along the Clarke Belt. The Clarke Belt is the part of space approximately 35,800 kilometers above the plane of the equator where geostationary orbit may be achieved. Various governments acquire rights to these orbital locations through filings made with the International Telecommunication Union, a sub-organization of the United Nations. We will continue to have rights to operate satellites at our orbital locations so long as we maintain our authorizations to do so.

Our rights to operate at orbital locations can be used and sold individually; however, since satellites and customers can be and are moved from one orbital location to another, our rights are used in conjunction with each other as a network that can be adapted to meet the changing needs of our customers and market demands. Due to the interchangeable nature of orbital locations, the aggregate value of all of the orbital locations is used to measure the extent of impairment, if any.

We determined the estimated fair value of our rights to operate at orbital locations by using the build-up method to determine cash flows for the income approach, with the resulting projected cash flows discounted at an appropriate weighted average cost of capital. Under the build-up approach, the amount a reasonable investor would be willing to pay for the right to operate a satellite business using orbital locations is calculated by first estimating the cash flows that typical market participants might assume could be available from the right to operate satellites using the subject location in a similar market. It is assumed that rather than acquiring such a business as a going concern, the buyer would hypothetically start with the right to operate satellites at orbital locations and build a new business with similar attributes from the beginning. Thus, the buyer is assumed to incur the start-up costs and losses typically associated with the going concern value and pay for all other tangible and intangible assets.

The key assumptions used in estimating the fair values of our rights to operate at our orbital locations included the following: (i) market penetration leading to revenue growth, (ii) profit margin, (iii) duration and profile of the build-up period, (iv) estimated start-up costs and losses incurred during the build-up period and (v) weighted average cost of capital.

We completed our analysis of the estimated fair value of our rights to operate at certain orbital locations in connection with the analysis of goodwill described above in the first quarter of 2020 and concluded that the fair value was greater than the carrying value, resulting in no impairment.

Trade Name. We have implemented the relief from royalty method to determine the estimated fair value of the Intelsat trade name. The relief from royalty analysis is comprised of two major steps: (i) a determination of the hypothetical royalty rate, and (ii) the subsequent application of the royalty rate to projected revenue. In determining the hypothetical royalty rate utilized in the relief from royalty approach, we considered comparable license agreements, an excess earnings analysis to determine aggregate intangible asset earnings, and other qualitative factors, each of which is considered a Level 3 input within the fair value hierarchy under ASC 820.

The key assumptions used in our model to estimate the fair value of the Intelsat trade name included forecasted revenues, the royalty rate, the tax rate and the discount rate. We completed our analysis of the estimated fair value of the Intelsat trade name in connection with the analysis of goodwill described above in the first quarter of 2020 and it resulted in an impairment of our trade name intangible asset of \$12.2 million, which is included within impairment of non-amortizable intangible and other assets in the condensed consolidated statements of operations.

The carrying amounts of acquired intangible assets not subject to amortization consisted of the following (in thousands):

	D	As of ecember 31, 2019	As of September 30, 2020
Orbital locations	\$	2,387,700	\$ 2,387,700
Trade name		65,200	 53,000
Total non-amortizable intangible assets	\$	2,452,900	\$ 2,440,700

Other Intangible Assets. The Company evaluated acquired intangible assets subject to amortization for potential impairment using internal projections of undiscounted cash flows expected to result from the use and eventual disposal of the assets. The key assumptions included in our model were projected growth rates, cost of capital, effective tax rates, and industry and economic trends. A change in estimated future cash flows or other assumptions could change our estimated fair values and result in future impairments. The conclusion of our analysis was that the undiscounted cash flows of the asset group were greater than its carrying value, resulting in no impairment.

The carrying amounts and accumulated amortization of acquired intangible assets subject to amortization consisted of the following (in thousands):

		As o	of December 31, 201	9			As o	f September 30, 202	0	
	 Gross Carrying Amount		Accumulated Amortization		Net Carrying Amount	 Gross Carrying Amount		Accumulated Amortization		Net Carrying Amount
Backlog and other	\$ 743,760	\$	(713,205)	\$	30,555	\$ 743,760	\$	(720,292)	\$	23,468
Customer relationships	534,030		(287,833)		246,197	534,030		(304,073)		229,957
Total	\$ 1,277,790	\$	(1,001,038)	\$	276,752	\$ 1,277,790	\$	(1,024,365)	\$	253,425

Intangible assets are amortized based on the expected pattern of consumption. Amortization expense was \$8.6 million and \$7.8 million for the three months ended September 30, 2019 and 2020, respectively, and \$25.8 million and \$23.3 million for the nine months ended September 30, 2019 and 2020, respectively.

Note 11 Debt

As discussed in Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters, the filing of the Chapter 11 Cases constituted an event of default that accelerated substantially all of our obligations under the documents governing the prepetition existing indebtedness of Intelsat S.A., Intelsat Luxembourg, Intelsat Connect and Intelsat Jackson. As such, we have reclassified all such debt obligations, other than debt subject to compromise, to current portion of long-term debt on our condensed consolidated balance sheet as of September 30, 2020. While the Chapter 11 Cases are pending, the Debtors do not anticipate making interest payments due under their respective unsecured debt instruments; however, the Debtors expect to make monthly interest payments on their senior secured debt instruments pursuant to the adequate protection requirements under the DIP Order.

The carrying values and fair values of our notes payable and debt were as follows (in thousands):

	As of December 31, 2019					As of Septer	nber :	30, 2020
	Ca	rrying Value		Fair Value	(Carrying Value		Fair Value
Intelsat S.A.:								
4.5% Convertible Senior Notes due June 2025 ⁽¹⁾	\$	402,500	\$	265,231	\$	402,500	\$	128,800
Unamortized prepaid debt issuance costs and discount on 4.5% Convertible Senior Notes		(133,310)		_		_		_
Total Intelsat S.A. obligations		269,190		265,231		402,500		128,800
Intelsat Luxembourg:								
7.75% Senior Notes due June 2021 ⁽¹⁾		421,219		336,975		421,219		14,743
Unamortized prepaid debt issuance costs on 7.75% Senior Notes		(1,257)		_		_		
8.125% Senior Notes due June 2023 ⁽¹⁾		1,000,000		590,000		1,000,000		35,000
Unamortized prepaid debt issuance costs on 8.125% Senior Notes		(5,838)		_				
12.5% Senior Notes due November 2024 ⁽¹⁾		403,350		277,152		403,350		50,536
Unamortized prepaid debt issuance costs and discount on 12.5% Senior Notes		(184,344)		_		_		_
Total Intelsat Luxembourg obligations		1,633,130		1,204,127		1,824,569		100,279
Intelsat Connect Finance:								
9.5% Senior Notes due February 2023 ⁽¹⁾		1,250,000		865,625		1,250,000		375,000
Unamortized prepaid debt issuance costs and discount on 9.5% Senior Notes		(27,741)				_		_
Total Intelsat Connect Finance obligations		1,222,259		865,625		1,250,000		375,000
Intelsat Jackson:		· · · ·						
9.5% Senior Secured Notes due September 2022		490,000		562,275		490,000		531,650
Unamortized prepaid debt issuance costs and discount on 9.5% Senior Secured Notes		(11,204)				(8,459)		
8% Senior Secured Notes due February 2024		1,349,678		1,380,046		1,349,678		1,368,236

	As of Decen	ıber 31, 2019	As of Septen	As of September 30, 2020			
	Carrying Value	Fair Value	Carrying Value	Fair Value			
Unamortized prepaid debt issuance costs and premium on 8% Senior Secured Notes	(3,903)	_	(3,286)	_			
5.5% Senior Notes due August 2023 ⁽¹⁾	1,985,000	1,687,250	1,985,000	1,230,700			
Unamortized prepaid debt issuance costs on 5.5% Senior Notes	(8,723)	_	_	—			
9.75% Senior Notes due July 2025 ⁽¹⁾	1,885,000	1,729,488	1,885,000	1,232,319			
Unamortized prepaid debt issuance costs on 9.75% Senior Notes	(20,487)	—	—	—			
8.5% Senior Notes due October 2024 ⁽¹⁾	2,950,000	2,669,750	2,950,000	1,902,750			
Unamortized prepaid debt issuance costs and premium on 8.5% Senior Notes	(12,916)	_	_	_			
Senior Secured Credit Facilities due November 2023	2,000,000	1,985,000	2,000,000	1,980,000			
Unamortized prepaid debt issuance costs and discount on Senior Secured Credit Facilities	(22,149)	_	(18,290)	_			
Senior Secured Credit Facilities due January 2024	395,000	398,950	395,000	391,050			
Unamortized prepaid debt issuance costs and discount on Senior Secured Credit Facilities	(1,600)	_	(1,331)	_			
6.625% Senior Secured Credit Facilities due January 2024	700,000	712,250	700,000	707,000			
Unamortized prepaid debt issuance costs and discount on Senior Secured Credit Facilities	(2,832)	_	(2,359)	_			
Superpriority Secured DIP Credit Facilities due July 2021	—	—	500,000	505,625			
Total Intelsat Jackson obligations	11,670,864	11,125,009	12,220,953	9,849,330			
Eliminations:							
8.125% Senior Notes of Intelsat Luxembourg due June 2023 owned by Intelsat Jackson ⁽¹⁾	(111,663)	(65,881)	(111,663)	(3,908)			
Unamortized prepaid debt issuance costs on 8.125% Senior Notes	652	—		—			
12.5% Senior Notes of Intelsat Luxembourg due November 2024 owned by Intelsat Connect, Intelsat Jackson and Intelsat Envision ⁽¹⁾	(403,245)	(277,080)	(403,245)	(50,522)			
Unamortized prepaid debt issuance costs and discount on 12.5% Senior Notes	184,296	_	_	_			
Total eliminations:	(329,960)	(342,961)	(514,908)	(54,430)			
Total Intelsat S.A. debt	14,465,483	13,117,031	15,183,114	10,398,979			
Less: current portion of long-term debt			5,400,953	5,483,561			
Less: debt included in liabilities subject to compromise			9,782,161	4,915,418			
Total Intelsat S.A. long-term debt	\$ 14,465,483	\$ 13,117,031	\$ _	\$ _			

(1) In connection with the Chapter 11 Cases, these balances have been reclassified as liabilities subject to compromise in our condensed consolidated balance sheet as of September 30, 2020. As of April 15, 2020, the Company ceased making principal and interest payments, and as of May 13, 2020 ceased accruing interest expense in relation to this long-term debt that was reclassified as liabilities subject to compromise. Further, \$197.0 million of debt discount, premium and issuance costs related to these notes was included within reorganization items in the condensed consolidated statements of operations for the nine months ended September 30, 2020.

The fair value for publicly traded instruments is determined using quoted market prices, and the fair value for non-publicly traded instruments is based upon composite pricing from a variety of sources, including market leading data providers, market makers, and leading brokerage firms. Substantially all of the inputs used to determine the fair value of our debt are classified as Level 1 inputs within the fair value hierarchy under ASC 820, except for our senior secured credit facilities and our 2025 Convertible Notes, the inputs for which are classified as Level 2.

Intelsat Jackson Superpriority Secured Debtor-in-Possession Term Loan Facility

On June 17, 2020 (the "Closing Date"), the DIP Debtors and DIP Lenders entered into the DIP Credit Agreement, a non-amortizing multiple draw superpriority secured debtor-in-possession term loan facility, in an aggregate principal amount of \$1.0 billion, on the terms and conditions set forth therein. See Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters.

Intelsat Jackson borrowed \$500.0 million of term loans under the DIP Facility on the Closing Date. Under the DIP Facility, Intelsat Jackson may, at its sole discretion, make incremental draws of the lesser of \$250.0 million and the remaining available commitments of the DIP Lenders. Drawn amounts under the DIP Facility bear interest at either (i) 4.50% per annum plus a base rate of the highest of (a) the Federal Funds Effective Rate plus ½ of 1.00%, (b) the Prime Rate as in effect on such day and (c) the London Inter-Bank Offered Rate ("LIBOR Rate") for a one-month interest period on such day (or if such day is not a business day, the immediately preceding business day) plus 1.00% or (ii) 5.50% plus the LIBOR Rate. For purposes of the DIP Facility, the LIBOR Rate has an effective floor rate of 1.0%. Undrawn amounts under the DIP Facility shall be subject to a ticking fee of 3.6% of the amount of commitments of the DIP Lenders from the entry of the DIP Order until such commitments terminate, which ticking fee shall be payable on the last day of each fiscal quarter prior to the date such commitments terminate and on the date of such termination. If an event of default under the DIP Facility occurs, the overdue amounts under the DIP Facility would bear interest at an additional 2.0% per annum above the interest rate otherwise applicable.

The proceeds of the DIP Facility may be used, among other things, to pay for (i) working capital needs of the DIP Debtors in the ordinary course of business, (ii) potential C-band relocation costs, (iii) investment and other general corporate purposes, and (iv) the costs and expenses of administering the Chapter 11 Cases. The maturity date of the DIP Facility is July 13, 2021, subject to certain extensions pursuant to the terms of the DIP Credit Agreement.

The DIP Credit Agreement includes usual and customary negative covenants for debtor-in-possession loan agreements of this type, including covenants limiting the Company's and its subsidiaries' ability to, among other things, incur additional indebtedness, create liens on assets, make investments, loans or advances, engage in mergers, consolidations, sales of assets and acquisitions, pay dividends and distributions and make payments in respect of junior or prepetition indebtedness, in each case subject to customary exceptions for debtor-in-possession loan agreements of this type.

The DIP Credit Agreement also includes certain customary representations and warranties, affirmative covenants and events of default, including, but not limited to, payment defaults, breaches of representations and warranties, covenant defaults, certain events under the Employee Retirement Income Security Act of 1974, as amended, and change of control. Certain bankruptcy-related events are also events of default, including, but not limited to, the dismissal by the Bankruptcy Court of any of the Chapter 11 Cases, the conversion of any of the Chapter 11 Cases to a case under Chapter 7 of the Bankruptcy Code and certain other events related to the impairment of the DIP Lenders' rights or liens granted under the DIP Credit Agreement.

On August 24, 2020, the DIP Debtors and DIP Lenders entered into DIP Amendment No. 1 to the DIP Credit Agreement, in connection with the Gogo Transaction (see Note 1—General—Gogo Inc. Transaction for additional information). DIP Amendment No. 1 was approved by the Bankruptcy Court on August 31, 2020.

The foregoing descriptions of the DIP Credit Agreement and DIP Amendment No. 1 do not purport to be complete and are qualified in their entirety by reference to the full text of the DIP Credit Agreement and DIP Amendment No. 1, as applicable.

Intelsat Jackson Senior Secured Credit Agreement and the Company and Certain of its Subsidiaries' Indentures

The commencement of the Chapter 11 Cases constituted an immediate event of default under Intelsat Jackson's secured credit agreement, dated as of January 12, 2011 (as amended, the "Intelsat Jackson Secured Credit Agreement"), as well as under the indentures governing certain of the Company and its subsidiaries' senior secured notes and senior notes, resulting in the automatic and immediate acceleration of substantially all of our outstanding debt. Any efforts to enforce payment obligations related to the acceleration of our debt have been automatically stayed as a result of the filing of the Chapter 11 Cases, and the creditors' rights of enforcement are subject to the applicable provisions of the Bankruptcy Code.

In addition, in April 2020, our LIBOR loans under the Intelsat Jackson Secured Credit Agreement were converted to Alternate Base Rate ("ABR") loans. We expect to pay interest on the floating rate term loans under the Intelsat Jackson Secured Credit Agreement at the rate applicable to ABR loans.

Note 12 Derivative Instruments and Hedging Activities

Interest Rate Cap Contracts

As of December 31, 2019 and September 30, 2020, we held interest rate cap contracts with an aggregate notional value of \$2.4 billion that mature in February 2021. These interest rate cap contracts, which were entered into in 2017 and amended in 2018, are designed to mitigate our risk of interest rate increases on the floating rate portion of our senior secured credit facilities (see Note 11—



Debt). The contracts have not been designated for hedge accounting treatment in accordance with ASC 815, *Derivatives and Hedging* ("ASC 815"), and the changes in fair value of these instruments, net of payments received, are recognized in the condensed consolidated statements of operations during the period of change. We received \$9.5 million in settlement payments related to the interest rate cap contracts for the nine months ended September 30, 2019 with no comparable amounts for the nine months ended September 30, 2020.

Preferred Stock Warrant and Common Stock Warrant

During 2017, we were issued a warrant to purchase preferred shares of one of our investments. We concluded that the warrant is a free-standing derivative in accordance with ASC 815. As of December 31, 2019 and September 30, 2020, the fair value of the preferred stock warrant was zero. During 2019, we were issued a warrant to purchase common shares of a separate investment. We concluded that the warrant is a free-standing derivative in accordance with ASC 815.

The following table sets forth the fair value of our derivatives by category (in thousands):

Derivatives not designated as hedging instruments	Classification		As o December 3		Sept	As of ember 30, 2020
Common stock warrant	Other assets	5	\$	3,239	\$	3,239
Interest rate cap contracts	Other assets			372		
Total derivatives			\$	3,611	\$	3,239

The following table sets forth the effect of the derivative instruments in our condensed consolidated statements of operations (in thousands):

Derivatives not designated as hedging instruments	Classification	ree Months Ended ptember 30, 2019	Three Months Ended September 30, 2020	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Interest rate cap contracts	Loss included in interest expense, net	\$ (1,625)	\$ (23)	\$ (21,104)	\$ (372)
Preferred stock warrant	Loss included in other income (expense), net	(247)	_	(4,100)	_
Total loss on derivative financial	instruments	\$ (1,872)	\$ (23)	\$ (25,204)	\$ (372)

Note 13 Income Taxes

In February 2018, the FASB issued ASU 2018-02, *Income Statement—Reporting Comprehensive Income (Topic 220)—Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income* ("ASU 2018-02"), which allows for an optional reclassification from accumulated other comprehensive income to retained earnings for stranded tax effects resulting from the U.S. Tax Cuts and Jobs Act (the "Act"), which was signed into law on December 22, 2017. Consequently, the amendments eliminated the stranded tax effects resulting from the Act for those entities that elect the optional reclassification. ASU 2018-02 is effective for all entities for interim and annual periods beginning after December 15, 2018. We adopted ASU 2018-02 in the first quarter of 2019, which resulted in a reclassification of stranded tax effects of \$16.2 million from accumulated other comprehensive loss to accumulated deficit.

The Act includes a number of provisions, including the lowering of the U.S. corporate tax rate from 35 percent to 21 percent, effective January 1, 2018. The Act limits our U.S. interest expense deductions to approximately 30 percent of EBITDA through December 31, 2021 and approximately 30 percent of earnings before net interest and taxes thereafter. The Act also introduced a new minimum tax, the Base Erosion Anti-Abuse Tax ("BEAT"). We are treating the BEAT as a period cost.

Effective January 1, 2019, the Luxembourg corporate tax rate decreased from 26.01% to 24.94%. This resulted in a decrease in deferred tax assets and corresponding valuation allowance.

The Company measures deferred tax assets and liabilities using enacted tax rates that will apply in the years in which the temporary differences are expected to be recovered or paid.

On July 2, 2018, we implemented a series of internal transactions and related steps that reorganized the ownership of certain assets among our subsidiaries (the "2018 Internal Reorganization"). The 2018 Internal Reorganization resulted in the majority of our operations being owned by a U.S.-based partnership, with certain of our wholly-owned Luxembourg and U.S. subsidiaries as partners.

In response to the COVID-19 pandemic, on March 18, 2020, the Families First Coronavirus Response Act (the "FFCR Act") was enacted, and on March 27, 2020, the Coronavirus Aid, Relief, and Economic Security Act (the "CARES Act") was enacted. The FFCR Act and the CARES Act contain numerous income tax provisions, such as increasing the 30 percent adjusted taxable income threshold to 50 percent for taxable years beginning in 2019 and 2020 for purposes of determining allowable business interest expense

deductions. The CARES Act repeals the 80 percent limitation for taxable years beginning before January 1, 2021 (enacted by the Act), and it further specifies that net operating losses arising in a taxable year beginning after December 31, 2017, and before January 1, 2021, are allowed as a carryback to each of the five taxable years preceding the taxable year of such losses. Modifications to the tax rules for the carryback of net operating losses and business interest limitations resulted in a federal tax refund of approximately \$11.4 million and \$26.9 million for the three and nine months ended September 30, 2020, respectively. In addition, the CARES Act includes refundable payroll tax credits and deferral of employment side social security payments. As of September 30, 2020, Intelsat's payroll deferral was approximately \$3.4 million.

The majority of our operations are located in taxable jurisdictions, including Luxembourg, the U.S. and the United Kingdom ("UK"). Due to our cumulative losses in recent years, and the inherent uncertainty associated with the realization of taxable income in the foreseeable future, we recorded a full valuation allowance against the cumulative net operating losses generated in Luxembourg. The difference between tax expense (benefit) reported in the condensed consolidated statements of operations and tax computed at statutory rates is attributable to the valuation allowance on losses generated in Luxembourg, the provision for foreign taxes, which were principally in the U.S. as a result of the final BEAT regulations and the UK, as well as withholding taxes on revenue earned in some of the foreign markets in which we operate, offset by the tax benefit resulting from the impacts of the CARES Act and a tax reserve established on certain tax benefits taken with respect the final BEAT regulations.

As of December 31, 2019 and September 30, 2020, our gross unrecognized tax benefits were \$25.0 million and \$42.4 million, respectively (including interest and penalties), of which \$21.5 million and \$38.5 million, respectively, if recognized, would affect our effective tax rate. As of December 31, 2019 and September 30, 2020, we had recorded reserves for interest and penalties in the amount of \$0.6 million and \$0.7 million, respectively. We continue to recognize interest and, to the extent applicable, penalties with respect to the unrecognized tax benefits as income tax expense. Since December 31, 2019, the change in the balance of unrecognized tax benefits has consisted of an increase of \$7.0 million related to current tax positions, and a decrease of \$2.5 million related to the expiration of a statute of limitations on the assessment of certain taxes.

We operate in various taxable jurisdictions throughout the world, and our tax returns are subject to audit and review from time to time. We consider Luxembourg, the U.S., the UK and Brazil to be our significant tax jurisdictions. Our subsidiaries in these jurisdictions are subject to income tax examination for periods after December 31, 2014. We believe that there are no jurisdictions in which the outcome of unresolved tax issues or claims is likely to be material to our results of operations, financial position or cash flows within the next twelve months.

During 2019, the Company made payments to the government of India in the amount of \$7.0 million with respect to ongoing administrative proceedings. We believe it is more likely than not that the positions which we have presented in these matters will result in a favorable outcome for the Company. As a result, the payments have been recorded in taxes receivable.

Effective January 31, 2020, the UK formally exited the European Union ("EU"). As a result of the withdrawal, existing tax reliefs and exemptions on intra-European transactions will likely cease to apply to transactions between UK entities and EU entities. In addition, transactions with non-EU countries, such as the U.S., may also be affected. As of September 30, 2020, all relevant tax laws and treaties remained unchanged and the tax consequences were unknown. Therefore, we have not recognized any impacts of the withdrawal in the income tax provision as of September 30, 2020. We will recognize any impacts to the tax provision when changes in tax laws or treaties between the UK and the EU or individual EU member states are enacted.

On December 2, 2019, the U.S. Department of Treasury and the U.S. Internal Revenue Service released final regulations with respect to BEAT as enacted by the 2017 Tax Reform Act. These regulations represent the final version of proposed regulations which were released in December 2018. The BEAT is a minimum tax established by the Act that subjects certain payments made by U.S. corporations or subsidiaries to foreign related parties to a secondary federal income tax regime in the U.S. The final regulations clarify which taxpayers are subject to the BEAT and how the BEAT rules apply to certain payments and transactions. We have adopted the final BEAT regulations as of the release date. These regulations are effective for the Company as of its tax year ended December 31, 2018. A second set of final regulations was issued on September 1, 2020, addressing among other topics, the application of the BEAT to partnerships and the application of the effectively connected income exception to depreciable or amortizable property contributed to a U.S. partnership by a foreign partner. Similar to the first set of final regulations issued in December 2019, the second set of final regulations are effective for the Company as of its tax year ended December 31, 2018. As of September 30, 2020, the Company recognized the BEAT tax impact associated with the second set of final regulations related to its tax years ended December 31, 2018 and 2019, and as of September 30, 2020.

Note 14 Commitments and Contingencies

On May 13, 2020, Intelsat S.A. and certain of its subsidiaries filed voluntary petitions for relief under title 11 of the Bankruptcy Code in the Bankruptcy Court. As a result of such bankruptcy filings, substantially all proceedings pending against the Debtors have been stayed and prepetition liabilities are subject to compromise. See Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters.

In the absence of the automatic stay due to the Chapter 11 Cases, we are subject to litigation in the ordinary course of business. Other than as disclosed below, management does not believe that the resolution of any of those pending proceedings would have a material adverse effect on our financial position or results of operations.

SES Claim

On July 14, 2020, SES Americom, Inc. ("SES") filed a proof of claim in the Bankruptcy Court in the amount of \$1.8 billion against each of the Debtors. SES asserts that the Debtors owe money (or will owe money) to SES pursuant to certain contractual and fiduciary obligations made in the context of the consortium agreement between Debtor Intelsat US LLC, SES, and other satellite operators (the "Consortium Agreement"). SES believes it is entitled to 50% of the combined payments that may eventually be payable to the Debtors and SES pursuant to the FCC Final Order, which provides for Acceleration Payments subject to the satisfaction of certain deadlines and other conditions set forth therein. SES's proof of claim alleges that the Debtors breached the Consortium. The proof of claim alleges breach of fiduciary duties and unjust enrichment and seeks monetary and punitive damages. We dispute the allegations in the proof of claim and on October 19, 2020, filed an objection to the claim, which we intend to litigate vigorously. To the extent that any portion of SES's claim is allowed, we have asked the Bankruptcy Court to 'equitably subordinate' such claim based on SES's conduct in matters related to the Consortium Agreement. While the ultimate resolution of the claim is not currently predictable, if there is an adverse ruling, the ruling could constitute a material adverse outcome on our future consolidated financial condition.

Note 15 Related Party Transactions

(a) Shareholders' Agreements

Certain shareholders of Intelsat S.A. entered into a shareholders' agreement in December 2018, which provides, among other things, specific rights to and limitations upon the holders of Intelsat S.A.'s share capital with respect to shares held by such holders.

(b) Governance Agreement

In December 2018, the Company entered into a governance agreement with its shareholder affiliated with Serafina S.A. The agreement contains provisions relating to the composition of the Company's board of directors and certain other matters.

(c) Indemnification Agreements

We have entered into agreements with our executive officers and directors to provide contractual indemnification in addition to the indemnification provided for in our articles of incorporation.

(d) Horizons Holdings

We have a 50% ownership interest in Horizons Holdings as a result of a joint venture with JSAT (see Note 9(a)-Investments-Horizons Holdings).

(e) Horizons-3 Satellite LLC

We have a 50% ownership interest in Horizons 3 as a result of a joint venture with JSAT (see Note 9(b)-Investments-Horizons-3 Satellite LLC).

Note 16 Condensed Combined Debtors' Financial Information

The following presents our Debtors' condensed combined balance sheet as of September 30, 2020, statements of operations for the three and nine months ended September 30, 2020 and statement of cash flows for the nine months ended September 30, 2020. Consolidating adjustments include eliminations of the following:

- investments in subsidiaries;
- intercompany accounts;
- · intercompany sales and expenses; and
- intercompany equity balances.

Intercompany balances with non-Debtor affiliates have not been eliminated. On the Debtors' condensed combined balance sheet, these primarily consist of net intercompany trade receivables generated under our Master Intercompany Service Agreement ("MISA")



and funding for the operations of non-Debtor affiliates. On the Debtors' condensed combined statements of operations, total reported revenue includes intercompany revenue of \$106.0 million and \$228.6 million for the three and nine months ended September 30, 2020, respectively, primarily consisting of satellite capacity charges and revenue recognized pursuant to intercompany agreements between certain Debtor entities and a newly-formed non-Debtor entity relating to the management of certain reorganization-related costs. Cost from affiliates primarily relates to sales and technical support services provided to Debtors as specified under the MISA. Investments in non-Debtor affiliates are presented under the equity method of accounting in the condensed combined financial statements set forth below.

DEBTORS' UNAUDITED CONDENSED COMBINED BALANCE SHEET

(in thousands, except per share amounts)

September 30, 2020

	Se	ptember 30, 2020
ASSETS		
Current assets:		
Cash and cash equivalents	\$	880,185
Restricted cash		19,091
Receivables, net of allowance of \$24,884		146,096
Contract assets		30,387
Prepaid expenses and other current assets		100,815
Intercompany receivables		187,613
Total current assets		1,364,187
Satellites and other property and equipment, net		4,631,470
Goodwill		2,620,627
Non-amortizable intangible assets		2,440,700
Amortizable intangible assets, net		253,425
Contract assets, net of current portion		40,148
Investment in affiliates		168,251
Other assets		411,348
Total assets	\$	11,930,156
LIABILITIES AND SHAREHOLDERS' DEFICIT	<u>*</u>	;; - •; •
Current liabilities:		
Accounts payable and accrued liabilities	\$	163,538
Taxes payable	Ψ	8,775
Employee related liabilities		31,047
Accrued interest payable		13,837
Current portion of long-term debt		5,400,953
Contract liabilities		125,415
Deferred satellite performance incentives		42,337
Other current liabilities		53,075
Total current liabilities		5,838,977
Contract liabilities, net of current portion		1,042,850
Deferred satellite performance incentives, net of current portion		146,559
Deferred income taxes		63,646
Accrued retirement benefits, net of current portion		113,258
Other long-term liabilities		115,258
Liabilities subject to compromise		10,170,344
Shareholders' deficit:		10,170,544
Common shares, nominal value \$0.01 per share		1,421
Paid-in capital		2,572,324
Accumulated deficit		(8,144,803)
Accumulated other comprehensive loss		(8,144,803) (61,191)
Total shareholders' deficit		(5,632,249)
	¢	
Total liabilities and shareholders' deficit	<u>\$</u>	11,930,156

DEBTORS' UNAUDITED CONDENSED COMBINED STATEMENTS OF OPERATIONS

(in thousands)

	Months Ended nber 30, 2020	ine Months Ended eptember 30, 2020
Revenue	\$ 473,556	\$ 1,314,307
Operating expenses:		
Direct costs of revenue (excluding depreciation and amortization)	69,189	200,538
Selling, general and administrative	57,659	181,931
Cost from affiliates	9,785	32,047
Depreciation and amortization	157,442	472,610
Impairment of non-amortizable intangible and other assets	—	46,243
Total operating expenses	 294,075	 933,369
Income from operations	 179,481	 380,938
Interest expense, net	137,444	677,124
Equity in loss of affiliates	(40,791)	(41,260)
Other income, net	2,973	15,007
Reorganization items	(36,367)	(335,059)
Loss before income taxes	 (32,148)	 (657,498)
Benefit from income taxes	(16,217)	(17,441)
Net loss	\$ (15,931)	\$ (640,057)

DEBTORS' UNAUDITED CONDENSED COMBINED STATEMENT OF CASH FLOWS

(in thousands)

Cash flows from an anti-itica		Months Ended Tember 30, 2020
Cash flows from operating activities: Net loss	\$	(640,057)
Adjustments to reconcile net loss to net cash provided by operating activities:	Э	(040,037)
Depreciation and amortization		472,610
Provision for doubtful accounts		34,465
Foreign currency transaction gain		(801)
Impairment of non-amortizable intangible and other assets		46,243
Share-based compensation		9,031
Deferred income taxes		5,470
Amortization of discount, premium, issuance costs and related costs		19,689
Non-cash reorganization items		19,089
Debtor-in-possession financing fees		52,182
Amortization of actuarial loss and prior service credits for retirement benefits Unrealized losses on derivative financial instruments		1,976
		372
Unrealized losses on investments and loans held-for-investment		721
Equity in loss of affiliates		41,260
Other non-cash items		(7
Changes in operating assets and liabilities:		10.041
Receivables		12,841
Intercompany receivables		(105,048
Prepaid expenses, contract and other assets		(81,814
Accounts payable and accrued liabilities		69,812
Accrued interest payable		48,713
Contract liabilities		(66,304
Accrued retirement benefits		(12,253
Other long-term liabilities		2,801
Net cash provided by operating activities		108,876
Cash flows from investing activities:		
Payments for satellites and other property and equipment (including capitalized interest)		(414,610
Dividends from affiliates		28,960
Proceeds from loans held-for-investment		973
Capital contribution to affiliates		(9,005
Other proceeds from satellites		5,625
Net cash used in investing activities		(388,057
Cash flows from financing activities:		
Proceeds from debtor-in-possession financing		500,000
Debtor-in-possession financing fees		(52,182)
Principal payments on deferred satellite performance incentives		(25,428
Net cash provided by financing activities		422,390
Effect of exchange rate changes on cash, cash equivalents and restricted cash		754
Net change in cash, cash equivalents and restricted cash		143,963
Cash, cash equivalents, and restricted cash, beginning of period		755,313
Cash, cash equivalents, and restricted cash, end of period	\$	899,276
	ψ	077,270
Reconciliation of cash, cash equivalents and restricted cash reported within the condensed consolidated Debtors' balance sheet to the total sum of these same amounts shown on the condensed consolidated Debtors' statement of cash flows:		
Cash and cash equivalents	\$	880,185
Restricted cash		19,091
Total cash, cash equivalents and restricted cash reported in the condensed consolidated Debtors' statement of cash flows	\$	899,276
		5,270

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in conjunction with our unaudited condensed consolidated financial statements and their notes included in Item 1—Financial Statements in this Quarterly Report, and with our audited consolidated financial statements and their notes included in our Annual Report on Form 10-K for the year ended December 31, 2019. In addition, see "Forward-Looking Statements" and Item 1A—Risk Factors for a discussion of factors that could cause our future financial condition and results of operations to be materially different from those discussed below.

Overview

We operate one of the world's largest satellite services businesses, providing a critical layer in the global communications infrastructure.

We provide diversified communications services to the world's leading media companies, fixed and wireless telecommunications operators, data networking service providers for enterprise and mobile applications in the air and on the seas, multinational corporations and internet service providers. We are also the leading provider of commercial satellite capacity to the United States ("U.S.") government and other select military organizations and their contractors.

Our customers use our global network for a broad range of applications, from global distribution of content for media companies to providing the transmission layer for commercial aeronautical consumer broadband connectivity, to enabling essential network backbones for telecommunications providers in high-growth emerging regions.

Our network solutions are a critical component of our customers' infrastructures and business models. Generally, our customers need the specialized connectivity that satellites provide so long as they are in business or pursuing their mission. In recent years, mobility services providers have contracted for services on our fleet that support broadband connections for passengers on commercial flights and cruise ships, connectivity that in some cases is only available through our network. In addition, our satellite neighborhoods provide our media customers with efficient and reliable broadcast distribution that maximizes audience reach, a technical and economic benefit that is difficult for terrestrial services to match. In developing regions, our satellite solutions often provide higher reliability than is available from local terrestrial telecommunications services and allow our customers to reach geographies that they would otherwise be unable to serve.

Recent Developments

Voluntary Reorganization under Chapter 11

On May 13, 2020, Intelsat S.A. and certain of its subsidiaries (each, a "Debtor" and collectively, the "Debtors") commenced voluntary cases (the "Chapter 11 Cases") under title 11 of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the Eastern District of Virginia (the "Bankruptcy Court"). Primary factors causing us to file for Chapter 11 protection included the Company's intention to participate in the accelerated clearing process of C-band spectrum set forth in the U.S. Federal Communications Commission's ("FCC") March 3, 2020 final order (the "FCC Final Order"), requiring the Company to incur significant costs related to clearing activities well in advance of receiving reimbursement for such costs and the need for additional financing to fund the C-band clearing process, service our current debt obligations, and meet our operating requirements, as well as the economic slowdown impacting the Company and several of its end markets due to the novel coronavirus ("COVID-19") pandemic.

On August 14, 2020, the Company filed its final C-band spectrum transition plan with the FCC. The FCC Final Order provides for monetary enticements for fixed satellite services ("FSS") providers to clear a portion of the C-band spectrum on an accelerated basis (the "Acceleration Payments"). On September 17, 2020, the Company announced it finalized all of its required contracts with satellite manufacturers and launch-vehicle providers to move forward and meet the accelerated C-band spectrum clearing timelines established by the FCC. Under the FCC Final Order, the Company is eligible to receive Acceleration Payments of approximately \$1.2 billion and \$3.7 billion on December 5, 2021 and December 5, 2023, respectively, subject to the satisfaction of certain deadlines and other conditions set forth therein.

The Chapter 11 process can be unpredictable and involves significant risks and uncertainties. As a result of these risks and uncertainties, the amount and composition of the Company's assets, liabilities, officers and/or directors could be significantly different following the outcome of the Chapter 11 Cases, and the description of the Company's operations, properties and liquidity and capital resources included in this Quarterly Report may not accurately reflect its operations, properties and liquidity and capital resources following the Chapter 11 process.

Pursuant to various orders from the Bankruptcy Court, the Debtors have received approval from the Bankruptcy Court to generally maintain their ordinary course operations and uphold certain commitments to their stakeholders, including employees, customers, and vendors during the restructuring process, subject to the jurisdiction of the Bankruptcy Court and in accordance with the applicable provisions of the Bankruptcy Code. While the Chapter 11 Cases are pending, the Debtors do not anticipate making interest payments due under their respective unsecured debt instruments; however, the Debtors expect to make interest payments on a monthly basis to holders of their senior secured debt instruments. For the three and nine months ended September 30, 2020, the contractual



interest expense pursuant to our unsecured debt instruments that was not recognized in our condensed consolidated statements of operations was \$196.4 million and \$298.9 million, respectively.

The filing of the Chapter 11 Cases constituted an event of default that accelerated substantially all of our obligations under the documents governing the prepetition existing indebtedness of Intelsat S.A., Intelsat Luxembourg, Intelsat Connect and Intelsat Jackson. For additional discussion regarding the impact of the Chapter 11 Cases on our debt obligations, see Item 1, Note 11—Debt.

On June 9, 2020, Intelsat Jackson received approval from the Bankruptcy Court (the "DIP Order") to enter into a non-amortizing multiple draw superpriority secured debtor-in-possession term loan facility (the "DIP Facility"), in an aggregate principal amount of \$1.0 billion on the terms and conditions as set forth in the DIP Facility credit agreement (the "DIP Credit Agreement") with certain of the Debtors' prepetition secured parties (the "DIP Lenders"), and on June 17, 2020, Intelsat Jackson and certain of its subsidiaries as guarantors (together with Intelsat Jackson, the "DIP Debtors") entered into the DIP Credit Agreement with the DIP Lenders, as amended by an amendment ("DIP Amendment No. 1") to the DIP Credit Agreement, dated as of August 24, 2020. DIP Amendment No. 1 was approved by the Bankruptcy Court on August 31, 2020. For additional information regarding the DIP Facility, DIP Credit Agreement and DIP Amendment No. 1, see Liquidity and Capital Resources—*Debt* below.

On July 11, 2020, the Debtors filed with the Bankruptcy Court schedules and statements setting forth, among other things, the assets and liabilities of each of the Debtors, subject to the assumptions filed in connection therewith. These schedules and statements may be subject to further amendment or modification after filing.

Update on the Impact of COVID-19 on the Company

The COVID-19 pandemic has had an adverse impact on our business, results of operations and financial condition, a trend we expect to continue. Among the impacts of the COVID-19 pandemic were a reduction of revenue and a decreased likelihood of collection from certain mobility customers. We continue to closely monitor the ongoing impact on our employees, customers, business and results of operations.

Gogo Inc. Transaction

On August 31, 2020, following approval from the Bankruptcy Court, Intelsat Jackson and Gogo Inc., a Delaware corporation ("Gogo"), entered into a purchase and sale agreement (the "Purchase and Sale Agreement") with respect to Gogo's commercial aviation business for \$400.0 million in cash, subject to customary adjustments (the "Purchase Price"). The transaction further propels the Company's efforts in the growing commercial in-flight connectivity market, pairing our high-capacity global satellite and ground network with Gogo's installed base of more than 3,000 commercial aircraft to redefine the connectivity experience. In connection with the transactions contemplated by the Purchase and Sale Agreement (the "Gogo Transaction"), the DIP Debtors and DIP Lenders entered into DIP Amendment No. 1 to our DIP Credit Agreement (see *—Voluntary Reorganization under Chapter 11* above). We expect to fund the Purchase Price with proceeds from our existing DIP Facility and cash on hand. The Gogo Transaction is expected to close before the end of the first quarter of 2021, subject to the receipt of certain regulatory approvals and the satisfaction or waiver of certain other customary closing conditions.

A. Results of Operations

Three Months Ended September 30, 2019 and 2020

The following table sets forth our comparative statements of operations for the periods shown with the increase (decrease) and percentage changes, except those deemed not meaningful ("NM"), between the periods presented (in thousands, except percentages):

	Three Months Ended September 30, 2019		Three Months Ender September 30, 2020		Increase (Decrease)	Percentage Change
Revenue	\$	506,658	\$ 489,44	19 \$	\$ (17,209)	(3)%
Operating expenses:						
Direct costs of revenue (excluding depreciation and amortization)		104,743	120,20	57	15,524	15%
Selling, general and administrative		60,750	69,21	15	8,465	14%
Depreciation and amortization		161,536	162,57	73	1,037	1%
Total operating expenses		327,029	352,05	55	25,026	8%
Income from operations		179,629	137,39	94	(42,235)	(24)%
Interest expense, net		315,964	138,07	75	(177,889)	(56)%
Other income (expense), net		(5,115)	3,00	57	8,182	NM
Reorganization items			(36,36	57)	(36,367)	NM
Loss before income taxes	(141,450)	(33,98	31)	107,469	(76)%
Provision for (benefit from) income taxes		6,248	(18,65	50)	(24,898)	NM
Net loss	(147,698)	(15,33	51)	132,367	(90)%
Net income attributable to noncontrolling interest		(594)	(60)0)	(6)	1%
Net loss attributable to Intelsat S.A.	\$ (148,292)	\$ (15,93	\$1) \$	\$ 132,361	(89)%

Revenue

We earn revenue primarily by providing services to our customers using our satellite transponder capacity. Our customers generally obtain satellite capacity from us by placing an order pursuant to one of several master customer service agreements. On-network services are comprised primarily of services delivered on our owned network infrastructure, as well as commitments for third-party capacity, generally long-term in nature, which we integrate and market as part of our owned infrastructure. In the case of third-party services in support of government applications, the commitments for third-party capacity are shorter and matched to the government contracting period, and thus remain classified as off-network services. Off-network services can include transponder services and other satellite-based transmission services, such as mobile satellite services ("MSS"), which are sourced from other operators, often in frequencies not available on our network. Under the category Off-Network and Other Revenues, we also include revenues from consulting and other services.

The following table sets forth our comparative revenue by service type, with Off-Network and Other Revenues shown separately from On-Network Revenues, for the periods below (in thousands, except percentages):

	ree Months Ended ptember 30, 2019	Three Months Ended September 30, 2020		Increase (Decrease)	Percentage Change
On-Network Revenues					
Transponder services	\$ 364,312	\$	353,758	\$ (10,554)	(3)%
Managed services	93,080		69,438	(23,642)	(25)%
Channel	601		280	(321)	(53)%
Total on-network revenues	457,993		423,476	 (34,517)	(8)%
Off-Network and Other Revenues					
Transponder, MSS and other off-network services	39,129		54,478	15,349	39%
Satellite-related services	9,536		11,495	1,959	21%
Total off-network and other revenues	48,665		65,973	17,308	36%
Total	\$ 506,658	\$	489,449	\$ (17,209)	(3)%

Total revenue for the three months ended September 30, 2020 decreased by \$17.2 million, or 3%, as compared to the three months ended September 30, 2019. By service type, our revenues increased or decreased due to the following:

On-Network Revenues:

• *Transponder services*—an aggregate decrease of \$10.6 million, primarily due to an \$8.9 million decrease in revenue from media customers and a \$2.4 million decrease from government customers, partially offset by a \$0.8 million increase from network services customers. The decline in revenue from media customers was mainly driven by non-renewals,

renewals at lower pricing, service contractions, and early terminations, as well as the recognition of deferred revenue in the prior period for certain prepaid capacity and service contracts that terminated in 2019 for which there are no comparable amounts in 2020. The decline from government customers was primarily driven by non-renewals, pricing compression, and the transfer of services to off-network capacity. The increase in revenue from network services customers was primarily driven by a renegotiated contract with a maritime mobility customer for which revenues were previously classified as managed services, and new wireless infrastructure services in the Asia-Pacific region. This increase was partially offset by non-renewals, renewals at lower pricing, service contractions, and services transferred to off-network services.

Managed services—an aggregate decrease of \$23.6 million, primarily due to a \$14.0 million decrease in revenue from network services customers and a \$10.5 million decrease in revenue from media customers, partially offset by a \$0.9 million increase in revenue from government customers. The decrease in revenue from network services customers was primarily driven by the transfer of revenue recognition from managed services to transponder services, largely in connection with the renegotiated contract with a maritime mobility customer referenced above, as well as credits provided to customers related to COVID-19. These declines were partially offset by increases in revenue due to new trunking applications and new FlexAir services. The decline in revenue from media customers was mainly related to the termination of managed video service contracts in 2019 and a decrease in occasional use video services. The increase in revenue from government customers was mainly attributable to new FlexGround services.

Off-Network and Other Revenues:

- Transponder, MSS and other off-network services—an aggregate increase of \$15.3 million, primarily due to a \$12.8 million increase in
 revenue from government customers, largely driven by the transfer of certain services from on-network to off-network and the sale of
 customer premise equipment.
- Satellite-related services—an aggregate of increase of \$2.0 million, largely related to professional fees recognized in connection with Mission Extension Vehicle-1 ("MEV-1") and Mission Extension Vehicle-2 ("MEV-2").

Operating Expenses

Direct Costs of Revenue (Excluding Depreciation and Amortization)

Direct costs of revenue increased by \$15.5 million, or 15%, to \$120.3 million for the three months ended September 30, 2020, as compared to the three months ended September 30, 2019. The increase was primarily due to a \$13.4 million increase in equipment costs largely incurred in connection with government customers and a \$5.3 million increase in staff-related expenses largely relating to our employee retention incentive plans, partially offset by a \$2.7 million decrease in third-party managed capacity costs largely related to a government customer.

Selling, General and Administrative

Selling, general and administrative expenses increased by \$8.5 million, or 14%, to \$69.2 million for the three months ended September 30, 2020, as compared to the three months ended September 30, 2019. The increase was primarily due to a \$7.2 million increase in professional fees and a \$4.5 million increase in staff-related expenses largely relating to our employee retention incentive plans, partially offset by a \$4.2 million decrease in costs for licenses.

Depreciation and Amortization

Depreciation and amortization expense increased by \$1.0 million, or 1%, to \$162.6 million for the three months ended September 30, 2020, as compared to the three months ended September 30, 2019. Significant items impacting depreciation and amortization included an increase of \$4.7 million in depreciation expense resulting from the impact of a satellite placed in service, partially offset by a decrease of \$3.1 million in depreciation expense due to the timing of a satellite becoming fully depreciated and a decrease of \$0.7 million in amortization expenses.

Interest Expense, Net

Interest expense, net consists of gross interest expense incurred together with gains and losses on interest rate cap contracts (which reflect the change in their fair value), offset by interest income earned and interest capitalized related to assets under construction. As of September 30, 2020, we held interest rate cap contracts with an aggregate notional amount of \$2.4 billion to mitigate the risk of interest rate increases on the floating-rate term loans under our senior secured credit facilities. The contracts have not been designated as hedges for accounting purposes.

Interest expense, net decreased by \$177.9 million, or 56%, to \$138.1 million for the three months ended September 30, 2020, as compared to the three months ended September 30, 2019, primarily due to a decrease of \$176.3 million in interest expense largely

resulting from Chapter 11 restructuring activities, partially offset by an increase in interest expense recognized on our senior secured credit facilities.

The non-cash portion of total interest expense, net was \$29.7 million for the three months ended September 30, 2020, primarily consisting of interest expense related to the significant financing component identified in customer contracts, amortization and accretion of discounts and premiums and amortization of deferred financing fees.

Other Income (Expense), Net

Other income, net was \$3.1 million for the three months ended September 30, 2020, as compared to other expense, net of \$5.1 million for the three months ended September 30, 2019. The net increase in other income primarily consisted of lower foreign currency losses of \$4.1 million largely related to our business conducted in Brazilian *reais* and a \$3.6 million net loss due to a change in value of certain investments in third parties for the three months ended September 30, 2019, with no similar activity in 2020.

Reorganization Items

Reorganization items reflect direct costs incurred in connection with the Chapter 11 Cases. Reorganization items of \$36.4 million for the three months ended September 30, 2020 primarily consisted of professional fees. There were no comparable amounts for the three months ended September 30, 2019.

Provision for (Benefit from) Income Taxes

Income tax expense decreased by \$24.9 million to income tax benefit of \$18.7 million for the three months ended September 30, 2020, as compared to the three months ended September 30, 2019. The decrease was principally attributable to a net benefit recognized for the Base Erosion Anti-Abuse Tax ("BEAT"), additional benefit from the Coronavirus Aid, Relief, and Economic Security Act (the "CARES Act") with regards to the relaxed limitations on the deductibility of interest, the use of net operating losses arising in taxable years beginning after 2018 and lower income from our U.S. subsidiaries for the three months ended September 30, 2020.

Cash paid for income taxes, net of refunds, totaled \$2.5 million and \$2.1 million for the three months ended September 30, 2019 and 2020, respectively.

Net Loss Attributable to Intelsat S.A.

Net loss attributable to Intelsat S.A. was \$15.9 million for the three months ended September 30, 2020, as compared to net loss attributable to Intelsat S.A. of \$148.3 million for the three months ended September 30, 2019. The change reflects the various items discussed above.

Nine Months Ended September 30, 2019 and 2020

The following table sets forth our comparative statements of operations for the periods shown with the increase (decrease) and percentage changes, except those deemed not meaningful ("NM"), between the periods presented (in thousands, except percentages):

	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020	Increase (Decrease)	Percentage Change
Revenue	\$ 1,544,514	\$ 1,430,303	\$ (114,211)	(7)%
Operating expenses:				
Direct costs of revenue (excluding depreciation and amortization)	305,595	331,771	26,176	9%
Selling, general and administrative	168,263	214,586	46,323	28%
Depreciation and amortization	496,438	488,235	(8,203)	(2)%
Satellite impairment loss	381,565	—	(381,565)	NM
Impairment of non-amortizable intangible and other assets		46,243	46,243	NM
Total operating expenses	1,351,861	1,080,835	(271,026)	(20)%
Income from operations	192,653	349,468	156,815	81%
Interest expense, net	953,246	678,937	(274,309)	(29)%
Other income (expense), net	(32,374)	8,564	40,938	NM
Reorganization items	—	(335,059)	(335,059)	NM
Loss before income taxes	(792,967)	(655,964)	137,003	(17)%
Provision for (benefit from) income taxes	3,884	(17,691)	(21,575)	NM
Net loss	(796,851)	(638,273)	158,578	(20)%
Net income attributable to noncontrolling interest	(1,785)	(1,784)	1	%
Net loss attributable to Intelsat S.A.	\$ (798,636)	\$ (640,057)	\$ 158,579	(20)%

Revenue

The following table sets forth our comparative revenue by service type, with Off-Network and Other Revenues shown separately from On-Network Revenues, for the periods shown (in thousands, except percentages):

	Nine Months Ended September 30, 2019		Nine Months Ended September 30, 2020	Increase (Decrease)	Percentage Change
On-Network Revenues					
Transponder services	\$ 1,111,182	\$	1,028,692	\$ (82,490)	(7)%
Managed services	277,616		226,749	(50,867)	(18)%
Channel	1,877		1,096	(781)	(42)%
Total on-network revenues	 1,390,675		1,256,537	 (134,138)	(10)%
Off-Network and Other Revenues					
Transponder, MSS and other off-network services	126,704		141,783	15,079	12%
Satellite-related services	27,135		31,983	4,848	18%
Total off-network and other revenues	 153,839		173,766	 19,927	13%
Total	\$ 1,544,514	\$	1,430,303	\$ (114,211)	(7)%

Total revenue for the nine months ended September 30, 2020 decreased by \$114.2 million, or 7%, as compared to the nine months ended September 30, 2019. By service type, our revenues increased or decreased due to the following:

On-Network Revenues:

• *Transponder services*—an aggregate decrease of \$82.5 million, primarily due to a \$43.3 million decrease from network services customers, a \$32.6 million decrease in revenue from media customers, and a \$6.6 million decrease in revenue from government customers. The decline from network services customers was primarily due to non-renewals, renewals at lower pricing and service contractions for enterprise, maritime and aero mobility, and wireless infrastructure application customers, and lost revenue resulting from the loss of Intelsat 29e, a portion of which services were restored

with off-network services. These losses were partially offset by new business from enterprise and wireless infrastructure application customers. The decrease in revenue from media customers was primarily due to non-renewals, renewals at lower pricing, service contractions and early terminations largely relating to distribution services, as well as the recognition of deferred revenue in the prior period for certain prepaid capacity and service contracts that terminated in 2019 for which there are no comparable amounts in 2020. The decrease in revenue from government customers was primarily related to non-renewals, renewals at lower pricing, and services moved to off-network services.

Managed services—an aggregate decrease of \$50.9 million, primarily due to a \$27.6 million decrease in revenue from media customers and a \$23.2 million decrease from network services customers. The decline in revenue from media customers was primarily due to a decrease in revenue from managed video services mainly related to an early termination of a contract in 2019 and a decrease from occasional use video services. The decline in revenue from network services customers primarily related to maritime and aero mobility customers due to non-renewals, a renegotiated contract with a maritime mobility customer, and lost revenue resulting from the loss of Intelsat 29e, a portion of which services were restored with off-network services. The decline in revenue from network services customers was partially offset by new Flex business from maritime mobility customers and new business from trunking applications.

Off-Network and Other Revenues:

- *Transponder, MSS and other off-network services*—an aggregate increase of \$15.1 million, primarily due to a \$22.5 million increase in revenue from government customers largely driven by the transfer of certain services from on-network to off-network and the sale of customer premise equipment, and a \$6.5 million increase in revenue due to the transfer of certain Intelsat 29e network services customers to off-network capacity. These increases were partially offset by a \$14.3 million decrease due to revenue recognized during the first quarter of 2019 from a network services customer accounted for as a sales-type lease under ASC 842, *Leases* ("ASC 842"), with no comparable amount for the same period in 2020.
- *Satellite-related services*—an aggregate increase of \$4.8 million, largely related to professional fees recognized in connection with MEV-1 and MEV-2, as well as increased revenues from services supporting government customers.

Operating Expenses

Direct Costs of Revenue (Excluding Depreciation and Amortization)

Direct costs of revenue increased by \$26.2 million, or 9%, to \$331.8 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The increase was primarily due to:

- a \$17.4 million increase in third-party FSS capacity costs largely incurred in connection with government customers, Intelsat 29e customer restoration on third-party satellites, and the Azercosmos (Intelsat 38) satellite;
- a \$9.6 million increase in staff-related expenses largely relating to our employee retention incentive plans; and
- a \$4.7 million increase in equipment costs primarily related to \$12.4 million attributable to a government customer, partially offset by a \$7.7 million decrease in third-party capacity costs recognized under ASC 842 for the nine months ended September 30, 2019, with no comparable costs for the nine months ended September 30, 2020; partially offset by
- a \$5.6 million decrease in third-party managed capacity costs largely related to a government customer.

Selling, General and Administrative

Selling, general and administrative expenses increased by \$46.3 million, or 28%, to \$214.6 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The increase was primarily due to a \$22.9 million increase in bad debt expense, largely relating to a certain customer that filed for Chapter 11 bankruptcy protection and certain customers in the North America, Asia-Pacific and Africa regions, a \$13.2 million increase in professional fees and a \$7.7 million increase in staff-related expenses largely relating to our employee retention incentive plans.

Depreciation and Amortization

Depreciation and amortization expense decreased by \$8.2 million, or 2%, to \$488.2 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. Significant items impacting depreciation and amortization included:

- a decrease of \$12.6 million in depreciation expense due to the timing of a satellite becoming fully depreciated;
- a decrease of \$10.0 million in depreciation expense due to the write-off of Intelsat 29e; and

- a decrease of \$1.1 million in amortization expense; partially offset by
- an increase of \$14.1 million in depreciation expense resulting from the impact of a satellite placed in service; and
- an increase of \$1.2 million in depreciation expense due to the impact of certain ground segment assets placed in service.

Satellite Impairment Loss

We recognized an impairment charge of \$381.6 million for the nine months ended September 30, 2019 relating to the loss of Intelsat 29e (see Item 1, Note 8—Satellites and Other Property and Equipment). The impairment charge consisted of approximately \$377.9 million related to the write-off of the carrying value of the satellite and associated deferred satellite performance incentive obligations, and approximately \$3.7 million related to prepaid regulatory fees. No comparable amounts were recognized for the nine months ended September 30, 2020.

Impairment of Non-Amortizable Intangible and Other Assets

We recognized impairment charges of \$34.0 million and \$12.2 million for the nine months ended September 30, 2020 relating to certain satellite and launch vehicle deposits and the Intelsat trade name intangible asset, respectively. No comparable amounts were recognized for the nine months ended September 30, 2019. See Item 1, Note 8—Satellites and Other Property and Equipment and Item 1, Note 10—Goodwill and Other Intangible Assets for further discussion.

Interest Expense, Net

Interest expense, net decreased by \$274.3 million, or 29%, to \$678.9 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The decrease in interest expense, net was principally due to the following:

- a decrease of \$258.6 million in interest expense primarily resulting from Chapter 11 restructuring activities, partially offset by an increase in interest expense recognized on our senior secured credit facilities; and
- a decrease of \$20.7 million corresponding to a larger decrease in fair value of the interest rate cap contracts during the nine months ended September 30, 2019 as compared to the nine months ended September 30, 2020; partially offset by
- an increase of \$6.3 million due to lower capitalized interest primarily resulting from decreased levels of satellites and related assets under construction.

The non-cash portion of total interest expense, net was \$103.2 million for the nine months ended September 30, 2020, primarily consisting of interest expense related to the significant financing component identified in customer contracts, amortization and accretion of discounts and premiums and amortization of deferred financing fees.

Other Income (Expense), Net

Other income, net was \$8.6 million for the nine months ended September 30, 2020, as compared to other expense, net of \$32.4 million for the nine months ended September 30, 2019. The net increase in other income was primarily driven by a \$36.5 million net loss for the nine months ended September 30, 2019 due to a change in value of certain investments in third parties with no similar activity in 2020, as well as a \$6.0 million gain on sale of assets for the nine months ended September 30, 2020.

Reorganization Items

Reorganization items reflect direct costs incurred in connection with the Chapter 11 Cases. Reorganization items of \$335.1 million for the nine months ended September 30, 2020 primarily consisted of \$197.0 million related to the write-off of debt discount, premium and issuance costs, \$85.2 million in professional fees and \$52.2 million in financing fees related to the DIP Facility. There were no comparable amounts for the nine months ended September 30, 2019.

Provision for (Benefit from) Income Taxes

Income tax expense decreased by \$21.6 million to income tax benefit of \$17.7 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The decrease was principally attributable to a net benefit recognized for the BEAT, additional benefit from the CARES Act with regards to the relaxed limitations on the deductibility of interest, the use of net operating losses arising in taxable years beginning after 2018 and lower income from our U.S. subsidiaries for the nine months ended September 30, 2020.

Cash paid for income taxes, net of refunds, totaled \$9.3 million and \$4.7 million for the nine months ended September 30, 2019 and 2020, respectively.

Net Loss Attributable to Intelsat S.A.

Net loss attributable to Intelsat S.A. was \$640.1 million for the nine months ended September 30, 2020, as compared to net loss attributable to Intelsat S.A. of \$798.6 million for the nine months ended September 30, 2019. The change reflects the various items discussed above.

EBITDA

EBITDA consists of earnings before net interest, loss (gain) on early extinguishment of debt, taxes and depreciation and amortization. Given our high level of leverage, refinancing activities are a frequent part of our efforts to manage our costs of borrowing. Accordingly, we consider loss (gain) on early extinguishment of debt an element of interest expense. EBITDA is a measure commonly used in the FSS sector, and we present EBITDA to enhance the understanding of our operating performance. We use EBITDA as one criterion for evaluating our performance relative to that of our peers. We believe that EBITDA is an operating performance measure, and not a liquidity measure, that provides investors and analysts with a measure of operating results unaffected by differences in capital structures, capital investment cycles and ages of related assets among otherwise comparable companies. However, EBITDA is not a measure of financial performance under U.S. GAAP, and our EBITDA may not be comparable to similarly titled measures of other companies. EBITDA should not be considered as an alternative to operating income (loss) or net income (loss) determined in accordance with U.S. GAAP, as an indicator of our operating performance, or as an alternative to cash flows from operating activities determined in accordance with U.S. GAAP, as an indicator of cash flows, or as a measure of liquidity.

A reconciliation of net loss to EBITDA for the periods shown is as follows (in thousands):

	Three Months Ended September 30, 2019	Three Months Ended September 30, 2020	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Net loss	\$ (147,698)	\$ (15,331)	\$ (796,851)	\$ (638,273)
Add (Subtract):				
Interest expense, net	315,964	138,075	953,246	678,937
Provision for (benefit from) income taxes	6,248	(18,650)	3,884	(17,691)
Depreciation and amortization	161,536	162,573	496,438	488,235
EBITDA	\$ 336,050	\$ 266,667	\$ 656,717	\$ 511,208

Adjusted EBITDA

In addition to EBITDA, we calculate a measure called Adjusted EBITDA to assess the operating performance of Intelsat S.A. Adjusted EBITDA consists of EBITDA of Intelsat S.A. as adjusted to exclude or include certain unusual items, certain other operating expense items and certain other adjustments as described in the table and related footnotes below. Our management believes that the presentation of Adjusted EBITDA provides useful information to investors, lenders and financial analysts regarding our financial condition and results of operations because it permits clearer comparability of our operating performance between periods. By excluding the potential volatility related to the timing and extent of non-operating activities, such as impairments of asset value and other non-recurring items, our management believes that Adjusted EBITDA provides a useful means of evaluating the success of our operating activities. We also use Adjusted EBITDA, together with other appropriate metrics, to set goals for and measure the operating performance of our business, and it is one of the principal measures we use to evaluate our management's performance in determining compensation under our incentive compensation plans. Adjusted EBITDA measures have been used historically by investors, lenders and financial analysts to estimate the value of a company, to make informed investment decisions and to evaluate performance. Our management believes that the inclusion of Adjusted EBITDA EBITDA facilitates comparison of our results with those of companies having different capital structures.

Adjusted EBITDA is not a measure of financial performance under U.S. GAAP and may not be comparable to similarly titled measures of other companies. Adjusted EBITDA should not be considered as an alternative to operating income (loss) or net income (loss) determined in accordance with U.S. GAAP, as an indicator of our operating performance, as an alternative to cash flows from operating activities determined in accordance with U.S. GAAP, as an indicator of cash flows, or as a measure of liquidity.



A reconciliation of net loss to EBITDA and EBITDA to Adjusted EBITDA is as follows (in thousands):

	ee Months Ended tember 30, 2019	Fhree Months Ended September 30, 2020	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Net loss	\$ (147,698)	\$ (15,331)	\$ (796,851)	\$ (638,273)
Add (Subtract):				
Interest expense, net	315,964	138,075	953,246	678,937
Provision for (benefit from) income taxes	6,248	(18,650)	3,884	(17,691)
Depreciation and amortization	161,536	162,573	496,438	488,235
EBITDA	 336,050	266,667	 656,717	 511,208
Add:				
Compensation and benefits ⁽¹⁾	3,924	15,484	10,215	39,338
Non-recurring and non-cash items ⁽²⁾	11,944	10,510	50,373	25,837
Satellite impairment loss ⁽³⁾	—	—	381,565	—
Impairment of non-amortizable intangible and other assets ⁽⁴⁾	—	—	_	46,243
Reorganization items ⁽⁵⁾	_	36,367	—	335,059
Proportionate share from unconsolidated joint venture ⁽⁶⁾ :				
Interest expense, net	1,345	1,074	3,825	3,199
Depreciation and amortization	2,814	2,815	7,505	8,444
Adjusted EBITDA ⁽⁷⁾⁽⁸⁾	\$ 356,077	\$ 332,917	\$ 1,110,200	\$ 969,328

(1) Reflects non-cash expenses incurred relating to our equity compensation plans and, for the three and nine months ended September 30, 2020, expenses incurred relating to our employee retention incentive plans in connection with our Chapter 11 proceedings.

- (2) Reflects certain non-recurring expenses, gains and losses and non-cash items, including the following: professional fees related to our liability management initiatives; costs associated with our C-band spectrum proposal; corporate development and strategy costs; certain research and development costs; severance, retention and relocation payments; changes in fair value of certain investments; certain foreign exchange gains and losses; and other various non-recurring expenses. In 2019, these costs were partially offset by non-cash income related to the recognition of deferred revenue on a straight-line basis for certain prepaid capacity service contracts.
- (3) Reflects a non-cash impairment charge recorded in connection with the Intelsat 29e satellite loss.
- (4) Reflects a non-cash impairment charge recorded in connection with the write-off of certain satellite and launch vehicle deposits (see Item 1, Note 8—Satellites and Other Property and Equipment) and a trade name impairment (see Item 1, Note 10—Goodwill and Other Intangible Assets).
- (5) Reflects direct costs incurred in connection with our Chapter 11 proceedings. See Item 1, Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters.
- (6) Reflects adjustments related to our interest in Horizons 3. See Item 1, Note 9(b)—Investments—Horizons-3 Satellite LLC.
- (7) Adjusted EBITDA included \$25.8 million and \$26.2 million for the three months ended September 30, 2019 and 2020, respectively, and \$76.1 million and \$78.7 million for the nine months ended September 30, 2019 and 2020, respectively, of revenue relating to the significant financing component identified in customer contracts in accordance with the adoption of ASC 606, *Revenue from Contracts with Customers*. These impacts are not permitted to be reflected in the applicable consolidated and Adjusted EBITDA definitions under our debt agreements.
- (8) Intelsat S.A. Adjusted EBITDA reflected \$5.1 million and \$4.4 million for the three months ended September 30, 2019 and 2020, respectively, and \$7.6 million and \$14.3 million for the nine months ended September 30, 2019 and 2020, respectively, of Adjusted EBITDA attributable to Intelsat Horizons-3 LLC, its subsidiaries and its proportionate share of Horizons 3. These entities are considered to be unrestricted subsidiaries under the definitions set forth in our applicable debt agreements.

B. Liquidity and Capital Resources

Overview

We are a highly leveraged company and our contractual obligations, commitments and debt service requirements over the next several years are significant. At September 30, 2020, the aggregate principal amount of our debt outstanding not held by affiliates was \$15.2 billion. Our interest expense, net for the three and nine months ended September 30, 2020 was \$138.1 million and \$678.9 million, respectively, which included \$29.7 million and \$103.2 million of non-cash interest expense, respectively. At September 30, 2020, cash, cash equivalents and restricted cash were approximately \$1.0 billion.

The commencement of the Chapter 11 Cases accelerated substantially all of our outstanding debt. Any efforts to enforce payment obligations related to the acceleration of our debt have been automatically stayed as a result of the filing of the Chapter 11 Cases, and the creditors' rights of enforcement are subject to the applicable provisions of the Bankruptcy Code.

During the pendency of the Chapter 11 Cases, as discussed above in —Recent Developments, *Voluntary Reorganization under Chapter 11*, we expect cash payments on our interest obligations to be lower than in prior years. In past years, our cash flows from operations and cash on hand have been sufficient to fund significant capital expenditures (\$255.7 million and \$229.8 million for the years ended December 31, 2018 and 2019, respectively). However, as discussed above in —Recent Developments, *Voluntary Reorganization under Chapter 11*, our ability to fund operating expenses is now, to some extent, subject to obtaining certain approvals from the Bankruptcy Court in connection with our Chapter 11 proceedings.

A significant factor driving the Company's decision to file for Chapter 11 protection was the Company's desire to participate in the FCC's process for accelerated clearing of the C-band spectrum, for which we need to incur significant upfront expenses for clearing activities well in advance of receiving reimbursement payments. On August 14, 2020, the Company filed its final C-band spectrum transition plan with the FCC.

In addition to the significant capital expenditures we expect to make in 2020 and beyond, we expect total clearing costs will be approximately \$1.7 billion over the next two to three years. Our primary source of liquidity is and will continue to be cash generated from operations, as well as existing cash. Also, as noted above in —Recent Developments, *Voluntary Reorganization under Chapter 11*, we have entered into a DIP Facility in an aggregate principal amount of \$1.0 billion. We currently expect to use cash on hand, cash flows from operations and borrowings under the DIP Facility to fund our most significant cash outlays, including debt service requirements, capital expenditures, and the Purchase Price for the Gogo Transaction (see —Recent Developments, *Gogo Inc. Transaction* above) in the next twelve months and beyond. We also expect to receive reimbursement payments for certain upfront C-band spectrum clearing expenses incurred and under the FCC Final Order, the Company is eligible to receive Acceleration Payments of approximately \$1.2 billion and \$3.7 billion on December 5, 2021 and December 5, 2023, respectively, subject to the satisfaction of certain deadlines and other conditions set forth therein.

Cash Flow Items

Our cash flows consisted of the following for the periods shown (in thousands):

	Nine Months Ended September 30, 2019	Nine Months Ended September 30, 2020
Net cash provided by operating activities	\$ 163,429	\$ 194,729
Net cash used in investing activities	(214,527)	(418,346)
Net cash provided by financing activities	370,816	417,431
Net change in cash, cash equivalents and restricted cash	317,477	189,486

Net Cash Provided by Operating Activities

Net cash provided by operating activities increased by \$31.3 million to \$194.7 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The increase was primarily due to a \$21.0 million decrease in net loss and changes in non-cash items and a \$10.3 million increase from changes in operating assets and liabilities. The increase in operating assets and liabilities was primarily due to lower outflows related to the amount and timing of accounts payable and interest payments, as well as higher inflows from receivables, partially offset by higher outflows related to prepaid and other assets and deferred revenues.



Net Cash Used in Investing Activities

Net cash used in investing activities increased by \$203.8 million to \$418.3 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The increase was primarily due to higher capital expenditures partially offset by lower origination of loans held-for-investment.

Net Cash Provided by Financing Activities

Net cash provided by financing activities increased by \$46.6 million to \$417.4 million for the nine months ended September 30, 2020, as compared to the nine months ended September 30, 2019. The increase was primarily due to higher net proceeds of \$52.4 million resulting from our DIP financing during the nine months ended September 30, 2020, as compared to the net proceeds from the debt offering completed during the nine months ended September 30, 2020, as compared to the net proceeds from the debt offering completed during the nine months ended September 30, 2019. This was partially offset by an increase of \$4.4 million in deferred financing payments on satellite performance incentives as compared to the nine months ended September 30, 2019.

Restricted Cash

As of September 30, 2020, \$19.4 million of cash was legally restricted, being held as a compensating balance for certain outstanding letters of credit.

Debt

Intelsat Jackson Superpriority Secured Debtor-in-Possession Term Loan Facility

On June 17, 2020 (the "Closing Date"), the DIP Debtors and DIP Lenders entered into the DIP Credit Agreement, a non-amortizing multiple draw superpriority secured debtor-in-possession term loan facility, in an aggregate principal amount of \$1.0 billion, on the terms and conditions set forth therein (see—Recent Developments—*Voluntary Reorganization under Chapter 11* above).

Intelsat Jackson borrowed \$500.0 million of term loans under the DIP Facility on the Closing Date. Under the DIP Facility, Intelsat Jackson may, at its sole discretion, make incremental draws of the lesser of \$250.0 million and the remaining available commitments of the DIP Lenders. Drawn amounts under the DIP Facility bear interest at either (i) 4.50% per annum plus a base rate of the highest of (a) the Federal Funds Effective Rate plus ½ of 1.00%, (b) the Prime Rate as in effect on such day and (c) the London Inter-Bank Offered Rate ("LIBOR Rate") for a one-month interest period on such day (or if such day is not a business day, the immediately preceding business day) plus 1.00%, or (ii) 5.50% plus the LIBOR Rate. For purposes of the DIP Facility, the LIBOR Rate has an effective floor rate of 1.0%. Undrawn amounts under the DIP Facility shall be subject to a ticking fee of 3.6% of the amount of commitments of the DIP Lenders from the entry of the DIP Order until such commitments terminate, which ticking fee shall be payable on the last day of each fiscal quarter prior to the date such commitments terminate and on the date of such termination. If an event of default under the DIP Facility occurs, the overdue amounts under the DIP Facility would bear interest at an additional 2.0% per annum above the interest rate otherwise applicable.

The proceeds of the DIP Facility may be used, among other things, to pay for (i) working capital needs of the DIP Debtors in the ordinary course of business, (ii) potential C-band relocation costs, (iii) investment and other general corporate purposes, and (iv) the costs and expenses of administering the Chapter 11 Cases. The maturity date of the DIP Facility is July 13, 2021, subject to certain extensions pursuant to the terms of the DIP Credit Agreement.

The DIP Credit Agreement includes usual and customary negative covenants for debtor-in-possession loan agreements of this type, including covenants limiting the Company's and its subsidiaries' ability to, among other things, incur additional indebtedness, create liens on assets, make investments, loans or advances, engage in mergers, consolidations, sales of assets and acquisitions, pay dividends and distributions and make payments in respect of junior or pre-petition indebtedness, in each case subject to customary exceptions for debtor-in-possession loan agreements of this type.

The DIP Credit Agreement also includes certain customary representations and warranties, affirmative covenants and events of default, including, but not limited to, payment defaults, breaches of representations and warranties, covenant defaults, certain events under the Employee Retirement Income Security Act of 1974, as amended, and change of control. Certain bankruptcy-related events are also events of default, including, but not limited to, the dismissal by the Bankruptcy Court of any of the Chapter 11 Cases, the conversion of any of the Chapter 11 Cases to a case under Chapter 7 of the Bankruptcy Code and certain other events related to the impairment of the DIP Lenders' rights or liens granted under the DIP Credit Agreement.

On August 24, 2020, the DIP Debtors and DIP Lenders entered into DIP Amendment No. 1 to the DIP Credit Agreement, in connection with the Gogo Transaction (see Note 1—General—Gogo Inc. Transaction for additional information). DIP Amendment No. 1 was approved by the Bankruptcy Court on August 31, 2020.

The foregoing descriptions of the DIP Credit Agreement and DIP Amendment No. 1 do not purport to be complete and are qualified in their entirety by reference to the full text of the DIP Credit Agreement and DIP Amendment No. 1, as applicable.

Intelsat Jackson Senior Secured Credit Agreement and the Company and Certain of its Subsidiaries' Indentures

The commencement of the Chapter 11 Cases constituted an immediate event of default under Intelsat Jackson's secured credit agreement, dated as of January 12, 2011 (as amended, the "Intelsat Jackson Secured Credit Agreement"), as well as under the indentures governing certain of the Company and its subsidiaries' senior secured notes and senior notes, resulting in the automatic and immediate acceleration of substantially all of our outstanding debt. Any efforts to enforce payment obligations related to the acceleration of our debt have been automatically stayed as a result of the filing of the Chapter 11 Cases, and the creditors' rights of enforcement are subject to the applicable provisions of the Bankruptcy Code.

In addition, in April 2020, our LIBOR loans under the Intelsat Jackson Secured Credit Agreement were converted to Alternate Base Rate ("ABR") loans. We expect to pay interest on the floating rate term loans under the Intelsat Jackson Secured Credit Agreement at the rate applicable to ABR loans.

Contracted Backlog

We have historically had, and currently have, a substantial contracted backlog, which provides some assurance regarding our future revenue expectations. Contracted backlog is our expected future revenue under customer contracts and includes both cancelable and non-cancelable contracts. Approximately 91% of our total contracted backlog as of September 30, 2020 related to contracts that were non-cancelable and approximately 8% related to contracts that were cancelable subject to substantial termination fees. In certain cases of breach for non-payment or customer bankruptcy, we may not be able to recover the full value of certain contracts or termination fees. Our contracted backlog as of September 30, 2020 was approximately \$6.2 billion. We believe this backlog and associated revenues result in more predictable cash flows in the FSS sector as compared to that of typical companies outside our industry.

Capital Expenditures

Our capital expenditures depend on our business strategies and reflect our commercial responses to opportunities and trends in our industry. Our actual capital expenditures may differ from our expected capital expenditures if, among other things, we enter into any currently unplanned strategic transactions. Levels of capital spending from one year to the next are also influenced by the nature of the satellite life cycle and by the capital-intensive nature of the satellite industry. For example, we incur significant capital expenditures during the years in which satellites are under construction. We typically procure a new satellite within a timeframe that would allow the satellite to be deployed at least one year prior to the end of the service life of the satellite to be replaced. As a result, we frequently experience significant variances in our capital expenditures from year to year. Further, following the Company's filing of its final C-band spectrum transition plan with the FCC on August 14, 2020, we expect to incur significant upfront expenses for clearing activities well in advance of receiving reimbursement payments. Payments for satellites and other property and equipment during the nine months ended September 30, 2020 were \$420.0 million. However, subject to the satisfaction of certain deadlines and other conditions set forth in the FCC Final Order, the Company is eligible to receive Acceleration Payments in an aggregate total amount of approximately \$4.9 billion over the next 3 years.

We intend to fund our capital expenditure requirements from cash on hand, cash provided from operating activities and the DIP Facility; however, our ability to fund capital expenditures in the ordinary course is, to some extent, subject to obtaining certain approvals from the Bankruptcy Court in connection with our Chapter 11 proceedings. On August 31, 2020 the Bankruptcy Court approved our intention to fund the Purchase Price of the Gogo Transaction with proceeds from our existing DIP Facility and cash on hand.

Off-Balance Sheet Arrangements

We have revenue sharing agreements with JSAT related to services sold on the Horizons 1, Horizons 2 and Horizons 3 satellites. We are responsible for billing and collection for such services and we remit 50% of the revenue, less applicable fees and commissions, to JSAT. Refer to Item 1, Note 9— Investments for disclosures relating to the revenue sharing agreements with JSAT.

Contractual Obligations

Other than as disclosed elsewhere in this report with respect to the filing of the Chapter 11 Cases and the acceleration of substantially all of our debt as a result, there have been no material changes outside the ordinary course of business to the information provided with respect to our contractual obligations as disclosed in our Annual Report on Form 10-K for the year ended December 31, 2019, filed with the SEC on February 20, 2020 (the "2019 Annual Report").

Critical Accounting Policies and Estimates

The preparation of these condensed consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities as of the date of these condensed consolidated financial statements, the reported amounts of revenues and expenses during the reporting periods, and the disclosures of contingent liabilities.

The Company's significant accounting policies are described in Note 1—Background and Summary of Significant Accounting Policies in our 2019 Annual Report. The Company's critical accounting estimates are described in Item 7—Management's Discussion and Analysis of Financial Condition and Results of Operations in our 2019 Annual Report and are further described below.

Bankruptcy Accounting

Our condensed consolidated financial statements included herein have been prepared as if we are a going concern and reflect the application of ASC 852, *Reorganizations* ("ASC 852"). ASC 852 requires the financial statements, for periods subsequent to the commencement of the Chapter 11 proceedings, to distinguish transactions and events that are directly associated with the reorganization from the ongoing operations of the business. Accordingly, we classify liabilities and obligations whose treatment and satisfaction are dependent on the outcome of the reorganization under the Chapter 11 proceedings as liabilities subject to compromise on our condensed consolidated balance sheets. In addition, we classify all income, expenses, gains or losses that are incurred or realized as a result of the Chapter 11 proceedings as reorganization items in our condensed consolidated statements of operations (see Item 1, Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters).

Asset Impairment Assessments

We account for goodwill and other non-amortizable intangible assets in accordance with ASC 350, *Intangibles—Goodwill and Other*, and have deemed these assets to have indefinite lives. Therefore, these assets are not amortized but are tested on an annual basis for impairment during the fourth quarter, or whenever events or changes in circumstances indicate that the carrying amount may not be fully recoverable. We review our long-lived and amortizable intangible assets to assess whether an impairment has occurred in accordance with the guidance provided under ASC 360—*Property, Plant and Equipment*, whenever events or changes in circumstances indicate, in our judgment, that the carrying amount of an asset may not be recoverable.

During the first quarter of 2020, the price of our common shares and trading values of our debt securities experienced sustained reductions. We also witnessed certain declines in financial performance as compared to previously prepared internal budget and forecast projections. Among the impacts of the COVID-19 pandemic were a reduction of revenue and a decreased likelihood of collection from certain mobility customers. Based on our examination of these and other qualitative factors, we concluded that further testing of goodwill and other non-amortizable assets as well as long-lived and amortizable intangible assets was required.

Goodwill. For the analysis of goodwill, we applied ASU 2017-04, which is further described in Item 1, Note 1—General. Intelsat has only one reporting unit for purposes of the analysis of goodwill, and accordingly, the analysis is undertaken at the enterprise level. We determined the estimated fair value of our reporting unit using a discounted cash flow analysis, along with independent source data related to comparative market multiples and, when available, recent transactions, each of which is considered a Level 3 input within the fair value hierarchy under ASC 820, *Fair Value Measurements and Disclosure* ("ASC 820"). The discounted cash flows were derived from a five-year projection of cash flows plus a residual value, with the resulting projected cash flows discounted at an appropriate weighted average cost of capital.

In estimating the undiscounted cash flows, we primarily used our internally prepared budgets and forecast information. The key assumptions included in our model were projected growth rates, cost of capital, effective tax rates, and industry and economic trends, along with the C-band Acceleration Payments expected to be received subject to the satisfaction of certain deadlines and other conditions set forth in the FCC Final Order and the discount rate applied to those cash flows. A change in estimated future cash flows or other assumptions could change our estimated fair values and result in future impairments. The conclusion of our analysis was that the fair value of our reporting unit was greater than its carrying value, resulting in no impairment of goodwill.

Orbital Locations. We determined the estimated fair value of our rights to operate at orbital locations by using the build-up method to determine cash flows for the income approach, with the resulting projected cash flows discounted at an appropriate weighted average cost of capital. Under the build-up approach, the amount a reasonable investor would be willing to pay for the right to operate a satellite business using orbital locations is calculated by first estimating the cash flows that typical market participants might assume could be available from the right to operate satellites using the subject location in a similar market. It is assumed that rather than acquiring such a business as a going concern, the buyer would hypothetically start with the right to operate satellites at

orbital locations and build a new business with similar attributes from the beginning. Thus, the buyer is assumed to incur the start-up costs and losses typically associated with the going concern value and pay for all other tangible and intangible assets.

The key assumptions used in estimating the fair values of our rights to operate at our orbital locations included the following: (i) market penetration leading to revenue growth, (ii) profit margin, (iii) duration and profile of the build-up period, (iv) estimated start-up costs and losses incurred during the build-up period and (v) weighted average cost of capital.

We completed our analysis of the estimated fair value of our rights to operate at certain orbital locations in connection with the analysis of goodwill described above and concluded that the fair value was greater than the carrying value, resulting in no impairment.

Trade Name. We have implemented the relief from royalty method to determine the estimated fair value of the Intelsat trade name. The relief from royalty analysis is comprised of two major steps: (i) a determination of the hypothetical royalty rate, and (ii) the subsequent application of the royalty rate to projected revenue. In determining the hypothetical royalty rate utilized in the relief from royalty approach, we considered comparable license agreements, an excess earnings analysis to determine aggregate intangible asset earnings, and other qualitative factors, each of which is considered a Level 3 input within the fair value hierarchy under ASC 820.

The key assumptions used in our model to estimate the fair value of the Intelsat trade name included forecasted revenues, the royalty rate, the tax rate and the discount rate. We completed our analysis of the estimated fair value of the Intelsat trade name in connection with the analysis of goodwill described above and it resulted in an impairment of our trade name intangible asset of \$12.2 million, which is included within impairment of non-amortizable intangible and other assets in the condensed consolidated statements of operations.

Long-Lived and Amortizable Intangible Assets. The Company evaluated the assets for potential impairment using internal projections of undiscounted cash flows expected to result from the use and eventual disposal of the assets. The key assumptions included in our model were projected growth rates, cost of capital, effective tax rates, and industry and economic trends. A change in estimated future cash flows or other assumptions could change our estimated undiscounted cash flows and result in future impairments. The conclusion of our analysis was that the undiscounted cash flows of the asset group was greater than its carrying value, resulting in no impairment.

Recently Issued Accounting Pronouncements

For disclosures related to recently issued accounting pronouncements, see Item 1, Note 1—General—Recently Issued Accounting Pronouncements.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

Except as set forth below, there have been no material changes to the Company's disclosures about market risk made in Part II, Item 7A. Quantitative and Qualitative Disclosures about Market Risk of the Company's 2019 Annual Report.

Interest Rate Risk

As discussed in Item 1, Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters, the filing of the Chapter 11 Cases constituted an event of default that accelerated substantially all of our obligations under the documents governing the prepetition existing indebtedness of Intelsat S.A., Intelsat Luxembourg, Intelsat Connect and Intelsat Jackson. As such, we have reclassified all such debt obligations, other than debt subject to compromise, to current portion of long-term debt on our condensed consolidated balance sheet as of September 30, 2020. While the Chapter 11 Cases are pending, the Debtors do not anticipate making interest payments due under their respective unsecured debt instruments; however, the Debtors expect to make monthly interest payments on their senior secured debt instruments pursuant to the adequate protection requirements under the DIP Order.

Item 4. Controls and Procedures

Disclosure Controls and Procedures

We have carried out an evaluation, under the supervision and with the participation of our management, including our principal executive officer and our principal financial officer, of the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rule 13a-15(e) and 15d-15(e) under the Exchange Act), as of the quarter ended September 30, 2020. Based upon that evaluation, our principal executive officer and our principal financial officer concluded that our disclosure controls and procedures were effective as of September 30, 2020.

Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting for the quarter ended September 30, 2020 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

Chapter 11 Cases

On May 13, 2020, the Debtors filed voluntary petitions for relief under Chapter 11 of the Bankruptcy Code in Bankruptcy Court. The information contained in Item 1, Note 2—Chapter 11 Proceedings, Ability to Continue as a Going Concern and Other Related Matters—*Voluntary Reorganization under Chapter 11* of the notes to the condensed consolidated financial statements is incorporated herein by reference. As a result of such bankruptcy filings, substantially all proceedings pending against the Debtors have been stayed. These matters will be subject to resolution in accordance with the Bankruptcy Code and applicable orders of the Bankruptcy Court.

As part of the Chapter 11 Cases, parties believing that they have claims or causes of action against the Debtors could file proofs of claim evidencing such claims. Pursuant to an order entered by the Bankruptcy Court, all proofs of claim were to be filed with the Bankruptcy Court by September 9, 2020, except for claims by governmental units. Governmental units must file proofs of claim by November 16, 2020. As claims are filed, the claims will be reconciled to amounts recorded in liabilities subject to compromise in our condensed consolidated balance sheet. The Debtors may ask the Bankruptcy Court to disallow claims that the Debtors believe are duplicative, have been later amended or superseded, are without merit, are overstated or should be disallowed for other reasons. In addition, as a result of this process, the Debtors may identify additional liabilities that will need to be recorded or reclassified to liabilities subject to compromise. In light of the substantial number of claims filed, and expected to be filed, the claims resolution process may take considerable time to complete and likely will continue after the Debtors emerge from bankruptcy.

SES Claim

On July 14, 2020, SES Americom, Inc. ("SES") filed a proof of claim in the Bankruptcy Court in the amount of \$1.8 billion against each of the Debtors. SES asserts that the Debtors owe money (or will owe money) to SES pursuant to certain contractual and fiduciary obligations made in the context of the consortium agreement between Debtor Intelsat US LLC, SES, and other satellite operators (the "Consortium Agreement"). SES believes it is entitled to 50% of the combined payments that may eventually be payable to the Debtors and SES pursuant to the FCC Final Order, which provides for Acceleration Payments subject to the satisfaction of certain deadlines and other conditions set forth therein. SES's proof of claim alleges that the Debtors breached the Consortium. The proof of claim alloges breach of fiduciary duties and unjust enrichment and seeks monetary and punitive damages. We dispute the allegations in the proof of claim and on October 19, 2020, filed an objection to the claim, which we intend to litigate vigorously. To the extent that any portion of SES's claim is allowed, we have asked the Bankruptcy Court to 'equitably subordinate' such claim based on SES's conduct in matters related to the Consortium Agreement. While the ultimate resolution of the claim is not currently predictable, if there is an adverse ruling, the ruling could constitute a material adverse outcome on our future consolidated financial condition.

Other Litigation Matters

We are subject to litigation in the ordinary course of business, but management does not believe that the resolution of any of those pending proceedings would have a material adverse effect on our financial position or results of operations.

Item 1A. Risk Factors

Except as set forth below, there have been no material changes or additions to the risk factors previously disclosed in Part I—Item 1A—Risk Factors of our 2019 Annual Report, as supplemented by Part II—Item 1A—Risk Factors of our Quarterly Report on Form 10-Q for the quarter ended March 31, 2020, filed with the SEC on June 4, 2020 and Part II—Item 1A—Risk Factors of our Quarterly Report on Form 10-Q for the quarter ended June 30, 2020, filed with the SEC on August 6, 2020.

Intelsat may fail to realize all of the anticipated benefits of the potential Gogo Transaction or those benefits may take longer to realize than expected. We may also encounter significant difficulties in integrating Gogo's commercial aviation business.

Our ability to realize the anticipated benefits of the potential Gogo Transaction will depend, to a large extent, on our ability to integrate Gogo's commercial aviation business. The combination of two independent businesses is a complex, costly and time-consuming process, particularly while we are undergoing the Chapter 11 proceedings. Further, the restrictions posed by the agreements governing our DIP Facility may limit our ability to integrate the two businesses. As a result, we will be required to devote significant management attention and resources to integrate the business practices and operations of Intelsat and Gogo's commercial aviation business. The integration process may disrupt the businesses and, if implemented ineffectively, would restrict the realization of the full expected benefits. The failure to meet the challenges involved in integrating the two businesses and to realize the anticipated



benefits of the potential transaction could cause an interruption of, or a loss of momentum in, the activities of the combined businesses and could adversely affect the results of operations of the combined businesses.

In addition, the overall integration of the businesses may result in material unanticipated problems, expenses, liabilities, competitive responses, loss of customers and other business relationships, and diversion of management's attention. The difficulties of combining the operations of the companies include, among others:

- the diversion of management's attention to integration matters;
- difficulties in achieving anticipated cost savings, synergies, business opportunities and growth prospects from the combination;
- difficulties in the integration of operations and systems;
- conforming standards, controls, procedures and accounting and other policies, business cultures and compensation structures between the two companies;
- difficulties in the assimilation of employees;
- difficulties in managing the expanded operations of a significantly larger and more complex company;
- challenges in keeping existing customers and obtaining new customers;
- challenges in attracting and retaining key personnel; and
- coordinating a geographically dispersed organization, particularly due to the effects of the COVID-19 pandemic.

Many of these factors will be outside of our control and any one of them could result in increased costs, decreases in the amount of expected revenues and diversion of management's time and energy, which could materially impact the business, financial condition and results of operations of the combined company. In addition, even if Gogo's commercial aviation business operations are integrated successfully, the full benefits of the potential transaction and other pending acquisitions may not be realized, including the synergies, cost savings or sales or growth opportunities that are expected. These benefits may not be achieved within the anticipated time frame, or at all. Further, additional unanticipated costs may be incurred in the integration of the businesses. As a result, it cannot be assured that the potential Gogo Transaction will result in the realization of the full benefits anticipated from such transaction.

Item 6. Exhibits

Exhibit No.	Document Description
2.1	Purchase and Sale Agreement, dated as of August 31, 2020, between Gogo Inc. and Intelsat Jackson Holdings S.A. (incorporated by reference to Exhibit 2.1 of Intelsat S.A.'s Current Report on Form 8-K, File No. 001-35878, filed on September 1, 2020).
10.1	Joinder Agreement No. 1, dated as of August 7, 2020, to Superpriority Secured Debtor In Possession Credit Agreement, dated as of June 17, 2020, by and among Intelsat Velocity Holdings LLC, as New Guarantor, Intelsat Invoice Services LLC, as New Guarantor, and Credit Suisse AG, Cayman Islands Branch as Administrative Agent.*
10.2	Amendment No. 1, dated as of August 24, 2020, to Superpriority Secured Debtor In Possession Credit Agreement, dated as of June 17, 2020, by and among Intelsat Jackson Holdings S.A., as borrower, the guarantor parties thereto, Credit Suisse AG, as administrative agent and collateral agent, and the lender parties thereto.*
31.1	Rule 13a-14(a)/15d-14(a). Certification of Principal Executive Officer.*
31.2	Rule 13a-14(a)/15d-14(a) Certification of Principal Financial Officer.*
32.1	Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.*
32.2	Certification of Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.*
101	The following financial information from the Company's Quarterly Report on Form 10-Q for the quarter ended September 30, 2020, formatted in Inline Extensible Business Reporting Language (iXBRL): (i) Consolidated Balance Sheets, (ii) Consolidated Statements of Operations, (iii) Consolidated Statements of Comprehensive Loss, (iv) Consolidated Statements of Changes in Shareholders' Deficit, (v) Consolidated Statements of Cash Flows, and (vi) Notes to Consolidated Financial Statements.*
104	Cover Page Interactive Data File - the cover page XBRL tags are embedded within the Inline XBRL document.

Filed herewith.

The certifications attached as Exhibits 32.1 and 32.2 that accompany this Quarterly Report on Form 10-Q are not deemed filed with the SEC and are not to be incorporated by reference into any filing of Intelsat S.A. under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, whether made before or after the date of this Quarterly Report on Form 10-Q, irrespective of any general incorporation language contained in such filing.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

INTELSAT S.A.

 Date: November 5, 2020
 By
 /s/ STEPHEN SPENGLER

 Stephen Spengler
Chief Executive Officer

 Date: November 5, 2020
 By
 /s/ DAVID TOLLEY

 David Tolley
Executive Vice President and Chief Financial Officer

Joinder Agreement No. 1

JOINDER AGREEMENT NO. 1, dated as of August 7, 2020 (this "Joinder") to the SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT, dated as of June 17, 2020 (the "Credit Agreement"), among INTELSAT JACKSON HOLDINGS S.A., a *société anonyme* existing under the laws of Luxembourg (the "Borrower"), the subsidiaries of the Borrower from time to time party thereto, as Guarantors (the "Guarantors"), the lending institutions from time to time parties thereto (each, a "Lender" and, collectively, the "Lenders"), Credit Suisse AG, Cayman Islands Branch, as the Administrative Agent and Collateral Agent and Credit Suisse Loan Funding LLC, as Lead Arranger. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned to such terms in the Credit Agreement.

WHEREAS, the Guarantors have entered into the Credit Agreement in order to induce the Administrative Agent, the Collateral Agent, the Lead Arranger and the Lenders to enter into the Credit Agreement and to induce the Lenders to make their respective Loans to the Borrower under the Credit Agreement. Section 9.11 of the Credit Agreement provides that additional Subsidiaries may become Guarantors under the Credit Agreement by execution and delivery of an instrument in the form of this Joinder. Each undersigned Subsidiary (each, a "<u>New Guarantor</u>") is executing this Joinder in accordance with the requirements of the Credit Agreement to become a Guarantor under the Credit Agreement in order to induce the Lenders to make additional Loans and as consideration for Loans previously made.

Accordingly, the Administrative Agent and each New Guarantor agree as follows:

SECTION 1. In accordance with Section 9.11 of the Credit Agreement, each New Guarantor by its signature below hereby becomes a Guarantor under the Credit Agreement with the same force and effect as if originally named therein as a Guarantor, and each New Guarantor hereby (a) agrees to all the terms and provisions of the Credit Agreement applicable to it as a Guarantor thereunder and (b) represents and warrants that the representations and warranties made by it as a Guarantor thereunder are true and correct on and as of the date hereof. Each reference to a Guarantor in the Guarantee shall be deemed to include each New Guarantor. The Guarantee is hereby incorporated herein by reference.

SECTION 2. Each New Guarantor represents and warrants to the Administrative Agent and the other Guaranteed Parties that this Joinder has been duly authorized, executed and delivered by it and, subject to the Final DIP Order (if applicable to such New Guarantor), constitutes its legal, valid and binding obligation, enforceable against it in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, insolvency or similar laws affecting creditors' rights generally and subject to general principles of equity and subject to mandatory Luxembourg law provisions.

SECTION 3. This Joinder may be executed by one or more of the parties to this Joinder on any number of separate counterparts and all of said counterparts taken together shall be deemed to constitute one and the same instrument. Any signature to this Joinder may be delivered by facsimile, electronic mail (including pdf) or any electronic signature complying with the U.S. federal ESIGN Act of 2000 or the New York Electronic Signature and Records Act

or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes to the fullest extent permitted by applicable law. Each of the Parties represents and warrants to the other Parties that it has the corporate capacity and authority to execute this Joinder through electronic means. A set of the copies of this Joinder signed by all the parties shall be lodged with the Borrower and the Administrative Agent. This Joinder shall become effective as to each New Guarantor when the Administrative Agent shall have received counterparts of this Joinder that, when taken together, bear the signatures of such New Guarantor and the Administrative Agent.

SECTION 4. Except as expressly supplemented hereby, the Credit Agreement shall remain in full force and effect.

SECTION 5. THIS JOINDER AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK AND, TO THE EXTENT APPLICABLE, THE BANKRUPTCY CODE.

SECTION 6. Any provision of this Joinder that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof and in the Credit Agreement, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. The parties hereto shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.

SECTION 7. All notices, requests and demands pursuant hereto shall be made in accordance with Section 14.2 of the Credit Agreement. All communications and notices hereunder to each New Guarantor shall be given to it in care of the Borrower at the Borrower's address set forth in Section 14.2 of the Credit Agreement.

SECTION 8. Each New Guarantor agrees to reimburse the Administrative Agent for its out-of-pocket expenses in connection with this Joinder, including the fees, disbursements and other charges of counsel for the Administrative Agent.

IN WITNESS WHEREOF, each New Guarantor and the Administrative Agent have duly executed this Joinder to the Credit Agreement as of the day and year first above written.

Intelsat Velocity Holdings LLC, as a New Guarantor

By: Intelsat Jackson Holdings S.A., its Sole Member

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Chairman & Chief Executive Officer

INTELSAT INVOICE SERVICES LLC, as a New Guarantor

By: Intelsat Jackson Holdings S.A., its Managing Member

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Chairman & Chief Executive Officer

[Signature page to Joinder Agreement]

CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Administrative Agent

By: <u>/s/ William O'Daly</u> Name: William O'Daly Title: Authorized Signatory

By: <u>/s/ D. Andrew Maletta</u> Name: D. Andrew Maletta Title: Authorized Signatory

[Signature page to Joinder Agreement]

AMENDMENT NO. 1

THIS AMENDMENT NO. 1 (this "Amendment"), dated as of August 24, 2020, to the SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT, dated as of June 17, 2020 (as amended, restated, amended and restated, supplemented or otherwise modified from time to time, the "Credit Agreement" and, as amended hereby, the "Amended Credit Agreement"), among INTELSAT JACKSON HOLDINGS S.A., a public limited liability company (société anonyme) existing as société anonyme under the laws of the Grand Duchy of Luxembourg, having its registered office at 4, rue Albert Borschette, L-1246 Luxembourg and registered with the Luxembourg trade and companies' register under number B149.959 and a debtor and debtor in possession under Chapter 11 of the Bankruptcy Code (the "Borrower"), the Subsidiaries of the Borrower from time to time party thereof, as Guarantors, each Guarantor on the date hereof, a debtor and debtor in possession under Chapter 11 of the Bankruptcy Code, the lending institutions from time to time parties hereto (each a "Lender" and, collectively, the "Lenders"), CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Administrative Agent, and CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Collateral Agent.

WITNESSETH

WHEREAS, the Borrower, the Administrative Agent and the Lenders are parties to the Credit Agreement, pursuant to which the Lenders have, subject to the terms and conditions set forth therein, made certain credit available to and on behalf of the Borrower.

WHEREAS, the Borrower is seeking authority from the Bankruptcy Court pursuant to a proposed order (the "*Order*") approving the Designated Target Transaction (as defined in the Amended Credit Agreement), substantially in the form exhibited hereto as <u>Exhibit B</u>, with any changes to such form as are reasonably satisfactory to the Borrower, on the one hand, and the Required Lenders, on the other hand.

WHEREAS, the Borrower has requested certain amendments to the Credit Agreement and the Lenders party hereto who constitute the Required Lenders have agreed to the requested amendments on the terms and conditions set forth herein.

NOW, THEREFORE, the parties hereto agree as follows:

1. <u>Defined Terms</u>. Capitalized terms used herein but not otherwise defined herein shall have the meanings provided to such terms in the Amended Credit Agreement.

2. <u>Amendments</u>. Effective as of the Amendment No. 1 Order Date (as defined below), each of the parties hereto agrees that the Credit Agreement shall be amended to delete the stricken text (indicated textually in the same manner as the following example: stricken text) and to add the double-underlined text (indicated textually in the same manner as the following example: double-underlined text) as set forth in the document attached as Exhibit A hereto.

3. <u>Conditions Precedent</u>. (i) This Amendment shall become a binding obligation of the parties hereto upon the Administrative Agent's receipt of the counterparts of this Amendment executed by the Borrower, the Subsidiary Guarantors and the Lenders constituting Required Lenders (such date, the "*Amendment No. 1 Signing Date*") and (ii) the amendments attached hereto as Exhibit A shall become effective (such date, the "*Amendment No. 1 Order Date*") when the Bankruptcy Court shall have entered the Order.

4. <u>Representations and Warranties</u>. Each Credit Party hereby represents and warrants to the Administrative Agent that on and as of the Amendment No. 1 Signing Date:

(a) (i) it has the corporate or other organizational power and authority to, execute, deliver and carry out the terms and provisions of this Amendment and the Amended Credit Agreement, and has taken all necessary corporate or other organizational action to authorize the execution, delivery and performance of this Amendment and the Amended Credit Agreement and (ii) it has duly executed and delivered this Amendment;

(b) this Amendment and the Amended Credit Agreement constitute the legal, valid and binding obligation of such Credit Party enforceable in accordance with its terms, except as the

enforceability thereof may be limited by bankruptcy, insolvency or similar laws affecting creditors' rights generally and subject to general principles of equity subject to mandatory Luxembourg law provisions;

(c) each of the representations and warranties made by any Credit Party set forth in Section 8 of the Credit Agreement or in any other Credit Document are true and correct in all material respects (except where such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct in all material respects as of such earlier date); and

(d) no Default or Event of Default has occurred and is continuing.

5. <u>Amendment is a "Credit Document"</u>. This Amendment is a Credit Document and all references to a "Credit Document" in the Credit Agreement and the other Credit Documents (including, without limitation, all such references in the representations and warranties in the Credit Agreement and the other Credit Documents) shall be deemed to include this Amendment.

6. <u>Reaffirmation of Obligations</u>. Each Credit Party (a) acknowledges and consents to all of the terms and conditions of this Amendment, (b) affirms all of its obligations under the Credit Documents and (c) agrees that this Amendment and all documents executed in connection herewith do not operate to reduce or discharge such Credit Party's obligations under the Credit Documents.

7. <u>Reaffirmation of Security Interests</u>. Each Credit Party (a) affirms that each of the Liens granted in or pursuant to the Credit Documents are valid and subsisting and (b) agrees that this Amendment shall in no manner impair or otherwise adversely affect any of the Liens granted in or pursuant to the Credit Documents.

8. <u>No Other Changes</u>. Except as modified hereby, all of the terms and provisions of the Credit Documents shall remain in full force and effect.

9. <u>Counterparts</u>; <u>Delivery</u>. This Amendment may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Delivery of an executed counterpart of this Amendment by facsimile or other electronic imaging means shall be effective as an original.

10. <u>Governing Law</u>. This Amendment shall be deemed to be a contract made under, and for all purposes shall be construed in accordance with, the laws of the State of New York.

[Signature Pages Follow]

IN WITNESS WHEREOF, each of the parties hereto has caused a counterpart of this Amendment to be duly executed and delivered as of the date first written above.

BORROWER: INTELSAT JACKSON HOLDINGS S.A.

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Chairman & Chief Executive Officer

SUBSIDIARY GUARANTORS: INTELSAT VENTURES S.À R.L.

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Chairman & Chief Executive Officer

INTELSAT ALIGN S.À R.L.

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Manager

INTELSAT GENESIS INC. INTELSAT GENESIS GP LLC INTELSAT US FINANCE LLC INTELSAT US LLC

By: <u>/s/ Sajid Ajmeri</u> Name: Sajid Ajmeri Title: VP, Corporate & Securities & Assistant Secretary

INTELSAT HOLDINGS LLC INTELSAT SATELLITE LLC INTELSAT LICENSE HOLDINGS LLC INTELSAT LICENSE LLC

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Deputy Chairman

INTELSAT VIRGINIA HOLDINGS LLC

By: <u>/s/ Sajid Ajmeri</u> Name: Sajid Ajmeri Title: VP, Corporate & Securities & Assistant Secretary

PANAMSAT EUROPE CORPORATION INTELSAT ASIA CARRIER SERVICES LLC SOUTHERN SATELLITE LICENSEE LLC SOUTHERN SATELLITE LLC PANAMSAT INDIA LLC INTELSAT SERVICE AND EQUIPMENT LLC INTELSAT INTERNATIONAL EMPLOYMENT LLC

By: <u>/s/ Mark Rasmussen</u> Name: Mark Rasmussen Title: Chairman, President & COO

PANAMSAT INTERNATIONAL SALES, LLC

By: <u>/s/ Mark Rasmussen</u> Name: Mark Rasmussen Title: President and Chief Operating Officer

PANAMSAT INTERNATIONAL HOLDINGS, LLC INTELSAT INTERNATIONAL SYSTEMS, LLC PANAMSAT INDIA MARKETING, L.L.C.

By: <u>/s/ Mark Rasmussen</u> Name: Mark Rasmussen Title: Manager

INTELSAT ALLIANCE LP By: INTELSAT GENESIS GP LLC Its General Partner

By: <u>/s/ David Tolley</u> Name: David Tolley Title: EVP & Chief Financial Officer

INTELSAT FINANCE BERMUDA LTD.

By: <u>/s/ Jean-Philippe Gillet</u> Name: Jean-Philippe Gillet Title: Director

EXECUTED AS A DEED BY INTELSAT SUBSIDIARY (GIBRALTAR) LIMITED ACTING BY TWO DIRECTORS

By: <u>/s/ Jean-Philippe Gillet</u> Name: Jean-Philippe Gillet Title: Director

By: <u>/s/ Matthew Oldham</u> Name: Matthew Oldham Title: Director

INTELSAT GLOBAL SALES & MARKETING LTD.INTELSAT UK FINANCIAL SERVICES LTD.

By: <u>/s/ Jean-Philippe Gillet</u> Name: Jean-Philippe Gillet Title: Chairman

INTELSAT VELOCITY HOLDINGS LLC,

By: INTELSAT JACKSON HOLDINGS S.A., its Sole Member

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Chairman & Chief Executive Officer

INTELSAT INVOICE SERVICES LLC,

By: INTELSAT JACKSON HOLDINGS S.A., its Managing Member

By: <u>/s/ José Toscano</u> Name: José Toscano Title: Chairman & Chief Executive Officer

[LENDERS:]*

By: Name: Title:

*LENDER SIGNATURES ON FILE WITH REGISTRANT

<u>Exhibit A</u>

Amendments to Credit Agreement

[attached]

SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT Dated as of June 17, 2020

(as amended by Amendment No. 1, dated as of August 24, 2020)

among

INTELSAT JACKSON HOLDINGS S.A., as a Debtor and Debtor in Possession under Chapter 11 of the Bankruptcy Code, as Borrower THE SUBSIDIARIES OF INTELSAT JACKSON HOLDINGS S.A. FROM TIME TO TIME PARTY HERETO each a Debtor and Debtor in Possession under Chapter 11 of the Bankruptcy Code, as Guarantors THE LENDERS PARTY HERETO CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Administrative Agent CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Collateral Agent

and CREDIT SUISSE LOAN FUNDING LLC, as Lead Arranger

TABLE OF CONTENTS

<u>Page</u>

SECTION 1 1.1. 1.2. 1.3. 1.4. 1.5. 1.6.	DEFINITIONS.1Defined Terms1Exchange Rates22Reserved22Accounting Terms22Ability to Redesignate22Divisions22
SECTION 2.	AMOUNT AND TERMS OF CREDIT. 23
2.1.	Commitments 23
2.2.	Minimum Amount of Each Borrowing; Maximum Number of Borrowings 23
2.3.	Notice of Borrowing 23
2.4.	Disbursement of Funds 23
2.5.	Repayment of Loans; Evidence of Debt 24
2.6.	Conversions and Continuations 24
2.7.	Pro Rata Borrowings 25
2.8.	Interest 25
2.9.	Interest Periods 25
2.10.	Increased Costs, Illegality, etc. 26
2.11.	Compensation 27
2.12.	Change of Lending Office 27
2.13.	Notice of Certain Costs 27
2.14.	Inability to Determine Rates 27
2.15.	[Reserved] 28
2.16.	MIRE Event 28
~~~~~	
SECTION 3.	PRIORITY AND LIENS. 28
3.1.	DIP Liens 28
3.2.	Primed Liens 29
3.3.	Lien Priority and Perfection 30
3.4. 3.5.	Real Estate Liens30No Discharge; Survival of Claims30
3.6.	Payment of Obligations 30
5.0.	ayment of obligations 50
SECTION 4.	FEES; COMMITMENTS. 30
4.1.	Fees 31
4.2.	Voluntary Reduction of Term Loan Commitments 31
4.3.	Mandatory Termination of Commitments 31
1.5.	
SECTION 5.	PAYMENTS. 32
5.1.	Voluntary Prepayments 32
5.2.	Mandatory Prepayments 32
5.3.	Method and Place of Payment 33
5.4.	Net Payments 33
5.5.	Computations of Interest and Fees 34
5.6.	Limit on Rate of Interest 34
SECTION 6.	CONDITIONS PRECEDENT TO EFFECTIVENESS ON THE CLOSING
<i>C</i> 1	DATE. 35 Executed Counterments of this Agreement 25
6.1. 6.2.	Executed Counterparts of this Agreement 35
6.2. 6.3.	13-Week DIP Budget35Corporate and Other Proceedings35
6.4.	Opinions of Counsel 35
0.4.	

	6.5.	Promissory Notes 35
6.6.	Fees	35
	6.7.	Collateral 35
	6.8.	Perfection Certificate 36
	6.9.	Intercompany Subordination Agreement 36
	6.10.	Petition Date 36
	6.11. 6.12.	Final DIP Order36First Dav Orders36
	6.12. 6.13.	First Day Orders 36 Trustee 36
	6.14.	Material Adverse Effect 36
	6.15.	Patriot Act 36
	6.16.	Liens 36
SECTIO		CONDITIONS PRECEDENT TO ALL CREDIT EVENTS. 36
	7.1	No Default; Representations and Warranties 36
	7.2. 7.3.	Notice of Borrowing37No Violation37
	7.5. 7.4.	No Violation37Fees and Expenses37
	7. <del>4</del> . 7.5.	Participation Election 37
	1.5.	
SECTIO	ON 8.	REPRESENTATIONS, WARRANTIES AND AGREEMENTS. 37
	8.1.	Corporate Status 37
	8.2.	Corporate Power and Authority 37
	8.3.	No Violation 37
	8.4.	Litigation 38
	8.5.	Margin Regulations 38
	8.6.	Governmental Approvals 38
	8.7. 8.8.	Investment Company Act 38
	8.9.	True and Complete Disclosure38Financial Condition; Financial Statements38
	8.9. 8.10.	Tax Returns and Payments 38
	8.11.	Compliance with ERISA 39
	8.12.	Subsidiaries 39
	8.13.	Patents, etc. 39
	8.14.	Environmental Laws 39
	8.15.	Properties 40
	8.16.	Use of Proceeds 40
	8.17.	FCC Licenses, Etc. 40
	8.18.	Satellites 40
	8.19. 8.20.	Centre of Main Interest 40 Orders 40
	8.20.	Status of Obligations; Perfection and Priority of Security Interests 40
	0.21.	Suide of Congulous, Perfection and Phoney of Security Indeeses
SECTIO	DN 9.	AFFIRMATIVE COVENANTS. 40
	9.1	Information Covenants 41
	9.2.	Books, Records and Inspections 44
	9.3.	Maintenance of Insurance 44
	9.4.	Payment of Taxes 45
	9.5.	Consolidated Corporate Franchises 45
	9.6.	Compliance with Statutes, Regulations, etc. 45
	9.7.	ERISA 46
	9.8. 9.9.	Maintenance of Properties46Transactions with Affiliates46
	9.9. 9.10.	End of Fiscal Years; Fiscal Quarters 46
	9.10. 9.11.	Additional Guarantors and Grantors 47
	9.12.	Pledges of Additional Stock and Evidence of Indebtedness 47
		~

9.13.	Use of Proceeds	48
9.14.	Changes in Business	48
9.15.	Further Assurances	48
9.16.	Access and Command Codes	49
9.17.	TTC&M Providers	49
9.18.	[Reserved]	50
9.19.	Government Business Subsid	
9.20.	Post-closing Covenants	50
9.20.	FCC Covenants	50
9.21.	Bankruptcy Matters	50
9.22.	Bankrupicy Matters	30
GEOTION 10		50
SECTION 10.	NEGATIVE COVENANTS.	50
10.1.	Limitation on Indebtedness	50
10.2.	Limitation on Liens	52
10.3.	Limitation on Fundamental C	
10.4.	Limitation on Sale of Assets	53
10.5.	Limitation on Investments	54
10.6.	Limitation on Dividends	56
10.7.	Limitations on Debt Payment	s 56
10.8.	Limitations on Sale Leasebac	ks 56
10.9.	Transfer to Parent Companies	56
10.10.	Additional Bankruptcy Matt	
SECTION 11.	[RESERVED].	57
SECTION 12.	EVENTS OF DEFAULT.	57
12.1.	Payments	57
12.2.	Representations, etc	57
12.3.	Covenants	57
12.4.	Default Under Other Agreem	
12.5.	Bankruptcy, etc	58
12.6.		58
12.0.	Guarantee	58
12.8.	[Reserved]	58
12.0.	Security Agreements	58
12.10.	FCC Matters	58
12.10.	Judgments	58
12.11.	Change of Control	59
12.12.	Dismissal; Conversion; App	
12.13.		
	Superpriority Claims Adverse Claims	59 59
12.15.		
12.16.	Adverse Orders	59
12.17.	Stay Relief	59
12.18.	Orders; Actions	59
12.19.	Pre-Petition Payments	60
12.20.	Adverse Actions	60
12.21.	Reorganization Plan	60
12.22.	Supporting Actions	60
12.23.	Sale Motions	60
12.24.	Acceptable Plan	60
12.25.	First Day Orders	60
OF OTION 12	THE ACENT	(1
SECTION 13.	THE AGENT.	61
13.1.	Appointment and Authority	61
13.2.	Rights as a Lender	61
13.3.	Exculpatory Provisions	61

13.4. 13.5.	Reliance by Agent61Delegation of Duties62
	8
13.6. 13.7.	$\partial$
	8
13.8.	[Reserved]. 63
13.9.	Administrative Agent May File Proofs of Claim 63
13.10.	Indemnification 63
13.11.	Collateral and Guaranty Matters 64
13.12.	[Reserved]. 64
13.13.	[Reserved]. 64
13.14.	Withholding Taxes 64
13.15.	Certain ERISA Matters 64
SECTION 14.	MISCELLANEOUS. 65
14.1.	Amendments and Waivers 65
14.2.	Notices 66
14.3.	No Waiver; Cumulative Remedies 67
14.4.	Survival of Representations and Warranties 67
14.5.	Payment of Expenses and Taxes 67
14.6.	Successors and Assigns; Participations and Assignments 68
14.7.	Replacements of Lenders Under Certain Circumstances 71
14.8.	Adjustments; Set-off 71
14.9.	Counterparts 72
14.10.	Severability 72
14.11.	Integration; Order Controls 72
14.12.	GOVERNING LAW 72
14.13.	Submission to Jurisdiction; Consent to Service; Waivers 72
14.14.	Acknowledgments 73
14.15.	WAIVERS OF JURY TRIAL 73
14.16.	Confidentiality 73
14.17.	Direct Website Communications 74
14.18.	USA PATRIOT Act 74
14.19.	Conversion of Currencies 74
14.20.	Acknowledgement and Consent to Bail-In of EEA Financial Institutions 75
SECTION 15.	GUARANTEE. 75
15.1.	Guarantee 75
15.2.	Right of Contribution 76
15.3.	Right of Set-off 76
15.4.	No Subrogation 76
15.5.	Amendments, etc. with Respect to the Obligations; Waiver of Rights 77
15.6.	Guarantee Absolute and Unconditional 77
15.0	Subardination 70

15.6.	Guarantee Absolute and Un	conditional
15.8.	Subordination	78
	-	

15.9.	Payments	78		
15.11.	[Reserved]	78		
15.12.	Luxembourg Guaran	tee Limitation		78
15.13.	UK Guarantee Limit	ations.	79	

# **ANNEXES**

Annex I Commitments

# **SCHEDULES**

Schedule 1.1(a)	ECA Financing and ECA Collateral
Schedule 1.1(b)	Subsidiary Guarantors

- Schedule 1.1(c) Unrestricted Subsidiaries Schedule 8.12 Subsidiaries Schedule 8.17 FCC Licenses Schedule 8.18 Satellites Schedule 9.9 Existing Affiliate Transactions Schedule 9.20 Post-closing Covenants Schedule 10.1 Indebtedness Schedule 10.2 Liens
- Schedule 10.5 Investments

# **EXHIBITS**

- Exhibit A Form of Final DIP Order
- Exhibit B Form of Interim Cash Collateral Order
- Exhibit C Form of Notice of Borrowings
- Exhibit D Form of Perfection Certificate
- Exhibit E Form of Security and Pledge Agreement
- Exhibit F Form of Assignment and Acceptance Exhibit G Form of Promissory Note

- Exhibit H Form of Joinder Agreement Exhibit I Form of Intercompany Subordination Agreement
- Form of Satellite Health Report Exhibit J

### SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT

This SUPERPRIORITY SECURED DEBTOR IN POSSESSION CREDIT AGREEMENT, is entered into as of June 17, 2020 (as <u>amended by</u> <u>Amendment No. 1, dated as of August 24, 2020 and as further</u> modified from time to time pursuant to the terms hereof, this "<u>Agreement</u>"), among INTELSAT JACKSON HOLDINGS S.A., a public limited liability company (*société anonyme*) existing as *société anonyme* under the laws of the Grand Duchy of Luxembourg, having its registered office at 4, rue Albert Borschette, L-1246 Luxembourg and registered with the Luxembourg trade and companies' register under number B149.959 and a debtor and debtor in possession under Chapter 11 of the Bankruptcy Code (the "<u>Borrower</u>"), the Subsidiaries of the Borrower from time to time party thereof, as Guarantors, each Guarantor on the date hereof, a debtor and debtor in possession under Chapter 11 of the Bankruptcy Code, the lending institutions from time to time parties hereto (each a "Lender" and, collectively, the "Lenders"), CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Administrative Agent, and CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Collateral Agent (such term and each other capitalized term used but not defined in this introductory statement having the meaning provided in Section 1.1).

#### INTRODUCTORY STATEMENTS

WHEREAS, the Borrower, certain Subsidiaries of the Borrower (together with the Borrower, each a "<u>DIP Debtor</u>" and collectively, the "<u>DIP Debtors</u>"), INTELSAT S.A., a public limited liability company (*société anonyme*) existing as *société anonyme* under the laws of the Grand Duchy of Luxembourg, having its registered office at 4, rue Albert Borschette, L-1246 Luxembourg and registered with the Luxembourg trade and companies" register under number B162.135 ("<u>Holdings</u>") and certain Subsidiaries of Holdings other than the DIP Debtors (together with Holdings collectively, the "<u>Parent Companies</u>" and together with the DIP Debtors, each a "<u>Debtor</u>" and collectively, the "<u>Debtors</u>") filed on the Petition Date voluntary petitions with the Bankruptcy Court commencing their respective cases under Chapter 11 of the Bankruptcy Code (the cases of the Debtors, each a "<u>Case</u>" and collectively, the "<u>Cases</u>") and will continue in the possession of their assets and in the management of their business pursuant to Sections 1107 and 1108 of the Bankruptcy Code.

WHEREAS, the Borrower has requested that the Lenders provide it with a term loan facility in the aggregate principal amount equal to \$1,000,000,000 (the "<u>DIP Facility</u>") subject to the conditions set forth herein. All of the Borrower's obligations under the DIP Facility are to be guaranteed by the Guarantors. The Lenders are willing to extend such credit to the Borrower on the terms and subject to the conditions set forth herein.

WHEREAS, the priorities of the DIP Facility and the other Obligations with respect to the Collateral shall be as set forth in the Final DIP Order upon entry thereof by the Bankruptcy Court.

NOW THEREFORE, in consideration of the mutual agreements herein contained and other good and valuable consideration, the sufficiency and receipt of which are hereby acknowledged, the parties hereto hereby agree as follows:

### SECTION 1. Definitions.

#### 1.1. Defined Terms.

(a) As used herein, the following terms shall have the meanings specified in this Section 1.1 (it being understood that defined terms in this Agreement shall include in the singular number the plural and in the plural the singular).

"<u>ABR</u>" shall mean, for any day, a fluctuating rate per annum equal to the highest of (a) the Federal Funds Effective Rate <u>plus</u> ½ of 1.00%, (b) the Prime Rate as in in effect on such day and (c) the LIBOR Rate for a one-month Interest Period on such day (or if such day is not a Business Day, the immediately preceding Business Day) plus 1.00%. Any change in ABR due to a change in the Federal Funds Effective Rate, the Prime Rate or the LIBOR Rate shall be effective on the effective date of such change in the Prime Rate, the Federal Funds Effective Rate or the LIBOR Rate, as the case may be. The Prime Rate is a reference rate and does not necessarily represent the lowest or best rate actually charged to any customer. The Administrative Agent or any other Lender may make commercial loans or other loans at rates of interest at, above or below the Prime Rate.

"ABR Loan" shall mean each Loan bearing interest at the rate provided in Section 2.8(a).

"Acceptable Plan" shall mean a Reorganization Plan that provides for the termination of the Commitments and the Payment in Full of the Obligations (other than contingent indemnification obligations not yet due and payable) on the Plan Effective Date of such Reorganization Plan or is otherwise acceptable to Non-Defaulting

Lenders having or holding at least two-thirds of the sum of (i) the Available Term Loan Commitments and (ii) outstanding principal amount of the Term Loans (excluding Term Loans held by Defaulting Lenders).

"<u>Acquisition</u>" shall mean any transaction or series of related transactions, whether, by purchase, merger, consolidation, contribution or otherwise, for the direct or indirect (x) acquisition of all or substantially all of the property of any Person, or all or substantially all of any business, product line, unit or division of any Person, (y) acquisition of in excess of 50% of the Equity Interests of any Person, and otherwise causing such Person to become a Subsidiary of such Person, or (z) merger or consolidation or any other combination with any Person, in each case, including as a result of any Investment in any Subsidiary that serves to increase the equity ownership of the Borrower or any Restricted Subsidiary therein.

"Administrative Agent" shall mean "Credit Suisse AG, Cayman Islands Branch" and its successors and assigns, as the administrative agent for the Lenders under this Agreement and the other Credit Documents.

"<u>Administrative Agent's Office</u>" shall mean in respect of all Credit Events for the account of the Borrower, the office of the Administrative Agent located at Eleven Madison Avenue, New York, New York, 10010, or such other office as the Administrative Agent may hereafter designate in writing as such to the other parties hereto.

"<u>Administrative Agent Fee Letter</u>" shall mean that certain letter agreement, dated as of June 1, 2020, among the Borrower, the Administrative Agent and Credit Suisse Loan Funding LLC.

"Administrative Expense Claims" shall have the meaning provided in Section 3.2(d).

"Administrative Questionnaire" shall have the meaning provided in Section 14.6(b)(iii).

"Adequate Protection Liens" shall have the meaning provided in the Orders.

"Adequate Protection Payments" shall have the meaning provided in the Orders.

"Affected Financial Institution" shall mean (a) any EEA Financial Institution or (b) any UK Financial Institution.

"<u>Affiliate</u>" shall mean, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under direct or indirect common control with such Person. A Person shall be deemed to control a corporation if such Person possesses, directly or indirectly, the power (a) to vote 10% or more of the securities having ordinary voting power for the election of directors of such corporation or (b) to direct or cause the direction of the management and policies of such corporation, whether through the ownership of voting securities, by contract or otherwise.

"Agent Parties" shall have the meaning provided in Section 14.17(c).

"Agents" shall mean the Lead Arranger, the Administrative Agent and the Collateral Agent.

"<u>Agreement</u>" shall mean this Superpriority Secured Debtor in Possession Credit Agreement, as the same may be amended, supplemented or otherwise modified from time to time.

"<u>Agreement Currency</u>" shall have the meaning provided in Section 14.19(b).

"Amendment No. 1" shall mean that certain Amendment No. 1, dated as of August 24, 2020, among the Borrower, the other Credit Parties party thereto, the Administrative Agent and the Lenders party thereto.

"Applicable ABR Margin" shall mean, at any date, with respect to each ABR Loan, 4.50% per annum.

"<u>Applicable Creditor</u>" shall have the meaning provided in Section 14.19(b).

"Applicable LIBOR Margin" shall mean, at any date, with respect to each LIBOR Loan, 5.50% per annum.

"Applicable Rate" shall mean with respect to ABR Loans, the Applicable ABR Margin, and with respect to LIBOR Term Loans, the Applicable LIBOR Margin.

"Approved Fund" shall have the meaning provided in Section 14.6.

"Asset Sale Prepayment Event" shall mean any sale, transfer or other disposition of any business units, assets or other property of the Borrower or any of the Restricted Subsidiaries not in the ordinary course of business (including any sale, transfer or other disposition of any capital stock of any Subsidiary of the Borrower owned by the Borrower or a Restricted Subsidiary, including any sale or issuance of any capital stock of any Restricted Subsidiary). Notwithstanding the foregoing, the term "Asset Sale Prepayment Event" shall not include any Permitted Sale Leaseback or any transaction permitted by Section 10.4, other than transactions permitted by Sections 10.4(b), (d)(x) and (e).

"Assignment and Acceptance" shall mean an assignment and acceptance substantially in the form of Exhibit F hereto.

"Authorized Officer" shall mean the President, the Chief Financial Officer, the Treasurer, the Controller or any other senior officer of the Borrower designated as such in writing to the Administrative Agent by the Borrower.

"Available Term Loan Commitment" shall mean, on any date of determination, an amount equal to the then unfunded amount of the outstanding Term Loan Commitments on such date, if any.

"Avoidance Actions" shall have the meaning provided in Section 3.1(d).

"Avoidance Proceeds" shall have the meaning provided in Section 3.1(d).

"Backstop Commitment Letter" shall mean that certain Debtor in Possession Term Loan Facility Commitment Letter, dated as of May 13, 2020, by and among the Backstop Parties and the Borrower.

"Backstop Party" shall mean any "Commitment Party" party, as of the Restatement Date, to the Restated Backstop Commitment Letter.

"Bail-In Action" shall mean the exercise of any Write-Down and Conversion Powers by the applicable Resolution Authority in respect of any liability of an Affected Financial Institution.

"<u>Bail-In Legislation</u>" shall mean, (a) with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law; regulation rule or requirement for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, Part I of the United Kingdom Banking Act 2009 (as amended from time to time) and any other law, regulation or rule applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (other than through liquidation, administration or other insolvency proceedings). "<u>Bankruptcy Code</u>" shall mean the Bankruptcy Reform Act of 1978, as heretofore and hereafter amended, and codified in 11 U.S.C. Section 101

et seq.

"<u>Bankruptcy Court</u>" shall mean the United States Bankruptcy Court for the Eastern District of Virginia or any other court having jurisdiction over the Cases from time to time and any Federal appellate court thereof.

"Beneficial Ownership Certification" means a certification regarding beneficial ownership or control as required by the Beneficial Ownership Regulation.

"Beneficial Ownership Regulation" means 31 C.F.R. § 1010.230.

"benefited Lender" shall have the meaning provided in Section 14.8(a).

"Benefit Plan" shall mean any of (a) an "employee benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in and subject to Section 4975 of the Code or (c) any Person whose assets include (for purposes of ERISA Section 3(42) or otherwise for purposes of Title I of ERISA or Section 4975 of the Code) the assets of any such "employee benefit plan" or "plan."

"Board" shall mean the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" shall have the meaning provided in the preamble to this Agreement.

"Borrower Materials" shall mean materials and/or information provided by or on behalf of the Borrower under this Agreement.

"Borrowing" shall mean and include the incurrence of one Type of Term Loan on or after the Closing Date (or resulting from conversions on a given date after the Closing Date) having the same Interest Period (provided that ABR Loans incurred pursuant to Section 2.10(b) shall be considered part of any related Borrowing of LIBOR Term Loans).

"<u>Business Day</u>" shall mean (i) for all purposes other than as covered by clause (ii) below, any day excluding Saturday, Sunday and any day that shall be in The City of New York or the Grand Duchy of Luxembourg a legal holiday or a day on which banking institutions are authorized by law or other governmental actions to close and (ii) with respect to all notices and determinations in connection with, and payments of principal and interest on, LIBOR Loans, any day which is a Business Day described in clause (i) above and which is also a day for trading by and between banks in U.S. dollar deposits in the New York or London interbank eurodollar market.

"C-Band Prepayment Event" shall mean any receipt by Borrower or any of the Restricted Subsidiaries of any proceeds of any C-Band Sweep Payments.

"<u>C-Band Payments</u>" shall mean reimbursement, compensation or other payments or the right to receive reimbursement, compensation or other payments in connection with the transition of the C-band spectrum in connection with (i) reimbursement of relocation costs, (ii) accelerated relocation payments and (iii) payments received for clearing of the C-band spectrum pursuant to private negotiations with third parties, as set forth in the FCC C-Band Rules.

"<u>C-Band Sweep Payments</u>" shall mean reimbursement, compensation or other payments or the right to receive reimbursement, compensation or other payments in connection with the transition of the C-band spectrum in connection with (i) reimbursement of relocation costs and (ii) accelerated relocation payments, as set forth in the FCC C-Band Rules.

"Capital Lease" shall mean, as applied to any Person, any lease of any property (whether real, personal or mixed) by that Person as lessee that, in conformity with GAAP, is, or is required to be, accounted for as a capital lease on the balance sheet of that Person.

"<u>Capitalized Lease Obligations</u>" shall mean, as applied to any Person, all obligations under Capital Leases of such Person or any of its Subsidiaries, in each case taken at the amount thereof accounted for as liabilities in accordance with GAAP.

"<u>Carve Out</u>" shall have the meaning provided in the Final DIP Order.

"Case" or "Cases" shall have the meaning provided in the Introductory Statement.

"Cash Management Agreement" shall mean any agreement to provide cash management services, including treasury, depository, overdraft, credit or debit card, electronic funds transfer and other cash management arrangements.

"<u>Casualty Event</u>" shall mean, with respect to any property (including any Satellite) other than any ECA Collateral of any Person, any loss of or damage to, or any condemnation or other taking by a Governmental Authority of, such property for which such Person or any of its Restricted Subsidiaries receives insurance proceeds, or proceeds of a condemnation award or other compensation.

"<u>Change in Law</u>" shall mean (a) the adoption of any law, treaty, order, policy, rule or regulation after the Signing Date, (b) any change in any law, treaty, order, policy, rule or regulation or in the interpretation or application thereof by any Governmental Authority after the Signing Date or (c) compliance by the Lender with any guideline, request or directive issued or made after the Signing Date by any central bank or other governmental or quasi-governmental authority (whether or not having the force of law). Notwithstanding anything to the contrary herein, it is understood and agreed that the Dodd–Frank Wall Street Reform and Consumer Protection Act (Pub. L. 111-203, H.R. 4173), all requests, rules, guidelines and directives relating thereto, all interpretations and applications thereof and any compliance by a Lender with any request or directive relating thereto, shall, for the purposes of this Agreement, be deemed to be adopted subsequent to the date hereof.

"<u>Change of Control</u>" shall mean the occurrence of any of the following: (a) the Borrower becomes aware of (by way of a report or any other filing pursuant Section 13(d) of the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"), proxy, vote, written notice or otherwise) the acquisition by any Person or group (within the meaning of Section 13(d) (3) or Section 14(d)(2) of the Exchange Act, or any successor provision), including any group acting for the purpose of acquiring, holding or disposing of securities (within the meaning of Rule 13d-5(b)(1) under the Exchange Act), other than the Permitted Holders, in a single transaction or in a related series of transactions, by way of merger, consolidation or other business combination or purchase of beneficial ownership (within the meaning of Rule 13d-3 under the Exchange Act, or any successor provision) of 50% or more of the total voting power of the Voting Stock of the Borrower or any of its direct or indirect parent entities; and/or (b) ICF shall cease to own, directly or indirectly, 100% of the outstanding capital stock of the Borrower.

"<u>Closing Date</u>" shall mean June 17, 2020, the date upon which all conditions precedent set forth in Section 6 are satisfied or waived pursuant to the terms hereof.

"<u>Code</u>" shall mean the Internal Revenue Code of 1986, as amended from time to time, and the regulations promulgated and rulings issued thereunder. Section references to the Code are to the Code, as in effect at the Signing Date, and any subsequent provisions of the Code, amendatory thereof, supplemental thereto or substituted therefor.

"<u>Collateral</u>" shall mean (i) all the "Collateral" (or equivalent term) as defined in any Security Document and also include all other property of whatever kind or nature over which a Lien is granted under any Security Document and (ii) the DIP Collateral, in each case, for the avoidance of doubt, including the C-Band Payments.

"Collateral Agent" shall mean Credit Suisse AG, Cayman Islands Branch.

"<u>Commitment Re-allocation</u>" shall mean the primary syndication of the Term Loan Commitments and/or Term Loans and the assignment of the Term Loan Commitments and/or Term Loans related thereto in connection with the offering of the right to participate in a portion of the Term Loan Commitments and/or Term Loans to the Prepetition Secured Parties (other than the Backstop Parties, the "<u>Other Prepetition Secured Parties</u>") contemplated in the Restated Backstop Commitment Letter.

"<u>Commitment Termination Date</u>" shall mean the earlier of (i) the date of the third Borrowing of the Term Loans upon the funding of such third Borrowing, (ii) the date on which the Term Loan Commitments are reduced to zero and (iii) the Maturity Date.

"Commitments" shall mean, with respect to each Lender, such Lender's Term Loan Commitment.

"Communications" shall have the meaning provided in Section 14.17(a).

"Confidential Information" shall have the meaning provided in Section 14.16.

"<u>Credit Agreement Obligations</u>" shall mean the collective reference to (i) the due and punctual payment of (x) the principal of and premium, if any, and interest at the applicable rate provided in this Agreement (including interest accruing during the pendency of any bankruptcy, insolvency, receivership or other similar proceeding and including, without limitation, in relation to any company incorporated under the laws of the Grand Duchy of

Luxembourg, bankruptcy (*faillite*), insolvency, its voluntary or judicial liquidation (*liquidation volontaire ou judiciaire*), composition with creditors (*concordat préventif de faillite*), reprieve from payment (*sursis de paiement*), controlled management (*gestion contrôlée*), fraudulent conveyance (*actio pauliana*), general settlement with creditors, reorganisation or other similar proceeding, regardless of whether allowed or allowable in such proceeding) on the Loans, when and as due, whether at maturity, by acceleration, upon one or more dates set for prepayment or otherwise, (y) [reserved], and (z) all other monetary obligations, including fees, costs, expenses and indemnities, whether primary, secondary, direct, contingent, fixed or otherwise (including monetary obligations incurred during the pendency of any bankruptcy, insolvency, receivership or other similar proceeding and including, without limitation, in relation to any company incorporated under the laws of the Grand Duchy of Luxembourg, bankruptcy (*faillite*), reprieve from payment (*sursis de paiement*), controlled management (*gestion contrôlée*), fraudulent conveyance (*actio pauliana*), general settlement with creditors, reorganization or other similar proceeding, regardless of whether allowed or allowable in such proceeding), of the Borrower or any other Credit Party to any of the Secured Parties, (ii) the due and punctual performance of all covenants, agreements, obligations and liabilities of each other Credit Documents, and (iii) the due and punctual payment and performance of all the covenants, agreements, obligations and liabilities of each other Credit Party under or pursuant to this Agreement and the other Credit Documents.

"<u>Credit Documents</u>" shall mean this Agreement, the Security Documents, the Orders, <u>Amendment No. 1</u> and any promissory notes issued by the Borrower hereunder.

"Credit Event" shall mean and include the making (but not the conversion or continuation) of a Loan.

"Credit Party" shall mean each of the Borrower, the Subsidiary Guarantors and each other Subsidiary of the Borrower that is a party to a Credit Document. For the avoidance of doubt, each DIP Debtor shall be a Credit Party hereunder.

"Credit Suisse" shall mean Credit Suisse Loan Funding LLC and Credit Suisse AG, Cayman Islands Branch (acting through such of its affiliates or branches as it deems appropriate).

"<u>Creditors' Committee</u>" mean any statutory committee of unsecured creditors appointed in the Cases pursuant to Bankruptcy Code section 1102. "Debtor" shall have the meaning provided in the Introductory Statement.

"Debt Incurrence Prepayment Event" shall mean any issuance or incurrence by the Borrower or any of the Restricted Subsidiaries of any Indebtedness but excluding any Indebtedness permitted to be issued or incurred under Section 10.1.

"Debtor Relief Laws" shall mean the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" shall mean any event, act or condition that with notice or lapse of time, or both, would constitute an Event of Default.

"Defaulting Lender" shall mean, subject to Section 3.8(b), any Lender that (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans, within five Business Days of the date required to be funded by it hereunder unless such failure is the result of such Lender's good faith determination that a condition precedent (specifically identified and including the particular default, if any) to funding a loan under this Agreement cannot be satisfied, (b) has notified the Borrower and the Administrative Agent that it does not intend to comply with its funding obligations or has made a public statement to that effect with respect to its funding obligations hereunder or under other agreements generally in which it commits to extend credit unless such writing or public statement indicates that such position is based on such Lender's good faith determination that a condition precedent (specifically identified and including the particular default, if any) to funding a loan under this Agreement cannot be satisfied, (c) has failed, within five Business Days after request by the Administrative Agent, to confirm in a manner satisfactory to the Administrative Agent that it will comply with its funding obligations unless such failure is the result of such Lender's good faith determination precedent (specifically identified and including a loan under this Agreement cannot be satisfied, if any) to funding a loan under this Agreement cannot be satisfied, if any to funding a loan under this Agreement cannot be satisfied, if any) to funding a loan under this Agreement cannot be satisfied, if any to funding a loan under this Agreement cannot be satisfied, if any) to funding a loan under this Agreement cannot be satisfied, if any to funding a loan under this Agreement cannot be satisfied, if any to funding a loan under this Agreement cannot be satisfied, if any to funding a loan under this Agreement cannot be satisfied, or (d) has, or has a direct or indirect parent company that has, (i) become th

Lender or any direct or indirect parent company thereof by a Governmental Authority or (iv) become the subject of a Bail-In Action.

"Designated Target Acquisition Vehicles" shall mean, collectively, any Restricted Subsidiary of the Borrower formed or used to consummate the Designated Target Transaction (excluding the Designated Target Entities and the subsidiaries thereof).

"Designated Target Buyer" shall mean the "Buyer" as used and defined in the Designated Target Purchase Agreement.

"Designated Target Intercompany Loan" shall mean a secured intercompany loan from one or more DIP Debtors to one or more Designated Target Subsidiaries to fund the consummation of the Designated Target Transaction in accordance with the terms of the Designated Target Transaction Agreements (it being understood, for the avoidance of doubt, such amount shall not include any amount for any Investments in the Designated Target Subsidiaries for working capital purposes other than the working capital and cash adjustments to the purchase price in accordance with the Designated Target Transaction Agreements); provided that such intercompany loan shall be borrowed and/or guaranteed by all of the Designated Target Subsidiaries that are wholly-owned Domestic Subsidiaries and secured, on a first lien basis, by substantially all assets of the borrower and guarantors party thereunder, in each case subject to customary exceptions, grace periods, limitations and materiality thresholds; provided further that the guarantees and security in favor of such intercompany loan shall not be released or terminated in connection with any incurrence of the Designated Target Working Capital Facility by any Designated Target Subsidiary.

"Designated Target Purchase Agreement" shall mean that certain Purchase and Sale Agreement, to be entered into by and among Designated Target Seller, Designated Target Buyer and the other parties thereto, including all schedules, exhibits and annexes thereto, in substantially the form filed in connection with that certain Motion For Entry Of An Order (I) Authorizing The Debtors To (A) Consummate A Proposed Transaction And (B) Enter Into An Amendment To The Dip Credit Agreement And (II) Granting Related Relief filed with the Bankruptcy Court on August 24, 2020 and consistent with the draft purchase and sale agreement delivered to Akin Gump Strauss Hauer & Feld LLP and Jones Day by Kirkland & Ellis LLP on August 24, 2020 at 9:52 p.m. New York City time (the "Agreed Form of Designated Target Purchase Agreement"), together with any changes to the Agreed Form of Designated Target Purchase to the Agreed Form of Designated Target Buyer and Designated Target Seller; provided that Designated Target Buyer and the Borrower shall not agree to any change to the Agreed Form of Designated Target Purchase Agreement that is materially adverse to the Lenders without the prior written consent of the Required Lenders.

"Designated Target Seller" shall mean the "Seller" as used and defined in the Designated Target Purchase Agreement.

"Designated Target Subsidiaries" shall mean, collectively, (i) the Designated Target Acquisition Vehicles and (ii) the Designated Target Entities and their respective subsidiaries that shall become the Restricted Subsidiaries of the Borrower upon the consummation of the Designated Target Transaction.

"Designated Target Entities" shall mean the Designated Target Transferred Entities and their respective Subsidiaries; provided, that, for the avoidance of doubt, the Designated Target Entities that constitute Subsidiaries shall become Restricted Subsidiaries of the Borrower upon the consummation of the Designated Target Transaction.

"Designated Target Transaction" shall mean the purchase by Designated Target Buyer of 100% of the issued and outstanding equity interests of the Designated Target Transferred Entities pursuant to the Designated Target Transaction Agreements and the payment of all fees, costs and expenses incurred or payable by the Borrower or its Restricted Subsidiaries in connection therewith.

"Designated Target Transaction Agreements" shall mean, collectively, (i) the Designated Target Purchase Agreement and (ii) each of the other agreements, documents or certificates to be entered into in connection with the closing of the transactions contemplated thereby, including, without limitation, (A) the Transition Services Agreement, (B) the Commercial Agreements and (C) the IP License Agreement, in each case (A)–(C) as such term is defined in the Designated Target Purchase Agreement and in form and substance agreed to by Designated Target Buyer and Designated Target Seller in accordance with and consistent with the terms provided in the Designated Target Purchase Agreement, including the respective forms and term sheets attached thereto; provided that Designated Target Buyer and the Borrower shall not agree to any change to such forms that is materially adverse to the Lenders without the prior written consent of the Required Lenders.

"Designated Target Transaction Order" shall mean an order of the Bankruptcy Court approving the Designated Target Transaction, substantially in the form exhibited to Amendment No. 1 as Exhibit B thereto, with any changes to such form as are satisfactory to the Borrower, on the one hand, and the Required Lenders, on the other hand.

<u>"Designated Target Transaction Order Date" shall mean the date on which the Designated Target Transaction Order is entered by the Bankruptcy</u> Court.

"Designated Target Transferred Entities" shall mean Gogo LLC, a Delaware limited liability company, and Gogo International Holdings LLC, a Delaware limited liability company.

"Designated Target Working Capital Facility" shall mean Indebtedness of any Designated Target Subsidiary incurred, on or prior to the consummation of the Designated Target Transaction, to finance the working capital of the Designated Target Subsidiaries; provided that such Indebtedness is not incurred or guaranteed in any respect by Borrower or any Restricted Subsidiary (other than the Designated Target Subsidiaries) and otherwise has no recourse against the Borrower or any Restricted Subsidiary (other than the Designated Target Subsidiaries) or any of their assets.

"<u>DIP Budget</u>" shall mean a projected statement of sources and uses of cash for the Borrower and the Guarantors on a weekly basis for the following 13 calendar weeks. As used herein, "DIP Budget" shall initially refer to the initial 13-week projection delivered in accordance with Section 6.2 and thereafter shall refer to the most recent 13-week projection delivered by the Borrower in accordance with Section 9.1(c).

"DIP Collateral" shall have the meaning provided in Section 3.1.

"<u>DIP Debtor</u>" shall have the meaning provided in the Introductory Statement.

"DIP Facility" shall have the meaning provided in the Introductory Statements.

"DIP Liens" shall have the meaning provided in Section 3.1.

"<u>DIP Superpriority Claims</u>" shall have the meaning provided in Section 3.1(d).

"Disqualified Lenders" means those competitors of the Borrower and its Subsidiaries separately identified in writing (including by email) as such by the Borrower to the Backstop Parties on or prior to the Petition Date (and provided to Credit Suisse on or prior to the Closing Date) or by the Borrower to the Administrative Agent (for the distribution to Lenders) from time to time following the Closing Date and any of its Affiliates that are either (x) separately identified in writing (including by email) by the Borrower to the Administrative Agent (for the distribution to Lenders) from time to time or (y) known or reasonably identifiable as Affiliates; provided that, notwithstanding the foregoing, in no event shall any Prepetition Secured Parties (or any Affiliate thereof) that participates in the Commitment Re-Allocation be a Disqualified Lender.

"Disqualified Preferred Stock" shall mean any preferred capital stock or preferred equity interest of the Borrower.

"dividends" shall have the meaning provided in Section 10.6.

"<u>Division</u>" shall have the meaning provided in Section 1.06.

"Dollar Equivalent" shall mean, on any date of determination, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in any Foreign Currency, the equivalent in Dollars of such amount, determined by the Administrative Agent at such time on the basis of the Exchange Rate (determined in respect of the most recent Revaluation Date) for the purchase of Dollars with such Foreign Currency. "Dollars" and "\$" shall mean dollars in lawful currency of the United States of America.

"<u>Domestic Subsidiary</u>" shall mean each Subsidiary of the Borrower that is organized under the laws of the United States, any state or territory thereof, or the District of Columbia.

"Drawing" shall have the meaning provided in Section 3.4(b).

"ECA Collateral" shall mean have the meaning provided in Section 9.15(b).

"<u>ECA Financing</u>" shall mean any Indebtedness including Indebtedness owing to or otherwise supported by any export credit agency, in each case, which was incurred by the Borrower and/or its Subsidiaries prior to the Signing Date and set forth on <u>Schedule 1.1(A)</u> to finance (i) the acquisition (by purchase, lease or otherwise) construction or improvement of a Satellite in an aggregate principal amount, at any time outstanding, not to exceed \$300.0 million and/or (ii) certain services in connection with the launch of a Satellite in an aggregate principal amount, at any time outstanding, not to exceed \$500.0 million.

"EEA Financial Institution" shall mean (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" shall mean any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" shall mean any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any EEA Financial Institution.

"Environment" shall mean ambient air, indoor air, surface water, groundwater, drinking water, land surface and subsurface strata and natural resources such as wetlands.

"Environmental Claims" shall mean any and all actions, suits, orders, decrees, demands, demand letters, claims, Liens, notices of noncompliance, violation or potential responsibility or investigation (other than internal reports prepared by the Borrower or any of the Subsidiaries (a) in the ordinary course of such Person's business or (b) as required in connection with a financing transaction or an acquisition or disposition of real estate) or proceedings relating in any way to any Environmental Law or any permit issued, or any approval given, under any such Environmental Law (hereinafter, "Claims"), including, without limitation, (i) any and all Claims by governmental or regulatory authorities for enforcement, cleanup, removal, response, remedial or other actions or damages pursuant to any applicable Environmental Law and (ii) any and all Claims by any third party seeking damages, contribution, indemnification, cost recovery, compensation or injunctive relief relating to the presence, release or threatened release of Hazardous Materials or arising from alleged injury or threat of injury to health or safety (to the extent relating to human exposure to Hazardous Materials), or the environment including, without limitation, ambient air, surface water, groundwater, land surface and subsurface strata and natural resources such as wetlands.

"Environmental Law" shall mean any applicable Federal, state, foreign or local statute, law, rule, regulation, ordinance, code and rule of common law now or hereafter in effect and in each case as amended, and any binding judicial or administrative interpretation thereof, including any binding judicial or administrative order, consent decree or judgment, relating to the protection of environment, including, without limitation, ambient air, surface water, groundwater, land surface and subsurface strata and natural resources such as wetlands, or human health or safety (to the extent relating to human exposure to Hazardous Materials), or Hazardous Materials.

"Equity Interests" shall mean capital stock and all warrants, options or other rights to acquire capital stock (but excluding any debt security that is convertible into, or exchangeable for, capital stock).

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder (unless the context requires otherwise).

"<u>ERISA Affiliate</u>" shall mean each person (as defined in Section 3(9) of ERISA) that together with the Borrower or a Subsidiary would be deemed to be a "single employer" within the meaning of Section 414(b) or (c) of the Code or, solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code.

"EU Bail-In Legislation Schedule" shall mean the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor Person), as in effect from time to time.

"Europs" or " $\underline{\mathbb{C}}$ " shall mean the single currency of the European Union as constituted by the treaty establishing the European Community being the Treaty of Rome, as amended from time to time.

"Event of Default" shall have the meaning provided in Section 12.

"Exchange Rate" for a currency shall mean the rate shown on the Bloomberg currency page for such currency or, if the same does not provide such exchange rate, by reference to such other publicly available service for displaying exchange rates as may be reasonably selected by the Administrative Agent or, in the event no such service is selected, on such other basis as is reasonably satisfactory to the Administrative Agent).

"Excluded Satellite" shall mean any Satellite (or, if the entire Satellite is not owned by the Borrower or any of its Subsidiaries, as the case may be, the portion of the Satellite it owns or for which it has risk of loss) (i) that is not expected or intended in the good faith determination of the Borrower to earn revenues from the operation of such Satellite (or portion, as applicable) in excess of \$75,000,000 for the immediately succeeding 12-month calendar period or (ii) that has a net book value not in excess of \$200,000,000 or (iii) that (1) the procurement of In-Orbit Insurance therefor in the amounts and on the terms required by Section 9.3 would not be available for a price that is, and on other terms and conditions that are, commercially reasonable or (2) the procurement of such In-Orbit Insurance therefor would be subject to exclusions or limitations of coverage that would make the terms of the insurance commercially unreasonable, in either case, in the good faith determination of the Borrower, or (iv) for which In-Orbit Contingency Protection is available or (v) whose primary purpose is to provide In-Orbit Contingency Protection for the satellites of the Borrower, to earn revenues from the operation of such Satellite (or portion, as applicable) in excess of \$75,000,000 for the immediately succeeding 12-month calendar period.

"Excluded Taxes" shall mean (a) with respect to the Administrative Agent or any Lender, net income taxes, franchise taxes and capital taxes (imposed in lieu of net income taxes) imposed in each case as a result of the Administrative Agent or such Lender (as applicable) being organized in, or having its principal office or applicable lending office in, such jurisdiction imposing such tax or any political subdivision or taxing authority thereof or

therein, (b) any Tax to the extent attributable to such Lender's failure to comply with Section 5.4(d),(c) the Luxembourg law of December 23, 2005, and (d) any withholding Taxes imposed under FATCA.

"<u>Fair Market Value</u>" of a specified asset shall mean the fair market value of assets as determined in good faith by the Borrower and (i) in the event the specified asset has a Fair Market Value in excess of \$20,000,000, shall be set forth in a certificate of an Authorized Officer or (ii) in the event the specified asset has a Fair Market Value in excess of \$50,000,000, shall be set forth in a resolution approved by a majority of the board of directors of the Borrower.

"<u>FATCA</u>" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable), any current or future regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the Code and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities and implementing such Sections of the Code.

"FCC" shall mean the Federal Communications Commission or any governmental authority substituted therefor.

"<u>FCC C-Band Rules</u>" shall mean Expanding Flexible Use of the 3.7 to 4.2 GHz Band, GN Docket No. 18-122, Report and Order and Order of Proposed Modification, FCC 20-22 (2020) and the rules adopted therein, to be codified in Title 47 of the Code of Federal Regulations, together with all other current or subsequently adopted FCC rules, orders and public notices pertaining to expanding flexible use of the 3.7 to 4.2 GHz band.

"FCC Licenses" shall mean all authorizations, orders, licenses and permits issued by the FCC to the Borrower or any of its Restricted Subsidiaries, under which the Borrower or any of its Restricted Subsidiaries is authorized to launch and operate any of its Satellites or to operate any of its transmit only, receive only or transmit and receive earth stations.

"<u>Federal Funds Effective Rate</u>" shall mean, for any day, the rate *per annum* equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; <u>provided</u> that (a) if such day is not a Business Day, the Federal Funds Effective Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on any Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three (3) Federal funds brokers of recognized standing selected by the Administrative Agent.

"Fees" shall mean all amounts payable pursuant to, or referred to in, Section 4.1.

"<u>Final DIP Order</u>" shall mean an order of the Bankruptcy Court approving the DIP Facility and authorizing the Debtors' use of cash collateral, substantially in the form exhibited hereto as <u>Exhibit A</u>, with any changes to such form as are satisfactory to the Borrower, on the one hand, and the Required Backstop Parties or, with respect to matters described hereunder that requires consent from or shall be acceptable or satisfactory to each Lender or

each Lender directly and adversely affected thereby, each Backstop Party or each Backstop Party directly and adversely affected thereby, on the other hand. "<u>Final DIP Order Entry Date</u>" shall mean the date on which the Final DIP Order is entered by the Bankruptcy Court.

"First Day Orders" shall mean all orders entered by the Bankruptcy Court on or, within five Business Days of, the Petition Date, based on motions filed by the Debtors on or about the Petition Date which orders, for the avoidance of doubt, shall include the Interim Cash Collateral Order.

"First Maturity Extension" shall have the meaning provided in the definition of "Scheduled Maturity Date."

"First Extension Fee" shall have the meaning provided in Section 4.1(d).

"Flood Insurance Laws" shall mean, collectively, (i) the National Flood Insurance Act of 1968 as now or hereafter in effect or any successor statute thereto, (ii) the Flood Disaster Protection Act of 1973 as now or hereafter in effect or any successor statue thereto, (iii) the National Flood Insurance Reform Act of 1994 as now or hereafter in effect or any successor statute thereto and (iv) the Flood Insurance Reform Act of 2004 as now or hereafter in effect or any successor statute thereto.

"Foreign Currency" shall mean Euro.

"Foreign Plan" shall mean any employee benefit plan, program, fund, policy, arrangement or agreement maintained or contributed to by the Borrower or any of its Subsidiaries with respect to employees employed outside the United States.

"Foreign Subsidiary" shall mean each Subsidiary of the Borrower that is not a Domestic Subsidiary.

"GAAP" shall mean generally accepted accounting principles in the United States of America, as in effect from time to time; provided, however, that if the Borrower notifies the Administrative Agent that the Borrower

requests an amendment to any provision hereof to eliminate the effect of any change occurring after the Signing Date in GAAP or in the application thereof on the operation of such provision (or if the Administrative Agent notifies the Borrower that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such change in GAAP or in the application thereof, then such provision shall be interpreted on the basis of GAAP as in effect and applied immediately before such change shall have become effective until such notice shall have been withdrawn or such provision amended in accordance herewith. Notwithstanding any other provision contained herein, the amount of any Indebtedness under GAAP with respect to Capital Lease Obligations shall be determined providing that all obligations of any Person that are or would have been characterized as an operating lease as determined in accordance with GAAP as in effect prior to February 25, 2016 (whether or not such operating lease was in effect on such date) shall continue to be accounted for as an operating lease (and not as a Capital Lease or Capital Lease Obligation) for purposes of the definition of "Indebtedness" under this Agreement regardless of any change in GAAP following such date that would otherwise require such obligation to be recharacterized as a Capital Lease Obligation, to the extent that financial reporting shall not be affected hereby.

"Government Business Subsidiary" shall mean any Restricted Subsidiary of the Borrower, including Intelsat General Corporation and its Subsidiaries for so long as Intelsat General Corporation is a Restricted Subsidiary of the Borrower, that (i) is engaged primarily in the business of providing services to customers similar to the services provided on the Signing Date by Intelsat General Corporation and services or activities that are reasonably similar thereto or a reasonable extension, development or expansion thereof, or is complementary, incidental, ancillary or related thereto and (ii) is subject to the Proxy Agreement or a substantially similar agreement substantially restricting the Borrower's control of such Restricted Subsidiary.

"<u>Governmental Authority</u>" shall mean the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Guarantee" shall mean the guaranty made by the Guarantors pursuant to Section 15 hereof, and any guaranty or guaranty supplement delivered with respect thereto (including by way of any Joinder Agreement).

"<u>Guarantee Obligations</u>" shall mean, as to any Person, any obligation of such Person guaranteeing or intended to guarantee any Indebtedness of any other Person (the "<u>primary obligor</u>") in any manner, whether directly or indirectly, including any obligation of such Person, whether or not contingent, (a) to purchase any such Indebtedness or any property constituting direct or indirect security therefor, (b) to advance or supply funds (i) for the purchase or payment of any such Indebtedness or (ii) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (c) to purchase property, securities or services primarily for the purpose of assuring the owner of any such Indebtedness against loss in respect thereof; <u>provided</u>, <u>however</u>, that the term "Guarantee Obligations" shall not include endorsements of instruments for deposit or collection in the ordinary course of business or standard contractual indemnities or guarantees (including performance guarantees) that are not direct guarantees of payments of Indebtedness. The amount of any Guarantee Obligation shall be deemed to be an amount equal to the stated or determinable amount of the Indebtedness in respect of which such Guarantee Obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determined by such Person in good faith.

### "Guarantor" shall mean each Subsidiary Guarantor.

"<u>Hazardous Materials</u>" shall mean (a) any petroleum or petroleum products, radioactive materials, friable asbestos, urea formaldehyde foam insulation, transformers or other equipment that contain dielectric fluid containing regulated levels of polychlorinated biphenyls, and radon gas; (b) any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous waste," "hazardous materials," "extremely hazardous waste," "restricted hazardous waste," "toxic substances," "toxic pollutants," "contaminants" or "pollutants," or words of similar import, under any applicable Environmental Law; and (c) any other chemical, material or substance, which is prohibited, limited or regulated by any Environmental Law.

"Hedge Agreements" shall mean interest rate swap, cap or collar agreements, interest rate future or option contracts, currency swap agreements, currency future or option contracts, commodity price protection agreements or other commodity price hedging agreements, and other similar agreements entered into by the Borrower or any Restricted Subsidiary that is a Credit Party in the ordinary course of business (and not for speculative purposes) in

order to protect the Borrower or any of such Restricted Subsidiaries against fluctuations in interest rates, currency exchange rates or commodity prices.

"<u>Historical Financial Statements</u>" shall mean, as of the Signing Date, the audited financial statements of the Borrower and its Subsidiaries, for the immediately preceding three fiscal years, consisting of balance sheets and the related consolidated statements of income, stockholders' equity and cash flows for such fiscal years.

Holdings" shall have the meaning given in the Introductory Statement to this Agreement.

"IFRS" shall mean the International Financial Reporting Standards set by the International Accounting Standards Board (or the Financial Accounting Standards Board of the American Institute of Certified Public Accountants or the SEC, as the case may be) or any successor thereto, as in effect from time to time; provided, however, that if the Borrower notifies the Administrative Agent that the Borrower requests an amendment to any provision hereof to eliminate the effect of any change occurring after the Signing Date in IFRS or in the application thereof on the operation of such provision (or if the Administrative Agent notifies the Borrower that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such change in IFRS or in the application thereof, then such provision shall be interpreted on the basis of IFRS as in effect and applied immediately before such change shall have become effective until such notice shall have been withdrawn or such provision amended in accordance herewith. Notwithstanding any other provision contained herein, the amount of any Indebtedness under IFRS with respect to Capital Lease Obligations shall be determined providing that all obligations of any Person that are or would have been characterized as an operating lease as determined in accordance with IFRS (IAS) as in effect prior to January 13, 2016 (whether or not such operating lease was in effect on such date) shall continue to be accounted for as an operating lease (and not as a Capital Lease Obligation) for purposes of the definition of "Indebtedness" under this Agreement regardless of any change in IFRS following such date that would otherwise require such obligation to be recharacterized as a Capital Lease Obligation, to the extent that financial reporting shall not be affected hereby.

"In-Orbit Contingency Protection" shall mean transponder capacity that, in the good faith determination of the Borrower, is available on a contingency basis from the Borrower or its Restricted Subsidiaries, or any Subsidiary of any parent of the Borrower, directly or from another satellite operator pursuant to a contractual arrangement, to accommodate the transfer of traffic representing at least 25% of the revenue-generating capacity with respect to any Satellite (or, if the entire Satellite is not owned by the Borrower or any of its Restricted Subsidiaries, as the case may be, the portion of the Satellite it owns or for which it has risk of loss) that may suffer actual or constructive total loss and that meets or exceeds the contractual performance specifications for the transponders that had been utilized by such traffic; it being understood that the Satellite (or portion, as applicable) shall be deemed to be insured for a percentage of the Satellite's (or applicable portion's) net book value for which In-Orbit Contingency Protection is available.

"In-Orbit Insurance" shall mean, with respect to any Satellite (or, if the entire Satellite is not owned by the Borrower or any of its Restricted Subsidiaries, as the case may be, the portion of the Satellite it owns or for which it has risk of loss), insurance (subject to a right of coinsurance in an amount up to \$150,000,000) or other contractual arrangement providing for coverage against the risk of loss of or damage to such Satellite (or portion, as applicable) attaching upon the expiration of the launch insurance therefor (or, if launch insurance is not procured, upon the initial completion of in-orbit testing) and attaching, during the commercial in-orbit service of such Satellite (or portion, as applicable), upon the expiration of the immediately preceding corresponding policy or other contractual arrangement, as the case may be, subject to the terms and conditions set forth herein.

"<u>In-Orbit Satellite</u>" shall mean a Satellite owned by the Borrower or any of its Restricted Subsidiaries that has been launched (or, if the entire Satellite is not owned by the Borrower or any of its Restricted Subsidiaries, the portion of the Satellite the Borrower and/or such Restricted Subsidiary owns), excluding any such Satellite that has been decommissioned or that has otherwise suffered a constructive or actual total loss.

"Indebtedness" of any Person shall mean, without duplication, (A) all indebtedness of such Person for borrowed money, (B) the deferred purchase price of assets or services that in accordance with GAAP would be included as liabilities in the balance sheet of such Person but excluding deferred rent to the extent not capitalized in accordance with GAAP, (C) the face amount of all letters of credit issued for the account of such Person and, without duplication, all drafts drawn thereunder, (D) all Indebtedness of a second Person secured by any Lien on any property owned by such first Person, whether or not such Indebtedness has been assumed, (E) all Disqualified Preferred Stock of such Person, (F) all obligations of such Person under interest rate swap, cap or collar agreements, interest rate future or option contracts, currency swap agreements, currency future or option contracts, commodity price protection agreements or other commodity price hedging agreements and other similar agreements, (G) without duplication, all Guarantee Obligations of such Person, (H) purchase price adjustments (that are in the nature of earn outs or similar deferred purchase price mechanisms not described in clause (iv) of the proviso to this

definition) in connection with acquisitions or sales of assets and/or businesses effected in accordance with the requirements of this Agreement, (I) to the extent not otherwise included, with respect to the Borrower and its Restricted Subsidiaries, the amount then outstanding (i.e., received by, and available for use by, the Borrower or any of its Restricted Subsidiaries) under any receivables financing (as set forth in the books and records of Borrower or any of its Restricted Subsidiaries and confirmed by the agent, trustee or other representative of the institution or group providing such receivables financing) and (J) all obligations of such Person in respect of Disqualified Preferred Stock; provided that Indebtedness shall not include (i) trade payables and accrued expenses, in each case payable directly or through a bank clearing arrangement and arising in the ordinary course of business, (ii) obligations under Satellite Purchase Agreements, launch service agreements, in each case, not overdue by more than 90 days, (iii) deferred or prepaid revenue, (iv) purchase price holdbacks in respect of a portion of the purchase price of an asset to satisfy warranty or other unperformed obligations of the respective seller and (v) obligations to make payments to one or more insurers under satellite insurance policies or post-closing working capital adjustments in respect of premiums or the requirement to remit to such insurer(s) a portion of the future revenues generated by a Satellite which has been declared a constructive total loss, in each case in accordance with the terms of the insurance policies relating thereto. Notwithstanding the foregoing, (i) Indebtedness shall not include, and shall be calculated without giving effect to, the effects of the Financial Accounting Standards Board Accounting Standards Codification 815 and related interpretations to the extent such effects would otherwise increase or decrease an amount of Indebtedness for any purpose under this Agreement as a result of accounting for any embedded derivatives created by the terms of such Indebtedness; and any such amounts that would have constituted Indebtedness under this Agreement but for the application of this sentence shall not be deemed an incurrence of Indebtedness under this Agreement and (ii) the amount of any Indebtedness that is non-recourse to the cash flows or assets of the Borrower and its Restricted Subsidiaries (other than the assets securing such Indebtedness and proceeds thereof) shall be deemed to be the lesser of the face amount of such Indebtedness and the fair market value of the collateral securing such Indebtedness.

- "Indemnified Taxes" shall mean all Taxes (other than Excluded Taxes).
- "Initial Maturity Date" shall have the meaning provided in the definition of "Scheduled Maturity Date."
- "Intercompany Subordination Agreement" shall have the meaning provided in Section 9.12(c).

"Interest Period" shall mean, with respect to any Term Loan, the interest period applicable thereto, as determined pursuant to Section 2.9.

"Interim Cash Collateral Order" shall mean an order to be entered by the Bankruptcy Court approving the Debtors' use of cash collateral of the Prepetition Secured Parties and the provision of adequate protection to the Prepetition Secured Parties, including the Adequate Protection Payments, on an interim basis and in the form attached hereto as <u>Exhibit B</u>, with any changes to such form as are satisfactory to the Borrower and the Required Backstop Parties.

"Interpolated Rate" shall mean, at any time, the rate per annum reasonably determined by the Administrative Agent (which determination, as to any Lender, shall be conclusive and binding absent manifest error) to be equal to the rate that results from interpolating on a linear basis between: (a) the LIBOR Screen Rate for the longest period (for which the LIBOR Screen Rate for the applicable currency is available) that is shorter than the Interest Period and (b) the LIBOR Screen Rate for the shortest period (for which the LIBOR Screen Rate for the applicable currency is available) that exceeds the Interest Period, in each case, as of the applicable quotation time in the applicable currency.

"Investment" shall mean, for any Person: (a) the acquisition (whether for cash, property, services or securities or otherwise) of capital stock, bonds, notes, debentures, partnership or other ownership interests or other securities of any other Person (including any "short sale" or any sale of any securities at a time when such securities are not owned by the Person entering into such sale); (b) the making of any deposit with (other than demand deposits with commercial banks made in the ordinary course of business), or advance, loan or other extension of credit to, any other Person (including the purchase of property from another Person subject to an understanding or agreement, contingent or otherwise, to resell such property to such Person), but excluding any such advance, loan or extension of credit having a term not exceeding 364 days arising in the ordinary course of business and excluding also any Investment in leases entered into in the ordinary course of business; or (c) the entering into of any guarantee of, or other contingent obligation with respect to, Indebtedness or other monetary liability of any other Person.

"Joinder Agreement" shall mean an agreement substantially in the form of Exhibit H hereto.

"Joint Ventures" shall mean any Person in which the Borrower or a Restricted Subsidiary maintains an equity investment (including those formed for the purpose of selling or leasing transponders or transponder capacity to third party customers in the ordinary course of business of the Borrower and its Restricted Subsidiaries), but which is not a Subsidiary of the Borrower.

"Judgment Currency" shall have the meaning provided in Section 14.19(b).

"Lead Arranger" shall mean Credit Suisse Loan Funding LLC.

"Lender" shall have the meaning provided in the preamble to this Agreement.

"LIBOR Loan" shall mean any LIBOR Term Loan.

"LIBO Rate" with respect to any LIBOR Loan for any Interest Period, the rate per annum equal to the London interbank offered rate administered by ICE Benchmark Administration Limited, or any other person which takes over the administration of that rate, (such page currently being the LIBOR01 page) for deposits in dollars (as set forth by any service selected by the Administrative Agent that has been nominated by ICE Benchmark Administration Limited as an authorized vendor for the purpose of displaying such rates) at approximately 11:00 a.m., London time, two (2) Business Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period (the "LIBOR Screen Rate"); provided that, if the LIBO Screen Rate shall not be available at such time for such Interest Period, then the LIBO Rate for such Interest Period shall be the Interpolated Rate. It is understood and agreed that all of the terms and conditions of this definition of "LIBO Rate" shall be subject to Section 2.14.

"LIBOR Rate" shall mean, with respect to any LIBOR Loan for any Interest Period, an interest rate per annum equal to the greater of (a) 1.00% per annum and (b) the product of (i) the LIBO Rate in effect for such Interest Period and (ii) the Statutory Reserve Rate.

"<u>LIBOR Term Loan</u>" shall mean any Term Loan bearing interest at a rate determined by reference to the LIBOR Rate. "<u>License Subsidiary</u>" shall mean Intelsat License LLC., a Delaware limited liability company, and any other Wholly Owned Subsidiary formed prior to the Signing Date and any Subsidiary Guarantor formed on or after the Petition Date, in each case, for the purpose of holding Subject Licenses to be used by the Borrower or any of its Restricted Subsidiaries in the operation of their respective businesses and all of the shares of capital stock and other ownership interests of which are held by a Subsidiary Guarantor.

'Lien'' shall mean any mortgage, pledge, security interest, hypothecation, conditional or security assignment, lien (statutory or other) or similar encumbrance (including any currently effective agreement to give any of the foregoing, any conditional sale or other title retention agreement or any lease in the nature thereof).

"Liquidity" shall mean, at any date of determination, the sum of (a) the Unrestricted Cash Amount plus (b) the then Available Term Loan Commitment.

"Loan" shall mean any Term Loan made by any Lender hereunder.

"Loan to Value Ratio" shall mean, as of any date of calculation, the ratio of (a) the aggregate principal amount of the Permitted Acquisition Debt incurred to (b) the Fair Market Value of the Acquired Person as of such date.

"Management Investors" shall mean directors and/or members of senior management of Intelsat Holdings S.A., any Subsidiaries of Intelsat Holdings S.A. and any parent company of Intelsat Holdings S.A., or any of their respective spouses, direct lineal descendants, heirs or trusts for the benefit of any of the foregoing.

"Material Adverse Effect" shall mean any circumstance or condition affecting the business, assets, operations, properties or financial condition of the Borrower and its Subsidiaries taken as a whole that would, individually or in the aggregate, have a material adverse effect (other than as a result of the events leading up to, and following the commencement of the Cases and the continuation and prosecution thereof, including circumstances or conditions customarily resulting from such events, commencement, continuation and prosecution, which shall not, individually or in the aggregate, constitute a Material Adverse Effect) on (i) the ability of the Credit Parties (taken as a whole) to perform their respective payment obligations under the Credit Documents or (ii) the rights and remedies of the Lenders or Agents under any Credit Documents, including the ability of the Agents and Lenders to enforce the Credit Documents; provided that nothing disclosed in any of the following filings or methods by the Borrower, Holdings and/or any Parent Company: (1) the annual report on Form 10-K for the year ended December 31, 2019, (2) any filings on Form 8-K or on Form 10-Q made through the Signing Date, (3) any matters publicly disclosed (including any such disclosure through a publicly accessible website affiliated with the Borrower, Holdings, and/or any Parent Company) prior to the Signing Date, and/or (4) any event relating to or resulting from any default under the Prepetition Debt, shall, in any case, in and of itself and based solely on facts as disclosed therein (without giving effect to any developments not disclosed therein) constitute a Material Adverse Effect.

"Material Subsidiary" shall mean, at any date of determination, (1) each License Subsidiary, (2) each Subsidiary Guarantor set forth on Schedule 1.1(b) and (3) each Restricted Subsidiary of the Borrower (a) whose total assets at the last day of the Test Period ending on the last day of the most recent fiscal period for which Section 9.1 Financials have been delivered were equal to or greater than 5% of the consolidated total assets of the Borrower and the Restricted Subsidiaries at such date or (b) whose gross revenues for such Test Period were equal to or greater

than 5% of the consolidated gross revenues of the Borrower and the Restricted Subsidiaries for such period, in each case determined in accordance with GAAP; <u>provided</u> that Government Business Subsidiaries and their respective Subsidiaries shall be deemed not to be a Material Subsidiary.

"<u>Maturity Date</u>" shall mean the earliest of (a) the Scheduled Maturity Date; (b) the substantial consummation (as defined in Section 1101 of the Bankruptcy Code and which for purposes hereof shall be no later than the "effective date" thereof) of a Reorganization Plan filed in the Cases that is confirmed pursuant to an order entered by the Bankruptcy Court; (d) the acceleration of the loans and the termination of the commitment with respect to the DIP Facility in accordance with the Credit Documents; and (e) a sale of all or substantially all of the assets of Borrower (or the Borrower and the Guarantors) pursuant to Section 363 of the Bankruptcy Code.

"<u>Minority Investment</u>" shall mean any Person (other than a Subsidiary) in which the Borrower or any Restricted Subsidiary owns capital stock or other equity interests.

"Moody's" shall mean Moody's Investors Service, Inc. or any successor by merger or consolidation to its business.

"<u>Mortgaged Property</u>" shall mean each parcel of Real Estate and improvements thereto with respect to which a mortgage or a security interest is granted to the Collateral Agent for the benefit of the Secured Parties pursuant to the Security Documents.

"<u>Net Cash Proceeds</u>" shall mean, with respect to any Prepayment Event, (a) the gross cash proceeds (including payments from time to time in respect of installment obligations, if applicable) received by or on behalf of the Borrower or any of the Restricted Subsidiaries in respect of such Prepayment Event, *less* (b) the sum of:

(i) in the case of any Prepayment Event, the amount, if any, of all taxes paid or estimated to be payable by the Borrower (including indirectly via a distribution that is permitted by Section 9.9(f) and Section 10.6(b)) or any of the Restricted Subsidiaries in connection with or as a result of such Prepayment Event,

(ii) in the case of any Prepayment Event (other than with respect to a C-Band Prepayment Event), the amount of any reasonable reserve (other than any taxes deducted pursuant to clause (i) above) and only for a period not to exceed one year; provided that in the event such amount of proceeds so reserved exceeds \$500,000, the Borrower shall deliver to the Administrative Agent the certificate of an Authorized Officer as to the reasonableness of such determination) established in good faith against any liabilities (x) associated with the assets that are the subject of such Prepayment Event and (y) retained by the Borrower or any of the Restricted Subsidiaries; provided that the amount of any subsequent reduction of such reserve (other than in connection with a payment in respect of any such liability) shall be deemed to be Net Cash Proceeds of such a Prepayment Event occurring on the date of such reduction,

(iii) in the case of any Prepayment Event (other than with respect to a C-Band Prepayment Event), the amount of any Indebtedness secured by a Lien on the assets that are the subject of such Prepayment Event (excluding any Prepetition Secured Debt, Prepetition Senior Notes and any Indebtedness of any Parent Companies) to the extent that the instrument creating or evidencing such Indebtedness or any agreement relating to such Prepayment Event requires that such Indebtedness be repaid upon consummation of such Prepayment Event and payments of liabilities relating to such assets which are retained by the Borrower or any Restricted Subsidiary,

(iv) in the case of any Casualty Event, the amount of any proceeds of such Casualty Event that the Borrower or any Subsidiary has reinvested (or intends to reinvest within the Reinvestment Period or has entered into a binding commitment prior to the last day of the Reinvestment Period to reinvest) in the replacement assets (subject to Section 9.14); provided that any portion of such proceeds that has not been so reinvested within such Reinvestment Period shall, unless the Borrower or a Subsidiary has entered into a binding commitment prior to the last day of such Reinvestment Period to reinvest such proceeds, (x) be deemed to be Net Cash Proceeds of a Casualty Event occurring on the last day of such Reinvestment Period and (y) be applied to the repayment of Term Loans in accordance with Section 5.2(a), and

(v) in the case of any Prepayment Event, reasonable and customary fees, commissions, expenses, issuance costs, discounts, premiums, consent payments and redemption, tender offer, defeasance and other costs paid by the Borrower or any of the Restricted Subsidiaries, as applicable, in connection with such Prepayment Event (other than those payable to the Borrower or any Subsidiary of the Borrower), in each case only to the extent not already deducted in arriving at the amount referred to in clause (a) above; provided that, with respect to any C-Band Prepayment Event, it is agreed that this clause (v) shall not apply to such items that are the underlying source of a relocation cost reimbursement).

Net Cash Proceeds shall not include (i) any trade-in-credits or purchase price reductions received by the Borrower or any of its Restricted Subsidiaries in connection with an exchange of equipment for replacement equipment that is the functional equivalent of such exchanged equipment, (ii) proceeds from business interruption insurance, third party

liability insurance, rent insurance and other payments, in each case, for interruption of operations or (iii) up to \$50,000,000 in Net Cash Proceeds from Asset Sale Prepayment Events since the Closing Date.

"<u>Non-Consenting Lender</u>" shall have the meaning provided in Section 14.7(b).

"Non-Defaulting Lender" shall mean and include each Lender other than a Defaulting Lender.

"Notice of Borrowing" shall have the meaning provided in Section 2.3(a), such notice shall be substantially in the form of Exhibit C hereto, as applicable.

"<u>Notice of Conversion or Continuation</u>" shall have the meaning provided in Section 2.6(a).

"Obligations" shall mean the Credit Agreement Obligations and the collective reference to the "Obligations" and/or "Secured Obligations" (or any terms of similar import) as defined in the various Security Documents.

"Orders" shall mean, collectively, the Interim Cash Collateral Order and the Final DIP Order.

"Other Prepetition Secured Parties" shall have the meaning provided in the definition of "Commitment Re-allocation."

"Other Taxes" shall mean any and all present or future stamp, documentary or any other excise, any value-added tax, property or similar Taxes (including interest, fines, penalties, additions to tax and related expenses with regard thereto) arising directly from any payment made or required to be made under this Agreement or any other Credit Document or from the execution or delivery of, registration (except where such registration is not necessary to evidence, prove, maintain, enforce, compel or otherwise assert the rights of the Lender or the Administrative Agent under a Credit Document) or enforcement of, consummation or administration of, or otherwise with respect to, this Agreement or any other Credit Document.

"Participant" shall have the meaning provided in Section 14.6(c)(i).

"Participation Election Form" shall have the meaning provided in Section 6.18.

"<u>Payment in Full</u>" shall mean the time at which no Lender shall have any Commitments, any Loan or other Obligations unpaid, unsatisfied or outstanding (other than in respect of contingent obligations, indemnities and expenses related thereto that are not then payable or in existence) as a result of all such Loans and other Obligations having been paid in full in cash.

"Payment Office" shall mean the office of the Administrative Agent located at Eleven Madison Avenue, New York, NY 10010 or such other office as the Administrative Agent may designate to the Borrower and the Lenders from time to time.

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Section 4002 of ERISA, or any successor thereto.

"<u>Perfection Certificate</u>" shall mean a certificate of the Borrower in the form of <u>Exhibit D</u> hereto or any other form approved by the Required Backstop Parties.

"<u>Permitted Acquisition</u>" shall mean an Acquisition, so long as (a) such acquisition and all transactions related thereto shall be consummated in accordance with applicable law; (b) such acquisition shall result in the issuer of such Equity Interests becoming a Restricted Subsidiary or such assets being acquired by a Restricted Subsidiary and, to the extent required by Section 9.11, a Subsidiary Guarantor; (c) such acquisition shall result in the Collateral Agent, for the benefit of the applicable Lenders, being granted a security interest in any such Equity Interest or assets so acquired, to the extent required by Sections 9.11, 9.12 and/or 9.15; and (d) after giving effect to such acquisition, no Default or Event of Default shall have occurred and be continuing.

"Permitted Acquisition Debt" shall have the meaning provided in Section 10.01(j).

"<u>Permitted Business</u>" shall have the meaning provided in Section 9.14(a).

"Permitted Holders" shall mean (i) the Sponsors, (ii) the Management Investors, and (iii) any group (within the meaning of Section 13(d)(3) or Section 14(d)(2) of the Exchange Act, or any successor provision), the members of which include any of the Permitted Holders specified in clauses (i) and/or (ii) above, that (directly or indirectly) holds or acquires beneficial ownership of the Voting Stock of the Borrower or any parent of the Borrower (a "Permitted Holder Group"), so long as no Person or other "group" (other than Permitted Holders specified in clauses (i) through (ii) above) beneficially owns more than 50% on a fully diluted basis of the Voting Stock held by such Permitted Holder Group.

"Permitted Investments" shall mean:

(a) securities issued or unconditionally guaranteed by the United States government or any agency or instrumentality thereof, in each case having maturities of not more than 24 months from the date of acquisition thereof;

(b) securities issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof or any political subdivision of any such state or any public instrumentality thereof having maturities of not more than 24 months from the date of acquisition thereof and, at the

time of acquisition, having an investment grade rating generally obtainable from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, then from another nationally recognized rating service);

(c) commercial paper issued by any Lender or any bank holding company owning any Lender;

(d) commercial paper maturing no more than 12 months after the date of creation thereof and, at the time of acquisition, having a rating of at least A-2 or P-2 from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, an equivalent rating from another nationally recognized rating service);

(e) domestic and LIBOR certificates of deposit or bankers' acceptances maturing no more than two years after the date of acquisition thereof issued by any Lender or any other bank having combined capital and surplus of not less than \$250,000,000 in the case of domestic banks and \$100,000,000 (or the Dollar Equivalent thereof) in the case of foreign banks;

(f) repurchase agreements with a term of not more than 30 days for underlying securities of the type described in clauses (a), (b) and (e) above entered into with any bank meeting the qualifications specified in clause (e) above or securities dealers of recognized national standing;

(g) marketable short-term money market and similar funds (x) either having assets in excess of 250,000,000 or (y) having a rating of at least A-2 or P-2 from either S&P or Moody's (or, if at any time neither S&P nor Moody's shall be rating such obligations, an equivalent rating from another nationally recognized rating service);

(h) shares of investment companies that are registered under the Investment Company Act of 1940 and substantially all the investments of which are one or more of the types of securities described in clauses (a) through (g) above; and

(i) in the case of Investments by the Borrower or any Restricted Foreign Subsidiary or Investments made in a country outside the United States of America, other customarily utilized high-quality Investments in the country where such Restricted Foreign Subsidiary is located or in which such Investment is made.

"Permitted Liens" shall mean:

(a) Liens for taxes, assessments or governmental charges or claims (i) not yet due, (ii) which are being contested in good faith and by appropriate proceedings for which appropriate reserves have been established in accordance with GAAP or (iii) for a tax claim for any pre-filing or straddle tax period, the nonpayment of which is permitted or required by the bankruptcy code;

(b) Liens in respect of property or assets of the Borrower or any of the Subsidiaries imposed by law, such as carriers', warehousemen's, repairmen's, bankers', landlords' and mechanics' Liens and other similar Liens, in each case so long as such Liens arise in the ordinary course of business and do not individually or in the aggregate have a Material Adverse Effect;

(c) Liens arising from judgments or decrees in circumstances not constituting an Event of Default under Section 12.11;

(d) Liens incurred or deposits made in connection with workers' compensation, unemployment insurance and other types of social security legislation, or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, performance and return-of-money bonds and other similar obligations incurred in the ordinary course of business;

(e) ground leases in respect of Real Estate on which facilities owned or leased by the Borrower or any of its Subsidiaries are located;

(f) easements, rights-of-way, restrictions, minor defects or irregularities in title, violations of zoning or other municipal ordinances, and other similar charges or encumbrances not interfering in any material respect with the business of the Borrower and its Subsidiaries, taken as a whole;

(g) any interest or title of a lessor or secured by a lessor's interest under any lease permitted by this Agreement;

(h) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;

(i) Liens on goods the purchase price of which is financed by a documentary letter of credit issued for the account of the Borrower or any of its Subsidiaries; provided that such Lien secures only the obligations of the Borrower or such Subsidiaries in respect of such letter of credit to the extent permitted under Section 10.1;

(j) licenses, sublicenses, leases or subleases granted to others not interfering in any material respect with the business of the Borrower and its Subsidiaries, taken as a whole;

(k) Liens on equipment of the Borrower or any Subsidiary granted in the ordinary course of business to customers at whose premises such equipment is located;

(1) Liens created in the ordinary course of business in favor of banks and other financial institutions over credit balances of any bank accounts of the Borrower and the Restricted Subsidiaries held at such banks or financial institutions, as the case may be, to facilitate the operation of cash pooling and/or interest set-off arrangements in respect of such bank accounts in the ordinary course of business; and

(m) (i) any agreement not to use all or any portion of C-band or any other restrictions on the use of C-band (including any agreement to impose restrictions) and (ii) any escrow arrangements entered into in connection with the foregoing.

"Permitted Refinancing Indebtedness" shall mean any Indebtedness issued in exchange for, or the net proceeds of which are used to extend, refinance, renew, replace, defease or refund (collectively, to "Refinance"), the Indebtedness being Refinanced (or previous refinancing thereof constituting Permitted Refinancing Indebtedness); provided that (a) the principal amount (or accreted value, if applicable) of such Permitted Refinancing Indebtedness); provided that (a) the principal amount (or accreted value, if applicable) of such Permitted Refinancing Indebtedness does not exceed the principal amount (or accreted value, if applicable) of the Indebtedness so Refinanced (plus unpaid accrued interest and premium thereon and reasonable fees and expenses payable in connection with such refinancing), (b) the weighted average life to maturity of such Permitted Refinancing Indebtedness at the time such Refinancing Indebtedness is incurred or issued is greater than or equal to the weighted average life to maturity at such time of the Indebtedness being Refinanced, (c) if the Indebtedness being Refinanced is subordinated in right of payment to the Obligations, such Permitted Refinancing Indebtedness being Refinanced, (d) no Permitted Refinancing Indebtedness shall be subordinated in right of payment to such Obligations on terms at least as favorable to the Lenders as those contained in the documentation governing the Indebtedness being Refinanced, except to the extent otherwise permitted hereunder and (e) if the Indebtedness being Refinanced, except to the extent otherwise permitted hereunder and (e) if the Indebtedness being Refinanced is secured by any assets securing the Obligations (whether equally and ratably with, or junior to, the Secured Parties than those contained in the documentation governing the Indebtedness being Refinanced; provided that Indebtedness incurred to refinance Indebtedness outstanding under Sections 10.1(A)(f), (g), (q), (s), (t), (u) and (v) shall be deemeed to have been incurred and to be outst

"<u>Permitted Sale Leaseback</u>" shall mean any Sale Leaseback consummated by the Borrower or any of the Restricted Subsidiaries after the Signing Date; <u>provided</u> that any such Sale Leaseback (other than any Sale Leaseback that is between the Borrower and any Guarantor or any Guarantor and another Guarantor) is consummated for fair value as determined at the time of consummation in good faith by the Borrower and, in the case of any Sale Leaseback (or series of related Sale Leasebacks) the aggregate proceeds of which exceed \$20,000,000, the board of directors of the Borrower (which such determination may take into account any retained interest or other Investment of the Borrower or such Restricted Subsidiary in connection with, and any other material economic terms of, such Sale Leaseback).

"Person" shall mean any individual, partnership, joint venture, firm, corporation, limited liability company, association, trust or other enterprise or any Governmental Authority.

"Petition Date" shall mean May 13, 2020, the date on which the Debtors filed voluntary petitions in the Bankruptcy Court to commence the Cases.

"<u>Plan</u>" shall mean any multiemployer or single-employer plan, as defined in Section 4001 of ERISA and subject to Title IV of ERISA, that is or was within any of the preceding six plan years maintained or contributed to by (or to which there is or was an obligation to contribute or to make payments to) the Borrower, a Subsidiary or an ERISA Affiliate.

"<u>Plan Effective Date</u>" means the date of the substantial consummation (as defined in section 1101(2) of the Bankruptcy Code, which for purposes hereof shall be no later than the effective date) of one or more plans of reorganization confirmed pursuant to a final order entered by the Bankruptcy Court.

"Platform" shall have the meaning provided in Section 14.17(b).

"Prepayment Event" shall mean any Asset Sale Prepayment Event, C-Band Prepayment Event, Debt Incurrence Prepayment Event, Casualty Event or any Permitted Sale Leaseback.

"Prepetition 51/2% Senior Notes" shall mean the Borrower's \$2,000.0 million 51/2% Senior Notes due 2023.

"Prepetition 8.0% Senior Secured Notes" shall mean the Borrower's \$1,350.0 million 8.0% Senior Secured Notes due 2024.

"Prepetition 81/2% Senior Notes" shall mean the Borrower's \$2,950.0 million 81/2% Senior Notes due 2024.

"Prepetition 91/2% Senior Secured Notes" shall mean the Borrower's \$490.0 million 91/2% Senior Notes due 2022.

"Prepetition 93/4% Senior Notes" shall mean the Borrower's \$1,485.0 million 93/4% Senior Notes due 2025.

"Prepetition Collateral" shall have the meaning provided in Section 3.1(b).

"Prepetition Credit Agreement" shall mean that certain Credit Agreement dated as of January 12, 2011 (as amended, supplement or otherwise modified from time to time prior to the date hereof), among Borrower, Intelsat (Luxembourg) S.A., the lenders party thereto, Bank of America, N.A., as Administrative Agent, and the other party thereto.

"Prepetition Credit Facilities" shall mean the credit facilities provided pursuant to the Prepetition Credit Agreement.

"Prepetition Debt" shall mean, collectively, the Indebtedness of each Credit Party outstanding and unpaid on the date on which such Person becomes a Credit Party.

"Prepetition Liens" shall have the meaning provided in Section 3.1(b).

"<u>Prepetition Payment</u>" shall mean a payment (by way of adequate protection or otherwise) of principal or interest or otherwise on account of any (i) Prepetition Debt, (ii) "critical or foreign vendor payments" or (iii) trade payables (including, without limitation, in respect of reclamation claims), or other pre-petition claims against any Credit Party.

"Prepetition Secured Debt" shall mean all obligations in respect of the Prepetition Credit Facilities and the Prepetition Secured Notes.

"<u>Prepetition Secured Notes</u>" shall mean, collectively, the Prepetition 8.0% Senior Secured Notes and Prepetition 9½% Senior Secured Notes "<u>Prepetition Secured Parties</u>" shall mean the lender parties under the Prepetition Credit Facilities and the beneficial holders of the Prepetition

Secured Notes.

"Prepetition Senior Notes" shall mean, collectively, the Prepetition 5½% Senior Notes, Prepetition 8½% Senior Notes and Prepetition 9¾% Senior Notes.

"<u>Prime Rate</u>" shall mean the rate of interest per annum determined from time to time by Credit Suisse AG as its prime rate in effect at its principal office in New York City and notified to the Borrower. The prime rate is a rate set by Credit Suisse AG based upon various factors including Credit Suisse AG's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such rate.

"Primed Liens" shall have the meaning provided in Section 3.1(b).

"Primed Parties" shall have the meaning provided in Section 3.2.

"Priming Liens" shall have the meaning provided in Section 3.1(b).

"Private Act" shall mean separate legislation enacted in Bermuda with the intention that such legislation apply specifically to a Credit Party, in whole or in part.

"Proxy Agreement" shall have the meaning provided in Section 9.19.

"<u>PTE</u>" shall mean a prohibited transaction class exemption issued by the U.S. Department of Labor, as any such exemption may be amended from time to time.

"<u>Rating Agencies</u>" shall mean Moody's and S&P or if Moody's or S&P or both shall not make a rating on the Loans or this Agreement publicly available, a nationally recognized statistical rating agency or agencies, as the case may be, selected by the Borrower which shall be substituted for Moody's or S&P or both, as the case may be.

"Real Estate" shall have the meaning provided in Section 9.1(f).

"Register" shall have the meaning provided in Section 14.6(b)(v).

"<u>Regulation D</u>" shall mean Regulation D of the Board as from time to time in effect and any successor to all or a portion thereof establishing reserve requirements.

"Regulation T" shall mean Regulation T of the Board as from time to time in effect and any successor to all or a portion thereof establishing margin requirements.

"<u>Regulation U</u>" shall mean Regulation U of the Board as from time to time in effect and any successor to all or a portion thereof establishing margin requirements.

"Regulation X" shall mean Regulation X of the Board as from time to time in effect and any successor to all or a portion thereof establishing margin requirements.

"Reinvestment Period" shall mean the period from the date of the applicable Casualty Event until the date that is 15 months following the date of such Prepayment Event.

"Related Parties" shall mean, with respect to any specified Person, such Person's Affiliates and the directors, officers, employees, agents, trustees, advisors of such Person and any Person that possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of such Person, whether through the ability to exercise voting power, by contract or otherwise.

"Reorganization Plan" shall mean a plan of reorganization in the Cases.

"Reportable Event" shall mean an event described in Section 4043 of ERISA and the regulations thereunder.

"<u>Required Backstop Parties</u>" shall mean, at any date, Backstop Parties having or holding a majority of the outstanding principal amount of the Term Loan Commitments at such date (determined without giving effect to the Commitment Re-allocation).

"<u>Required Lenders</u>" shall mean, at any date, Non-Defaulting Lenders having or holding a majority of the sum of (i) the Available Term Loan Commitments at such time and (ii) outstanding principal amount of the Term Loans (excluding Term Loans held by Defaulting Lenders) at such date.

"<u>Requirement of Law</u>" shall mean, as to any Person, the certificate of incorporation and by-laws or other organizational or governing documents of such Person, and any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its property or assets or to which such Person or any of its property or assets is subject.

"Resolution Authority" shall mean an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Restatement Date" shall mean June 1, 2020, which shall be the date of the Restated Backstop Commitment Letter.

"Restated Backstop Commitment Letter" shall mean that certain Amended & Restated Debtor in Possession Term Loan Facility Commitment Letter, dated as of June 1, 2020, by and among the Backstop Parties and the Borrower.

"Restricted Foreign Subsidiary" shall mean a Foreign Subsidiary that is a Restricted Subsidiary.

"<u>Restricted Subsidiary</u>" shall mean any Subsidiary of the Borrower other than an Unrestricted Subsidiary; <u>provided</u> that in any event each License Subsidiary shall be a Restricted Subsidiary.

"<u>Sale Leaseback</u>" shall mean any transaction or series of related transactions pursuant to which the Borrower or any of the Restricted Subsidiaries (a) sells, transfers or otherwise disposes of any property, real or personal, whether now owned or hereafter acquired, and (b) as part of such transaction, thereafter rents or leases such property or other property that it intends to use for substantially the same purpose or purposes as the property being sold, transferred or disposed.

"<u>Satellite</u>" shall mean any satellite owned by, or leased to, the Borrower or any of its Restricted Subsidiaries and any satellite purchased pursuant to the terms of a Satellite Purchase Agreement, whether such satellite is in the process of manufacture, has been delivered for launch or is in orbit (whether or not in operational service).

"<u>Satellite Health Report</u>" shall mean a satellite health report, prepared by the Borrower and certified by an Authorized Officer, in the form of <u>Exhibit J</u> (appropriately completed).

"Satellite Manufacturer" shall mean, with respect to any Satellite, the prime contractor and manufacturer of such Satellite.

"Satellite Purchase Agreement" shall mean, with respect to any Satellite, the agreement between the applicable Satellite Purchaser and the applicable Satellite Manufacturer relating to the manufacture, testing and delivery of such Satellite.

"<u>Satellite Purchaser</u>" shall mean the Borrower or Restricted Subsidiary that is a party to a Satellite Purchase Agreement or a launch service agreement, as the case may be.

"S&P" shall mean Standard & Poor's Ratings Services or any successor by merger or consolidation to its business.

"Second Maturity Extension" shall have the meaning provided in the definition of "Scheduled Maturity Date."

"Second Extension Fee" shall have the meaning provided in Section 4.1(e).

"<u>Scheduled Maturity Date</u>" shall mean the date that is fourteen (14) months after the Petition Date (or if such day shall not be a Business Day, the next succeeding Business Day) (the "<u>Initial Maturity Date</u>"); provided that the Borrower may (i) make, prior to the Initial Maturity Date, a one-time election to extend the Initial Scheduled Maturity Date by up to six (6) months so long as no Event of Default shall have occurred and be continuing and the Borrower shall have paid the First Extension Fee (the "<u>First Maturity Extension</u>") and (ii) to the extent the First Maturity Extension shall have been exercised, make, prior to the Scheduled Maturity Date (after giving effect to the First Maturity Extension), another one-time election to further extend the Scheduled Maturity Date (by up to six (6) months solely to the extent (v) an order of the Bankruptcy Court confirming an Acceptable Plan shall have been entered on or prior to the then Scheduled Maturity Date (after giving effect to the First Maturity Extension, but without giving effect to the proposed extension set forth in this clause (ii)), (x) the sole remaining conditions precedent to the Plan Effective Date with respect to an Acceptable Plan are to obtain regulatory approvals necessary

to consummate such Acceptable Plan, (y) no Event of Default shall have occurred and be continuing and (z) the Borrower shall have paid the Second Extension Fee (the "Second Maturity Extension").

"SEC" shall mean the Securities and Exchange Commission or any successor thereto.

"Section 9.1 Financials" shall mean the financial statements delivered, or required to be delivered, pursuant to Section 9.1(a) or (b) together with the accompanying officer's certificate delivered, or required to be delivered, pursuant to Section 9.1(d).

"<u>Secured Parties</u>" shall mean collectively, (i) the Lenders, (ii) the Administrative Agent, (iii) the Collateral Agent, (iii) the beneficiaries of each indemnification obligation undertaken by any Credit Party under this Agreement or any other Credit Document and (iv) any successors, indorsees, transferees and assigns of each of the foregoing and also the Persons described in the term "Secured Parties" in the applicable Security Documents, or, where the context so requires, shall be the collective reference to all such Persons.

"Security Agreement" shall mean the Security and Pledge Agreement, to be entered into on the Closing Date by the Credit Parties, the Administrative Agent and the Collateral Agent for the benefit of the Secured Parties, substantially in the form of Exhibit E hereto, the same may be amended, supplemented or otherwise modified from time to time

"Security Documents" shall mean, collectively, (a) the Guarantee, (b) the Security Agreement, (c) the Final DIP Order and (d) each other security agreement or other instrument or document executed and delivered pursuant to Section 9.11 or 9.12 or pursuant to any of the Security Documents to secure any of the Obligations. The Security Documents (other than the Final DIP Order) shall supplement, and shall not limit, the grant of a Lien on and security interest in the Collateral pursuant to the Final DIP Order.

"Series" shall have the meaning as provided in Section 2.14.

"Signing Date" shall mean May 13, 2020, which shall be the date of the Backstop Commitment Letter.

"Sponsor" shall mean (1) one or more investment funds advised, managed or controlled by BC Partners Holdings Limited or any Affiliate thereof, (2) one or more investment funds advised, managed or controlled by Silver Lake or any Affiliate thereof, and (3) one or more investment funds advised, managed or controlled by any of the Persons described in clauses (1) and (2) of this definition, and, in each case (whether individually or as a group), their Affiliates.

"Statutory Reserve Rate" shall mean, for any day as applied to any LIBOR Loan, a fraction (expressed as a decimal), the numerator of which is the number one and the denominator of which is the number one minus the aggregate of the maximum reserve percentages (including any marginal, special, emergency or supplemental reserves) applicable to any member bank of the Federal Reserve System of the United States of America in respect of Eurocurrency Liabilities (as defined in Regulation D of the Board of Governors of the Federal Reserve System of the United States of America). Term Loans shall be deemed to be subject to such reserve requirements. The Statutory Reserve Rate shall be adjusted automatically on and as of the effective date of any change in any reserve percentage.

"Subject Licenses" shall mean all FCC Licenses for the launch and operation of Satellites or for the operation of any TT&C Station (other than any FCC License held by Intelsat General Corporation or any of its Subsidiaries).

"Subordinated Indebtedness" shall mean Indebtedness of the Borrower or any Guarantor that is by its terms subordinated in right of payment to the obligations of the Borrower and such Subsidiary Guarantor, as applicable, under this Agreement.

"Subsidiary" of any Person shall mean and include (a) any corporation more than 50% of whose stock of any class or classes having by the terms thereof ordinary voting power to elect a majority of the directors of such corporation (irrespective of whether or not at the time stock of any class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) is at the time owned by such Person directly or indirectly through Subsidiaries and (b) any partnership, association, joint venture or other entity in which such Person directly through Subsidiaries has more than a 50% equity interest at the time. Unless otherwise expressly provided, all references herein to a "Subsidiary" shall mean a Subsidiary of the Borrower.

"Subsidiary Guarantors" shall mean (a) each Subsidiary listed of Schedule 1.1(b), and (b) each Subsidiary that becomes a party to the Guarantee after the Signing Date pursuant to Section 9.11.

"Taxes" shall mean any and all present or future taxes, duties, levies, imposts, assessments, deductions, withholdings or other similar charges imposed by any Governmental Authority whether computed on a separate, consolidated, unitary, combined or other basis and any and all liabilities (including interest, fines, penalties or additions to tax) with respect to the foregoing.

"Term Loan Commitment" shall mean, with respect to each Lender, (a) in the case of each Backstop Party, the amount of such Backstop Party's commitment to make Term Loans in the amount set forth opposite such

Backstop Party's name on <u>Annex I</u> to the Restated Backstop Commitment Letter as such Backstop Party's "Term Loan Commitment" (as subsequently adjusted to give effect to the Commitment Re-allocation) and (b) the amount specified as such Lender's "Term Loan Commitment" set forth opposite such Lender's name in <u>Annex I</u> or in the documentation relating to the Commitment Re-allocation reasonably satisfactory to the Borrower, the Administrative Agent and the Required Backstop Parties or in the Assignment and Acceptance, in each case, pursuant to which such Lender acquired a portion of the Term Loan Commitment, as the case may be, in each case, as the same may be increased or decreased from time to time pursuant to the terms hereof.

"Term Loans" shall mean a Loan made pursuant to Section 2.01(a).

"<u>Ticking Fee</u>" shall have the meaning provided in Section 4.1(c).

"Total Credit Exposure" shall mean, at any date, the sum of (a) Available Term Loan Commitment at such date and (b) the outstanding principal amount of all Term Loans at such date.

"Transaction Expenses" shall mean any fees or expenses incurred or paid by the Borrower or any of its Subsidiaries in connection with this Agreement and the other Credit Documents and the transactions contemplated hereby and thereby.

"Transferee" shall have the meaning provided in Section 14.6(e).

"Treasury Rate" shall mean, as of a given date, the yield to maturity, as of such date and as determined by the Borrower, interpolated on a straightline basis between United States Treasury securities with constant maturities (as compiled and published in the most recent Federal Reserve Statistical Release H.15 (519) that has become publicly available at least two business days prior to such date (or, if such Statistical Release is no longer published, any publicly available source of similar market data)) most nearly equal to the period from such redemption date to July 2, 2020; provided, however, that if the period from such date to July 2, 2020, is less than one year, the weekly average yield on one year constant maturity United States Treasury securities (as compiled and published in the most recent Federal Reserve Statistical Release H.15 (519) that has become publicly available at least two days prior to such redemption date (or, if such Statistical Release is no longer published, any publicly available source of similar market data)).

"TT&C Station" shall mean an earth station operated by the Borrower or any of its Restricted Subsidiaries for the purpose of providing tracking, telemetry, control and monitoring of any Satellite.

"Type" shall mean, as to any Term Loan, its nature as an ABR Loan or a LIBOR Term Loan.

"<u>UK Financial Institution</u>" shall mean any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended form time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.

"UK Guarantor" shall mean any Guarantor organized under the laws of England and Wales.

"UK Resolution Authority" shall mean the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.

"Unfunded Current Liability" of any Plan shall mean the amount, if any, by which the present value of the accrued benefits under the Plan as of the close of its most recent plan year, determined in accordance with Statement of Financial Accounting Standards No. 87 as in effect on the Signing Date, based upon the actuarial assumptions that would be used by the Plan's actuary in a termination of the Plan, exceeds the fair market value of the assets allocable thereto.

"Unrestricted Cash Amount" shall mean, as of any date of determination, the aggregate amount of cash and cash equivalents (in each case, free and clear of all liens, other than Liens permitted pursuant to Section 10.02) (excluding cash and cash equivalent that are listed as "restricted" on the consolidated balance sheet of the applicable Persons as of such date unless "restricted" in favor of the DIP Facility or the Prepetition Secured Debt (which may also secure other Indebtedness along with the DIP Facility to the extent permitted pursuant to Section 10.2)).

"<u>Unrestricted Subsidiary</u>" shall mean (a) each Subsidiary of the Borrower set forth in <u>Schedule 1.1(c)</u> and (b) each Subsidiary of an Unrestricted Subsidiary set forth in the preceding clause (a); <u>provided</u>, <u>however</u>, that at the time of any written re-designation by the Borrower to the Administrative Agent that any Unrestricted Subsidiary shall no longer constitute an Unrestricted Subsidiary, such Unrestricted Subsidiary shall cease to be an Unrestricted Subsidiary to the extent no Default or Event of Default would result from such re-designation; provided, further, that, for the avoidance of doubt, no additional Subsidiary of the Borrower may be added to <u>Schedule 1.1(c)</u> on and after the Signing Date.

"<u>Upfront Payment</u>" shall have the meaning provided in Section 4.1(b).

"Upfront Payment Percentage" shall have the meaning provided in Section 4.1(b).

"Voting Stock" shall mean, with respect to any Person, shares of such Person's capital stock having the right to vote for the election of directors of such Person under ordinary circumstances.

"Waivable Mandatory Repayment" shall have the meaning provided in Section 5.2(g).

"<u>Wholly Owned Subsidiary</u>" of any Person shall mean a Subsidiary of such Person 100% of the outstanding capital stock or other ownership interests of which (other than directors' qualifying shares or shares or interests required to be held by foreign nationals) shall at the time be owned by such Person or by one or more Wholly Owned Subsidiaries of such Person.

"Write-Down and Conversion Powers" shall mean, (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule, and (b) with respect to the United Kingdom, any powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section references are to Sections of this Agreement unless otherwise specified. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." Each reference to an agreement or document herein shall mean such agreement or document as from time to time amended, supplemented or modified in accordance with its terms, unless expressly stated otherwise.

1.2. Exchange Rates. For purposes of determining compliance under Sections 10.4, 10.5, 10.6 and 11 with respect to any amount in a Foreign Currency, such amount shall be deemed to equal the Dollar Equivalent thereof based on the average Exchange Rate for a Foreign Currency for the most recent twelve-month period immediately prior to the date of determination determined in a manner reasonably acceptable to the Required Lenders. For purposes of determining compliance with Sections 10.1 and 10.2, with respect to any amount of Indebtedness in a Foreign Currency, compliance will be determined at the time of incurrence thereof using the Dollar Equivalent thereof at the Exchange Rate in effect at the time of such incurrence.

# 1.3. Reserved.

### 1.4. Accounting Terms.

(a) The Borrower may notify the Administrative Agent in writing at any time that it has elected to so use IFRS in lieu of GAAP and, upon any such written notice (a copy of which shall be provided by the Administrative Agent to the Lenders), references herein to GAAP shall thereafter be construed to mean IFRS as in effect from time to time; provided that, to the extent that such election would affect any financial ratio set forth in this Agreement or requirement set forth in Section 11, (i) the Borrower shall provide to the Administrative Agent (for delivery to the Lenders) financial statements and other documents reasonably requested by the Administrative Agent or any Lender setting forth a reconciliation with respect to such ratio or requirement made before and after giving effect to such election and (ii) if the Borrower, the Administrative Agent or the Required Lenders shall so request, the Administrative Agent, the Required Lenders and the Borrower shall negotiate in good faith to amend such ratio to preserve the original intent thereof in light of such change.

(b) Anything in this Agreement to the contrary notwithstanding, no effect shall be given to any change in GAAP arising out of a change described in the Proposed Accounting Standards Update to Leases (Topic 840) dated August 17, 2010 or a substantially similar pronouncement.

1.5. <u>Ability to Redesignate</u>. For purposes of determining compliance with any one of Sections 9.9, 10.1, 10.2, 10.3, 10.4, 10.5, 10.6, 10.7, 10.8, 10.9 and 1.1(a), in the event that any Lien, Investment, Indebtedness, merger, consolidation, amalgamation or similar fundamental change, Disposition, dividend, affiliate transaction, contractual obligation or prepayment of Indebtedness meets the criteria of more than one of the categories of transactions permitted pursuant to any clause of such Section, such transaction (or portion thereof) at any time shall be permitted under one or more of such clauses as determined by the Borrower (and the Borrower shall be entitled to redesignate use of any such clauses from time to time) in its sole discretion at such time.

1.6. <u>Divisions</u>. Any reference herein to a merger, transfer, consolidation, amalgamation, assignment, sale, disposition or transfer, or similar term, shall be deemed to apply to a division of or by a limited liability company or other Person, or an allocation of assets to a series of a limited liability company or other Person (or the unwinding of such a division or allocation) (any such transaction, a "<u>Division</u>"), as if it were a merger, transfer, consolidation, amalgamation, assignment, sale, disposition or transfer, or similar term, as applicable, to, of or with a



separate Person. Any Division of a limited liability company or other Person shall constitute a separate Person hereunder (and each Division of any limited liability company or other Person that is a Subsidiary, Restricted Subsidiary, Unrestricted Subsidiary, joint venture or any other like term shall also constitute such a Person or entity).

#### SECTION 2. Amount and Terms of Credit.

### 2.1. Commitments.

(a) <u>Term Loan Commitments</u>. Subject to the terms and conditions set forth herein and in the Final DIP Order, each Lender agrees, severally and not jointly, to make Term Loans to the Borrower denominated in Dollars in up to three draws on any Business Day on or after the Closing Date, in an aggregate principal amount not to exceed its respective Term Loan Commitment.

(b) All such Term Loans made by each of the Lenders may be repaid or prepaid in accordance with the provisions hereof, but once repaid or prepaid, may not be reborrowed. For the avoidance of doubt, any Term Loan Commitments (x) shall be reduced dollar for dollar at the time of funding of any Term Loans thereunder and (y) shall terminate in full upon the third Borrowing hereunder.

(c) Each Lender may at its option make any LIBOR Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that (A) any exercise of such option shall not affect the obligation of the Borrower to repay such Loan and (B) in exercising such option, such Lender shall use its reasonable efforts to minimize any increased costs to the Borrower resulting therefrom (which obligation of the Lender shall not require it to take, or refrain from taking, actions that it determines would result in increased costs for which it will not be compensated hereunder or that it determines would be otherwise disadvantageous to it and in the event of such request for costs for which compensation is provided under this Agreement, the provisions of Section 3.5 shall apply).

2.2. <u>Minimum Amount of Each Borrowing; Maximum Number of Borrowings</u>. (x) The first Borrowing of Term Loans shall be made on or after the Closing Date in the aggregate principal amount of \$500,000,000, (y) the second Borrowing of Term Loans shall be made after the date of the first Borrowing but prior to the Maturity Date in an aggregate principal amount equal to the lesser of the then Available Term Loan Commitment and \$250,000,000 and (z) the third Borrowing of Term Loans shall be made after the date of the second Borrowing but prior to the Maturity Date in the aggregate principal amount equal to the then Available Term Loans shall be made after the date of the second Borrowing but prior to the Maturity Date in the aggregate principal amount equal to the then Available Term Loans commitment. Each Borrowing of Term Loans shall consist of Term Loans of the same Type made on the same day by the Term Lenders ratably according to their respective Term Commitments; provided that there shall not at any time be more than a total of six (6) LIBOR Loans outstanding.

2.3. Notice of Borrowing. The Borrower shall give the Administrative Agent at the Administrative Agent's Office (i) prior to 12:00 Noon (New York City time) at least three Business Days' (or, with respect to the second and third Borrowings, ten (10) Business Days') prior written notice of the Borrowing of Term Loans. Such notice (a "Notice of Borrowing") shall be irrevocable and shall specify (i) the aggregate principal amount of the Term Loans to be made, (ii) the date of the Borrowing and (iii) whether the Term Loans shall consist of ABR Loans or LIBOR Term Loans and, if the Term Loans are to include LIBOR Term Loans, the Interest Period to be initially applicable thereto; provided that the Borrowing shall be a LIBOR Loan. If no Interest Period is specified with respect to any requested LIBOR Loan, then the Borrower shall be deemed to have selected an Interest Period of one (1) month's duration.

The Administrative Agent shall promptly give each Lender written notice of the proposed Borrowing of Term Loans, of such Lender's proportionate share thereof and of the other matters covered by the related Notice of Borrowing.

#### 2.4. Disbursement of Funds.

(a) No later than 12:00 Noon (New York City time) on the date specified in each Notice of Borrowing, each Lender will make available its *pro* rata portion, if any, of each Borrowing requested to be made on such date in the manner provided below.

(b) Each Lender shall make available all amounts it is to fund to the Borrower under any Borrowing for its applicable Commitments, and in immediately available funds to the Administrative Agent at the Administrative Agent's Office and the Administrative Agent will make available to the Borrower, by promptly crediting the amounts so received, in like funds, to the account of the Borrower designated in the applicable Notice of Borrowing. Unless the Administrative Agent shall have been notified by any Lender prior to the date of any such Borrowing that such Lender does not intend to make available to the Administrative Agent its portion of the Borrowing or Borrowings to be made on such date, the Administrative Agent may assume that such Lender has

made such amount available to the Administrative Agent on such date of Borrowing, and the Administrative Agent, in reliance upon such assumption, shall make available to the Borrower a corresponding amount. If such corresponding amount is not in fact made available to the Administrative Agent by such Lender and the Administrative Agent has made available same to the Borrower, the Administrative Agent shall be entitled to recover such corresponding amount forthwith upon the Administrative Agent's demand therefor the Administrative Agent shall promptly notify the Borrower and the Borrower shall immediately pay such corresponding amount to the Administrative Agent shall also be entitled to recover from such Lender or the Borrower interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to the Borrower to the date such corresponding amount is recovered by the Administrative Agent, at a rate *per annum* equal to (i) if paid by such Lender, the Federal Funds Effective Rate or (ii) if paid by the Borrower, the then-applicable rate of interest or fees, calculated in accordance with Section 2.8, for the respective Loans.

(c) Nothing in this Section 2.4 shall be deemed to relieve any Lender from its obligation to fulfill its commitments hereunder or to prejudice any rights that the Borrower may have against any Lender as a result of any default by such Lender hereunder (it being understood, however, that no Lender shall be responsible for the failure of any other Lender to fulfill its commitments hereunder).

2.5. Repayment of Loans; Evidence of Debt.

(a) The Borrower shall repay to the Administrative Agent, for the benefit of the Lenders, on the Maturity Date, the then unpaid Loans, in Dollars.

(b) [Reserved].

(c) [Reserved].

(d) [Reserved].

(e) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to the appropriate lending office of such Lender resulting from each Loan made by such lending office of such Lender from time to time, including the amounts of principal and interest payable and paid to such lending office of such Lender from time to time under this Agreement.

(f) The Administrative Agent shall maintain the Register pursuant to Section 14.6(b), and a subaccount for each Lender, in which Register and subaccounts (taken together) shall be recorded (i) the amount of each Loan made hereunder, the Type of each Loan made and the Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder and (iii) the amount of any sum received by the Administrative Agent hereunder from the Borrower and each Lender's share thereof.

(g) The entries made in the Register and accounts and subaccounts maintained pursuant to paragraphs (e) and (f) of this Section 2.5 shall, to the extent permitted by applicable law, be prima facie evidence of the existence and amounts of the obligations of the Borrower therein recorded; provided, however, that the failure of any Lender or the Administrative Agent to maintain such account, such Register or such subaccount, as applicable, or any error therein, shall not in any manner affect the obligation of the Borrower to repay (with applicable interest) the Loans made to the Borrower by such Lender in accordance with the terms of this Agreement. In the event of any conflict between the Register and the account or accounts of each Lender, the Register shall control absent manifest error.

#### 2.6. Conversions and Continuations.

(a) The Borrower shall have the option on any Business Day to convert all or a portion of the outstanding principal amount of Term Loans made to the Borrower (as applicable) of one Type into a Borrowing or Borrowings of another Type and the Borrower shall have the option on any Business Day to continue the outstanding principal amount of any LIBOR Term Loans as LIBOR Term Loans for an additional Interest Period; provided that (i) ABR Loans may not be converted into LIBOR Term Loans if a Default or Event of Default is in existence on the date of the conversion and the Administrative Agent has or the Required Lenders have determined in its or their sole discretion not to permit such conversion, and (ii) LIBOR Loans may not be continued as LIBOR Loans for an additional Interest Period if a Default or Event of Default is in existence on the date of the proposed continuation and the Administrative Agent has or the Required Lenders have determined in its or their sole discretion not to permit such continuation. Each such conversion or continuation shall be effected by the Borrower by giving the Administrative Agent at the Administrative Agent's Office prior to 12:00 Noon (New York City time) at least three Business Days' (or one Business Day's notice in the case of a conversion into ABR Loans) prior written notice (each, a "Notice of Conversion or Continuation") specifying the Term Loans to be so converted or continued, the Type of Term Loans to be converted or continued into and, if such Term Loans are to be converted

into or continued as LIBOR Loans, the Interest Period to be initially applicable thereto. If no Interest Period is specified in the Notice of Conversion or Continuation, then the Borrower shall be deemed to have selected an Interest Period of one (1) month's duration. The Administrative Agent shall give each Lender notice as promptly as practicable of any such proposed conversion or continuation affecting any of its Term Loans.

(b) If any Default or Event of Default is in existence at the time of any proposed continuation of any LIBOR Loans and the Administrative Agent has or the Required Lenders have determined in its or their sole discretion not to permit such continuation, such LIBOR Loans shall be automatically converted on the last day of the current Interest Period into ABR Loans. If upon the expiration of any Interest Period in respect of LIBOR Loans, the Borrower has failed to elect a new Interest Period to be applicable thereto as provided in paragraph (a) above, the Borrower shall be deemed to have elected to continue such Borrowing of LIBOR Loans into a Borrowing of ABR Loans, effective as of the expiration date of such current Interest Period.

2.7. <u>Pro Rata Borrowings</u>. Each Borrowing under this Agreement shall be granted by the Lenders pro rata on the basis of their then applicable Term Loan Commitments. No Lender shall be responsible for any default by any other Lender in its obligation to make Loans hereunder and that each Lender shall be obligated to make the Loans provided to be made by it hereunder, regardless of the failure of any other Lender to fulfill its commitments hereunder.

#### 2.8. Interest.

(a) The unpaid principal amount of each ABR Loan shall bear interest from the date of the Borrowing thereof until maturity (whether by acceleration or otherwise) at a rate *per annum* that shall at all times be the Applicable ABR Margin plus the ABR in effect from time to time.

(b) The unpaid principal amount of each LIBOR Loan shall bear interest from the date of the Borrowing thereof until maturity thereof (whether by acceleration or otherwise) at a rate *per annum* that shall at all times be the Applicable LIBOR Margin in effect from time to time plus the relevant LIBOR Rate.

(c) If all or a portion of (i) the principal amount of any Loan or (ii) any interest or commitment fee after the cure period set forth in Section 12.1(b) payable thereon shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate *per annum* that is (x) in the case of overdue principal, the rate that would otherwise be applicable thereto <u>plus</u> 2% or (y) in the case of any overdue interest or commitment fee, to the extent permitted by applicable law, the rate described in Section 2.8(a) plus 2% from and including the date of such non-payment to but excluding the date on which such amount is paid in full (after as well as before judgment).

(d) Interest on each Loan shall accrue from and including the date of any Borrowing to but excluding the date of any repayment thereof and shall be payable (i) in respect of each ABR Loan, quarterly in arrears on the last day of each March, June, September and December, (ii) in respect of each LIBOR Loan, on the last day of each Interest Period applicable thereto and, in the case of an Interest Period in excess of three months, on each date occurring at three-month intervals after the first day of such Interest Period, or (iii) in respect of each Loan (except, other than in the case of prepayments, any ABR Loan), on any prepayment (on the amount prepaid), at maturity (whether by acceleration or otherwise) and, after such maturity, on demand.

(e) All computations of interest hereunder shall be made in accordance with Section 5.5.

(f) The Administrative Agent, upon determining the interest rate for any Borrowing of LIBOR Loans, shall promptly notify the Borrower and the relevant Lenders thereof. Each such determination shall, absent clearly demonstrable error, be final and conclusive and binding on all parties hereto.

2.9. Interest Periods. At the time the Borrower gives a Notice of Borrowing or Notice of Conversion or Continuation in respect of the making of, or conversion into or continuation as, a Borrowing of LIBOR Loans (in the case of the initial Interest Period applicable thereto) or prior to 10:00 a.m. (New York City time) on the third Business Day prior to the expiration of an Interest Period applicable to a Borrowing of LIBOR Loans, the Borrower shall have the right to elect by giving the Administrative Agent written notice the Interest Period applicable to such Borrowing, which Interest Period shall, except as contemplated by this Section 2.9, at the option of the Borrower be a one, two, three or six month period; provided that the initial Interest Period may be for a period less than one month if agreed upon by the Borrower and the Required Lenders.

Notwithstanding anything to the contrary contained above:

(a) the initial Interest Period for any Borrowing of LIBOR Loans shall commence on the date of such Borrowing (including the date of any conversion from a Borrowing of ABR Loans or) and each Interest Period occurring thereafter in respect of such Borrowing shall commence on the day on which the next preceding Interest Period expires;

(b) [reserved];

(c) if any Interest Period would otherwise expire on a day that is not a Business Day, such Interest Period shall expire on the next succeeding Business Day; <u>provided</u> that if any Interest Period in respect of a LIBOR Loan would otherwise expire on a day that is not a Business Day but is a day of the month after which no further Business Day occurs in such month, such Interest Period shall expire on the next preceding Business Day; and

(d) the Borrower shall not be entitled to elect any Interest Period in respect of any LIBOR Loan if such Interest Period would extend beyond the applicable Scheduled Maturity Date of such Loan then in effect; and

(e) unless the Borrower elects otherwise in its sole discretion, all Term Loans funded in the second and third Borrowings will be of the same Type and, in the case of LIBOR Loans, have the same Interest Periods and LIBOR Rate as all other LIBOR Term Loans then outstanding (on a ratable basis if there is more than one Borrowing of Term Loans then outstanding); provided that the initial Interest Period of any such LIBOR Term Loans funded shall commence on the applicable funding date and shall end on the last day of the then-current Interest Period for all other LIBOR Term Loans then outstanding (on a ratable basis if there is more than one Borrowing of other LIBOR Term Loans then outstanding).

2.10. Increased Costs, Illegality, etc.

(a) In the event that any Lender shall have reasonably determined (which determination shall, absent clearly demonstrable error, be final and conclusive and binding upon all parties hereto):

(i) at any time, that such Lender shall incur increased costs or reductions in the amounts received or receivable hereunder with respect to any LIBOR Loans (other than any such increase or reduction attributable to Indemnified Taxes and Other Taxes indemnifiable under Section 5.4 and Excluded Taxes) because of (x) any change since the Signing Date in any applicable law, governmental rule, regulation, guideline or order (or in the interpretation or administration thereof and including the introduction of any new law or governmental rule, regulation, guideline or order), such as, for example, without limitation, a change in official reserve requirements, and/or (y) other circumstances affecting the interbank LIBOR market or the position of such Lender in such market; or

(ii) at any time, that the making or continuance of any LIBOR Loan has become unlawful by compliance by such Lender in good faith with any law, governmental rule, regulation, guideline or order (or would conflict with any such governmental rule, regulation, guideline or order not having the force of law even though the failure to comply therewith would not be unlawful), or has become impracticable as a result of a contingency occurring after the Signing Date that materially and adversely affects the interbank LIBOR market;

then, and in any such event, such Lender shall within a reasonable time thereafter give notice to the Borrower and to the Administrative Agent of such determination (which notice the Administrative Agent shall promptly transmit to each of the other Lenders). Thereafter (x) in the case of clause (i) above, the Borrower shall pay to such Lender, promptly after receipt of written demand therefor such additional amounts (in the form of an increased rate of, or a different method of calculating, interest or otherwise as such Lender in its reasonable discretion shall determine) as shall be required to compensate such Lender for such increased costs or reductions in amounts receivable hereunder (it being agreed that a written notice as to the additional amounts owed to such Lender, showing in reasonable detail the basis for the calculation thereof, submitted to the Borrower by such Lender shall, absent clearly demonstrable error, be final and conclusive and binding upon all parties hereto) and (y) in the case of clause (ii) above, the Borrower shall take one of the actions specified in Section 2.10(b) as promptly as possible and, in any event, within the time period required by law.

(b) At any time that any LIBOR Loan is affected by the circumstances described in Section 2.10(a)(i) or (ii), the Borrower may (and in the case of a LIBOR Loan affected pursuant to Section 2.10(a)(i) shall) either (x) if the affected LIBOR Loan is then being made pursuant to a Borrowing, cancel said Borrowing by giving the Administrative Agent written notice thereof on the same date that the Borrower was notified by a Lender pursuant to Section 2.10(a)(i) or (ii) or (y) if the affected LIBOR Loan is then outstanding, upon at least three Business Days' notice to the Administrative Agent, require the affected Lender to convert each such LIBOR Term Loan into an ABR Loan; provided that if more than one Lender is affected at any time, then all affected Lenders must be treated in the same manner pursuant to this Section 2.10(b).

(c) If, after the Signing Date, the adoption of any applicable law, rule or regulation regarding capital adequacy, or any change therein, or any change in the interpretation or administration thereof by any governmental authority, the National Association of Insurance Commissioners, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by a Lender or its parent with any request or directive made or adopted after the Signing Date regarding capital adequacy (whether or not having the force of law) of any such authority, association, central bank or comparable agency, has or would have the effect of reducing the rate of return on such Lender's or its parent's or its Affiliate's capital or assets as a consequence of such Lender's

commitments or obligations hereunder to a level below that which such Lender or its parent or its Affiliate could have achieved but for such adoption, effectiveness, change or compliance (taking into consideration such Lender's or its parent's policies with respect to capital adequacy), then from time to time, promptly after demand by such Lender (with a copy to the Administrative Agent), the Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender or its parent for such reduction, it being understood and agreed, however, that a Lender shall not be entitled to such compensation as a result of such Lender's compliance with, or pursuant to any request or directive to comply with, any such law, rule or regulation as in effect on the Signing Date. Each Lender, upon determining in good faith that any additional amounts will be payable pursuant to this Section 2.10(c), will give prompt written notice thereof to the Borrower which notice shall set forth in reasonable detail the basis of the calculation of such additional amounts, although the failure to give any such notice shall not, subject to Section 2.13, release or diminish the Borrower's obligations to pay additional amounts pursuant to this Section 2.10(c) upon receipt of such notice. For the avoidance of doubt, this Section 2.10(c) shall apply to all requests, rules, guidelines or directives concerning capital adequacy promulgated by the United States financial regulatory authorities (including regulations implementing the recommendations of the Bank for International Settlements), regardless of the date adopted, issued, promulgated or implemented.

(d) It is understood that to the extent duplicative of Section 5.4, this Section 2.10 shall not apply to Taxes.

2.11. <u>Compensation</u>. If (a) any payment of principal of any LIBOR Loan is made by the Borrower to or for the account of a Lender other than on the last day of the Interest Period for such LIBOR Loan as a result of a payment or conversion pursuant to Section 2.5, 2.6, 2.10, 5.1, 5.2 or 14.7, as a result of acceleration of the maturity of the Loans pursuant to Section 12 or for any other reason, (b) any Borrowing of LIBOR Loans is not made as a result of a withdrawn Notice of Borrowing, (c) any ABR Loan is not converted into a LIBOR Loan as a result of a withdrawn Notice of Conversion or Continuation, (d) any LIBOR Loan is not continued as an LIBOR Loan, as the case may be, as a result of a withdrawn Notice of Conversion or Continuation or (e) any prepayment of principal of any LIBOR Loan is not made as a result of a withdrawn notice of prepayment pursuant to Section 5.1 or 5.2, the Borrower shall, after receipt of a written request by such Lender (which request shall set forth in reasonable detail the basis for requesting such amount), pay to the Administrative Agent for the account of such Lender any amounts required to compensate such Lender for any additional losses, costs or expenses that such Lender may reasonably incur as a result of such payment, failure to convert, failure to prepay, including any loss, cost or expense (excluding loss of anticipated profits) actually incurred by reason of the liquidation or reemployment of deposits or other funds acquired by any Lender to fund or maintain such LIBOR Loan.

2.12. <u>Change of Lending Office</u>. Each Lender agrees that, upon the occurrence of any event giving rise to the operation of Section 2.10(a)(i), 2.10(a)(ii), 2.10(b), 3.5 or 5.4 with respect to such Lender, it will, if requested by the Borrower use reasonable efforts (subject to overall policy considerations of such Lender) to designate another lending office for any Loans affected by such event; <u>provided</u> that such designation is made on such terms that such Lender and its lending office suffer no economic, legal or regulatory disadvantage, with the object of avoiding the consequence of the event giving rise to the operation of any such Section. Nothing in this Section 2.12 shall affect or postpone any of the obligations of the Borrower or the right of any Lender provided in Section 2.10, 3.5 or 5.4.

2.13. <u>Notice of Certain Costs</u>. Notwithstanding anything in this Agreement to the contrary, to the extent any notice required by Section 2.10, 2.11, 3.5 or 5.4 is given by any Lender more than 90 days after such Lender has knowledge (or should have had knowledge) of the occurrence of the event giving rise to the additional cost, reduction in amounts, loss, tax or other additional amounts described in such Sections, such Lender shall not be entitled to compensation under Section 2.10, 2.11, 3.5 or 5.4, as the case may be, for any such amounts incurred or accruing prior to the 91st day prior to the giving of such notice to the Borrower.

### 2.14. Inability to Determine Rates.

(a) If prior to the commencement of any Interest Period for a LIBOR Rate Borrowing:

(i) the Administrative Agent determines (which determination shall be conclusive absent manifest error) that adequate and reasonable means do not exist for ascertaining the LIBO Rate for such Interest Period; or

(ii) the Administrative Agent is advised by the Required Lenders that the LIBO Rate for such Interest Period will not adequately and fairly reflect the cost to such Lenders of making or maintaining their Loans included in such Borrowing for such Interest Period;

then the Administrative Agent shall give notice thereof to the Borrower and the Lenders by facsimile, telephone or electronic means as promptly as practicable thereafter and, until the Administrative Agent notifies the Borrower and the Lenders that the circumstances giving rise to such notice no longer exist, (i) any Notice of Conversion or

Continuation that requests the conversion of any Borrowing to, or continuation of any Borrowing as, a LIBO Rate Borrowing shall be ineffective and such Borrowing shall be converted to or continued as on the last day of the Interest Period applicable thereto an ABR Borrowing, and (ii) if any Notice of Borrowing requests a LIBO Rate Borrowing, such Borrowing shall be made as an ABR Borrowing.

(b) Notwithstanding anything to the contrary in this Agreement or the other Credit Documents, if at any time there ceases to exist a LIBO Rate or other interbank rate in the London Market regulated or otherwise overseen or authorized by the ICE Benchmark Administration or U.K. Financial Conduct Authority for interest periods greater than one (1) Business Day or the Administrative Agent determines in good faith (which determination shall be conclusive absent manifest error) that (i) the circumstances set forth in Section 2.14(a)(i) or (ii) above have arisen and such circumstances are unlikely to be temporary or (ii) the circumstances above have not arisen but the supervisor for the administrator of the LIBO Rate or a Governmental Authority having jurisdiction over the Administrative Agent has made a public statement identifying a specific date after which the LIBO Rate shall no longer be used for determining interest rates for loans, then the Administrative Agent and the Borrower shall endeavor to establish an alternate rate of interest to the LIBO Rate that gives due consideration to the then prevailing market convention for determining a rate of interest for fixed periods for syndicated loans in the United States at such time, and they shall enter into an amendment to the Credit Documents to reflect such alternate rate of interest and such other related changes as may be applicable and that are agreed by the Borrower and the Administrative Agent at such time, and shall include any technical, administrative or operational changes that the Administrative Agent reasonably requests and that are acceptable to the Borrower to reflect the adoption and implementation of such alternate rate of interest and to permit the administration thereof by the Administrative Agent in a manner substantially consistent with market practice; provided that, if such alternate rate of interest shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement. Notwithstanding anything to the contrary in the Credit Documents, such amendment shall become effective without any further action or consent of any other party to Credit Documents so long as the Administrative Agent shall not have received, within five (5) Business Days of the date that notice of such alternate rate of interest is provided to the Lenders, a written notice from the Required Lenders stating that they object to such amendment.

#### 2.15. [<u>Reserved</u>].

2.16. <u>MIRE Event</u>. Notwithstanding anything to the contrary herein, the making, increasing, extension or renewal of any Loans pursuant to this Agreement shall be subject to flood insurance due diligence and flood insurance compliance in accordance with Section 9.3(c) hereto and shall otherwise be reasonably satisfactory to the Administrative Agent and the Lenders.

### SECTION 3. Priority and Liens.

3.1. <u>DIP Liens</u>. Subject to the Carve Out, the Obligations shall be secured by valid, binding, continuing enforceable, fully-perfected, nonavoidable, automatically and properly perfected Liens on, and security interests in (such liens and security interests, the "<u>DIP Liens</u>"), all present and after acquired property (whether tangible, intangible, real, personal or mixed) of the DIP Debtors wherever located, including, without limitation, all accounts, as-extracted collateral, deposit accounts, cash and cash equivalents, inventory, equipment, capital stock in subsidiaries of the DIP Debtors, and the proceeds thereof, investment property, instruments, chattel paper, real estate, leasehold rights and leasehold interests, contracts, patents, copyrights, trademarks and other general intangibles, and all products and proceeds thereof, including proceeds from any directors/officers' insurance policies, and including (i) to the maximum extent permitted by law, all rights incident or appurtenant to the FCC Licenses and the right to receive all proceeds derived from or in connection with the sale, assignment or transfer of the FCC Licenses, (ii) the C-Band Payments and (iii) the Avoidance Proceeds (all such property, the "<u>DIP</u> <u>Collateral</u>") as follows:

(a) Pursuant to Bankruptcy Code section 364(c)(2), be secured by a valid, binding, continuing, enforceable, fully-perfected first priority senior security interest in and Lien upon all prepetition and postpetition property of the DIP Debtors, whether existing on the Petition Date or thereafter acquired, that, on or as of the Petition Date is not subject to valid, perfected and non-avoidable Liens (or perfected after the Petition Date to the extent permitted by Bankruptcy Code section 546(b)), including, without limitation, any unencumbered cash of the DIP Debtors (whether maintained with any Agent or otherwise) and any investment of such cash, accounts, inventory, good, contract rights, mineral rights, instruments, documents, chattel paper, patents, trademarks, copyrights, and licenses therefor, accounts receivable, receivables and receivables records, general intangible, payment intangibles, tax or other refunds, insurance proceeds, letters of credit, intercompany claims, contracts, owned real estate, real property leaseholds, fixtures, deposit accounts, commercial tort claims, securities accounts,

instruments, investment property, letter-of-credit rights, supporting obligations, vehicles, machinery and equipment, real property, leases (and proceeds from the disposition thereof), all of the issued and outstanding capital stock of each DIP Debtor, other equity or ownership interests, including equity interests in subsidiaries and non-wholly owned subsidiaries, money, investment property, causes of action (including the Avoidance Proceeds), and all cash and non-cash proceeds, rents, products, substitutions, accessions, profits and supporting obligations of any of the collateral described above, whether in existence on the Petition Date or thereafter created, acquired, or arising and wherever located, subject only to the Carve Out;

(b) Pursuant to section 364(d)(1) of the Bankruptcy Code, a valid, binding, continuing, enforceable, fully perfected first priority priming security interest and Lien (the "<u>Priming Liens</u>") on all prepetition and postpetition property of the DIP Debtors of the same nature, scope, and type as the collateral securing the Prepetition Secured Debt (the "<u>Prepetition Collateral</u>") whether in existence on the Petition Date or thereafter created, acquired, or arising and wherever located, that is subject to any of the Liens securing the obligations under the Prepetition Secured Debt (the "<u>Prepetition Liens</u>"), regardless of whether or not any Prepetition Liens on the assets are voided, avoided, invalidated, lapsed, or unperfected subject only to the Carve Out. The Priming Liens shall prime in all respects the Liens and security interests of the Prepetition Secured Parties, with respect to the Prepetition Secured Debt (including, without limitation, the Prepetition Liens and the Adequate Protection Liens granted to the Prepetition Secured Parties) (the "<u>Primed Liens</u>"). Notwithstanding anything herein to the contrary, the Priming Liens (i) shall be subject and junior to the Carve Out in all respects, (ii) shall be senior in all respects to the Prepetition Liens and (iii) shall also be senior to the Adequate Protection Liens;

(c) Pursuant to section 364(c)(3) of the Bankruptcy Code, be secured by a valid, binding, continuing, enforceable, fully perfected junior priority security interest and Lien on all prepetition and postpetition property of the DIP Debtors to the extent that such assets are subject to valid, perfected and unavoidable Liens in favor of third parties that were in existence immediately prior to the Petition Date (other than the Primed Liens), or to valid and unavoidable Liens in favor of third parties that were in existence immediately prior to the Petition Date that were perfected subsequent to the Petition Date as permitted by Section 546(b) of the Bankruptcy Code (other than Primed Liens), which Liens shall be (a) junior and subordinate to any such valid, perfected, and non-avoidable Liens (other than Primed Liens) in existence immediately prior to the Petition Date and/or (b) any such valid and non-avoidable Liens in existence immediately prior to the Petition Date as permitted by section 546(b) of the Bankruptcy Code (other that are perfected subsequent to the Petition Date as permitted by section 546(b) of the Petition Date that are perfected subsequent to the Petition Date as permitted by section 546(b) of the Bankruptcy Code; provided that nothing in the foregoing shall limit the rights of the DIP Secured Parties under the DIP Documents to the extent any such Liens are not permitted thereunder; and

(d) Pursuant to section 364(c)(1) of the Bankruptcy Code, all of the Obligations shall constitute allowed superpriority administrative expense claims against the DIP Debtors on a joint and several basis (without the need to file any proof of claim) with priority over any and all claims against the DIP Debtors, now existing or hereafter arising, of any kind whatsoever, including, without limitation, all administrative expenses of the kind specified in sections 503(b) and 507(b) of the Bankruptcy Code and any and all administrative expenses or other claims ("Administrative Expense Claims") arising under sections 105, 326, 327, 328, 330, 331, 365, 503(b), 506(c), 507(a), 507(b), 726, 1113, or 1114 of the Bankruptcy Code (including the Adequate Protection Obligations), whether or not such expenses or claims may become secured by a judgment Lien or other non-consensual Lien, levy, or attachment, which allowed claims (the "DIP Superpriority Claims") shall for purposes of section 1129(a)(9)(A) of the Bankruptcy Code be considered administrative expenses allowed under sections 503(b) of the Bankruptcy Code, and which DIP Superpriority Claims shall be payable from and have recourse to all prepetition and postpetition property of the DIP Debtors and all proceeds thereof (excluding claims and causes of action under sections 502(d), 544, 545, 547, 548, and 550 of the Bankruptcy Code, or any other avoidance actions under the Bankruptcy Code (collectively, "Avoidance Actions") but including any proceeds or property recovered, unencumbered, or otherwise, from Avoidance Actions, whether by judgment, settlement, or otherwise ("Avoidance Proceeds")) in accordance with the other DIP Documents, subject only to the Carve Out. The DIP Superpriority Claims shall be entitled to the full protection of section 364(e) of the Bankruptcy Code if the Final DIP Order or any provision hereof is vacated, reversed, or modified, on appeal or otherwise;

3.2. <u>Primed Liens</u>. The Priming Liens (i) shall be subject and junior to the Carve Out in all respects, (ii) shall be senior in all respects to the interests in such property that secures the obligations in respect of the Primed Liens (such holders, as applicable, the "<u>Primed Parties</u>") and (iii) shall be senior to any Liens granted after the Petition Date to provide adequate protection in respect of any of the Primed Liens. The Primed Liens shall be primed by and made subject and subordinate to the Carve Out and the Priming Liens, but the Priming Liens shall not prime Liens, if any, to which the Prepetition Liens are subject to on the Petition Date (other than Liens which are themselves Prepetition Liens).

3.3. Lien Priority and Perfection. The relative priorities of the Liens described in this Section 3 with respect to the Collateral shall be as set forth in the Final DIP Order and the Security Agreement. In accordance with the Final DIP Order, all of the Liens described in this Section 3 shall be effective and perfected upon entry of the Final DIP Order, without the necessity of the execution, recordation or filings by the Credit Parties of security agreements, control agreements, pledge agreements, mortgages, intellectual property filings, notice of Liens, financing statements or other similar instruments or documents, or the possession or control by the Collateral Agent of, or over, any Collateral, or take any other action in order to validate or perfect the Liens and security interests granted by or pursuant to this Agreement, the Final DIP Order or any other Credit Document, as set forth in the Final DIP Order; provided, however, subject to the terms hereof, such effectiveness and perfection shall not preclude the Collateral Agent or the Required Lenders from requesting such execution, recordation or filing that the Collateral Agent or the Required Lenders may deem desirable (including any security documents or actions under laws of the applicable foreign jurisdictions).

3.4. Real Estate Liens. Further to Section 3.1 and the Final DIP Order, to secure the full and timely payment and performance of the Obligations, each Credit Party hereby MORTGAGES, GRANTS, BARGAINS, ASSIGNS, SELLS, CONVEYS and CONFIRMS, to the Collateral Agent, for the benefit of the Agents and the ratable benefit of the Secured Parties, the Real Estate and property and interests relating thereto (which, for the avoidance of doubt, shall include all of such Credit Party's right, title and interest now or hereafter acquired in and to (A) all improvements now owned or hereafter acquired by such Credit Party, (B) all materials, supplies, equipment, apparatus and other items of personal property now owned or hereafter acquired by such Credit Party and now or hereafter attached to, installed in or used in connection with the Real Estate, and all utilities whether or not situated in easements, and all equipment, inventory and other goods in which such Credit Party now has or hereafter acquires any rights or any power to transfer rights and that are or are to become fixtures (as defined in the Uniform Commercial Code) related to the Real Estate, (C) [reserved], (D) all reserves, escrows or impounds and all deposit accounts maintained by such Credit Party with respect to the Real Estate, (E) all leases, licenses, concessions, occupancy agreements or other agreements (written or oral, now or at any time in effect) which grant to any Person a possessory interest in, or the right to use, all or any part of the Real Estate, together with all related security and other deposits, (F) all of the rents, revenues, royalties, income, proceeds, profits, accounts receivable, security and other types of deposits, and other benefits paid or payable by parties to the leases for using, leasing, licensing possessing, operating from, residing in, selling or otherwise enjoying the Real Estate, (G) all other agreements, such as construction contracts, architects' agreements, engineers' contracts, utility contracts, maintenance agreements, management agreements, service contracts, listing agreements, guaranties, warranties, permits, licenses, certificates and entitlements in any way relating to the construction, use, occupancy, operation, maintenance, enjoyment or ownership of the Real Estate, (H) all rights, privileges, tenements, hereditaments, rights-of-way, easements, appendages and appurtenances appertaining to the foregoing, (I) all property tax refunds payable with respect to the Real Estate, (J) all accessions, replacements and substitutions for any of the foregoing and all proceeds thereof, (K) all insurance policies, unearned premiums therefor and proceeds from such policies covering any of the above property now or hereafter acquired by such Credit Party as an insured party, and (L) all awards, damages, remunerations, reimbursements, settlements or compensation heretofore made or hereafter to be made to any Credit Party by any governmental authority pertaining to any condemnation or other taking (or any purchase in lieu thereof) of all or any Real Estate), TO HAVE AND TO HOLD to the Collateral Agent, and such Credit Party does hereby bind itself, its successors and assigns to WARRANT AND FOREVER DEFEND the title to such property, assets and interests unto the Collateral Agent.

3.5. <u>No Discharge</u>; <u>Survival of Claims</u>. Each of the Credit Parties agrees that prior to Payment in Full of the Obligations and termination of the Commitments, (a) its obligations under the Credit Documents shall not be discharged by the entry of an order confirming a Reorganization Plan (and each of the Credit Parties, pursuant to Section 1141(d)(4) of the Bankruptcy Code, hereby waives any such discharge) and (b) the DIP Superpriority Claims granted to the Agents and the Lenders pursuant to the Final DIP Order and the DIP Liens granted to the Agents and the Lenders pursuant to the Final DIP Order shall not be affected in any manner by the entry of an order confirming a Reorganization Plan.

3.6. <u>Payment of Obligations</u>. Subject to the Final DIP Order, upon the maturity (whether by acceleration or otherwise) of any of the Obligations of the Credit Parties under this Agreement or any of the other Credit Documents, the Agents and the Lenders shall be entitled to immediate payment of such Obligations without further application to or order of the Bankruptcy Court.

SECTION 4. Fees; Commitments.

#### 4.1. Fees.

(a) [Reserved].

(b) The Borrower agrees to pay to the Administrative Agent for the ratable account of the Lenders entitled thereto, an upfront payment (the "<u>Upfront Payment</u>") in cash (which, at the discretion of the Required Lenders (with written notice provided to the Administrative Agent), may take the form of original issue discount) in an amount equal to 1.5% of (the "<u>Upfront Payment Percentage</u>") of (i) the aggregate principal amount of the Term Loans funded hereunder in respect of the first Borrowing referred to in Section 2.2, which first Upfront Payment shall be earned, due and payable on the date of such first Borrowing and calculated by multiplying the Upfront Payment Percentage by the aggregate principal amount of Term Loans funded by such Lender on the date of such first Borrowing, (ii) the aggregate principal amount of the Term Loans funded hereunder in respect of the second Borrowing referred to in Section 2.2, which second Upfront Payment shall be earned, due and payable on the date of such first Borrowing and calculated by the aggregate principal amount of the Term Loans funded by such Lender on the date of such first Borrowing and calculated by multiplying the Upfront Payment shall be earned, due and payable on the date of such second Borrowing and calculated by multiplying the Upfront Payment shall be earned, due and payable on the date of such second Borrowing and calculated by multiplying the Upfront Payment Percentage by the aggregate principal amount of the Term Loans funded hereunder in respect of the third Borrowing referred to in Section 2.2, which third Upfront Payment shall be earned, due and payable on the date of such third Borrowing; and (iii) the aggregate principal amount of the Term Loans funded by such Lender on the date of such third Borrowing; provided that notwithstanding the foregoing, for the avoidance of doubt no applicable original issue discount shall reduce the amount of Obligations required to be prepaid or repaid with foregoing, for the avoidance of doubt no applicable original i

(c) The Borrower agrees to pay to the Administrative Agent for the ratable account of the Lenders entitled thereto, (i) on the last day of each fiscal quarter prior to the Commitment Termination Date and (ii) on the Commitment Termination Date, a ticking fee with respect to the then Available Term Loan Commitments (the "<u>Ticking Fee</u>"), in arrears, in cash equal to 3.60% per annum multiplied by the amount of the Available Term Loan Commitments accruing from the Final DIP Order Entry Date for any period then ending for which the Ticking Fee shall not have previously been paid.

(d) If the Borrower shall elect to exercise the First Maturity Extension, the Borrower shall pay, to the Administrative Agent for the ratable account of the Lenders entitled thereto, on the date of such election, as fee compensation for such extension of the Scheduled Maturity Date of the Term Loans, an extension fee (the "First Extension Fee") in cash in an amount equal to 0.50% of the sum of (i) the aggregate principal amount of the Term Loans then outstanding and (ii) the aggregate amount of the then Available Term Loan Commitments.

(e) If the Borrower shall elect to exercise the Second Maturity Extension, the Borrower shall pay, to the Administrative Agent for the ratable account of the Lenders entitled thereto, on the date of such election, as fee compensation for such extension of the Scheduled Maturity Date of the Term Loans, an extension fee (the "Second Extension Fee") in cash in an amount equal to 0.50% of the sum of (i) the aggregate principal amount of the Term Loans then outstanding and (ii) the aggregate amount of the then Available Term Loan Commitments.

(f) The Borrower agrees to pay the Agents, for their own account, fees payable in the amounts and at the times separately agreed upon between the Borrower and the applicable Agents (including pursuant to the Administrative Agent Fee Letter).

(g) All fees shall be paid on the dates due, in immediately available funds, to the Administrative Agent for appropriate distribution in accordance herewith. Once paid, none of the fees shall be refundable under any circumstances. Notwithstanding the foregoing, the Borrower shall not be obligated to pay any amounts to any Defaulting Lender pursuant to this Section 4.1.

4.2. <u>Voluntary Reduction of Term Loan Commitments</u>. Upon at least one Business Day's prior written notice to the Administrative Agent at the Administrative Agent's Office (which notice the Administrative Agent shall promptly transmit to each of the Lenders), the Borrower (on behalf of itself) shall have the right, without premium or penalty, on any day, permanently to terminate or reduce the Term Loan Commitments in whole or in part; provided that (a) any such reduction shall apply proportionately and permanently to reduce the Term Loan Commitment of each of the Lenders and (b) any partial reduction pursuant to this Section 4.2 shall be in the amount of at least \$5,000,000.

4.3. Mandatory Termination of Commitments.

(a) The Term Loan Commitments shall be permanently reduced dollar for dollar at the time of funding of any Term Loans thereunder and terminate on the Commitment Termination Date.

(b) The Term Loan Commitments shall be reduced pursuant to Section 5.2(a)(ii) below.

(c) The Term Loan Commitments shall terminate on or after the earliest to occur of (i) 45 days after the Petition Date unless prior to such time the Final DIP Order shall have been entered by the Bankruptcy Court, (ii)(x) prior to the Closing Date, the date when any DIP Debtors file a motion with the Bankruptcy Court for an alternative debtor-in-possession financing or use of cash collateral not contemplated by the Interim Cash Collateral Order and (y) on and after the Closing Date, the date when any DIP Debtors file a motion with the Bankruptcy Court for an alternative debtor-in-possession financing or use of cash collateral Order or Final Cash Collateral Order, as applicable, (iii) the Credit Documents shall not have been executed by the Borrower and the other Credit Parties and the conditions precedent contained in Section 6 of the this Agreement have not been satisfied within eight (8) Business Days after the Final Order Entry Date and (iv) the Borrower's failure to request the initial Borrowing within one (1) Business Day upon all conditions precedent to the initial Borrowing as set forth in this Agreement (other than the delivery of a Notice of Borrowing by the Borrower and other conditions of a nature that shall be satisfied concurrently with the funding of the initial Borrowing) having been satisfied.

### SECTION 5. Payments.

5.1. <u>Voluntary Prepayments</u>. The Borrower shall have the right to prepay Term Loans, without premium or penalty, in whole or in part from time to time on the following terms and conditions and subject to clause (b) and (c) below: (a) the Borrower shall give the Administrative Agent and at the Administrative Agent's Office written notice of its intent to make such prepayment, the amount of such prepayment and (in the case of LIBOR Loans) the specific Borrowing(s) pursuant to which made, which notice shall be given by the Borrower no later than (i) 10:00 a.m. (New York City time) one Business Day (or, in the case of LIBOR Loans, three (3) Business Days) prior to, the date of such prepayment and shall promptly be transmitted by the Administrative Agent to each of the Lenders; (b) each partial prepayment of any Borrowing of Term Loans shall be in a multiple of \$1,000,000 and in an aggregate principal amount of at least \$5,000,000; and (c) any prepayment of LIBOR Term Loans pursuant to this Section 5.1 on any day other than the last day of an Interest Period applicable thereto shall be subject to compliance by the Borrower with the applicable provisions of Section 2.11. At the Borrower's election, prepayments of Terms Loans pursuant to this Section 5.1 may be applied pro rata among the Term Loans. Each prepayment in respect of any tranche of Term Loans pursuant to this Section 5.1 shall be applied to reduce Term Loans in such order as the Borrower may determine. At the Borrower's election in connection with any prepayment pursuant to this Section 5.1, such prepayment shall not be applied to any Term Loan of a Defaulting Lender.

# 5.2. Mandatory Prepayments.

# (a) Term Loan Prepayments.

(i) On each occasion that a Prepayment Event (other than a C-Band Prepayment Event) occurs, the Borrower shall, within one Business Day after the occurrence of a Debt Incurrence Prepayment Event and within five Business Days after the occurrence of any other Prepayment Event (other than a C-Band Prepayment Event) and the receipt of Net Cash Proceeds by the Borrower or any Restricted Subsidiary in connection with such Prepayment Event, prepay, in accordance with paragraph (c) below, the principal amount of Term Loans in an amount equal to 100% of the Net Cash Proceeds from such Prepayment Event; provided that any proceeds received in escrow shall not be deemed to be received for purposes of this Section 5.2(a) until such proceeds are released from escrow.

(ii) On the tenth Business Day of each month, the Borrower shall, <u>first</u>, reduce the then Available Term Loan Commitment in an amount equal to 100% of the Net Cash Proceeds of the C-Band Sweep Payment from all C-Band Prepayment Events that occurred during the prior month; and <u>second</u>, to the extent the then Available Term Loan Commitment equals zero (\$0), prepay the principal amount of Term Loans in an amount equal to 100% of the Net Cash Proceeds of the C-Band Sweep Payment from all C-Band Prepayment Events that occurred during the prior month (minus the amount of the Net Cash Proceeds of the C-Band Sweep Payment from the applicable C-Band Prepayment Event that shall have reduced the then Available Term Loan Commitments pursuant to the preceding "first" clause), in each case, in accordance with paragraph (c) below; <u>provided</u> that any proceeds are released to the Borrower or any Restricted Subsidiary in escrow shall not be deemed to be received for purposes of this Section 5.2(a)(ii) until such proceeds are released to the Borrower or any Restricted Subsidiary from escrow; <u>provided further</u> that, notwithstanding the foregoing, with respect to each individual month, making the full amount of any such required commitment reduction or prepayment shall only be required to the extent that, after giving pro forma effect to such commitment reduction and/or prepayment, the Liquidity of the Credit Parties on such date would be greater than or equal to \$500,000,000 with such required commitment reduction and/or prepayment amount being reduced for each month to permit the DIP Debtors to maintain pro forma Liquidity of \$500,000,000.

## (b) [<u>Reserved</u>].

(c) <u>Application of Prepayment Amounts</u>. Each prepayment of Term Loans and reduction of Available Term Loan Commitments required by Section 5.2(a) shall be applied pro rata among the Term Loans and the Available Term Loan Commitments, as the case may be. With respect to each such prepayment or commitment reduction, the Borrower will, not later than the date specified in Section 5.2(a) for making such prepayment or commitment reduction, give the Administrative Agent written notice requesting that the Administrative Agent provide notice of such prepayment or commitment reduction to each applicable Lender.

(d) <u>Application to Term Loans</u>. With respect to each prepayment of Term Loans required by Section 5.2(a), the Borrower may designate the Types of Loans that are to be prepaid and the specific Borrowing(s) pursuant to which made. In the absence of a designation by the Borrower as described in the preceding sentence, the Administrative Agent shall, subject to the above, make such designation in its reasonable discretion with a view, but no obligation, to minimize breakage costs owing under Section 2.11.

(e) [<u>Reserved</u>].

(f) [<u>Reserved</u>].

(g) Notwithstanding anything to the contrary contained in this Section 5.2 or elsewhere in this Agreement (including, without limitation, in Section 14.1), each Lender with outstanding Term Loans or Available Term Loan Commitments may reject all (but not less than all) of its applicable share of a mandatory repayment of Term Loans or commitment reduction of the Available Term Loan Commitments which is to be made pursuant to Section 5.2 (such declined amounts, the "Declined Proceeds") by providing written notice to the Administrative Agent no later than 4:00 P.M. (New York time) on the date which is two Business Days after the date of such notice from the Administrative Agent (and the Administrative Agent shall promptly thereafter notify the Borrower thereof). If any Lender with outstanding Term Loans or Available Term Loan Commitments does not reply to the Administrative Agent within such two Business Day period, such Lender will be deemed to have accepted the total amount of such mandatory prepayment or commitment reduction. Any Declined Proceeds shall be retained by the Borrower.

5.3. Method and Place of Payment.

(a) Except as otherwise specifically provided herein, all payments under this Agreement shall be made by the Borrower, without set-off, counterclaim or deduction of any kind, to the Administrative Agent for the ratable account of the Lenders entitled thereto, not later than 12:00 Noon (New York City time) on the date when due and shall be made in immediately available funds at the Administrative Agent's Office or at such other office as the Administrative Agent shall specify for such purpose by notice to the Borrower, it being understood that written or facsimile notice by the Borrower to the Administrative Agent to make a payment from the funds in the Borrower's account at the Administrative Agent's Office shall constitute the making of such payment to the extent of such funds held in such account. All payments under each Credit Document (whether of principal, interest or otherwise) shall be made in Dollars. The Administrative Agent will thereafter cause to be distributed on the same day (if payment was actually received by the Administrative Agent prior to 2:00 p.m. (New York City time) on such day) like funds relating to the payment of principal or interest or Fees ratably to the Lenders entitled thereto.

(b) Any payments under this Agreement that are made later than 2:00 p.m. (New York City time) shall be deemed to have been made on the next succeeding Business Day. Whenever any payment to be made hereunder shall be stated to be due on a day that is not a Business Day, the due date thereof shall be extended to the next succeeding Business Day and, with respect to payments of principal, interest shall be payable during such extension at the applicable rate in effect immediately prior to such extension.

# 5.4. Net Payments.

(a) Any and all payments made by or on behalf of any Credit Party under this Agreement or any other Credit Document shall be made free and clear of, and without deduction or withholding for or on account of, any Indemnified Taxes; provided that if any Credit Party or other applicable withholding agent shall be required by law to deduct or withhold any Indemnified Taxes from such payments, then (i) the sum payable by the applicable Credit Party shall be increased as necessary so that after all required deductions and withholdings have been made (including deductions or withholdings applicable to additional sums payable under this Section 5.4) the Administrative Agent or any Lender, as the case may be, receives an amount equal to the sum it would have received had no such deductions or withholdings been made, (ii) the applicable withholding agent shall make such deductions or withholdings and (iii) the applicable withholding agent shall pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable law. Whenever any Indemnified Taxes are payable by any Credit Party, as promptly as possible thereafter, the Borrower shall send to the Administrative Agent for its own account or for the account of such Lender, as the case may be, a certified copy of an original official

receipt (or other evidence acceptable to such Lender, acting reasonably) received by the applicable Credit Party showing payment thereof.

(b) Borrower shall timely pay any Other Taxes.

(c) Borrower shall indemnify and hold harmless the Administrative Agent and each Lender within 15 Business Days after written demand therefor, for the full amount of any Indemnified Taxes imposed on the Administrative Agent or such Lender as the case may be, on or with respect to any payment by or on account of any obligation of any Credit Party hereunder or under any other Credit Document and any Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section 5.4) and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender or by the Administrative Agent on its own behalf or on behalf of a Lender shall be conclusive absent manifest error.

(d) A Lender that is entitled to an exemption from or reduction in a withholding tax imposed under the laws of Luxembourg with respect to any payments under this Agreement or any other Credit Document shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent such properly completed and executed documentation as will permit such payments to be made without withholding or at a reduced rate; provided that such Lender is legally entitled to complete, execute and deliver such documentation. To the extent it is legally entitled to do so, each Lender agrees to use reasonable efforts (consistent with legal and regulatory restrictions and subject to overall policy considerations of such Lender) to file or deliver to the Borrower and the Administrative Agent any certificate or document, as reasonably requested by the Borrower or the Administrative Agent, that may be necessary to establish any available exemption from, or reduction in the amount of, any withholding taxes imposed by a jurisdiction other than Luxembourg; provided, however, that a Lender shall not be required to file or deliver any such certificate or document if in such Lender's reasonable judgment such completion, execution or delivery would be disadvantageous to such Lender or would subject such Lender to any unreimbursed cost.

(e) If the Borrower determines in good faith that a reasonable basis exists for contesting any taxes for which indemnification has been demanded hereunder, the relevant Lender or the Administrative Agent, as applicable, shall cooperate with the Borrower in challenging such taxes at the Borrower's expense if so requested by the Borrower. If any Lender or the Administrative Agent, as applicable, receives a refund of a tax (in cash or applied as an offset against other cash tax liabilities) for which a payment has been made by any Credit Party pursuant to this Agreement, which refund in the good faith judgment of such Lender or Administrative Agent, as the case may be, is attributable to such payment made by such Credit Party, then the Lender or the Administrative Agent, as the case may be, shall reimburse the Borrower for such amount (together with any interest received thereon) as the Lender or Administrative Agent, as the case may be, determines to be the proportion of the refund as will leave it, after such reimbursement, in no better or worse position (taking into account expenses or any taxes imposed on the refund) than it would have been in if the payment had not been required; provided that the Borrower shall return any such amounts (along with any applicable interest) to the extent that the Administrative Agent or applicable Lender is required to repay any such refund to the applicable taxing authority. A Lender or Administrative Agent shall claim any refund that it determines is available to it, unless it concludes in its reasonable discretion that it would be adversely affected by making such a claim. Neither the Lender nor the Administrative Agent shall be obliged to disclose any information regarding its tax affairs or computations to the Borrower in connection with this paragraph (e) or any other provision of this Section 5.4.

(f) The agreements in this Section 5.4 shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

5.5. Computations of Interest and Fees.

(a) Interest on (x) LIBOR Loans shall be calculated on the basis of a 360-day year and (y) ABR Loans shall be calculated on the basis of a 365-(or 366-, as the case may be) day year for the actual days elapsed.

(b) Fees shall be calculated on the basis of a 365- (or 366-, as the case may be) day year for the actual days elapsed.

5.6. Limit on Rate of Interest.

(a) <u>No Payment Shall Exceed Lawful Rate</u>. Notwithstanding any other term of this Agreement, the Borrower shall not be obliged to pay any interest or other amounts under or in connection with this Agreement in excess of the amount or rate permitted under or consistent with any applicable law, rule or regulation.

(b) <u>Payment at Highest Lawful Rate</u>. If the Borrower is not obliged to make a payment which it would otherwise be required to make, as a result of Section 5.6(a), the Borrower shall make such payment to the maximum extent permitted by or consistent with applicable laws, rules and regulations.

(c) <u>Adjustment if Any Payment Exceeds Lawful Rate</u>. If any provision of this Agreement or any of the other Credit Documents would obligate the Borrower to make any payment of interest or other amount payable to any Lender in an amount or calculated at a rate which would be prohibited by any applicable law, rule or regulation, then notwithstanding such provision, such amount or rate shall be deemed to have been adjusted with retroactive effect to the maximum amount or rate of interest, as the case may be, as would not be so prohibited by law, such adjustment to be effected, to the extent necessary, by reducing the amount or rate of interest required to be paid by the Borrower to the affected Lender under Section 2.8.

Notwithstanding the foregoing, and after giving effect to all adjustments contemplated thereby, if any Lender shall have received from the Borrower an amount in excess of the maximum permitted by any applicable law, rule or regulation, then the Borrower shall be entitled, by notice in writing to the Administrative Agent to obtain reimbursement from that Lender in an amount equal to such excess, and pending such reimbursement, such amount shall be deemed to be an amount payable by that Lender to the Borrower.

## SECTION 6. Conditions Precedent to Effectiveness on the Closing Date.

The effectiveness of this Agreement on the Closing Date shall be subject to the prior or concurrent satisfaction of the Required Backstop Parties of each of the conditions precedent set forth in this Section 6 (or waiver thereof in accordance with Section 14.1).

6.1. Executed Counterparts of this Agreement. The Administrative Agent and counsel to the Backstop Parties shall have received this Agreement, duly executed by (A) each Lender, (B) each Credit Party and (C) each of the other parties hereto.

6.2. <u>13-Week DIP Budget</u>. The Administrative Agent and the financial advisor and counsel to the Backstop Parties shall have received the initial DIP Budget for the 13-week period ending after the Signing Date for informational purposes only.

6.3. <u>Corporate and Other Proceedings</u>. The Administrative Agent and counsel to the Backstop Parties shall have received from each Credit Party a certificate, executed by an officer of such Credit Party in form and substance reasonably satisfactory to the Required Backstop Parties, attaching: (i) a copy of the resolutions, in form and substance reasonably satisfactory to the Required Backstop Parties, of the board of directors (or similar body) of such Credit Party (or a duly authorized committee thereof) authorizing (A) the execution, delivery and performance of this Agreement and the Security Documents to which such Credit Party is party (and any other agreements relating thereto) and (B) in the case of the Borrower, the extensions of credit contemplated hereunder; (ii) the certificate of incorporation and bylaws (or memorandum and articles, or other documents of similar import pursuant to the laws of such Credit Party's jurisdiction of organization) of such Credit Party; and (iii) a certificate of good standing (or such other document of similar import as may be reasonably acceptable to the Required Backstop Parties) with respect to such Credit Party from the secretary of state (or comparable body) of the jurisdiction in which such Credit Party is organized, dated as of a recent date.

6.4. <u>Opinions of Counsel</u>. The Administrative Agent shall have received (i) a legal opinion from Kirkland & Ellis LLP, counsel to the Borrower and (ii) a legal opinion from Elvinger Hoss Prussen *société anonyme*, Luxembourg counsel to the Borrower, in each case, in form and substance reasonable satisfactory to the Required Backstop Parties.

6.5. <u>Promissory Notes</u>. Each applicable Lender shall have received, if requested at least two (2) Business Days prior to the Closing Date, one or more promissory notes payable to the order of such Lender duly executed by the Borrower in substantially the form of <u>Exhibits E</u> evidencing its Term Loans.

6.6. Fees. The Administrative Agent and Lenders shall have received the fees required to be paid on or prior to the Closing Date (including, without limitation, the backstop premium payable on the Signing Date pursuant to the Backstop Commitment Letter and the Restated Backstop Commitment Letter) and all reasonable, documented and invoiced out-of-pocket expenses (including the reasonable fees, disbursements and other charges of Akin Gump Strauss Hauer & Feld LLP, as counsel to the Lenders, and Centerview Partners, as financial advisors to the Lenders) for which invoices have been presented at least one (1) Business Days prior to the Closing Date shall have been paid.

6.7. <u>Collateral</u>. The Administrative Agent and counsel to the Backstop Parties shall have received (in each case in form and substance reasonably satisfactory to the Required Backstop Parties):

(a) the duly executed Security Agreement;

(b) UCC financing statements in appropriate form for filing under the UCC and such other documents under applicable Requirements of Law in each jurisdiction as may be, in the opinion of the Required Backstop Parties, reasonably desirable to perfect the Liens created, or purported to be created, by the Security Documents; and

(c) evidence (including a customary perfection certificate) that all other actions, recordings and filings that the Collateral Agent may deem desirable shall have been taken, completed or otherwise provided for in a manner reasonably satisfactory to the Required Backstop Parties;

provided, however, that without limiting the grant of a Lien on and security interest in the Collateral pursuant to the Orders and the Security Documents, with respect to and solely with respect to the occurrence of the Closing Date, (x) the Credit Parties will not be obligated to enter into any mortgages, authorize any fixture filing, enter into any agreement requiring "control" (as defined in Section 9-104, 9-105, 9-106 and 9-107 of the UCC as in effect in any relevant jurisdiction) or to undertake any registration in respect of assets subject to a certificate of title and (y) no Credit Party shall be required to complete any filings or other action with respect to the perfection of security interests in any jurisdiction outside of the United States; provided, further, that the preceding proviso shall not preclude the Agent or the Required Lenders from reasonably requesting such documents, filings or other actions after the Closing Date.

6.8. <u>Perfection Certificate</u>. The Collateral Agent and counsel to the Backstop Parties shall have received a completed Perfection Certificate substantially in the form of <u>Exhibit D</u>, together with all attachments contemplated thereby, in form and substance reasonably satisfactory to the Required Backstop Parties.

6.9. Intercompany Subordination Agreement. The Administrative Agent and counsel to the Backstop Parties shall have received a duly executed Intercompany Subordination Agreement substantially in the form of Exhibit I.

6.10. Petition Date. The Petition Date shall have occurred no later than May 15, 2020.

6.11. <u>Final DIP Order</u>. The Final DIP Order Entry Date shall have occurred not later than 45 days following the Petition Date, and the Final DIP Order shall not have been vacated, reversed, modified, amended or stayed.

6.12. <u>First Day Orders</u>. All First Day Orders, including, without limitation, all final orders with respect to the relief sought in the Debtors' motions requesting entry of the First Day Orders, shall have been entered by the Bankruptcy Court and all such entered First Day Orders shall be reasonably satisfactory in form and substance to the Required Backstop Parties, it being understood that drafts approved by counsel to the Required Backstop Parties, on or prior to the Signing Date are reasonably satisfactory.

 $6.13. \underline{\text{Trustee}}$ . No trustee under Chapter 7 or Chapter 11 of the Bankruptcy Code or examiner with enlarged powers beyond those set forth in Section 1106(a)(3) and (4) of the Bankruptcy Code shall have been appointed in any of the Cases.

6.14. Material Adverse Effect. Since the Signing Date, there shall have been no Material Adverse Effect.

6.15. Patriot Act. The Administrative Agent and each Lender that has requested the same shall have received on or prior to the Closing Date (i) all documentation and other information reasonably requested in writing by them at least ten (10) Business Days prior to the Closing Date that they shall have reasonably determined is required by the applicable regulatory authorities to comply with applicable "know your customer" and anti-money laundering rules and regulations, including the USA PATRIOT Act and (ii) to the extent the Borrower qualifies as a "legal entity customer" under the Beneficial Ownership Regulation, a Beneficial Ownership Certification to the extent requested in writing at least ten (10) Business Days prior to the Closing Date by the Administrative Agent or such Lender, as applicable.

6.16. Liens. The Collateral Agent, for the benefit of the Secured Parties, shall have valid and perfected Liens on all Collateral, to the extent contemplated hereby, and pursuant to the other Credit Documents, including the Final DIP Order, in each case, subject to the provisos at the end of Section 6.7.

For purposes of determining whether the conditions set forth in this Section 6 have been satisfied, by releasing its signature page hereto, the Administrative Agent and each Lender party hereto shall be deemed to have consented to, approved, accepted or be satisfied with each document or other matter required hereunder to be consented to or approved by, or acceptable or satisfactory to, the Administrative Agent or such Lender, as the case may be.

## SECTION 7. Conditions Precedent to All Credit Events.

The agreement of each Lender to make any Loan requested to be made by it on or after the Closing Date is subject to the satisfaction of the following conditions precedent:

7.1. <u>No Default; Representations and Warranties</u>. At the time of each Credit Event and also after giving effect thereto (a) no Default or Event of Default shall have occurred and be continuing and (b) all representations and warranties made by any Credit Party contained herein or in the other Credit Documents shall be true and correct in all material respects with the same effect as though such representations and warranties had been

made on and as of the date of such Credit Event (except where such representations and warranties expressly relate to an earlier date, in which case such representations and warranties shall have been true and correct in all material respects as of such earlier date).

7.2. Notice of Borrowing. Prior to the making of each Term Loan, the Administrative Agent shall have received a written Notice of Borrowing meeting the requirements of Section 2.3.

7.3. No Violation. The making of such Loan shall not violate any requirement of material Law applicable to the Credit Parties, after giving effect to the Final DIP Order and any other order of the Bankruptcy Court entered on or prior to the date of the applicable Credit Extension, and shall not be enjoined, temporarily, preliminarily or permanently.

7.4. Fees and Expenses. The Administrative Agent, the Backstop Parties and the Lenders shall have received the fees in the amounts previously agreed in writing by the Administrative Agent or otherwise pursuant to the Credit Documents, if any, on or prior to such date and all reasonable, documented and invoiced out-of-pocket expenses (including the reasonable fees, disbursements and other charges of counsel) payable by the Credit Parties for which, with respect to expenses, invoices have been presented at least one (1) Business Day prior to such date shall have been paid; which amounts may be paid from the proceeds of the Term Loans funded on the date of such Borrowing unless required to be paid by the Borrower prior to such date.

7.5. <u>Participation Election</u>. With respect to the first Borrowing, the Borrower or any of its Restricted Subsidiary that is a Guarantor party hereto and a guarantor party under the Prepetition Secured Debt shall have submitted to the FCC an election that is agreeable in form to the Borrower to participate in accelerated clearing of the C-band; *provided* that the entity receiving the accelerated relocation payments shall be a Debtor subsidiary of the Borrower (the "<u>Participation Election Form</u>").

In determining the satisfaction of the conditions specified in this Section 7, each Borrowing shall be deemed to constitute a representation and warranty by the Borrower on the date thereof as to the matters specified in Section 7.1.

### SECTION 8. Representations, Warranties and Agreements.

In order to induce the Lenders to enter into this Agreement, to make the Loans as provided for herein, each Credit Party, on behalf of itself and its Restricted Subsidiaries, makes the following representations and warranties to, and agreements with, the Lenders, on the Closing Date and the date of each Credit Event, all of which shall survive the execution and delivery of this Agreement and the making of the Loans:

8.1. <u>Corporate Status</u>. The Borrower and each Material Subsidiary (a) is a duly organized and validly existing corporation or public limited liability company or private limited liability company (as applicable) or other entity in good standing (to the extent that concept exists in the relevant jurisdiction) under the laws of the jurisdiction of its organization and, subject to the entry of the Final DIP Order and any other applicable orders of the Bankruptcy Court and subject to the terms thereof, has the corporate or other organizational power and authority to own its property and assets and to transact the business in which it is engaged and (b) has duly qualified and is authorized to do business and is in good standing in all jurisdictions where it is required to be so qualified, except where the failure to be so qualified could not reasonably be expected to result in a Material Adverse Effect.

8.2. <u>Corporate Power and Authority</u>. Subject to the entry of the Final DIP Order and subject to the terms thereof, each Credit Party has the corporate or other organizational power and authority to, execute, deliver and carry out the terms and provisions of the Credit Documents to which it is a party and has taken all necessary corporate or other organizational action to authorize the execution, delivery and performance of the Credit Documents to which it is a party. Subject to the entry of the Final DIP Order and subject to the terms thereof, each Credit Party has duly executed and delivered each Credit Document to which it is a party and each such Credit Document which is currently in effect constitutes the legal, valid and binding obligation of such Credit Party enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, insolvency or similar laws affecting creditors' rights generally and subject to general principles of equity subject to mandatory Luxembourg law provisions.

8.3. No Violation. Subject to the entry of the Final DIP Order and subject to the terms thereof, each Credit Party, neither the execution, delivery or performance by any Credit Party of the Credit Documents to which it is a party and which is currently in effect nor compliance with the terms and provisions thereof nor the consummation of the transactions contemplated hereby or thereby will (a) contravene any applicable provision of any material law, statute, rule, regulation, order, writ, injunction or decree of any court or governmental instrumentality, (b) except to the extent arising under the documents governing the Prepetition Debt, result in any breach of any of the terms, covenants, conditions or provisions of, or constitute a default under, or result in the

creation or imposition of (or the obligation to create or impose) any Lien upon any of the property or assets of the Borrower or any of the Restricted Subsidiaries (other than Liens created under the Credit Documents) pursuant to, the terms of any material indenture (including the Indentures governing the Prepetition Senior Notes and the Prepetition Secured Notes), loan agreement, lease agreement, mortgage, deed of trust, agreement or other material instrument to which the Borrower or any of the Restricted Subsidiaries is a party or by which it or any of its property or assets is bound or (c) violate any provision of the certificate of incorporation, by-laws or other constitutional documents (to the extent applicable) of the Borrower or any of the Restricted Subsidiaries.

8.4. <u>Litigation</u>. Except for the Cases, there are no actions, suits or proceedings (including Environmental Claims) pending or, to the knowledge of the Borrower, threatened with respect to the Borrower or any of its Subsidiaries that could reasonably be expected to result in a Material Adverse Effect.

8.5. <u>Margin Regulations</u>. Neither the making of any Loan hereunder nor the use of the proceeds thereof will violate the provisions of Regulation T, U or X of the Board.

8.6. Governmental Approvals. Subject to the entry of the Final DIP Order and subject to the terms thereof, the execution, delivery and performance of any Credit Document currently in effect does not require any consent or approval of, registration or filing with, or any other action by, any Governmental Authority, except for (i) such as have been obtained or made and are in full force and effect, (ii) filings and recordings in respect of the Liens created pursuant to the Security Documents, (iii) filings with the United States Patent and Trademark Office and the United States Copyright Office and comparable offices in foreign jurisdictions and equivalent filings in foreign jurisdictions, (iv) such FCC consents, approvals, registrations, and filings as may be required in connection with the exercise of rights under the Security Documents following an Event of Default, (v) such FCC consents, approvals, registrations, and filings as may be required in the ordinary course of business of the Borrower and its Subsidiaries in connection with the use of proceeds of the Loans hereunder, (vi) such licenses, approvals, authorizations and consents as may be required by the U.S. Department of State pursuant to the International Traffic in Arms Regulations, the U.S. Department of Commerce pursuant to the Export Administration Regulations and the U.S. Department of Treasury pursuant to Foreign Asset Control Regulations in connection with the exercise of rights hereunder and under the Security Documents following an Event of Default, and (vii) such licenses, approvals, authorizations or consents the failure to obtain or make could not reasonably be expected to have a Material Adverse Effect.

8.7. Investment Company Act. The Borrower is not required to register as an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

8.8. True and Complete Disclosure.

(a) None of the factual information and data (taken as a whole) heretofore or contemporaneously furnished by the Borrower, any of the Subsidiaries or any of their respective authorized representatives in writing to the Administrative Agent and/or any Lender in connection with this Agreement for purposes of or in connection with this Agreement or any transaction contemplated herein contained any untrue statement or omitted to state any material fact necessary to make such information and data (taken as a whole) not misleading at such time in light of the circumstances under which such information or data was furnished (subject, in the case of quarterly or interim financial statements, to normal year-end audit adjustments), it being understood and agreed that for purposes of this Section 8.8(a), such factual information and data shall not include projections and pro forma financial information.

(b) The DIP Budget has been based on good faith estimates and assumptions believed by such Persons to be reasonable at the time made, it being recognized by the Lenders that such projections as to future events are not to be viewed as facts and that actual results during the period or periods covered by any such projections may differ significantly from the projected results and such differences may be material.

8.9. Financial Condition; Financial Statements.

(a) The Historical Financial Statements, in each case present or will, when provided, present fairly in all material respects the combined financial position of the Borrower at the respective dates of said information, statements and results of operations for the respective periods covered thereby (subject, in the case of quarterly or interim financial statements, to normal year-end audit adjustments). The financial statements referred to in clause (a) of this Section 8.9 have been prepared in accordance with GAAP consistently applied except to the extent provided in the notes to said financial statements.

(b) There has been no Material Adverse Effect since the Signing Date.

8.10. Tax Returns and Payments.

(a) The Borrower and each of the Subsidiaries has filed all federal income tax returns and all other material tax returns, domestic and foreign, required to be filed by it and has paid all material Taxes payable by it that have become due (including in its capacity as a withholding agent), other than those (a) not yet delinquent or (b) the payment of which are stayed by the Cases, the nonpayment of which is permitted or required by the Bankruptcy

Code or that are contested in good faith as to which adequate reserves have been provided in accordance with GAAP and which could not reasonably be expected to result in a Material Adverse Effect. The Borrower and each of the Subsidiaries have paid, or have provided adequate reserves (in the good faith judgment of the management of the Borrower) in accordance with GAAP for the payment of, all material federal, state, provincial and foreign income taxes applicable for all prior fiscal years and for the current fiscal year.

(b) None of the Borrower or any of its Subsidiaries has ever been a party to any understanding or arrangement constituting a "tax shelter" within the meaning of Section 6662(d)(2)(C)(iii) of the Code or within the meaning of Section 6111(c) or Section 6111(d) of the Code as in effect immediately prior to the enactment of the American Jobs Creation of 2004, or has ever "participated" in a "reportable transaction" within the meaning of Treasury Regulation Section 1.6011-4, except as could not reasonably be likely to, individually or in the aggregate, have a Material Adverse Effect.

8.11. Compliance with ERISA.

(a) Each Plan is in compliance with ERISA, the Code and any applicable Requirement of Law; no Reportable Event has occurred (or is reasonably likely to occur) with respect to any Plan; no Plan is insolvent or in reorganization (or is reasonably likely to be insolvent or in reorganization), and no written notice of any such insolvency or reorganization has been given to the Borrower, any Subsidiary or any ERISA Affiliate; no failure to satisfy the minimum funding standard under Section 430 of the Code, whether or not waived, has occurred (or is reasonably expected to occur) with respect to a Plan or failure to make a required contribution to a Multiemployer Plan; none of the Borrower, any Subsidiary or any ERISA Affiliate has incurred (or is reasonably likely expected to incur) any liability to or on account of a Plan pursuant to Section 409, 502(i), 502(i), 515, 4062, 4063, 4064, 4069, 4201 or 4204 of ERISA or Section 4971 or 4975 of the Code or has been notified in writing that it will incur any liability under any of the foregoing Sections with respect to any Plan; no proceedings have been instituted (or are reasonably likely to be instituted) to terminate or to reorganize any Plan or to appoint a trustee to administer any Plan, and no written notice of any such proceedings has been given to the Borrower, any Subsidiary or any ERISA Affiliate; and no Lien imposed under the Code or ERISA on the assets of the Borrower or any Subsidiary or any ERISA Affiliate exists (or is reasonably likely to exist) nor has the Borrower, any Subsidiary or any ERISA Affiliate been notified in writing that such a Lien will be imposed on the assets of the Borrower, any Subsidiary or any ERISA Affiliate on account of any Plan, except to the extent that a breach of any of the representations, warranties or agreements in this Section 8.11 would not result, individually or in the aggregate, in an amount of liability that would be reasonably likely to have a Material Adverse Effect or relates to any matter disclosed in the financial statements of the Borrower contained in the Confidential Information Memorandum. No Plan (other than a multiemployer plan) has an Unfunded Current Liability that would, individually or when taken together with any other liabilities referenced in this Section 8.11, be reasonably likely to have a Material Adverse Effect. With respect to Plans that are multiemployer plans (as defined in Section 3(37) of ERISA), the representations and warranties in this Section 8.11(a), other than any made with respect to (i) liability under Section 4201 or 4204 of ERISA or required contributions to a Multiemployer Plan or (ii) liability for termination or reorganization of such Plans under ERISA, are made to the best knowledge of the Borrower

(b) All Foreign Plans are in compliance with, and have been established, administered and operated in accordance with, the terms of such Foreign Plans and applicable law, except for any failure to so comply, establish, administer or operate the Foreign Plans as would not reasonably be expected to have a Material Adverse Effect. All contributions or other payments which are due with respect to each Foreign Plan have been made in full and there are no funding deficiencies thereunder, except to the extent any such events would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(c) The Borrower represents and warrants as of the Amendment 3 Effective Date that the Borrower is not and will not be using "plan assets" (within the meaning of 29 CFR § 2510.3-101, as modified by Section 3(42) of ERISA) in connection with the Loans or the Commitments;

8.12. <u>Subsidiaries</u>. <u>Schedule 8.12</u> to this Agreement lists each Subsidiary of the Borrower (and the direct and indirect ownership interest of the Borrower therein), in each case existing on the Signing Date. All of the outstanding capital stock of the Borrower is owned directly or indirectly by Holdings.

8.13. <u>Patents, etc.</u> The Borrower and each of the Restricted Subsidiaries have obtained all patents, trademarks, servicemarks, trade names, copyrights, licenses and other rights, free from burdensome restrictions, that are necessary for the operation of their respective businesses as currently conducted and as proposed to be conducted, except where the failure to obtain any such rights could not reasonably be expected to have a Material Adverse Effect.

8.14. Environmental Laws.

(a) Except as could not reasonably be expected to have a Material Adverse Effect: (i) the Borrower and each of the Subsidiaries and all Real Estate are in compliance with all Environmental Laws; (ii) neither the Borrower nor any of the Subsidiaries is subject to any Environmental Claim or any other liability under any Environmental Law; (iii) the Borrower and its Subsidiaries are not conducting any investigation, removal, remedial or other corrective action pursuant to any Environmental Law at any location; and (iv) no underground storage tank or related piping, or any impoundment or other disposal area containing Hazardous Materials is located at, on or under any Real Estate currently owned or leased by the Borrower or any of its Subsidiaries.

(b) Neither the Borrower nor any of the Subsidiaries has treated, stored, transported, released or disposed or arranged for disposal or transport for disposal of Hazardous Materials at, on, under or from any currently or formerly owned or leased Real Estate or facility in a manner that could reasonably be expected to have a Material Adverse Effect.

8.15. <u>Properties</u>. (a) The Borrower and each of the Subsidiaries have good and marketable title to or leasehold interest in all properties that are necessary for the operation of their respective businesses as currently conducted and as proposed to be conducted, free and clear of all Liens (other than any Liens permitted by this Agreement) and except where the failure to have such good title could not reasonably be expected to have a Material Adverse Effect and (b) no mortgage encumbers improved Real Estate that is located in an area that has been identified by the Secretary of Housing and Urban Development as an area having special flood hazards within the meaning of the National Flood Insurance Act of 1968 unless flood insurance available under such Act has been obtained in accordance with Section 9.3.

8.16. Use of Proceeds. The Borrower will use proceeds of the DIP Facility (i) for the payment of working capital of the Credit Parties in the ordinary course of business, (ii) for C-Band relocation costs, (iii) investment and other general corporate purposes, (iv) for the payment of the costs and expenses of administering the Cases, and (v) to make the Adequate Protection Payments.

8.17. FCC Licenses, Etc. As of the Signing Date, <u>Schedule 8.17</u> hereto accurately and completely lists for each Satellite (a) all space station licenses for the launch and operation of Satellites with C-band or Ku-band transponders issued by the FCC to the Borrower or any Restricted Subsidiary and (b) all licenses and all other approvals, orders or authorizations issued or granted by any Governmental Authority outside of the United States of America to launch and operate any such Satellite. As of the Signing Date, the FCC Licenses and the other licenses, approvals or authorizations listed on <u>Schedule 8.17</u> hereto with respect to any Satellite include all material authorizations, licenses and permits issued by the FCC or any other Governmental Authority that are required or necessary to launch or operate such Satellite, as applicable. Except as could not reasonably be expected to have a Material Adverse Effect, each of the Subject Licenses is held in the name of a License Subsidiary and is validly issued and in full force and effect, and the Borrower and its Restricted Subsidiaries have fulfilled and performed in all respects all of their obligations with respect thereto and have full power and authority to operate thereunder.

8.18. <u>Satellites</u>. As of the Signing Date, <u>Schedule 8.18</u> hereto accurately and completely lists each of the Satellites owned by the Borrower and its Restricted Subsidiaries on the Signing Date, and sets forth for each such Satellite that is in orbit, the orbital slot and number and frequency band of the transponders on such Satellite.

8.19. <u>Centre of Main Interest</u>. Each Credit Party incorporated under the laws of the Grand Duchy of Luxembourg has its principal place of business (*principal établissement*), the seat of its central administration (*siège de l'administration centrale*) and its centre of main interests (*centre des intérêts principaux*) located at the place of its registered office (*siège statutaire*) in the Grand Duchy of Luxembourg and has no establishment (as defined by Regulation (EU) 2015/848 of the European Parliament and of the Council of 20 May 2015 on insolvency proceedings (recast)) outside the Grand Duchy of Luxembourg.

8.20. Orders. The Final DIP Order is effective to create in favor of the Collateral Agent, for the benefit of the Secured Parties, a legal, valid, binding and enforceable perfected security interest in the Collateral without the necessity of the execution of mortgages, security agreements, pledge agreements, financing statements or other agreements or documents.

8.21. <u>Status of Obligations; Perfection and Priority of Security Interests</u>. Subject to entry of the Final DIP Order and the terms of the Credit Documents, the Obligations shall have the status and priority set forth in Section 3 and, for the avoidance of doubt, are subject to the Carve Out in all respects.

## SECTION 9. Affirmative Covenants.

Each Credit Party hereby covenants and agrees that on the Closing Date and thereafter (with respect to Sections 9.1(j), 9.9, 9.14, 9.15(c), 9.21 and 9.22, on the Signing Date and thereafter), until the Commitments have terminated and the Loans, together with interest, Fees and all other Obligations incurred hereunder, are paid in full:

9.1. <u>Information Covenants</u>. The Borrower will furnish to the Administrative Agent for distribution to the Lenders (except with respect to clauses (c), (d) and (e) below, to the Administrative Agent, which may provide to any Lender upon request):

(a) <u>Annual Financial Statements</u>. As soon as available and in any event on or before the date on which such financial statements are required to be filed with the SEC or delivered to the holders of the Prepetition Senior Notes (or, if such financial statements are not required to be filed with the SEC or delivered to the holders of the Prepetition Senior Notes, on or before the date that is 120 days after the end of each such fiscal year), (A) the consolidated balance sheets of the Borrower and its Subsidiaries as at the end of such fiscal year, and the related consolidated statement of operations and cash flows for such fiscal year, setting forth comparative consolidated figures for the preceding fiscal year, and certified by independent certified public accountants of recognized national standing whose opinion shall not be qualified as to the scope of audit (except with respect to any Government Business Subsidiary), together in any event with a certificate of such accounting firm stating that in the course of its regular audit of the business of the Borrower and the Material Subsidiaries, which audit was conducted in accordance with generally accepted auditing standards, such accounting firm has obtained no knowledge of any Default or Event of Default relating to Section 11 that has occurred and is continuing or, if in the opinion of such accounting firm such a Default or Event of Default has occurred and is continuing, a statement as to the nature thereof, and (B) if the Borrower had any Unrestricted Subsidiaries during any period covered by the financial information set forth in clause (A), a reasonably detailed break-out of such financial information showing amounts attributable to the Restricted Subsidiaries as a whole and the Unrestricted Subsidiaries as a whole. The filing by Intelsat S.A. or any other direct or indirect parent entity of the Borrower of its Form 20-F or Form 10-K (or any successor or comparable forms) with the SEC as at the end of and for any fiscal year, reported on as aforesaid, shall be deemed to satisfy the obligations under the reporting portion of this paragraph with respect to such year so long as such filing includes (i) a "Management's Discussion and Analysis of Financial Condition and Results of Operations" (in the case of a Form 10-K) or an "Operating and Financial Review and Prospects" discussion (in the case of a Form 20-F) that includes a reasonably detailed analysis of the operating results and financial condition (considered separately from the other Subsidiaries of Intelsat S.A., where material) of the Borrower and its Subsidiaries; provided that such detailed analysis of the Borrower and its Subsidiaries shall not be required if Intelsat S.A.'s only material operations or assets in addition to the Borrower and its Subsidiaries includes one or more businesses, each of which discloses a "Management's Discussion and Analysis of Financial Condition and Results of Operations" or an "Operating and Financial Review and Prospects" discussion for such companies or substantially similar disclosure required by a non-U.S. jurisdiction (considered separately from other Subsidiaries of Intelsat S.A.) publicly on or through the website of Intelsat S.A. or through the EDGAR system and (ii) if the Borrower had any Unrestricted Subsidiaries during any period covered by the financial information set forth in clause (i), a reasonably detailed break-out of such financial information showing amounts attributable to the Restricted Subsidiaries as a whole and the Unrestricted Subsidiaries as a whole.

(b) <u>Quarterly Financial Statements</u>. As soon as available and in any event on or before the date on which such financial statements are required to be filed with the SEC or delivered to the holders of the Prepetition Senior Notes with respect to each of the first three quarterly accounting periods in each fiscal year of the Borrower (or, if such financial statements are not required to be filed with the SEC or delivered to the holders of the Prepetition Senior Notes, on or before the date that is 60 days after the end of each such quarterly accounting period), (A) the consolidated balance sheet of (i) the Borrower and the Restricted Subsidiaries and (ii) the Borrower and its Subsidiaries, in each case as at the end of such quarterly period and the related consolidated statement of operations for such quarterly accounting period and for the elapsed portion of the fiscal year ended with the last day of such quarterly period, and the related consolidated statement of cash flows for the elapsed portion of the fiscal year ended with the last day of such quarterly period, and setting forth comparative consolidated figures for the related periods in the prior fiscal year or, in the case of such consolidated balance sheet, for the last day of the prior fiscal year, all of which shall be certified by an Authorized Officer of the Borrower subject to changes resulting from audit and normal year-end audit adjustments, and (B) if the Borrower had any Unrestricted Subsidiaries during any period covered by the financial information set forth in clause (A), a reasonably detailed break-out of such financial information showing amounts attributable to the Restricted Subsidiaries as a whole and the Unrestricted Subsidiaries as a whole. The furnishing by Intelsat S.A. or any other direct or indirect parent entity of the Borrower of its Form 6-K (or any successor or comparable forms) relating to its quarterly financial statements or the filing of a Form 10-Q (or any successor or comparable forms) with the SEC as at the end of and for any fiscal quarter, certified as aforesaid, shall be deemed to satisfy the reporting obligations under this paragraph with respect to such quarter so long as such filing includes (i) a "Management's Discussion and Analysis of Financial Condition and Results of Operations" (in the case of a Form 10-Q) or, in the case of a Form 6-K, an "Operating and Financial Review and

Prospects" discussion complying with the requirements of Form 20-F (adjusted to reflect quarterly rather than annual reporting, consistent with the differences in the form requirements of Form 10-K and Form 10-Q) that includes a reasonably detailed analysis of the operating results and financial condition (considered separately from the other Subsidiaries of Intelsat S.A. where material) of Borrower and its Subsidiaries; provided that such detailed analysis of the Borrower and its Subsidiaries shall not be required if Intelsat S.A.'s only material operations or assets in addition to the Borrower and its Subsidiaries includes one or more businesses, each of which discloses a "Management's Discussion and Analysis of Financial Condition and Results of Operations" or an "Operating and Financial Review and Prospects" discussion for such companies or substantially similar disclosure required by a non-U.S. jurisdiction (considered separately from other Subsidiaries of Intelsat S.A.) publicly on or through the website of Intelsat S.A. or through the EDGAR system and (ii) if the Borrower had any Unrestricted Subsidiaries during any period covered by the financial information set forth in clause (i), a reasonably detailed break-out of such financial information showing amounts attributable to the Restricted Subsidiaries as a whole and the Unrestricted Subsidiaries as a whole.

(c) <u>DIP Budget</u>. On the Thursday following each four-week period (with the first such delivery on June 11, 2020), an updated cash flow forecast for the 13-week period ending after the end of the immediately preceding calendar week (such updated budget shall thereafter be the DIP Budget) for information purposes only.

(d) <u>Budget Variance Report</u>. On the Thursday following each four-week period (with the first such delivery on June 11, 2020 with respect to the four-week period ending on June 5, 2020), a report for information purposes, in each case, in form reasonably satisfactory to the Required Lenders (it being understood that any form previously agreed with the Required Backstop Parties is satisfactory), for the immediately preceding four-week period, that (i) sets forth the variances for the Credit Parties (on a line item basis, as a percentage and as a dollar amount) between the actual results and the corresponding projected amounts reflected in the DIP Budget then in effect for the corresponding period, and (ii) provides an explanation in reasonable detail of the reason for any such variance.

(e) <u>C-band Relocation Costs</u>. On the 21st day after the end of each fiscal month (with the first such delivery on June 21, 2020 with respect to the fiscal month ending on May 31, 2020), a report for information purposes, in form reasonably satisfactory to the Required Lenders (it being understood that any form previously agreed with the Required Backstop Parties is satisfactory), detailing the C-band relocation costs incurred and/or paid during the immediately preceding fiscal month.

(f) Officer's Certificates. At the time of the delivery of the financial statements provided for in Sections 9.1(a) and (b), a certificate of an Authorized Officer of the Borrower to the effect that no Default or Event of Default exists or, if any Default or Event of Default does exist, specifying the nature and extent thereof. At the time of the delivery of the financial statements provided for in Section 9.1(a), a certificate of an Authorized Officer and the chief financial or legal officer (separate from the foregoing Authorized Officer) of the Borrower or Intelsat S.A. setting forth the information required pursuant to Section 1(a) of the Perfection Certificate or confirming that there has been no change in such information since the Closing Date or the date of the most recent certificate delivered pursuant to this subsection (g), as the case may be.

(g) Notice of Default or Litigation. Promptly after an Authorized Officer of the Borrower or any of the Subsidiaries obtains actual knowledge thereof, notice of (i) the occurrence of any event that constitutes a Default or Event of Default, which notice shall specify the nature thereof, the period of existence thereof and what action the Borrower proposes to take with respect thereto, (ii) any litigation or governmental proceeding pending against the Borrower or any of the Subsidiaries (other than in connection with the Cases) that could reasonably be expected to result in a Material Adverse Effect, and (iii) any actual or constructive total or material partial loss event with respect to any Satellite.

(h) <u>Environmental Matters</u>. The Borrower will promptly advise the Lenders in writing after obtaining actual knowledge of any one or more of the following environmental matters, unless such environmental matters would not, individually or when aggregated with all other such matters, be reasonably expected to result in a Material Adverse Effect:

(i) Any pending or threatened Environmental Claim against the Borrower or any of the Subsidiaries or any Real Estate;

(ii) Any condition or occurrence on any Real Estate that (x) could reasonably be expected to result in noncompliance by the Borrower or any of the Subsidiaries with any applicable Environmental Law or (y) could reasonably be anticipated to form the basis of an Environmental Claim against the Borrower or any of the Subsidiaries or any Real Estate;

(iii) Any condition or occurrence on any Real Estate that could reasonably be anticipated to cause such Real Estate to be subject to any restrictions on the ownership, occupancy, use or transferability of such Real Estate under any Environmental Law; and

(iv) The conduct of any investigation, or any removal, remedial or other corrective action in response to the actual or alleged presence, release or threatened release of any Hazardous Material on, at, under or from any Real Estate.

All such notices shall describe in reasonable detail the nature of the claim, investigation, condition, occurrence or removal or remedial action and the response thereto. The term "<u>Real Estate</u>" shall mean land, buildings and improvements owned or leased by the Borrower or any of the Subsidiaries, but excluding all operating fixtures and equipment, whether or not incorporated into improvements.

(i) <u>Other Information</u>. Promptly upon filing thereof, copies of any filings (including on Form 10-K, 20-F, 10-Q, 6-K or 8-K) or registration statements with, and reports to, the SEC or any analogous Governmental Authority in any relevant jurisdiction by the Borrower (and Holdings, if Holdings is a public filing company outside the United States) or any of the Subsidiaries (other than amendments to any registration statement (to the extent such registration statement, in the form it becomes effective, is delivered to the Lenders), exhibits to any registration statement and, if applicable, any registration statements on Form S-8) and copies of all financial statements, proxy statements, notices and reports that the Borrower (and Holdings, if Holdings is a public filing company outside the United States) or any of the Subsidiaries shall send to the holders of any publicly issued debt of Holdings, the Borrower and/or any of the Subsidiaries (including any Senior Notes (whether publicly issued or not)) in their capacity as such holders (in each case to the extent not theretofore delivered to the Lenders pursuant to this Agreement) and, with reasonable promptness, such other information (financial or otherwise) as the Administrative Agent or the Required Lenders may reasonably request in writing from time to time; <u>provided</u> that Holdings and the Borrower shall not be required to furnish any such reports and other materials to the Administrative Agent or any Lender to the extent the same is publicly available on the website of Intelsat S.A. or the Borrower or through the EDGAR system.

(j) <u>Orders and Pleadings</u>. (i) as soon as reasonably practicable in advance of filing with the Bankruptcy Court or delivering to the Creditors' Committee or the U.S. Trustee, as the case may be, any Orders and all other proposed orders and pleadings related to the Loans and the Credit Documents, any Reorganization Plan and/or any disclosure statement related thereto and (ii) provide copies of all of the Debtors' proposed material substantive pleadings and orders in the Cases to the Administrative Agent and counsel to the Required Backstop Parties, in each case, with reasonably sufficient time for review and comment by such counsel no later than 3 calendar days prior to filing; provided that if the Borrower reasonably determines that, in light of the relevant circumstances at such time, providing a copy of a proposed pleading or order with sufficient time for review and comment would be impracticable, the Borrower shall instead be obligated to provide such copy as promptly as is reasonably practicable or (B) at the same time as such documents are provided by any of the Credit Parties to any statutory committee appointed in the Cases or the U.S. Trustee, all other notices, filings, motions, pleadings or other information concerning the financial condition of the Borrower or any of its Subsidiaries or other Indebtedness of the Credit Parties or any request for relief under Section 363, 365, 1113 or 1114 of the Bankruptcy Code or Section 9019 of the Federal Rules of Bankruptcy Procedure.

(k) <u>FCC Reports</u>. Promptly upon their becoming available, copies of any and all periodic or special reports filed by the Borrower or any of its Restricted Subsidiaries with the FCC or with any other Federal, state or local governmental authority, if such reports indicate any material adverse change in the business, operations, affairs or condition of the Borrower or any of its Restricted Subsidiaries, and copies of any and all notices and other communications from the FCC or from any other Federal, state or local governmental authority with respect to the Borrower, any of its Subsidiaries or any Satellite relating to any matter that could reasonably be expected to result in a Material Adverse Effect.

(1) <u>Satellite Health Report</u>. No less than annually with respect to each In-Orbit Satellite that has a net book value exceeding \$50,000,000, and upon the occurrence of an Event of Default at any time upon the reasonable request of the Administrative Agent, (i) with respect to any one or more In-Orbit Satellites operated by the Borrower or any of its Subsidiaries, a Satellite Health Report and (ii) with respect to any In-Orbit Satellite that is operated by any Person other than the Borrower or any of its Subsidiaries, any satellite health reports received by the Borrower from such Person, it being understood that to the extent that any such Satellite Health Report or other satellite health report contains any forward looking statements, estimates or projections, such statements, estimates or projections are subject to significant uncertainties and contingencies, many of which are beyond the Borrower's or any of its Subsidiaries' control, and no assurance can be given that such forward looking statements, estimates or projections will be realized, and provided that nothing in this clause (j) shall require the Borrower to deliver any

information to the Administrative Agent or any Lender to the extent delivery of such information is restricted by applicable law or regulation.

9.2. Books, Records and Inspections. The Borrower will, and the Borrower will cause each of the Subsidiaries to, permit officers and designated representatives of the Administrative Agent or the Required Lenders to visit and inspect any of the properties or assets of the Borrower and any such Subsidiary in whomsoever's possession to the extent that it is within such party's control to permit such inspection, and to examine the books and records of the Borrower and any such Subsidiary and discuss the affairs, finances and accounts of the Borrower and of any such Subsidiary with, and be advised as to the same by, its and their officers and independent accountants, all at such reasonable times and intervals and to such reasonable extent as the Administrative Agent or the Required Lenders may desire; provided that so long as no Default or Event of Default is then in existence, the Borrower and any Credit Party shall have the right to participate in any discussions of the Agents or the Lenders with any independent accountants of the Borrower. Notwithstanding anything to the contrary herein, neither the Borrower nor any Restricted Subsidiary will be required under this Section 9.2 or otherwise to disclose or permit the inspection or discussion of any document, information or other matter to the extent that such action would violate any attorney-client privilege (as reasonably determined by counsel to the Credit Parties), law, rule or regulation, or any contractual obligation of confidentiality binding on the Credit Parties or their respective Affiliates (provided that (x) the Administrative Agent shall be notified if any such information is being withheld and (y) the Credit Parties shall use commercially reasonable efforts to obtain a consent to such disclosure or action being withheld as a result of any contractual obligation of confidentiality (other than, for purposes of this clause (y), any such information being withheld as a result of attorney-client privilege).

9.3. Maintenance of Insurance.

(a) The Borrower will, and the Borrower will cause each of its Restricted Subsidiaries to, obtain, maintain and keep in full force and effect at all times (i) with respect to each Satellite procured by the Borrower or any of its Restricted Subsidiaries for which the risk of loss passes to the Borrower or such Restricted Subsidiary at or before launch, and for which launch insurance or commitments with respect thereto are not in place as of the Signing Date, launch insurance with respect to each such Satellite covering the launch of such Satellite and a period of time thereafter, but only to the extent, if at all, and on such terms (including coverage period, exclusions, limitations on coverage, co-insurance, deductibles and coverage amount) as is determined by the Borrower to be in the best interests of the Borrower, (ii) with respect to each Satellite it currently owns or for which it has risk of loss (or, if the entire Satellite is not owned, the portion it owns or for which it has risk of loss), other than any Excluded Satellite, In-Orbit Insurance and (iii) at all times subsequent to the coverage period of the launch insurance described in clause (i) above, if any, or if launch insurance is not procured, at all times subsequent to the initial completion of in-orbit testing, in each case with respect to each Satellite it then owns or for which it has risk of loss (or portion, as applicable), other than any Excluded Satellite, In-Orbit Insurance; provided, however, that at any time with respect to a Satellite that is not an Excluded Satellite, none of the Borrower or any of its Subsidiaries shall be required to maintain In-Orbit Insurance in excess of 33% of the aggregate net book value of all in-orbit Satellites (and portions it owns or for which it has risk of loss) insured (it being understood that any Satellite (or portion, as applicable) protected by In-Orbit Contingency Protection shall be deemed to be insured for a percentage of its net book value as set forth in the definition of "In-Orbit Contingency Protection"). In the event that the expiration and non-renewal of In-Orbit Insurance for such a Satellite (or portion, as applicable) resulting from a claim of loss under such policy causes a failure to comply with the proviso in the immediately preceding sentence, the Borrower and its Restricted Subsidiaries shall be deemed to be in compliance with such proviso for the 120 days immediately following such expiration or non-renewal; provided that the Borrower or any of its Restricted Subsidiaries, as the case may be, procures such In-Orbit Insurance or provides such In-Orbit Contingency Protection as necessary to comply with such proviso within such 120-day period. In the event of the unavailability of any In-Orbit Contingency Protection for any reason, the Borrower or any of its Restricted Subsidiaries, as the case may be, shall, subject to the first proviso above, within 120 days of such unavailability, be required to have in effect In-Orbit Insurance complying with clause (ii) or (iii) above, as applicable, with respect to all Satellites (or portions, as applicable), other than Excluded Satellites that the unavailable In-Orbit Contingency Protection was intended to protect and for so long as such In-Orbit Contingency Protection is unavailable; provided that the Borrower and its Restricted Subsidiaries shall be considered in compliance with this insurance covenant for the 120 days immediately following such unavailability.

(b) The Borrower will, and the Borrower will cause each of its Restricted Subsidiaries to, use its commercially reasonable efforts to at all times keep the respective property of the Borrower and its Restricted Subsidiaries (except (x) real or personal property leased or financed through third parties in accordance with this Agreement and (y) satellites) insured in favor of the Collateral Agent for the benefit of the Secured Parties, and all

policies or certificates with respect to such insurance (and any general liability, umbrella liability coverage and workers' compensation insurance (to the extent permitted by law) maintained by, or on behalf of, the Borrower or any Restricted Subsidiary of the Borrower) (i) shall be endorsed to the Collateral Agent's reasonable satisfaction for the benefit of the Collateral Agent (including, without limitation, by naming the Collateral Agent as certificate holder, mortgagee and loss payee with respect to Real Estate, certificate holder and loss payee with respect to personal property, additional insured with respect to general liability and umbrella liability coverage and (to the extent permitted by law) certificate holder with respect to workers' compensation insurance), (ii) shall state that such insurance policies shall not be cancelled or materially changed without at least 30 days' prior written notice thereof by the respective insurer to the Collateral Agent; provided that with respect to any launch insurance or In-Orbit Insurance, such notice is available, and (iii) shall, upon the request of the Collateral Agent, be deposited with the Collateral Agent for the benefit of the Secured Parties. Notwithstanding the foregoing or anything to the contrary contained elsewhere in this Agreement, the lender or collateral agent with respect to any ECA Financing may be named as additional insured, mortgagee or loss payee with respect to any insurance covering ECA Collateral which insurance shall not be required to be in accordance with or subject to the terms of this Section 9.3.

(c) If any portion of any Mortgaged Property is at any time located in an area identified by the Federal Emergency Management Agency (or any successor agency) as a Special Flood Hazard Area with respect to which flood insurance has been made available under the National Flood Insurance Act of 1968 (as now or hereafter in effect or successor act thereto), then the Borrower shall, or shall cause each Credit Party to (i) maintain, or cause to be maintained, with a financially sound and reputable insurer, flood insurance in an amount and otherwise sufficient to comply with all applicable rules and regulations promulgated pursuant to the Flood Insurance Laws and (ii) deliver to the Administrative Agent evidence of such compliance in form and substance reasonably acceptable to the Administrative Agent.

(d) If the Borrower or any of its Restricted Subsidiaries shall fail to maintain all insurance in accordance with this Section 9.3, or if the Borrower or any of its Restricted Subsidiaries shall fail to so name the Collateral Agent for the benefit of the Secured Parties as an additional insured, mortgagee or loss payee, as the case may be, or so deposit all certificates with respect thereto, the Collateral Agent shall have the right (but shall be under no obligation), upon reasonable prior notice to the Borrower of its intention to do so, to procure such insurance on such terms and against such risks as are required hereby, and the Borrower agrees to reimburse the Administrative Agent for any premium paid therefor.

9.4. <u>Payment of Taxes</u>. The Borrower will pay and discharge, and the Borrower will cause each of the Subsidiaries to pay and discharge, all material Taxes imposed upon it or upon its income or profits, or upon any properties belonging to it (in the case of any such Person that is a DIP Debtor under the Cases, solely to the extent arising after the Petition Date), prior to the date on which material penalties attach thereto, and all lawful material claims that, if unpaid, could reasonably be expected to become a material Lien upon any properties of the Borrower or any of the Restricted Subsidiaries; provided that none of the Borrower or any of the Subsidiaries shall be required to pay any such Tax or claim that is (a) being contested in good faith and by proper proceedings if it has maintained adequate reserves (in the good faith judgment of the management of the Borrower) with respect thereto in accordance with GAAP and the failure to pay could not reasonably be expected to result in a Material Adverse Effect or (b) with respect to any Person that is a DIP Debtor under the Cases, the nonpayment of such Tax is permitted or required by the Bankruptcy Code.

9.5. <u>Consolidated Corporate Franchises</u>. The Borrower will do, and the Borrower will cause each Restricted Subsidiary to do, or cause to be done, all things necessary to preserve and keep in full force and effect its existence, corporate rights and authority, except to the extent that the failure to do so could not reasonably be expected to have a Material Adverse Effect; <u>provided</u>, <u>however</u>, that the Borrower and its Subsidiaries may consummate any transaction permitted under Section 10.3, 10.4 or 10.5.

9.6. <u>Compliance with Statutes, Regulations, etc.</u> Except as otherwise excused by the Bankruptcy Code with respect to any DIP Debtor, the Borrower will, and the Borrower will cause each Subsidiary to, comply with all applicable laws, rules, regulations and orders applicable to it or its property (including all FCC Licenses and all other governmental approvals or authorizations required to launch and operate the Satellites and the TT&C Stations related thereto) and to transmit signals to and receive transmissions from the Satellites, and to maintain all such FCC Licenses and other governmental approvals or authorizations in full force and effect, in each case except where the failure to do so could not reasonably be expected to have a Material Adverse Effect (it being understood that any failure as it may relate to any FCC License for a Satellite that is yet to be launched shall not, in itself, be considered or deemed to result in a Material Adverse Effect).

9.7. ERISA. Promptly after the Borrower or any Subsidiary or any ERISA Affiliate knows or has reason to know of the occurrence of any of the following events that, individually or in the aggregate (including in the aggregate such events previously disclosed or exempt from disclosure hereunder, to the extent the liability therefor remains outstanding), would be reasonably likely to have a Material Adverse Effect, the Borrower will deliver to each of the Lenders a certificate of an Authorized Officer or any other senior officer of the Borrower setting forth details as to such occurrence and the action, if any, that the Borrower, such Subsidiary or such ERISA Affiliate is required or proposes to take, together with any notices (required, proposed or otherwise) given to or filed with or by the Borrower, such Subsidiary, such ERISA Affiliate, the PBGC, a Plan participant (other than notices relating to an individual participant's benefits) or the Plan administrator with respect thereto: that a Reportable Event has occurred; that a failure to satisfy the minimum funding standard under Section 412 of the Code has occurred or an application is to be made to the Secretary of the Treasury for a waiver or modification of the minimum funding standard (including any required installment payments) or an extension of any amortization period under Section 412 of the Code with respect to a Plan; that a Plan having an Unfunded Current Liability has been or is to be terminated, reorganized, partitioned or declared insolvent under Title IV of ERISA (including the giving of written notice thereof); that a Plan has an Unfunded Current Liability that has or will result in a Lien under ERISA or the Code; that proceedings will be or have been instituted to terminate a Plan having an Unfunded Current Liability (including the giving of written notice thereof); that a proceeding has been instituted against the Borrower, a Subsidiary or an ERISA Affiliate pursuant to Section 515 of ERISA to collect a delinquent contribution to a Plan; that the PBGC has notified the Borrower, any Subsidiary or any ERISA Affiliate of its intention to appoint a trustee to administer any Plan; that the Borrower, any Subsidiary or any ERISA Affiliate has failed to make a required installment or other payment pursuant to Section 412 of the Code with respect to a Plan; or that the Borrower, any Subsidiary or any ERISA Affiliate has incurred or will incur (or has been notified in writing that it will incur) any liability (including any contingent or secondary liability) to or on account of a Plan pursuant to Section 409. 502(i), 502(l), 515, 4062, 4063, 4064, 4069, 4201 or 4204 of ERISA or Section 4971 or 4975 of the Code.

9.8. <u>Maintenance of Properties</u>. The Borrower will, and the Borrower will cause each of its Restricted Subsidiaries to, keep and maintain all property material to the conduct of its business in good working order and condition, ordinary wear and tear excepted, which shall include, in the case of Satellites (other than Satellites yet to be launched), the provision of tracking, telemetry, control and monitoring of Satellites in their designated orbital positions in accordance with prudent and diligent standards in the commercial satellite industry, except to the extent that the failure to do so could reasonably be expected to have a Material Adverse Effect.

9.9. Transactions with Affiliates. Subject to the Order and any other order of the Bankruptcy Court, the Borrower will conduct, and cause each of the Restricted Subsidiaries to conduct, all transactions with any of its Affiliates (other than the Borrower or its Restricted Subsidiaries) involving aggregate consideration in excess of \$1,500,000 on terms that are substantially as favorable to the Borrower or such Restricted Subsidiary as it would obtain in a comparable arm's-length transaction with a Person that is not an Affiliate; provided that the foregoing restrictions shall not apply to (a) [reserved] (b) the transactions permitted pursuant to Section 10.9, (c) customary fees paid to and customary indemnities provided to members of the board of directors of the Borrower and the Subsidiaries, (d) transactions permitted by Section 10.6, (e) employment and other compensation arrangements with respect to the procurement of services of officers, consultants and employees of the Borrower and the Subsidiaries in the ordinary course of business, (f) any tax sharing agreement or arrangement relating to payments, whether directly or by dividend, by the Borrower or a Restricted Subsidiary to any parent of the Borrower if such parent is required to file a consolidated, unitary or similar tax return reflecting income of the Borrower or its Restricted Subsidiaries, in an amount equal to the portion of such taxes attributable to the Borrower and/or its Restricted Subsidiaries that are not payable directly by the Borrower and/or its Restricted Subsidiaries, but not to exceed the amount that the Borrower or such Restricted Subsidiaries would have been required to pay in respect of such taxes if the Borrower and such Restricted Subsidiaries had been required to pay such taxes directly as standalone taxpayers (or a standalone group separate from such parent), (g) agreements in effect on the Signing Date and listed on Schedule 9.9 and amendments thereto not materially disadvantageous to the Lenders, and (h) transactions between the Borrower or any of its Restricted Subsidiaries and any Person a director or directors of which is (are) also a director of any Parent Companies; provided that such director(s) abstain(s) from voting as a director of such Parent Company, as the case may be, on any matter involving such Person.

9.10. End of Fiscal Years; Fiscal Quarters. The Borrower will, for financial reporting purposes, cause (a) each of their, and each of their respective Subsidiaries', fiscal years to end on December 31 of each year and (b) each of their, and each of their respective Subsidiaries', fiscal quarters to end on dates consistent with such fiscal year-end and the Borrower's past practice; provided, however, that the Borrower may, upon written notice to the Administrative Agent, change the financial reporting convention specified above to any other financial reporting

convention reasonably acceptable to the Administrative Agent (it being agreed that a conversion from GAAP to IFRS shall be reasonably acceptable to the Administrative Agent), in which case the Borrower and the Administrative Agent will, and are hereby authorized by the Lenders to, make any adjustments to this Agreement that are necessary in order to reflect such change in financial reporting.

9.11. Additional Guarantors and Grantors. Except as set forth in Section 10.1(A)(i) or (A)(j), the Borrower will cause (i) each direct or indirect Subsidiary (other than any Unrestricted Subsidiary and any Designated Target Subsidiary) formed or otherwise purchased or acquired after the Signing Date (unless, with respect to any such Subsidiary acquired in or formed for the sole purpose of consummating a Permitted Acquisition that is subject to or formed for the sole purpose of entering one or more agreements governing Permitted Acquisition Debt, which agreements prohibit the granting of a guarantee by such Subsidiary as contemplated by this clause (i), in which case the guarantee otherwise required by this Section 9.11(i) with respect to such Subsidiary shall not be required until such prohibitions cease to be applicable), (ii) each Subsidiary (other than any Unrestricted Subsidiary and any Designated Target Subsidiary) that is not a Material Subsidiary on the Signing Date but subsequently becomes a Material Subsidiary, (iii) each inactive Subsidiary (unless such Subsidiary is designated an Unrestricted Subsidiary in accordance with terms of this Agreement_or is a Designated Target Subsidiary) which acquires any material assets or is otherwise no longer deemed inactive, and (iv) each Subsidiary that becomes a Debtor under the Cases, in each case to execute a Joinder Agreement and a supplement to the Security Agreement; provided, however, that no Foreign Subsidiary shall be required to take such actions if, and to the extent that, based upon written advice of local counsel reasonably satisfactory to the Required Backstop Parties, the Borrower and/or such Foreign Subsidiary concludes that the taking of such actions would either (i) violate the laws of the jurisdiction in which such Foreign Subsidiary is organized or (ii) the cost, burden, difficulty or consequence of taking such actions (taking into account any adverse tax consequences to the Borrower and its Affiliates (including the imposition of withholding or other material taxes)), outweighs the benefits to be obtained by the Lenders therefrom; provided further, that if steps (for example, limiting the amount guaranteed) can be taken so that such violation, cost, burden, difficulty or consequence would not exist, then, if requested by the Required Backstop Parties, the respective Foreign Subsidiary shall enter into a modified Guarantee and/or modified Security Documents that provide, to the maximum extent permissible under applicable law, as many of the benefits as possible as are provided pursuant to the Guarantee and the Security Documents executed and delivered on the Closing Date and taking into account such cost, burden, difficulty or consequence.

### 9.12. Pledges of Additional Stock and Evidence of Indebtedness.

(a) The Borrower will pledge, and, if applicable, will cause each Subsidiary Guarantor to pledge, to the Collateral Agent for the benefit of the Secured Parties, (i) all the capital stock of each Subsidiary (other than any Unrestricted Subsidiary) and Minority Investments held by the Borrower or a Subsidiary Guarantor, in each case, formed or otherwise purchased or acquired after the Signing Date, in each case pursuant to a supplement to the Security Agreement (or, if necessary, a new pledge agreement or security agreement) in form and substance reasonably satisfactory to the Required Lenders, (ii) all evidences of Indebtedness in excess of \$1,000,000 received by the Borrower or any Subsidiary Guarantor in connection with any disposition of assets pursuant to Section 10.4(b), in each case pursuant to a supplement to the Security Agreement (or, if necessary, a new pledge agreement or security agreement) in form and substance reasonably satisfactory to the Required Lenders, and (iii) any global promissory notes executed after the Signing Date evidencing Indebtedness of the Borrower, each Subsidiary and each Minority Investment that is owing to the Borrower or any Subsidiary Guarantor, in each case pursuant to the Security Agreement (or, if necessary, a new pledge agreement to the Security Agreement (or, if necessary, a new pledge agreement or security agreement) in form and substance reasonably satisfactory to the Required Lenders, and (iii) any global promissory notes executed after the Signing Date evidencing Indebtedness of the Borrower, each Subsidiary and each Minority Investment that is owing to the Borrower or any Subsidiary Guarantor, in each case pursuant to a supplement (or, if necessary, a new pledge agreement) in form and substance reasonably satisfactory to the Required Lenders.

(b) The Borrower agrees that all Indebtedness in excess of \$10,000,000 of the Borrower and each Subsidiary that is owing to any Credit Party shall be evidenced by one or more global promissory notes.

(c) At all times on and after the Signing Date, the Borrower will cause each obligor and obligee of any loan or advance (including, without limitation, pursuant to guarantees thereof or security thereof) which are (x) made to the Borrower by any Parent Company or any of the Borrower's Subsidiaries or (y) made to any Credit Party by a Subsidiary of the Borrower that is not a Credit Party prior to the extension or incurrence of such loan or advance, to execute and deliver to the Administrative Agent an intercompany subordination agreement substantially in the form of Exhibit I (as reasonably determined by the Required Lenders) (as modified, amended or supplemented from time to time, the "Intercompany Subordination Agreement") or, to the extent the Intercompany Subordination Agreement has previously come into effect, a joinder agreement in respect of the Intercompany Subordination Agreement and, in connection therewith, promptly execute and deliver all further instruments, and take all further action, that the Administrative Agent may reasonably require; provided that no such Intercompany Subordination

Agreement is required to be entered into in respect of intercompany loans and advances in the aggregate not exceeding \$10,000,000, and with such further exceptions as the Required Lenders may agree.

(d) Notwithstanding the foregoing, no Foreign Subsidiary shall be required to take any such actions pursuant to this Section 9.12 if, and to the extent that, based upon written advice of local counsel reasonably satisfactory to the Required Backstop Parties, the Borrower and/or such Foreign Subsidiary concludes that the taking of such actions would either (i) violate the laws of the jurisdiction in which such Foreign Subsidiary is organized or (ii) the cost, burden, difficulty or consequence of taking such actions (taking into account any adverse tax consequences to the Borrower and its Affiliates (including the imposition of withholding or other material taxes)), outweighs the benefits to be obtained by the Lenders therefrom; provided further, that if steps (for example, limiting the amount guaranteed) can be taken so that such violation, cost, burden, difficulty or consequence would not exist, then, if requested by the Required Backstop Parties, the respective Foreign Subsidiary shall enter into a modified Guarantee and/or modified Security Documents that provide, to the maximum extent permissible under applicable law, as many of the benefits as possible as are provided pursuant to the Guarantee and the Security Documents executed and delivered on the Closing Date and taking into account such cost, burden, difficulty or consequence.

9.13. Use of Proceeds. The Borrower will use the all Loans for the purposes set forth in Section 8.16.

9.14. Changes in Business.

(a) The Borrower and the Subsidiaries, taken as a whole, will not fundamentally and substantively alter the character of their business, taken as a whole, from the business conducted or proposed to be conducted by the Borrower and the Subsidiaries, taken as a whole, on the Signing Date and other business activities that are complementary, ancillary incidental or related to, reasonably similar to or a reasonable extension, development or expansion of any of the foregoing (a "<u>Permitted Business</u>").

(b) No License Subsidiary will engage in any line or lines of business activity other than to hold FCC Licenses issued to it and to enter into arrangements with the Borrower or other Restricted Subsidiaries (other than other License Subsidiaries) to manage and operate such FCC Licenses under its direction and control, in each case to the maximum extent permitted by applicable law. The Borrower will cause all Subject Licenses at all times to be held in the name of a License Subsidiary (which shall be the sole legal and beneficial owner thereof). Any license issued after the Signing Date by the FCC that constitutes a Subject License shall be held in the name of a License Subsidiary (which shall be the sole legal and beneficial owner thereof).

# 9.15. Further Assurances.

(a) The Borrower will, and will cause each other Credit Party to, execute any and all further documents, financing statements, agreements and instruments, and take all such further actions (including the filing and recording of financing statements, fixture filings, mortgages, deeds of trust and other documents and promptly provide evidence thereof to the Collateral Agent), which may be required under the laws of the United States or any applicable foreign country, which the Administrative Agent or the Required Lenders may reasonably request, in order to grant, preserve, protect and perfect the validity and priority of the security interests created or intended to be created by the Security Documents, all at the expense of the Borrower and the Restricted Subsidiaries; provided, however, that notwithstanding anything to the contrary contained in this Agreement or in any other Credit Document, but without limiting the grant of a Lien on and security interest in the Collateral pursuant to the Orders and the Security Documents, the Credit Parties will not be obligated to enter into any mortgages, authorize any fixture filing, enter into any agreement requiring "control" (as defined in Section 9-104, 9-105, 9-106 and 9-107 of the UCC as in effect in any relevant jurisdiction) or to undertake any registration in respect of assets subject to a certificate of title.

(b) If any assets (including any real estate or improvements thereto or any interest therein) with a book value or fair market value in excess of \$10,000,000 are acquired by the Borrower or any other Credit Party after the Signing Date (other than assets constituting Collateral under the Security Documents that become subject to the Lien of the Security Agreement upon acquisition thereof) that are of the nature secured by the Security Documents, the Borrower will notify the Administrative Agent and the Lenders thereof, and, if requested by the Administrative Agent or the Required Lenders, the Borrower will cause such assets to be subjected to a Lien securing the applicable Obligations and will take, and cause the other Credit Parties to take such actions as shall be necessary or reasonably requested by the Administrative Agent or Required Lenders to grant and perfect such Liens (unless, with respect to any such asset, same is subject to one or more agreements or Liens permitted hereunder which agreements or Lien(s) prohibit the granting of a security interest thereon as contemplated by this clause (b), in which case the actions otherwise required by this Section 9.15(b) with respect to such asset shall not be required to be taken until such prohibitions cease to be applicable) consistent with the applicable requirements of the Security Documents, including actions described in paragraph (a) of this Section 9.15, all at the expense of the Credit Parties. Any provision contained herein or in the Security Documents to the contrary notwithstanding, the Collateral shall

not include at any time (i) any FCC License to the extent (but only to the extent) that at such time the Collateral Agent may not validly possess a security interest therein pursuant to the Communications Act of 1934, as amended, and the regulations promulgated thereunder, as in effect at such time, but the Collateral shall include, to the maximum extent permitted by law, all rights incident or appurtenant to the FCC Licenses and the right to receive all proceeds derived from or in connection with the sale, assignment or transfer of the FCC Licenses or (ii) Satellites set forth on Schedule 1.1(b), insurance policies thereon and related construction and manufacturing contracts, launch insurance policies (to the extent of coverage for the launch and associated services and any premiums related thereto), launch services agreements (but excluding customer contracts) or proceeds thereof that secure obligations under any ECA Financing (this clause (ii), "ECA Collateral").

(c) Each Credit Party organized under the laws of Luxembourg will not own Satellites, orbital slots or hold customer contracts to the extent the Collateral Agent cannot hold a valid and perfected security interest in such property.

(d) Notwithstanding anything herein to the contrary, if the Required Lenders reasonably determine (taking into account the existence and effect of the Orders) in their reasonable judgment (confirmed in writing to the Borrower and the Administrative Agent) that the cost or other consequences (including adverse tax and accounting consequences) of creating or perfecting any Lien on any property is excessive in relation to the benefits afforded to the Secured Party thereby, then taking such actions to create or perfect such Lien on such property shall be not be required for all purposes of the Credit Documents.

#### 9.16. Access and Command Codes.

(a) The Borrower will, and will cause each of the Restricted Subsidiaries, at the request of the Administrative Agent to use commercially reasonable efforts to obtain promptly from each provider (other than the Borrower) of tracking, telemetry, control and monitoring services for any Satellite, consents and agreements with the Administrative Agent to:

(i) deliver expeditiously to the Administrative Agent, upon notification by the Administrative Agent that an acceleration pursuant to Section 12 has occurred, subject to having obtained any consent or approval of, or registration or filing with, any Governmental Authority for such delivery, all access codes, command codes and command encryption necessary to establish access to and perform tracking, telemetry, control and monitoring of any such Satellite, including activation and control of any spacecraft subsystems and payload components and the transponders thereon;

(ii) take commercially reasonable steps necessary, upon notification by the Administrative Agent that an acceleration pursuant to Section 12 has occurred, to obtain any consent or approval of, or registration or filing with, any Governmental Authority required to effect any transfer of operational control over any such Satellite and related technical data (including any license approving the export or re-export of such Satellite to any Person or Persons as designated by the Administrative Agent); and

(iii) deliver to the Administrative Agent written evidence of the issuance of any such consent, approval, registration or filing once such consent, approval, registration or filing has been obtained;

(b) If, after having used its commercially reasonable efforts to obtain the consents and agreements referred to in clause (i) above, any such consents or agreements shall not have been so obtained, instruct each such provider of tracking, telemetry, control and monitoring services (and each Satellite Manufacturer in respect of Satellites that have yet to be launched, to the extent that the Borrower or a Restricted Subsidiary does not have in its possession all items referred to in clause (ii) above) to cooperate in providing the access codes, command codes and command encryption referred to in said clause (i), in each case subject to having obtained any consent or approval of, or registration or filing with, any Governmental Authority for such delivery; and

(c) At any time upon an acceleration pursuant to Section 12, and upon notification thereof by the Administrative Agent, to promptly deliver to the Administrative Agent, subject to having obtained any requisite consent or approval of, or registration or filing with, any Governmental Authority for such delivery, all access codes, command codes and command encryption necessary, in the sole judgment of the Required Lenders, to establish access to and perform tracking, telemetry, control and monitoring of any Satellite, including activation and control of any spacecraft subsystems and payload components and the transponders thereon and any changes to or modifications of such codes and encryption.

9.17. TTC&M Providers. The Borrower will, and will use its commercially reasonable efforts to cause each provider (other than the Borrower) of tracking, telemetry, control and monitoring services for any Satellite to agree to, not change any access codes, command codes or command encryption necessary to establish access to and perform tracking, telemetry, control and monitoring of each Satellite at any time that an Event of Default exists and such provider of tracking, telemetry, control and monitoring services, as the case may be, has been notified by the Borrower or the Administrative Agent thereof, without promptly furnishing to the Administrative Agent the new

access codes, command codes and command encryption necessary to establish access to and perform tracking, telemetry, control and monitoring of such Satellite, once such access codes, command codes and command encryption have been delivered to the Administrative Agent pursuant to this Section 9.17.

9.18. [Reserved].

9.19. Government Business Subsidiaries. The Borrower will use its commercially reasonable efforts (as may be permitted under that certain proxy agreement (the "Proxy Agreement") among Intelsat General Communications LLC and the other parties thereto, and as may be permitted under any substantially similar agreement), and will use its commercially reasonable efforts (as may be permitted under the Proxy Agreement, and as may be permitted under any substantially similar agreement) to cause each of its Subsidiaries (other than Intelsat General Communications LLC and any other Government Business Subsidiary), not to allow or permit, directly or indirectly, Intelsat General Communications LLC or such other Government Business Subsidiary to take, or fail to take, any action that would violate the covenants and terms of this Agreement or cause any representation or warranty contained in this Agreement to be untrue.

9.20. Post-closing Covenants. As promptly as practicable, and in any event within the time periods after the Closing Date specified in Schedule 9.20 or such later date as reasonably agreed by the Required Lenders, including to reasonably accommodate circumstances unforeseen on the Closing Date, the Credit Parties shall deliver the documents or take the actions specified on Schedule 9.20 that would have been required to be delivered or taken on the Closing Date, in each case except to the extent otherwise agreed by the Administrative Agent pursuant to its authority as set forth herein with respect to collateral and guarantee matters.

9.21. FCC Covenants.

(a) The Credit Parties shall use commercially reasonable efforts to timely prepare, submit, and prosecute all requests for reimbursement of C-band relocation costs to the Relocation Payment Clearinghouse pursuant to the FCC C-Band Rules; and

(b) The Credit Parties shall use commercially reasonable efforts to comply with the deadlines related to accelerated clearing of the C-band spectrum as prescribed by the FCC as set forth in the FCC C-Band Rules.

9.22. Bankruptcy Matters. Each DIP Debtor shall:

(a) cause all proposed (i) First Day Orders, (ii) "second day" orders, (iii) orders related to or affecting the Loans and other Obligations, the Prepetition Debt and the Credit Documents, any other financing or use of cash collateral, any sale or other disposition of Collateral outside the ordinary course, cash management, adequate protection, any plan of reorganization and/or any disclosure statement related thereto, (iv) orders concerning the financial condition of the Borrower or any of its Subsidiaries or other Indebtedness of the DIP Debtors or seeking relief under section 363, 364 or 365 of the Bankruptcy Code or rule 9019 of the Federal Rules of Bankruptcy Procedure, (v) orders authorizing additional payments to critical vendors (outside of the relief approved in the First Day Orders and "second day" orders) and (vi) orders establishing procedures for administration of the Cases or approving significant transactions submitted to the Bankruptcy Court, in each case, proposed by the Debtors to be in accordance with and permitted by the terms of this Agreement;

(b) comply in all material respects with each order entered by the Bankruptcy Court in connection with the Cases; and comply in a timely manner with their obligations and responsibilities as debtors in possession under the Bankruptcy Code, the Bankruptcy Rules, the Orders, as applicable, and any other order of the Bankruptcy Court.

SECTION 10. Negative Covenants.

Each Credit Party hereby covenants and agrees that on the Signing Date and thereafter, until the Commitments have terminated and the Loans, together with interest, Fees and all other Obligations incurred hereunder, are paid in full:

10.1. Limitation on Indebtedness.

(A) The Credit Parties will not, and will not permit any of the their respective Restricted Subsidiaries to, create, incur, assume or suffer to exist any Indebtedness, except:

(a) Indebtedness arising under the Credit Documents;

(b) Indebtedness of (i) the Borrower to any Subsidiary of the Borrower and (ii) subject to compliance with Section 10.5(d) and/or (g), any Subsidiary to the Borrower or any Restricted Subsidiary of the Borrower;

(c) Indebtedness in respect of any banker's acceptances, bank guarantees, letter of credit, warehouse receipt or similar facilities entered into in the ordinary course of business:

(d) except as provided in clauses (i) and (j) below, subject to compliance with Section 10.5(g), Guarantee Obligations incurred by (i) Restricted Subsidiaries in respect of Indebtedness of the Borrower or other Restricted Subsidiaries that is permitted to be incurred under this Agreement and (ii) the Borrower in respect of Indebtedness of the Restricted Subsidiaries that is permitted to be incurred under this Agreement; provided that there

shall be no Guarantee by any Restricted Subsidiary that is not a Subsidiary Guarantor of any Indebtedness of the Borrower;

(e) Guarantee Obligations incurred in the ordinary course of business in respect of obligations of suppliers, customers, franchisees, lessors and licensees;

(f) (i) Indebtedness (including Indebtedness arising under Capital Leases) incurred within 270 days of (A) the acquisition (by purchase, lease or otherwise), construction or improvement of fixed or capital assets (including Real Estate), and (B) any ECA Financings incurred prior to the Signing Date, (ii) Indebtedness arising under Capital Leases entered into in connection with Permitted Sale Leasebacks and (iii) Indebtedness arising under Capital Leases, other than Capital Leases in effect on the Signing Date and Capital Leases entered into pursuant to subclauses (i) and (ii) above; provided that the aggregate amount of Indebtedness incurred pursuant to this subclause (iii) shall not exceed \$50,000,000 at any time outstanding;

(g) (i) the Prepetition Secured Debt; (ii) the Prepetition Senior Notes; and (iii) other Indebtedness outstanding on the Signing Date and listed on <u>Schedule 10.1</u> hereto;

(h) Indebtedness in respect of Hedge Agreements;

(i) [reserved];

(i) Indebtedness of any Designated Target Subsidiary or any Restricted Subsidiary that is not a Guarantor to the Borrower or any Restricted Subsidiary of the Borrower to the extent such Indebtedness is incurred to facilitate the Designated Target Transaction or a Permitted Acquisition, to make an Investment in a Designated Target Subsidiary or an Acquired Person or its Subsidiaries as an Investment permitted by Section 10.5, in each case, subject to compliance with Section 10.5;

(j) Indebtedness to finance Permitted Acquisitions (other than the Designated Target Transaction) incurred by any Person acquired or formed for the purpose of acquiring such assets as a result of such Permitted Acquisition (each, an "Acquired Person" and such Indebtedness, "Permitted Acquisition Debt"), on or prior to the consummation of such Permitted Acquisition; provided that (a) such Indebtedness is not incurred or guaranteed in any respect by Borrower or any Restricted Subsidiary (other than the Acquired Persons and their Subsidiaries) and otherwise has no recourse against the Borrower or any Restricted subsidiary (other than the Acquired Persons and their Subsidiaries) or any of their assets and (b) at the time incurred, the aggregate amount of such Indebtedness incurred pursuant to this clause (j) with respect to an Acquired Person shall not exceed a Loan to Value Ratio of 40%; provided, however, that, upon consummation of the Designated Target Transaction, neither the Credit Parties nor any Restricted Subsidiaries shall be permitted to incur any Indebtedness under this Section 10.1(j);

(k) Indebtedness in respect of performance bonds, bid bonds, appeal bonds, customs bonds, surety bonds and completion guarantees and similar obligations not in connection with money borrowed, in each case provided in the ordinary course of business, including those incurred to secure health, safety and environmental obligations in the ordinary course of business;

(1) Indebtedness incurred in connection with any Permitted Sale Leaseback (provided that the Net Cash Proceeds thereof are promptly applied to the extent required by Section 5.2);

(m) [reserved]; Indebtedness (i) constituting indemnification obligations or obligations in respect of purchase price or other similar adjustments (including deferred payments or similar obligations) incurred in the Designated Target Transaction and (ii) consisting of obligations under deferred compensation or other similar arrangements incurred in Designated Target Transaction;

(n) Indebtedness of the Designated Target Subsidiaries under the Designated Target Working Capital Facility not to exceed, at any time outstanding, an aggregate amount equal to \$200,000,000 minus the amount by which (x) the amount of Investments in the Designated Target Subsidiaries made in reliance on clause (i) of the Section 10.5(s) exceeds (y) \$400,000,000;

(no) [reserved];

(op) Indebtedness of the Borrower or any of its Subsidiaries which may be deemed to exist in connection with agreements providing for indemnification and similar obligations in connection with acquisitions or sales of assets and/or businesses effected in accordance with the requirements of this Agreement;

(pq) Indebtedness of the Borrower or any Subsidiary Guarantor not otherwise permitted hereunder in an aggregate principal amount which, when aggregated with the principal amount or liquidation preference of all other Indebtedness then outstanding and incurred pursuant to this clause (p), does not exceed \$50,000,000 at any one time outstanding;

(qr) Guarantee Obligations (i) of the Borrower in favor of its Subsidiaries to permit foreign currency transactions or fund transfers in an aggregate amount not to exceed \$2,000,000 at any time outstanding, (ii) of the Borrower or any of its Subsidiaries as a guarantor of the lessee under any lease pursuant to which the Borrower or any of its Subsidiaries is the lessee, other than any capital lease pursuant to which a Subsidiary that is not a

Subsidiary Guarantor is the lessee, so long as such lease is otherwise permitted hereunder, (iii) of the Borrower or any of its Subsidiaries as a guarantor of any Capitalized Lease Obligation to which a Joint Venture is a party or any contract entered into by such Joint Venture in the ordinary course of business in an aggregate amount not to exceed \$2,000,000 at any time outstanding; provided that the maximum liability of the Borrower or any of its Subsidiaries in respect of any obligations as described in this clause (iii) is permitted as an Investment pursuant to the requirements of Section 10.5, and (iv) of the Borrower or any of its Subsidiaries which may be deemed to exist pursuant to the acquisition agreements entered into in connection with any Investment permitted pursuant to Section 10.5 (including any obligation to pay the purchase price therefor and any indemnification, purchase price adjustment and similar obligations to the extent otherwise permitted hereunder);

 $(\underline{fs})$  obligations of the Borrower or any Subsidiary consisting of (x) the financing of insurance premiums or (y) take-or-pay obligations contained in supply arrangements, in each case, in the ordinary course of business, in each case to the extent constituting Indebtedness;

(st) [reserved];

(tu) Indebtedness of Subsidiaries that are not Subsidiary Guarantors for working capital purposes, so long as the Indebtedness under this clause (t) does not exceed \$10,000,000 in the aggregate at any time outstanding;

(<u>uv</u>) [reserved];

(<u>w</u>) letters of credit and bank guarantees so long as the aggregate U.S. Dollar equivalent of all such letters of credit and bank guarantees does not exceed \$25,000,000 at any time;

(wx) Permitted Refinancing Indebtedness in respect of any Indebtedness permitted under clauses (f), (g)(iii), (l), (t) and (v) of this Section 10.1(A); and

(xy) all premiums (if any), interest (including post-petition interest), fees, expenses, charges and additional or contingent interest on obligations described in clauses (a) through (w) above.

(B) The Borrower will not issue any preferred stock or other preferred equity interests.

10.2. <u>Limitation on Liens</u>. The Credit Parties will not, and will not permit any of their respective Restricted Subsidiaries to, create, incur, assume or suffer to exist any Lien upon any property or assets of any kind (real or personal, tangible or intangible) of the Borrower or any Restricted Subsidiary, whether now owned or hereafter acquired, except:

(a) Liens arising (i) under the Credit Documents, including the Final DIP Order and (ii) the Interim Cash Collateral Order (which, for the avoidance of doubt, includes Liens granted as adequate protection to Prepetition Secured Parties);

(b) Permitted Liens;

(c) Liens securing Indebtedness permitted pursuant to Section 10.1(A)(f); provided that such Liens attach at all times only to the assets so financed, ECA Collateral and/or other assets subject to Indebtedness incurred pursuant to Section 10.1(A)(f) owing to the same Person as such Indebtedness so secured;

(d) (i) Liens securing the Prepetition Secured debt and (ii) other Liens existing on the Signing Date and listed on <u>Schedule 10.2</u> hereto;

(e) the replacement, extension or renewal of any Lien permitted by clauses (a), (b), (c) and (d)(ii) above of this Section 10.2 upon or in the same assets theretofore subject to such Lien or the replacement, extension or renewal of the Indebtedness secured thereby (in each case, without increase in the amount or change in any direct or contingent obligor except to the extent otherwise permitted hereunder);

(f) [reserved]; Liens securing the Designated Target Working Capital Facility placed upon the Equity Interests in, or assets of, any Designated Target Subsidiary (other than the Equity Interests in any Designated Target Subsidiary directly held by any Credit Party); provided that, no such Liens may be placed in any assets of the Borrower or any Restricted Subsidiary (other than any Designated Target Subsidiaries);

(g) Liens securing (i) the Permitted Acquisition Debt and (ii) Indebtedness permitted under Section 10.1(A)(i), in each case, placed upon the Equity Interests in, or assets of, any Acquired Person or its Subsidiaries; provided that, no such Liens may be placed in any assets of the Borrower or any Restricted Subsidiary (other than the Acquired Persons and their Subsidiaries) other than the Equity Interests in such Acquired Person;

(h) additional Liens so long as the aggregate principal amount of the obligations so secured does not exceed \$50,000,000 at any time outstanding; (i) Liens on the Equity Interests of Unrestricted Subsidiaries;

(j) [reserved];

(k) deposits of cash and cash equivalents not to exceed \$3,000,000 in the aggregate to (i) secure the performance of bids, trade contracts, governmental contracts and leases (other than Indebtedness for borrowed money), statutory obligations, surety, stay, customs and appeal bonds, performance bonds and other obligations of a like nature (including those to secure health, safety and environmental obligations) incurred in the ordinary course of

business or (ii) to provide "adequate assurances of payment" as that term is used in Section 366 of the Bankruptcy Code;

(l) Liens securing Indebtedness permitted pursuant to Section 10.1(A)(v); and

(m) Liens deemed to exist in connection with a transaction permitted under Section 10.4.

10.3. <u>Limitation on Fundamental Changes</u>. Except as expressly permitted by Section 10.4 or 10.5, the Credit Parties will not, and will not permit any of their respective Restricted Subsidiaries to, enter into any merger, consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution), or convey, sell, lease, assign, transfer or otherwise dispose of, all or substantially all its business units, assets or other properties, except that:

(a) any Subsidiary (other than a License Subsidiary) of the Borrower or any other Person may be merged or consolidated with or into the Borrower; provided that (i) the Borrower shall be the continuing or surviving corporation, and (ii) no Default or Event of Default would result from the consummation of such merger or consolidation;

(b) any Subsidiary of the Borrower (other than a License Subsidiary) or any other Person may be merged, amalgamated or consolidated with or into any one or more Subsidiaries of the Borrower; provided that (i) in the case of any merger, amalgamation or consolidation involving one or more Restricted Subsidiaries, (A) a Restricted Subsidiary shall be the continuing or surviving corporation or (B) the Borrower shall take all steps necessary to cause the Person formed by or surviving any such merger, amalgamation or consolidation (if other than a Restricted Subsidiary) to become a Restricted Subsidiary, (ii) in the case of any merger, amalgamation or consolidation involving one or more Subsidiary Guarantors, a Subsidiary Guarantor shall be the continuing or surviving corporation or the Person formed by or surviving any such merger, amalgamation or consolidation (if other than a Restricted Subsidiary Guarantor shall be the continuing or surviving corporation or the Person formed by or surviving any such merger, amalgamation or consolidation (if other than a Subsidiary Guarantor shall be the continuing or surviving corporation or the Person formed by or surviving any such merger, amalgamation or consolidation (if other than a Subsidiary Guarantor shall be the continuing or surviving corporation or the Cases and execute a Joinder Agreement and a supplement to the Security Agreement, (iii) no Default or Event of Default would result from the consummation of such merger, amalgamation or consolidation, and (iv) the Borrower shall have delivered to the Administrative Agent an officers' certificate stating that such merger, amalgamation or consolidation and such supplements to any Security Document comply with this Agreement;

(c) any Restricted Subsidiary (other than a License Subsidiary) that is not a Guarantor may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to the Borrower, a Guarantor or any other Restricted Subsidiary of the Borrower;

(d) any Guarantor may sell, lease, transfer or otherwise dispose of any or all of its assets (upon voluntary liquidation or otherwise) to the Borrower or any other Guarantor;

(e) any Restricted Subsidiary (other than a License Subsidiary) may liquidate, wind up or dissolve if (x) the Borrower determines in good faith that such liquidation or dissolution is in the best interests of the Borrower and is not materially disadvantageous to the Lenders and (y) to the extent such Restricted Subsidiary is a Credit Party, any assets or business not otherwise disposed of or transferred in accordance with Section 10.4 or 10.5, or, in the case of any such business, discontinued, shall be transferred to, or otherwise owned or conducted by, another Credit Party after giving effect to such liquidation or dissolution;

(f) any License Subsidiary may (i) be merged or consolidated with any other License Subsidiary that is a Credit Party, (ii) sell, lease, transfer or otherwise dispose of any or all of its property (upon voluntary liquidation or otherwise) only to another License Subsidiary that is a Credit Party, (iii) sell, transfer or otherwise dispose of capital stock or other ownership interest of such License Subsidiary only to a Credit Party; and

(g) without limiting the ability of the Borrower or any of its Subsidiaries to form a new Subsidiary under the laws of any jurisdiction, the Borrower or any of its Subsidiaries may change its jurisdiction of organization to the United States of America (or any State thereof)-; and

(h) the Borrower and the Restricted Subsidiaries may consummate the Designated Target Transaction.

10.4. Limitation on Sale of Assets. The Credit Parties will not, and will not permit any of their respective Restricted Subsidiaries to, (i) convey, sell, lease, assign, transfer or otherwise dispose of any of its property, business or assets (including receivables and leasehold interests), whether now owned or hereafter acquired (other than any such sale, transfer, assignment or other disposition resulting from any casualty or condemnation, of any assets of the Borrower or the Restricted Subsidiaries) or (ii) sell to any Person (other than the Borrower or a Guarantor) any shares owned by it of any Restricted Subsidiary's capital stock, except that:

(a) the Borrower and the Restricted Subsidiaries may sell, transfer or otherwise dispose of (i) cash and other Permitted Investments, (ii) property no longer used or useful, or economically practicable to maintain, in the conduct of the business of the Borrower and the Restricted Subsidiaries (including allowing any registration or application for registration of any intellectual property that is no longer used or useful, or economically practicable

to maintain, to lapse, go abandoned, or be invalidated) and (ii) used or surplus equipment, vehicles, inventory and other assets in the ordinary course of business;

(b) the Borrower and the Restricted Subsidiaries may sell, transfer or otherwise dispose of other assets (other than accounts receivable) for fair value; provided that (i) the total non-cash consideration received since the Signing Date in respect of any sales, transfers and or dispositions pursuant to this clause (b) for consideration in excess of \$10,000,000 shall not be less than 75% of aggregate consideration received in such sale, transfer or disposition, (ii) any non-cash proceeds received are pledged to the Collateral Agent to the extent required under Section 9.12 as Collateral to the extent the persons selling, transferring or disposing such asset was the Borrower or a Guarantor, (iii) to the extent applicable, the Net Cash Proceeds thereof received by the Borrower and its Restricted Subsidiaries are promptly applied to the prepayment and/or commitment reductions as provided for in Section 5.2, (iv) after giving effect to any such sale, transfer or disposition, no Default or Event of Default shall have occurred and be continuing;

(c) the Borrower and the Restricted Subsidiaries may make sales of assets to the Borrower or to any Restricted Subsidiary; <u>provided</u> that with respect to any such sales to Restricted Subsidiaries that are not Guarantors (i) such sale, transfer or disposition shall be for fair value, (ii) to the extent constituting an Investment, such Investment must be a permitted Investment in a Restricted Subsidiary that is not a Credit Party in accordance with Section 10.5 and (iii) any non-cash proceeds received are pledged to the Collateral Agent to the extent required under Section 9.12 as Collateral to the extent the persons selling, transferring or disposing such asset was the Borrower or a Guarantor;

(d) the Borrower or any Restricted Subsidiary may effect any transaction permitted by (x) Section 10.2(b) pursuant to clause (m) of the definition of "Permitted Lien" or (y) Section 10.3;

(e) the Borrower and the Restricted Subsidiaries may sell or discount without recourse accounts receivable arising in the ordinary course of business in connection with the compromise or collection thereof;

(f) the Borrower and its Restricted Subsidiaries may lease, or sublease, any Real Estate or personal property in the ordinary course of business;

(g) [reserved];

(h) [reserved];

(i) the Borrower and its Restricted Subsidiaries may exchange operating assets for other operating assets (including a combination of assets and cash and cash equivalents) related to a Permitted Business of comparable or greater market value or usefulness to the business of the Borrower and its Restricted Subsidiaries as a whole, as determined in good faith by the Borrower, which in the event of an exchange of operating assets with a Fair Market Value in excess of (1) \$20,000,000 shall be evidenced by an certificate of an Authorized Officer, and (2) \$50,000,000 shall be set forth in a resolution approved in good faith by at least a majority of the board of directors of the Borrower; provided that the aggregate amount of assets exchanged pursuant to this Section 10.4(i) (determined based on the Fair Market Value thereof) shall not exceed \$500,000,000; provided, further, that, to the extent any operating assets being exchanged constitute Collateral, the exchanging operating assets so received shall become Collateral upon the consummation of the exchange subject to the terms hereof;

(j) [reserved]; any disposition of cash and cash equivalents and/or the assignment of any rights and obligations under the Designated Target Transaction Agreements to any Designated Target Acquisition Vehicle, in each case, made as part of the Designated Target Transaction;

(k) any disposition of assets pursuant to any First Day Orders; and

(1) the Borrower and its Restricted Subsidiaries may sell Equity Interests in, or Indebtedness or other securities of, an Unrestricted Subsidiary.

10.5. Limitation on Investments. The Credit Parties, subject to the Order and any other order of the Bankruptcy Court, will not, and will not permit any of their respective Restricted Subsidiaries to, make any advance, loan, extensions of credit or capital contribution to, or purchase any stock, bonds, notes, debentures or other securities of or any assets of, or make any other Investment in, any Person, except:

(a) extensions of trade credit and asset purchases in the ordinary course of business;

(b) Permitted Investments;

(c) loans and advances to officers, directors and employees of the Borrower or any of its Subsidiaries for reasonable and customary business related travel, entertainment, relocation and analogous ordinary business purposes in an aggregate principal amount at any time outstanding under this clause (c) not exceeding \$2,500,000;

(d) Investments existing or contractually committed on the Signing Date and listed on <u>Schedule 10.5</u> to this Agreement and any extensions, renewals or reinvestments thereof, so long as the aggregate amount of all Investments pursuant to this clause (d) is not increased at any time above the amount of such Investments existing on the Signing Date;

(e) Investments received in connection with the bankruptcy or reorganization of suppliers or customers and in settlement of delinquent obligations of, and other disputes with, customers arising in the ordinary course of business;

(f) [reserved];

(g) Investments in (i) any Subsidiary Guarantor or the Borrower; (ii) Restricted Subsidiaries that are not Guarantors, in the case of this clause (g) (ii), in an aggregate amount not to exceed (x) in the case of Government Business Subsidiaries, \$15,000,000 at any one time outstanding and (y) in the case of all other Restricted Subsidiaries that are not Guarantors, \$20,000,000 in the aggregate at any one time outstanding; and (iii) any Unrestricted Subsidiaries that are not Guarantors, in the case of this clause (g)(iii), in an aggregate amount not to exceed \$10,000,000 at any one time outstanding and (iv) Restricted Subsidiaries that are not Guarantors to the extent such Investments are made by Restricted Subsidiaries that are not Guarantors;

(h) [reserved];

(i) [reserved];

(j) Investments constituting non-cash proceeds of sales, transfers and other dispositions of assets to the extent permitted by Section 10.4(b) or (c); (k) [reserved];

(1) Investments permitted under Section 10.6;

(m) [reserved];

(n) Investments constituting advances in the form of a prepayment of expenses, so long as such expenses were incurred in the ordinary course of business and are being paid in accordance with customary trade terms of the Borrower or such Subsidiary;

(o) to the extent constituting Investments, any payments under any contracts to construct, launch, operate or insure Satellites which contracts are entered into in the ordinary course of business;

(p) loans and advances for purposes for which a dividend is otherwise permitted pursuant to Section 10.6, including, without limitation dividends of the type contemplated in Section 10.6(j);

(q) [reserved];

(r) [reserved];

(s) other Investments (including Investments in any Restricted Subsidiary that is not a Guarantor solely as an intermediate step of any such Investment; it being agreed that such intermediate steps shall not constitute separate Investments requiring additional capacity under this Section 10.5) in an amount of up to the sum of (i) \$350,000,000 at any one time outstanding, which amount shall be increased to an amount equal to the lesser of (A) \$600,000,000 and (B) an amount, as necessary, to consummate the Designated Target Transaction in accordance with the terms of the Designated Target Transaction Agreements (it being understood, for the avoidance of doubt, such amount shall not include any Investments in the Designated Target Subsidiaries for working capital purposes other than the working capital and cash adjustments to the purchase price in accordance with the Designated Target Transaction Agreements) ("Increased Investment Basket") concurrently with and in order to consummate the Designated Target Transaction (provided that, (w) the Designated Target Transaction Agreements shall not have been amended, modified or waived by the Borrower or its Subsidiaries in a manner materially adverse to the Lenders without the prior written consent of the Required Lenders (provided that each Lender shall be deemed to have consented to such amendment, waiver or consent unless it shall object in writing thereto to the Borrower within five (5) Business Day after notice and receipt of such proposed amendment, waiver or consent); (x) Investments in the Designated Target Subsidiaries in reliance on this clause (i) shall be funded by the Designated Target Intercompany Loan; and (y) the Designated Target Intercompany Loan and the Equity Interests in any Designated Target Subsidiary directly held by any Credit Party shall be subject to the DIP Liens and the Adequate Protection Liens) and (ii) except with respect to the Designated Target Transaction, an amount funded by any Permitted Acquisition Debt incurred in connection with any Permitted Acquisitions; provided, that, to the extent any Investment pursuant to this clause (FS) constitutes an Acquisition, such Investment shall be a Permitted Acquisition; provided further that, no Investments pursuant to this clause (FS) may be made in any Restricted Subsidiary that is not a Guarantor unless such Investment is to facilitate an Acquisition or to make an Investment in an Acquired Person or its Subsidiaries; provided, further, that, upon consummation of the Designated Target Transaction, neither the Credit Parties nor any Restricted Subsidiaries shall be permitted to make any Investments under clause (ii) of this Section 10.5(s).

(t) [reserved];

(u) Investments subject to and permitted under Section 10.3;

(v) Investments constituting Guarantee Obligations permitted under Section 10.1(A)(e); and

(w) Guarantee Obligations of Borrower and any Guarantor in respect of any Permitted Refinancing Indebtedness in respect thereof.

Notwithstanding the foregoing, no Investments may be made, on and after the Signing Date, to (i) any Unrestricted Subsidiary (other than pursuant to Section 10.5(g)(iii)) or (ii) any License Subsidiary that is not a Subsidiary Guarantor.

10.6. <u>Limitation on Dividends</u>. The Borrower will not declare or pay any dividends (other than dividends payable solely in its capital stock) or return any capital to its stockholders or make any other distribution, payment or delivery of property or cash to its stockholders as such, or redeem, retire, purchase or otherwise acquire, directly or indirectly, for consideration, any shares of any class of its capital stock or the capital stock of any direct or indirect parent now or hereafter outstanding (or any options or warrants or stock appreciation rights issued with respect to any of its capital stock), or set aside any funds for any of the foregoing purposes, or permit any of the Restricted Subsidiaries to purchase or otherwise acquire for consideration (other than in connection with an Investment permitted by Section 10.5) any shares of any class of the capital stock of the Borrower, now or hereafter outstanding (or any options or warrants or stock appreciation rights issued with respect to any of its capital stock) (all of the foregoing "dividends"); provided that, so long as no Default or Event of Default exists or would exist after giving effect thereto:

(a) the Borrower may pay any fees and expenses of any parent of the Borrower or commonly controlled Affiliate of any parent of the Borrower (as deemed dividends or distributions) related to ownership and operation of the Borrower and its Subsidiaries (including fees and expenses (including franchise or similar taxes) required to maintain its corporate existence, customary salary, bonus and other benefits payable to, and indemnities provided on behalf of, officers and employees of any parent of the Borrower and general corporate overhead expenses of any parent of the Borrower); and

(b) for any taxable year, the Borrower may declare and pay dividends or other distributions to any parent of the Borrower if such parent is required to file a consolidated, unitary or similar tax return reflecting income of the Borrower or its Restricted Subsidiaries, in an amount equal to the portion of such taxes attributable to the Borrower and/or its Restricted Subsidiaries that are not payable directly by the Borrower and/or its Restricted Subsidiaries, but not to exceed the amount that the Borrower and/or such Restricted Subsidiaries would have been required to pay in respect of such taxes if the Borrower and/or such Restricted Subsidiaries (as applicable) had been required to pay such taxes directly as standalone taxpayers (or a standalone group separate from such parent).

10.7. <u>Limitations on Debt Payments</u>. Except as expressly permitted by the terms and conditions set forth in the Orders, the First Day Orders or any other order of the Bankruptcy Court reasonably acceptable to the Required Lenders, the Borrower or any Restricted Subsidiary shall not make any prepayment on account of (or set aside funds for the prepayment of) any Indebtedness of a Debtor that was incurred or arose prior to filing of the Cases.

10.8. Limitations on Sale Leasebacks. The Borrower will not, and will not permit any of the Restricted Subsidiaries to, enter into or effect any Sale Leasebacks, other than Permitted Sale Leasebacks; provided that the aggregate fair value of all assets disposed of pursuant to this Section 10.8 shall not exceed \$60,000,000 in the aggregate.

10.9. <u>Transfer to Parent Companies</u>. The Credit Parties will not, and will not permit their respective Restricted Subsidiaries to, transfer or otherwise provide, directly or indirectly, any credit support, guarantee, cash and/or cash equivalents or any proceeds of the Loans, to or for the benefit of any Parent Company (in the form of dividends, Investments, intercompany advances, guarantee of obligations (including interest thereon) or otherwise), without the prior written consent of the Required Lenders, other than as permitted pursuant to Section 10.6 and the payment by the Credit Parties and their respective Restricted Subsidiaries of any costs of administration of the Cases allocable to any Parent Company based upon a to be agreed-upon allocation as between the DIP Debtors and the Required Backstop Parties (the "Parent Case Expenses"); provided however, that, the payment pursuant to Section 10.6(a) and the Parent Case Expenses shall be subject to reimbursement by such Parent Company to the Credit Parties or their respective Restricted Subsidiary, which reimbursement claims shall be treated as administrative expense claims as provided in the Orders.

10.10. Additional Bankruptcy Matters. The Credit Parties will not permit, and will not permit any of their respective Restricted Subsidiaries to, without the Required Lenders' prior written consent (unless otherwise specified), do any of the following:

(a) use any portion or proceeds of the Loans or the Collateral for payments or purposes that would violate the terms of the Orders;

(b) incur, create, assume, suffer to exist or permit, except for the Carve Out or as otherwise expressly permitted by the Orders or any other order of the Bankruptcy Court reasonably acceptable to the Required Lenders,

any other superpriority administrative claim which is pari passu with or senior to the claim of the Secured Parties against any DIP Debtor;

(c) subject to the Orders, assert, join, investigate, support or prosecute any claim or cause of action against any of the Secured Parties (in their capacities as such), unless such claim or cause of action is in connection with the enforcement of the Credit Documents against any of the Secured Parties;

(d) other than as provided in any First Day Order, enter into any agreement to return any of its inventory to any of its creditors for application against any pre-petition Indebtedness, pre-petition trade payables or other pre-petition claims under Section 546(c) of the Bankruptcy Code if, after giving effect to any such agreement, the aggregate amount applied to pre-petition Indebtedness, pre-petition trade payables and other pre-petition claims subject to all such agreements, setoffs and recoupments since the Petition Date would exceed \$250,000;

(e) seek, consent to, or permit to exist any order granting authority to take any action that is prohibited by the terms of this Agreement, the Orders, the or the other Credit Documents or refrain from taking any action that is required to be taken by the terms of this Agreement, the Orders or any of the other Credit Documents;

(f) subject to the terms of the Orders, object to, contest, delay, prevent or interfere with in any material manner the exercise of rights and remedies by the Agents, the Lenders or other Secured Parties with respect to the Collateral following the occurrence of an Event of Default, including without limitation a motion or petition by any Secured Party to lift an applicable stay of proceedings to do the foregoing (*provided* that any DIP Debtor may contest or dispute whether an Event of Default has occurred in accordance with the terms of the Orders and the Credit Documents);

(g) except for the Carve Out or as otherwise expressly permitted by the Orders, incur, create, assume, suffer to exist or permit (or file an application for the approval of) any other Superpriority Claim which is pari passu with or senior to the claims of the Administrative Agent, the Collateral Agent, Lenders and the other Secured Parties constituting Obligations against the DIP Debtors or the adequate protection Liens or claims granted under the Orders; or

(h) make or permit to be made any change to the Final DIP Order.

SECTION 11. [Reserved].

## SECTION 12. Events of Default.

Upon the occurrence of any of the following specified events (each an "Event of Default"):

12.1. <u>Payments</u>. The Borrower shall (a) default in the payment when due of any principal of the Loans or (b) default, and such default shall continue for five or more Business Days, in the payment when due of any interest or stamping fees on the Loans or any Fees or of any other amounts owing hereunder or under any other Credit Document; or

12.2. <u>Representations, etc.</u> Any representation, warranty or statement made or deemed made by any Credit Party herein or in any Security Document or any certificate delivered or required to be delivered pursuant hereto or thereto shall prove to be untrue in any material respect on the date as of which made or deemed made; <u>provided</u>, <u>however</u>, that to the extent any such representation, warranty or statement is untrue solely as a result of an action or inaction by Government Business Subsidiaries, and the Borrower has otherwise complied with the terms and conditions of Section 9.19 hereof, no Default or Event of Default shall occur; or

12.3. Covenants. Any Credit Party shall

(a) default in the due performance or observance by it of any term, covenant or agreement contained in Section 9.1(e), Section 10 or Section 11; or (b) default in the due performance or observance by it of any term, covenant or agreement (other than those referred to in Section 12.1 or 12.2 or clause (a) of this Section 12.3) contained in this Agreement, any Security Document or the Administrative Agent Fee Letter and such default shall continue unremedied for a period of at least 30 days after receipt of written notice by the Borrower from the Administrative Agent or the Required Lenders; provided, however, that to the extent such failure relates solely to an action or inaction by Government Business Subsidiaries, and the Borrower has otherwise complied with the terms and conditions of Section 9.19 hereof, no Default or Event of Default shall occur; or

12.4. Default Under Other Agreements. (a) The Borrower or any of the Restricted Subsidiaries shall (i) default in any payment with respect to any Indebtedness (other than the Obligations) in excess of \$50,000,000 in the aggregate, for the Borrower and such Restricted Subsidiaries, beyond the period of grace, if any, provided in the instrument or agreement under which such Indebtedness was created or (ii) default in the observance or performance of any agreement or condition relating to any such Indebtedness or contained in any instrument or agreement

evidencing, securing or relating thereto, or any other event shall occur or condition exist (other than, with respect to Indebtedness consisting of any Hedge Agreements, termination events or equivalent events pursuant to the terms of such Hedge Agreements), the effect of which default or other event or condition is to cause, or to permit the holder or holders of such Indebtedness (or a trustee or agent on behalf of such holder or holders) to cause, any such Indebtedness to become due prior to its stated maturity; or (b) without limiting the provisions of clause (a) above, any such Indebtedness shall be declared to be due and payable, or required to be prepaid other than by a regularly scheduled required prepayment or as a mandatory prepayment (and, with respect to Indebtedness consisting of any Hedge Agreements, other than due to a termination event or equivalent event pursuant to the terms of such Hedge Agreements), prior to the stated maturity thereof; provided that this Section 12.4 shall not apply to any Indebtedness of the DIP Debtors under the Cases that was incurred prior to the Petition Date unless such Indebtedness has been accelerated and the enforcement of remedies with respect to such Indebtedness shall not have been stayed by the commencement of the Cases; or

12.5. <u>Bankruptcy, etc</u>. An involuntary proceeding shall be commenced or a voluntary or involuntary petition shall be filed in a court of competent jurisdiction seeking (1) relief in respect of any Subsidiary that did not become a Debtor on the Petition Date, or of a substantial part of the assets of any Subsidiary that did not become a Debtor on the Petition Date, under any Debtor Relief Laws (unless (A) promptly (but in any event not later than fifteen (15) Business Days) following the commencement of such case such Subsidiary becomes a Credit Party hereunder, (B) within ten (10) Business Days after the commencement of such case such Subsidiary's case becomes jointly administered with the Cases and (C) each of the Orders are made applicable to such Subsidiary that did not become a Debtor on the Petition Date or for a substantial part of the assets of any Subsidiary that did not become a Debtor on the Petition Date or for a substantial part of the assets of any Subsidiary that did not become a Debtor on the Petition Date or for a substantial part of the assets of any Subsidiary that did not become a Debtor on the Petition Date or for a substantial part of the assets of any Subsidiary that did not become a Debtor on the Petition Date or for a substantial part of the assets of any Subsidiary that did not become a Debtor on the Petition Date or (3) the winding up or liquidation of any Subsidiary that did not become a Debtor on the Petition Date (except in a transaction permitted by Section 10.3) and any such proceeding or petition shall continue undismissed for sixty (60) days or an order or decree approving or ordering any of the foregoing shall be entered; or

12.6. <u>ERISA</u>. (a) Any Plan shall fail to satisfy the minimum funding standard required for any plan year or part thereof or a waiver of such standard or extension of any amortization period is sought or granted under Section 412 of the Code; any Plan is or shall have been terminated or is the subject of termination proceedings under ERISA (including the giving of written notice thereof); an event shall have occurred or a condition shall exist in either case entitling the PBGC to terminate any Plan or to appoint a trustee to administer any Plan (including the giving of written notice thereof); any Plan shall have an accumulated funding deficiency (whether or not waived), the Borrower or any Subsidiary or any ERISA Affiliate has incurred or is likely to incur a liability to or on account of a Plan under Section 409, 502(i), 502(l), 515, 4062, 4063, 4064, 4069, 4201 or 4204 of ERISA or Section 4971 or 4975 of the Code (including the giving of written notice thereof); (b) there could result from any event or events set forth in clause (a) of this Section 12.6 the imposition of a Lien, the granting of a security interest, or a liability, or the reasonable likelihood of incurring a Lien, security interest or liability; and (c) such Lien, security interest or liability will or would be reasonably likely to have a Material Adverse Effect; or

12.7. <u>Guarantee</u>. The Guarantees or any material provision thereof shall cease to be in full force or effect or any Guarantor thereunder or any Credit Party shall deny or disaffirm in writing any Guarantor's obligations under the Guarantee; or

12.8. [<u>Reserved</u>]; or

12.9. <u>Security Agreements</u>. Any Security Agreements or any material provision thereof shall cease to be in full force or effect (other than pursuant to the terms hereof or thereof or as a result of acts or omissions of the Administrative Agent or any Lender) or any grantor thereunder or any Credit Party shall deny or disaffirm in writing any grantor's obligations under any Security Agreement; or

# 12.10. FCC Matters

(a). (i) The Credit Parties shall have failed to submit to the FCC the Participation Election Form within 90 days of the Final DIP Order Entry Date or (ii) the Credit Parties shall revoke or withdraw the Participation Election Form from the FCC; or

12.11. Judgments. One or more judgments or decrees (excluding any First Day Order or any order fixing the amount of any claim in the Cases) as to any post-petition obligation shall be entered against the Borrower or any of the Restricted Subsidiaries involving a liability of \$50,000,000 or more in the aggregate for all such judgments and decrees for the Borrower and the Restricted Subsidiaries (to the extent not paid or fully covered by insurance provided by a carrier not disputing coverage) (and in the case of such a judgment against any of the DIP Debtors, such judgment arose post-petition) and any such judgments or decrees shall not have been satisfied, vacated,

discharged or stayed (including pursuant to the Bankruptcy Code) or bonded pending appeal within 60 days from the entry thereof; or

12.12. Change of Control. A Change of Control shall occur; or

12.13. <u>Dismissal; Conversion; Appointment of Trustee.</u> (i) Any of the Cases of the Debtors shall be dismissed or converted to a case under Chapter 7 of the Bankruptcy Code or any Debtors shall file a motion or other pleading seeking the dismissal of any of Case of any Debtor under Section 1112 of the Bankruptcy Code or otherwise, or seeking the conversion of any such Case to a case under Chapter 7 of the Bankruptcy Code, without causing Payment in Full of all Obligations hereunder and immediate termination of all Commitments or (ii) a trustee under Chapter 11 of the Bankruptcy Code or an examiner with enlarged powers relating to the operation of the business (powers beyond those set forth in Section 1106(a)(3) and (4) of the Bankruptcy Code under Section 1104(b) of the Bankruptcy Code shall be appointed in any of the Cases of the Debtors (or the Credit Parties or their Affiliates shall have acquiesced to the entry of such order) unless consented to by the Required Lenders; or

12.14. <u>Superpriority Claims</u>. An application shall be filed by any Debtor for the approval of (i) any other Superpriority Claim or Lien, or an order of the Bankruptcy Court shall be entered granting any other Superpriority Claim or Lien (other than the Carve Out), in any of the Cases of the Debtors that is pari passu with or senior to the claims (as such word is defined in the Bankruptcy Code) or Liens of the Collateral Agent, the Lenders and the other Secured Parties against the Borrower or any other Credit Party hereunder or under any of the other Credit Documents (including the adequate protection Liens and claims provided for in the Orders) or (ii) any Liens senior or pari passu with (A) the Liens under the Prepetition Secured Debt or (B) the adequate protection Liens granted on account of the Primed Liens, or there shall arise or otherwise be granted any such pari passu or senior Superpriority Claim or senior Lien, in each case other than adequate protection Liens granted by the Bankruptcy Court pursuant to the Orders and otherwise reasonably acceptable to the Required Lenders; or

12.15. <u>Adverse Claims</u>. Any Person shall assert any claim in the Cases arising under Section 506(c) of the Bankruptcy Code against any Agent, any Lender or the Collateral or any Prepetition Agent (as defined in the form of Final Order attached hereto as Exhibit A), any Prepetition Secured Parties (as defined in the form of Final Order attached hereto as Exhibit A) or the Prepetition Collateral (as defined in the form of Final Order attached hereto as Exhibit A), and, solely with respect to such claim asserted by any Person other than a Credit Party or any Affiliate thereof, either (i) the same shall remain unopposed by the Borrower for more than 5 Business Days, or (ii) in any event, any such claim shall not be disallowed, dismissed or withdrawn, with prejudice, within 60 days after the assertion thereof; or if any Agent, Lenders or the Collateral are surcharged (other than with respect to the Carve Out) pursuant to Sections 105, 506(c), 552 or any other section of the Bankruptcy Code without the prior written consent of the Required Lenders; or

12.16. <u>Adverse Orders</u>. Solely with respect to the DIP Facility, any order is entered by the Bankruptcy Court sustaining any objection or challenge of any kind or nature to the validity, priority, or amount of the Liens with created pursuant to the Credit Documents in favor of or claims held by, or an action to recharacterize or subordinate any Agent or any Lender; or

12.17. <u>Stay Relief</u>. The Bankruptcy Court shall enter an order or orders granting relief from the automatic stay applicable under Section 362 of the Bankruptcy Code to the holder or holders of any security interest to (i) permit foreclosure (or the granting of a deed in lieu of foreclosure or the like) on any assets of any of the Debtors which have a value in excess of \$50,000,000 in the aggregate or (ii) permit other actions that would have a Material Adverse Effect on the DIP Debtors or their estates (taken as a whole); or

12.18. Orders; Actions. (i) the Final DIP Order Entry Date shall not have occurred by the date that is 45 days following the Petition Date; (ii) an order of the Bankruptcy Court shall be entered reversing, amending, supplementing, staying, vacating or otherwise amending, supplementing or modifying the Orders or the Borrower or any Subsidiary of the Borrower shall apply for the authority to do so, in each case in a manner that is adverse in any respect to the Agents or the Lenders, without the prior written consent of the Administrative Agent and the Required Lenders or, with respect to matters described hereunder that requires consent from or shall be acceptable or satisfactory to each Lender or each Lender directly and adversely affected thereby; (iii) an order of the Bankruptcy Court shall be entered denying or terminating use of Cash Collateral (as defined in the Orders) by the Credit Parties or imposing any additional conditions on such use (and such order remains unstayed for more than three (3) Business Days) and the Credit Parties shall have not obtained use of Cash Collateral pursuant to an order consented to by, and in form and substance reasonably acceptable to, the Required Lenders; (iv) the Final DIP Order shall cease to create a valid and perfected Lien on the Collateral described therein or the Final DIP Order shall cease to be in full force and effect; (v) any of the Credit Parties or any Subsidiary of the Borrower shall fail to comply with the Orders in any material respect; (vi) the Final DIP Order shall not authorize the borrowing by the Borrower of the full amount of the

Commitments provided for hereunder; (vii) the entry of an order in the Cases seeking to use cash collateral or obtain financing pursuant to Section 364 of the Bankruptcy Code (other than the DIP Facility), unless such financing would (and actually does) provide for Payment in Full of all Obligations and terminate all Commitments upon the consummation thereof; (viii) any order shall be entered in the Cases providing adequate protection, other than any Orders or pursuant to any First Day Order or any other order acceptable to the Required Lenders; (ix) an order of the Bankruptcy Court shall be entered avoiding or permitting recovery of any portion of the payments made on account of the Obligations owing under this Agreement; (x) an order of the Bankruptcy Court shall be entered terminating or modifying the exclusive right of any DIP Debtor to file a Reorganization Plan pursuant to Section 1121 of the Bankruptcy Code or (xi) the Borrower or any of its Subsidiaries shall take any action in support of the items referred to in the foregoing clauses (i)-(x); or

12.19. <u>Pre-Petition Payments</u>. Except as permitted by the Orders or as otherwise agreed to by the Required Lenders and permitted by the Bankruptcy Court, any Debtor shall make any Pre-Petition Payment other than Pre-Petition Payments (i) authorized by the Bankruptcy Court in accordance with the Orders, or the First Day Orders of the Bankruptcy Court reasonably satisfactory to the Required Lenders, or (ii) authorized by other orders entered by the Bankruptcy Court in amounts reasonably acceptable to the Required Lenders; or

12.20. <u>Adverse Actions</u>. The Credit Parties or any of their Subsidiaries, or any Person claiming by or through the Credit Parties or any of their Subsidiaries, shall obtain court authorization to commence, or shall commence, join in, assist or otherwise participate as an adverse party in any suit or other proceeding against any Agent or any Lender (in any of their respective capacities as such) relating to the DIP Facility, unless such suit or other proceeding is in connection with the enforcement of the Credit Documents against any Agent or any Lender; or

12.21. <u>Reorganization Plan</u>. A Reorganization Plan that is not an Acceptable Plan shall be confirmed in any of the Cases of the Debtors, or any of the Credit Parties or any of their Subsidiaries shall file, propose, support or fail to contest in good faith the filing or confirmation of any such plan or entry of any such order; or

12.22. <u>Supporting Actions</u>. Any Credit Party or any of their Subsidiaries shall file any pleading with the Bankruptcy Court in support of any matter set forth in Sections 12.13, 12.14, 12.15, 12.16, 12.17, 12.19 or 12.21 and such pleading is not withdrawn after five (5) Business Days' the notice from the Administrative Agent or the Required Lenders; or

12.23. <u>Sale Motions</u>. The Borrower shall file (or fail to oppose) any motion seeking an order authorizing the sale of all or substantially all of the assets of the Credit Parties, unless such sale would (and actually does) result in the Payment in Full of all Obligations and terminate all Commitments under this Agreement and the other Credit Documents upon consummation thereof; or

12.24. Acceptable Plan. The DIP Debtors shall fail to obtain entry of an order of the Bankruptcy Court confirming an Acceptable Plan by the date that is 180 days prior to the then Scheduled Maturity Date; provided that, to the extent the Borrower makes a valid election of the First Maturity Extension prior to the date that is 180 days prior to the Initial Scheduled Maturity Date, the then Scheduled Maturity Date shall be determined giving effect to the First Maturity Extension; provided further, that to the extent the Borrower makes a valid election of the Second Maturity Extension, an Event of Default shall occur upon the failure of the DIP Debtors to obtain entry of an order of the Bankruptcy Court confirming an Acceptable Plan by the date that is the Scheduled Maturity Date after giving effect to the First Maturity Extension; or

12.25. <u>First Day Orders</u>. The Debtors shall file any first day motion (including, without limitation, any motion related to cash management any critical vendor or supplier motions) or a related order is entered by the Bankruptcy Court that are not in form and substance reasonably satisfactory to the Required Lenders.

then, and in any such event, and at any time thereafter, if any Event of Default shall then be continuing, the Administrative Agent shall, upon the written request of the Required Lenders, by five (5) Business Days' prior written notice to the Borrower, take any or all of the following actions (subject to the terms of the Final DIP Order), without prejudice to the rights of the Administrative Agent or any Lender to enforce its claims against the Borrower, except as otherwise specifically provided for in this Agreement: (i) declare the Term Loan Commitment terminated, whereupon the Commitments, if any, of each Lender shall forthwith terminate immediately and any Fees theretofore accrued shall forthwith become due and payable without any other notice of any kind; (ii) declare the principal of and any accrued interest and fees in respect of all Loans and all Obligations owing hereunder and thereunder to be, whereupon the same shall become, forthwith due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower and (iii) exercise on behalf of itself and the Lenders all rights and remedies available to it and the Lenders under the Credit Documents or applicable Law.

## SECTION 13. The Agent.

#### 13.1. Appointment and Authority.

(a) Each of the Lenders hereby irrevocably appoints Credit Suisse AG, Cayman Islands Branch to act on its behalf as the Administrative Agent and Credit Suisse AG, Cayman Islands Branch to act on its behalf as the Collateral Agent hereunder and under the other Credit Documents and authorizes each of the Administrative Agent and the Collateral Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent and the Collateral Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Section are solely for the benefit of the Administrative Agent, the Collateral Agent and the Lenders, and, except as provided under Section 13.6 and 13.11, neither the Borrower nor any other Credit Party shall have rights as a third party beneficiary of any of such provisions.

(b) The Collateral Agent and any co-agents, sub-agents and attorneys-in-fact appointed by the Collateral Agent pursuant to Section 13.5 for purposes of holding or enforcing any Lien on the Collateral (or any portion thereof) granted under the Security Documents, or for exercising any rights and remedies thereunder at the direction of the Collateral Agent), shall be entitled to the benefits of all provisions of this Section 13 and Section 14 as if set forth in full herein with respect thereto.

13.2. <u>Rights as a Lender</u>. The Person serving as an Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not an Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as an Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not an Agent hereunder and without any duty to account therefor to the Lenders.

13.3. <u>Exculpatory Provisions</u>. No Agent shall have any duties or obligations except those expressly set forth herein and in the other Credit Documents. Without limiting the generality of the foregoing, an Agent:

(a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

(b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Credit Documents that any Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Credit Documents); provided that an Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose such Agent to liability or that is contrary to any Credit Document or applicable law;

(c) shall not, except as expressly set forth herein and in the other Credit Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as an Agent or any of its Affiliates in any capacity;

(d) shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as such Agent shall believe in good faith shall be necessary, under the circumstances as provided in Sections 12 and 14.1) or (ii) in the absence of its own gross negligence or willful misconduct. Neither Agent shall be deemed to have knowledge of any Default unless and until written notice describing such Default is given to such Agent by the Borrower or a Lender.

(e) shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Credit Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Credit Document or any other agreement, instrument or document, or the creation, perfection or priority of any Lien purported to be created by the Security Documents, (v) the value or the sufficiency of any Collateral, or (vi) the satisfaction of any condition set forth in Section 6 or 7 or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to such Agent.

13.4. <u>Reliance by Agent</u>. The Agents shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document, order, judgment or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Agents also may rely

upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, that by its terms must be fulfilled to the satisfaction of a Lender, any Agent may presume that such condition is satisfactory to such Lender unless the Administrative Agent shall have received notice to the contrary from such Lender prior to the making of such Loan. The Agents may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall be entitled to rely on, and shall not be liable for any action taken or not taken by it in accordance with, the advice of any such counsel, accountants or experts. In addition, the Lenders hereby authorize Akin Gump Strauss Hauer & Feld LLP ("Akin") to provide written directions (which may be made by electronic mail) to each Agent on behalf of the Required Lenders in regards to the Credit Documents and such other matters until such time as the Required Lenders. The giving of any such direction by Akin is no longer authorized to provide any written directions to the Agents on behalf of the Required Lenders in regards to the crediting by Akin that such authorization from the Required Lenders. The giving of any such direction by Akin shall be deemed a reaffirmation by Akin that such authorization from the Required Lenders has been duly given. The Administrative Agent may rely on and act upon any such direction given by Akin and need not inquire as to the due authorization thereof. Notwithstanding the foregoing, to the extent any Agent determines that any direction form Akin requires any clarification or supplementation, such Agent shall promptly inform the Required Lenders or Akin, as applicable, of such determination and the Required Lenders or Akin, as applicable, shall promptly provide to such Agent one or more additional clarifying or supplementing directions

Anything herein to the contrary notwithstanding, whenever reference is made in this Agreement or any other Credit Document to any action by, consent, designation, specification, requirement or approval of, notice, request or other communication from, or other direction given or action to be undertaken or to be (or not to be) suffered or omitted by the Agents or to any election, decision, opinion, acceptance, use of judgment, expression of satisfaction or other exercise of discretion, rights or remedies to be made (or not to be made) by the Agents hereunder or thereunder, it is understood that in all cases the Agents shall be acting, giving, withholding, suffering, omitting, taking or otherwise undertaking and exercising the same) as directed by the Required Lenders or the Required Backstop Parties, as applicable, (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Credit Documents) and shall not be required to take any action unless it has received such direction from the Required Lenders or the Required Backstop Parties, as applicable, (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Credit Documents).

13.5. <u>Delegation of Duties</u>. The Administrative Agent and Collateral Agent may perform any and all of their respective duties and exercise their respective rights and powers hereunder or under any other Credit Document by or through any one or more sub-agents appointed by the Administrative Agent or Collateral Agent. The Administrative Agent, Collateral Agent and any such sub-agent may perform any and all of their respective duties and exercise their respective rights and powers by or through their respective Related Parties. The exculpatory provisions of this Section 13 shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and Collateral Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent and Collateral Agent.

13.6. <u>Resignation of Administrative Agent</u>. The Administrative Agent may at any time give notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, in consultation with the Borrower, to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (a) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Credit Documents (except that in the case of any collateral security held by the Administrative Agent on behalf of the Lenders under any of the Credit Documents, the retiring Administrative Agent shall continue to hold such collateral security until such time as a successor Administrative Agent is appointed) and (b) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender directly, until such time as the Required Lenders appoint a successor Administrative

Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Credit Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Credit Documents, the provisions of this Section 13 and Section 14.5 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

13.7. Non-Reliance on Administrative Agent and Other Lenders. Each Lender acknowledges that it has, independently and without reliance upon any Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon any Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Credit Document or any related agreement or any document furnished hereunder or thereunder.

### 13.8. [Reserved].

13.9. <u>Administrative Agent May File Proofs of Claim</u>. In case of the pendency of any proceeding under any Debtor Relief Law or any other judicial proceeding relative to any Credit Party, the Administrative Agent (irrespective of whether the principal of any Loan shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise

(a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans Outstanding and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders and the Administrative Agent under Sections 4.1 and 14.5) allowed in such judicial proceeding; and

(b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender to make such payments to the Administrative Agent and, if the Administrative Agent shall consent to the making of such payments directly to the Lenders, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 4.1 and 14.5.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender to authorize the Administrative Agent to vote in respect of the claim of any Lender or in any such proceeding.

13.10. Indemnification. The Lenders agree to indemnify each Agent in its capacity as such (to the extent not reimbursed by the Borrower and without limiting the obligation of the Borrower to do so), ratably according to their respective portions of the Total Credit Exposure in effect on the date on which indemnification is sought (or, if indemnification is sought after the date upon which the Commitments shall have terminated and the Loans shall have been paid in full, ratably in accordance with their respective portions of the Total Credit Exposure in effect immediately prior to the date on which payment or termination occurred), from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever that may at any time (including at any time following the payment of the Loans) be imposed on, incurred by or asserted against any Agent in any way relating to or arising out of, the Commitments, this Agreement, any of the other Credit Documents or any documents contemplated by or referred to herein or the transactions contemplated hereby or thereby or any action taken or omitted by any Agent under or in connection with any of the foregoing (including the enforcement of this Section 13.10), provided that no Lender shall be liable for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, suits, costs, expenses or disbursements resulting from such Agent's gross negligence or willful misconduct as determined by a final and non-

appealable judgment of a court of competent jurisdiction. The agreements in this Section 13.10 shall survive the payment of the Loans and all other amounts payable hereunder.

13.11. <u>Collateral and Guaranty Matters</u>. Each of the Lenders irrevocably authorizes the Collateral Agent, at the request of the Borrower,

(a) to release any Lien on any property granted to or held by the Collateral Agent under any Credit Document (i) upon termination of the aggregate Commitments of all the Lenders and Payment In Full of all Obligations, (ii) that is sold or to be sold as part of or in connection with any sale permitted hereunder or under any other Credit Document to a Person that is not a Credit Party, (iii) that constitutes "Excluded Property" (as such term is defined in the Security Agreement), (iv) that is excluded pursuant to Section 9.15(b), or (v) if approved, authorized or ratified in writing in accordance with Section 14.1;

(b) to release any Guarantor from its obligations under the Guarantee if such Person ceases to be a Subsidiary as a result of a transaction permitted hereunder; and

(c) to subordinate any Lien on any property granted to or held by the Administrative Agent under any Credit Document to the holder of any Lien on such property that is permitted by Section 10.1(A)(f).

Upon request by the Collateral Agent at any time, the Required Lenders will confirm in writing the Collateral Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Guarantor from its obligations under the Guarantee pursuant to this Section 13.11. In each case as specified in this Section 13.11, the Collateral Agent will, at the Borrower's expense, execute and deliver to the applicable Credit Party such documents as such Credit Party may reasonably request to evidence the release of such item of Collateral from the assignment and security interest granted under the Security Documents or to subordinate its interest in such item, or to release such Guarantor from its obligations under the Guarantee, in each case in accordance with the terms of the Credit Documents and this Section 13.11; provided that such release shall not in any manner discharge, affect or impair the Obligations or any Liens upon (or obligations of the Borrower or any of its Subsidiaries, including, without limitation, the proceeds of the sale, all of which shall continue to constitute part of the Collateral. In the event of any foreclosure or similar enforcement action with respect to any of the Collateral, the Collateral Agent shall be authorized to deduct all of the costs and expenses reasonably incurred by the Collateral Agent from the proceeds of any such sale, transfer or foreclosure.

13.12. [<u>Reserved]</u>.

13.13. [Reserved].

13.14. <u>Withholding Taxes</u>. To the extent required by any applicable Requirement of Law, the Administrative Agent may withhold from any payment to any Lender an amount equivalent to any applicable withholding Tax. Without limiting or expanding the provisions of Section 5.4, each Lender shall indemnify and hold harmless the Administrative Agent against, and shall make payable in respect thereof within 15 days after demand therefor, any and all Taxes and any and all related losses, claims, liabilities and expenses (including fees, charges and disbursements of any counsel for the Administrative Agent) incurred by or asserted against the Administrative Agent by the Internal Revenue Service or any other Governmental Authority as a result of the failure of the Administrative Agent to properly withhold Tax from amounts paid to or for the account of such Lender for any reason (including, without limitation, because the appropriate form was not delivered or not properly executed, or because such Lender failed to notify the Administrative Agent of a change in circumstance that rendered the exemption from, or reduction of withholding Tax ineffective). A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent shall be conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Credit Document against any amount due the Administrative Agent under this Section 13.14. The agreements in this Section 13.14 shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender, the termination of the Commitments and the repayment, satisfaction or discharge of all other Obligations.

## 13.15. Certain ERISA Matters.

(a) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Credit Party, that at least one of the following is and will be true:

(i) such Lender is not using "plan assets" (within the meaning of Section 3(42) of ERISA or otherwise) of one or more Benefit Plans with respect to such Lender's entrance into, participation in, administration of and performance of the Loans or the Commitments;

(ii) the transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance company general accounts), PTE 90-1 (a class exemption for certain transactions involving insurance company pooled separate accounts), PTE 91-38 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Commitments and this Agreement;

(iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Loans, the Commitments and this Agreement, (C) the entrance into, participation in, administration of and performance of the Loans, the Commitments of subsections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Commitments and this Agreement; or

(iv) such other representation, warranty and covenant as may be agreed in writing between the Administrative Agent, such Lender and Borrower, provided that Borrower shall not unreasonably withhold its consent.

(b) In addition, unless either (1) sub-clause (i) in the immediately preceding clause (a) is true with respect to a Lender or (2) a Lender has provided another representation, warranty and covenant in accordance with subclause (iv) in the immediately preceding clause (a), such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto, for the benefit of, the Administrative Agent and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Credit Party, that the Administrative Agent is not a fiduciary with respect to the assets of such Lender involved in such Lender's entrance into, participation in, administrative Agent under this Agreement, any Credit Document or any documents related hereto or thereto).

## SECTION 14. Miscellaneous.

14.1. Amendments and Waivers. Neither this Agreement nor any other Credit Document (other than the Orders), nor any terms hereof or thereof may be amended, supplemented or modified except in accordance with the provisions of this Section 14.1. The Required Lenders may, or, with the written consent of the Required Lenders, the Administrative Agent may, from time to time, (a) enter into with the relevant Credit Party or Credit Parties written amendments, supplements or modifications hereto and to the other Credit Documents for the purpose of adding any provisions to this Agreement or the other Credit Documents or changing in any manner the rights of the Lenders or of the Credit Parties hereunder or thereunder or (b) waive, on such terms and conditions as the Required Lenders or the Administrative Agent, as the case may be, may specify in such instrument, any of the requirements of this Agreement or the other Credit Documents or any Default or Event of Default and its consequences; provided, however, that no such waiver and no such amendment, supplement or modification shall directly (i) forgive or reduce any portion of any Loan or extend the final scheduled maturity date of any Loan or reduce the stated rate, or forgive any portion, or extend the date for the payment, of any interest or fee payable hereunder (other than as a result of waiving the applicability of any post-default increase in interest rates), or increase the aggregate amount of the Commitments of any Lender, or amend or modify any provisions of Section 5.3(a) (with respect to the ratable allocation of any payments only) and 14.8(a), in each case without the written consent of each Lender directly and adversely affected thereby, or (ii) amend, modify or waive any provision of this Section 14.1 or reduce the percentages specified in the definitions of the terms "Required Lenders," "Required Backstop Parties," or consent to the assignment or transfer by the Borrower of its rights and obligations under any Credit Document to which it is a party (except as permitted pursuant to Section 10.3), in each case without the written consent of each Lender directly and adversely affected thereby, or (iii) amend, modify or waive any provision affecting the rights or duties of the Administrative Agent or Collateral Agent, as applicable, under this Agreement or any other Credit Document without the written consent of the then current Administrative Agent or Collateral Agent, as applicable, or (iv) amend, modify or waive any provision of this Agreement in order to permit the incurrence of any financing pursuant to Section 364 of the Bankruptcy Code (other than the DIP Facility in the maximum amounts permitted after giving effect to the Final DIP Order) that would be secured by the Collateral (or

any portion thereof) on a pari passu or senior basis with the Obligations or that would benefit from any Superpriority Claim in the Cases that is pari passu or senior to the Superpriority Claims with respect to the Obligations as provided in the Orders, without the written consent of each Lender , or (v) [reserved], or (vi) change any Term Loan Commitment to a revolving credit commitment, in each case without the prior written consent of each Lender directly and adversely affected thereby, or (vii) release all or substantially all of the Guarantors under the Guarantee (except as expressly permitted by the Guarantee) or release all or substantially all of the Collateral under the Security Documents, in each case without the prior written consent of each Lender, or (viii) amend Section 2.9 so as to permit Interest Period intervals greater than six months without regard to availability to Lenders, without the written consent of each Lender directly and adversely affected thereby; provided that, notwithstanding anything to the contrary herein, with respect to the First Maturity Extension and the Second Maturity Extension, the applicable conditions thereto other than payment of the First Extension Fee and the Second Extension Fee may be amended, supplemented or modified or waived with the written consent of the Required Lenders. Any such waiver and any such amendment, supplement or modification shall apply equally to each of the affected Lenders and shall be binding upon the Borrower, such Lenders, the Agents and all future holders of the affected Loans. In the case of any waiver, the Borrower, the Lenders and the Agents shall be restored to their former positions and rights hereunder and under the other Credit Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing, it being understood that no such waiver shall extend to any subsequent or other Default or Event of Default or impair any right consequent thereon.

Notwithstanding anything to the contrary herein, no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder (and any amendment, waiver or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of the applicable Lenders other than Defaulting Lenders), except that (x) the Commitment of any Defaulting Lender may not be increased or extended without the consent of such Lender and (y) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that by its terms affects any Defaulting Lender more adversely than other affected Lenders shall require the consent of such Defaulting Lender.

In addition, notwithstanding the foregoing, (i) the Administrative Agent, the Collateral Agent and the relevant Credit Parties may amend, supplement or modify the Security Documents to make such ministerial changes as may be required to effect the provisions of Section 10.2(a) without the consent of any Lender so long as such amendments do not adversely affect the Lenders and (ii) the Administrative Agent, the Agent and the relevant Credit Parties may amend, supplement or modify this Agreement or any of the Security Documents and any other document delivered in connection therewith at the request of the Borrower without the need to obtain the consent of any other Lender if such amendment, supplement or waiver is delivered in order (i) to comply with local law or advice of local counsel, (ii) to cure ambiguities, omissions, mistakes or defects, (iii) to cause such any such Security Document or other document to be consistent with this Agreement and the other Credit Documents or (iv) add syndication or documentation agents and make customary changes and references related thereto.

14.2. <u>Notices</u>. Except as set forth in Section 14.17, all notices, requests and demands to or upon the respective parties hereto to be effective shall be in writing (including by facsimile or electronic mail), and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when delivered, or three days after being deposited in the mail, postage prepaid, or, in the case of telecopy or electronic mail notice, when received, addressed as follows in the case of the Borrower and the Administrative Agent, or to such other address as may be hereafter notified by the respective parties hereto:

The Borrower:

Intelsat Jackson Holdings S.A. Société anonyme Attention: General Counse 4, rue Albert Borschette L-1246 Luxembourg RCS Luxembourg n° B 149.959 Telecopier: 352.2784.169 Email:

With a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP Attention: Eric J. Wedel, P.C., Ben Steadman, Steven N. Serajeddini, P.C. and Anthony Grossi 601 Lexington Avenue New York, NY 10022 Facsimile: 1-212-446-4900 E-mail:

The Administrative Agent:

Credit Suisse AG, Attention of: Agency Manager Eleven Madison Avenue New York, NY 10010 Fax No. 212-322-2291 Email:

provided that any notice, request or demand to or upon the Administrative Agent or the Lenders pursuant to Sections 2.3, 2.6, 2.9, 4.2 and 5.1 shall not be effective until received.

14.3. <u>No Waiver; Cumulative Remedies</u>. No failure to exercise and no delay in exercising, on the part of the Administrative Agent or any Lender, any right, remedy, power or privilege hereunder or under the other Credit Documents shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

14.4. <u>Survival of Representations and Warranties</u>. All representations and warranties made hereunder, in the other Credit Documents and in any document, certificate or statement delivered pursuant hereto or in connection herewith shall survive the execution and delivery of this Agreement and the making of the Loans hereunder.

14.5. Payment of Expenses and Taxes. Each Credit Party agrees (a) to pay or reimburse the Agents, the Backstop Parties and the Lenders for all their reasonable out-of-pocket costs and expenses incurred in connection with the development, preparation, negotation and execution of, and any amendment, supplement or modification to, this Agreement and the other Credit Documents and any other documents prepared in connection herewith or therewith, and the consummation and administration of the transactions (including the Commitment Reallocation) contemplated hereby and thereby, including the reasonable fees, disbursements and other charges of counsel to the Agents and the Backstop Parties, (b) to pay or reimburse each Backstop Party, Lender and Agent for all its reasonable and documented costs and expenses incurred in connection with the enforcement or preservation of any rights under this Agreement, the other Credit Documents and any such other documents, including the reasonable fees, disbursements and other charges of counsel to the Agent and Seat Agent for all its reasonable to the Agents, (c) to pay, indemnify, and hold harmless each Backstop Party, Lender and Agent from, any and all recording and filing fees and (d) to pay, indemnify, and hold harmless each Lender and Agent from, any and all recording and filing fees and (d) to pay, indemnify, and hold harmless each Lender and Agent, sufficient, expresses, investment advisors and agents from and against any and all other liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses of any kind or nature whatsoever, including reasonable and documented fees, disbursements and other charges of counsel, with respect to the execution, delivery,

enforcement (including this Section 14.5), performance and administration of this Agreement, the other Credit Documents and any such other documents, including, without limitation, any of the foregoing relating to the violation of, noncompliance with or liability under, any Environmental Law or to any actual or alleged presence, release or threatened release of Hazardous Materials involving or attributable to the operations of the Borrower, any of its Subsidiaries or any of the Real Estate (all the foregoing in this clause (d), collectively, the "indemnified liabilities"); provided that such Credit Party shall have no obligation hereunder to the Administrative Agent or any Lender nor any of their respective directors, officers, employees and agents with respect to indemnified liabilities to the extent attributable to (i) the gross negligence or willful misconduct of the party to be indemnified as determined in a final and non-appealable judgment by a court of competent jurisdiction or (ii) disputes among the Administrative Agent, the Lenders and/or their transferees. The agreements in this Section 14.5 shall survive repayment of the Loans and all other amounts payable hereunder.

14.6. Successors and Assigns; Participations and Assignments.

(a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that (i) the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender (and any attempted assignment or transfer by the Borrower without such consent shall be null and void) and (ii) no Lender may assign or otherwise transfer its rights or obligations hereunder except in accordance with this Section 14.6. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants (to the extent provided in paragraph (c) of this Section 14.6) and, to the extent expressly contemplated hereby, the Related Parties of each of the Agents and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) (i) Subject to the conditions set forth in paragraphs (b)(ii) and (b)(iii) below, any Lender may assign to one or more assignees (other than any Disqualified Lender) all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitments and the Loans at the time owing to it) with the prior written consent (such consent not be unreasonably withheld or delayed; it being understood that, without limitation, the Borrower shall have the right to withhold its consent to any assignment if, in order for such assignment to comply with applicable law, the Borrower would be required to obtain the consent of, or make any filing or registration with, any Governmental Authority) of:

(A) the Borrower (which consent shall not be unreasonably withheld or delayed); <u>provided</u> that no consent of the Borrower shall be required for an assignment of Term Loans to a Lender, an Affiliate of a Lender (unless increased costs would result therefrom except if an Event of Default has occurred and is continuing), or an Approved Fund; <u>provided</u>, <u>however</u>, if an Event of Default has occurred and is continuing, an assignment pursuant to Section 14.6(b)(i) to any assignee shall be permitted; <u>provided</u>, <u>further</u>, that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within ten (10) Business Days after having received notice thereof; and

(B) the Administrative Agent (which consent shall not be unreasonably withheld or delayed); <u>provided</u> that no consent of the Administrative Agent shall be required for an assignment of (1) any Commitment to an assignee that is a Lender, an Affiliate of a Lender or Approved Fund of a Lender immediately prior to giving effect to such assignment; or (2) any Term Loan to a Lender, an Affiliate of a Lender or an Approved Fund of a Lender immediately prior to giving effect to such assignment.

Notwithstanding the foregoing, any Backstop Party may assign its Term Loan Commitment and/or Term Loans to one or more Other Prepetition Secured Parties pursuant to the Commitment Re-allocation, in each case, without the consent of the Borrower or the Administrative Agent.

(ii) No Lender may assign any portion of its rights and obligations under this Agreement to the Sponsors, Holdings or any of their

respective Affiliates;

(iii) Assignments shall be subject to the following additional conditions:

(A) except in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund or an assignment of the entire remaining amount of the assigning Lender's Commitment or Loans, the amount of the Commitment or Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to the Administrative Agent) shall not be less than \$5,000,000 (or, in the case of a Term Loan Commitment or Term Loan, \$1,000,000), and increments of \$1,000,000 in excess thereof, unless each of the Borrower and the Administrative Agent otherwise consents (which consents shall not be unreasonably withheld or delayed); provided that no such consent of the Borrower shall be required if an Event of Default has occurred and is continuing; provided, further, that contemporaneous assignments

to a single assignee made by Affiliates of Lenders and related Approved Funds shall be aggregated for purposes of meeting the minimum assignment amount requirements stated above;

(B) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement;

(C) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Acceptance, together with (other than any assignment in connection with the Commitment Re-allocation) a processing and recordation fee of \$3,500 (which fee may be waived or reduced in the sole discretion of the Administrative Agent in the case of any Assignment); provided that only one such fee shall be payable in the event of simultaneous assignments to or from two or more Approved Funds; provided further that any such Assignment and Assumption shall include a representation by the assignee that the assignee is not a Disqualified Lender or an Affiliate of a Disqualified Lender; and

(D) the assignee, if it shall not already be a Lender hereunder, shall deliver to the Administrative Agent an administrative questionnaire in a form approved by the Administrative Agent (the "Administrative Questionnaire").

For the purpose of this Section 14.6(b), the term "<u>Approved Fund</u>" shall mean any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course and that is administered, advised or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers, advises or manages a Lender.

(iv) Subject to acceptance and recording thereof pursuant to paragraph (b)(vi) of this Section 14.6, from and after the effective date specified in each Assignment and Acceptance, the assignee thereunder shall be a party hereto and, to the extent of the interest assigned by such Assignment and Acceptance but subject to paragraph (b)(viii) of this Section 14.6, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Acceptance, be released from its obligations under this Agreement (and, in the case of an Assignment and Acceptance covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 2.10, 2.11, 3.5, 5.4 and 14.5). Any assignment as a sale by such Lender of rights or obligations under this Agreement that does not comply with this Section 14.6 shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with paragraph (c) of this Section 14.6.

(v) The Administrative Agent, acting for this purpose as an agent of the Borrower shall maintain at the Administrative Agent's Office a copy of each Assignment and Acceptance delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amount of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). Further, the Register shall contain the name and address of the Administrative Agent and the lending office through which each such Person acts under this Agreement. The entries in the Register shall be conclusive absent manifest error, and the Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

(vi) Upon its receipt of a duly completed Assignment and Acceptance executed by an assigning Lender and an assignee, the assignee's completed Administrative Questionnaire (unless the assignee shall already be a Lender hereunder), the processing and recordation fee referred to in paragraph (b) of this Section 14.6 and any written consent to such assignment required by paragraph (b) of this Section 14.6, the Administrative Agent shall accept such Assignment and Acceptance and record the information contained therein in the Register. No assignment shall be effective for purposes of this Agreement unless it has been recorded in the Register as provided in this paragraph.

(vii) In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable *pro rata* share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full *pro rata* share of all Loans. Notwithstanding the foregoing, in the event that any assignment of

rights and obligations of any Defaulting Lender hereunder shall become effective under applicable law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

(viii) An assignee Lender shall not be entitled to receive any greater payment under Section 2.10 or 5.4 in respect of Luxembourg Taxes than the applicable assignor Lender would have been entitled to receive with respect to the Loan sold to the assignee Lender, unless either (1) the assignment of such Loan to such assignee Lender is made with the Borrower's prior written consent (which consent shall not be unreasonably withheld) or (2) such greater payment under Section 2.10 or 5.4 arises solely as the result of a Change in Law following the date on which the assignee becomes a Lender hereunder.

(c) (1) Any Lender may, without the consent of the Borrower or the Administrative Agent, sell participations to one or more banks or other entities (other than a Disqualified Lender) (each, a "<u>Participant</u>") in all or a portion of such Lender's rights and obligations under this Agreement (including all or a portion of its Commitments and the Loans owing to it); <u>provided</u> that (A) such Lender's obligations under this Agreement shall remain unchanged, (B) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (C) the Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver of any provision of the Participant, agree to any amendment, modification or waiver described in the first proviso to Section 14.1 that affects such Participant. Subject to paragraph (c)(ii) of this Section 14.6, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 2.10, 2.11 and 5.4 to the same extent as if it were a Lender (subject to the requirements of those Sections) and had acquired its interest by assignment pursuant to paragraph (b) of this Section 14.6. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 14.8(b) as though it were a Lender; <u>provided</u> such Participant agrees to be subject to Section 14.8(a) as though it were a Lender.

(i) A Participant shall not be entitled to receive any greater payment under Section 2.10 or 5.4 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent (which consent shall not be unreasonably withheld).

(d) Any Lender may, without the consent of the Borrower or the Administrative Agent, at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank, and this Section 14.6 shall not apply to any such pledge or assignment of a security interest; provided that no such pledge or assignment of a security interest shall release a Lender from any of its obligations hereunder or substitute any such pledge or assignee for such Lender as a party hereto. In order to facilitate such pledge or assignment, the Borrower hereby agrees that, upon request of any Lender at any time and from time to time after the Borrower has made its initial borrowing hereunder, the Borrower shall provide to such Lender, at the Borrower's own expense, a promissory note, substantially in the form of Exhibit G, as the case may be, evidencing the Term Loans owing to such Lender.

(e) Subject to Section 14.16, the Borrower authorizes each Lender to disclose to any Participant, secured creditor of such Lender or assignee (each, a "<u>Transferee</u>") and any prospective Transferee any and all financial information in such Lender's possession concerning the Borrower and its Affiliates that has been delivered to such Lender by or on behalf of the Borrower and its Affiliates pursuant to this Agreement or which has been delivered to such Lender by or on behalf of the Borrower and its Affiliates in connection with such Lender's credit evaluation of the Borrower and its Affiliates prior to becoming a party to this Agreement.

(f) Each of the Credit Parties which are incorporated under the laws of the Grand Duchy of Luxembourg expressly accepts and confirms for the purposes of articles 1278 to 1281 of the Luxembourg civil code that, notwithstanding any assignment, transfer and/or novation made pursuant to this Agreement, the guarantee given by it guarantees all Obligations (including without limitation, all obligations with respect to all rights and/or obligations so assigned, transferred or novated) and that any security interest created under any Security Document to which it is a party shall be preserved for the benefit of any successor and assign of the Lenders, Administrative Agent and/or Secured Parties.

(g) Notwithstanding the foregoing, no assignment may be made or participation sold to a Disqualified Lender without the prior written consent of the Borrower. Notwithstanding anything contained in this Agreement or any other Credit Document to the contrary, if any Lender was a Disqualified Lender at the time of the

assignment of any Loans or Commitments to such Lender, following written notice from the Borrower to such Lender and the Administrative Agent: (1) such Lender shall promptly assign all Loans and Commitments held by such Lender to an eligible assignee; provided that (A) the Administrative Agent shall not have any obligation to the Borrower, such Lender or any other Person to find such a replacement Lender, (B) the Borrower shall not have any obligation to such Disqualified Lender or any other Person to find such a replacement Lender or consent to any such assignment to itself or any other Person subject to the Borrower's consent and (C) the assignment of such Loans and/or Commitments, as the case may be, shall be at par plus accrued and unpaid interest and fees; (2) such Lender shall not have any voting or approval rights under the Credit Documents and shall be excluded in determining whether all Lenders, all affected Lenders or the Required Lenders have taken or may take any action hereunder (including any consent to any amendment or waiver pursuant to this Section 14.6); provided that (x) the Commitment of any Disqualified Lender may not be increased or extended without the consent of such Lender adversely and in a manner that is disproportionate to other affected Lenders shall require the consent of such Disqualified Lender; and (3) no Disqualified Lender is entitled to receive information provided solely to Lenders by the Administrative Agent or any Lender or will be permitted to attend or participate in meetings attended solely by the Lenders and the Administrative Agent, other than the right to receive notices or Borrowings, notices or prepayments and other administrative notices in respect of its Loans or Commitments required to be delivered to Lenders pursuant to Section 2 hereof.

14.7. Replacements of Lenders Under Certain Circumstances.

(a) The Borrower shall be permitted to replace any Lender that (a) requests reimbursement for amounts owing pursuant to Section 2.10, 2.11, 3.5 or 5.4, (b) is affected in the manner described in Section 2.10(a)(ii) and as a result thereof any of the actions described in such Section is required to be taken or (c) becomes a Defaulting Lender, with a replacement bank or other financial institution; provided that (i) such replacement does not conflict with any Requirement of Law, (ii) no Event of Default shall have occurred and be continuing at the time of such replacement, (iii) the Borrower shall repay (or the replacement bank or institution shall purchase, at par) all Loans and other amounts (other than any disputed amounts), pursuant to Section 2.10, 2.11, 2.12, 3.5 or 5.4, as the case may be) owing to such replaced Lender prior to the date of replacement, (iv) the replacement bank or institution, if not already a Lender, and the terms and conditions of such replacement, shall be reasonably satisfactory to the Administrative Agent, (v) the replaced Lender shall be obligated to make such replacement in accordance with the provisions of Section 14.6 (provided that the Borrower shall be obligated to pay the registration and processing fee referred to therein) and (vi) any such replacement shall not be deemed to be a waiver of any rights that the Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender.

(b) If any Lender (such Lender, a "<u>Non-Consenting Lender</u>") has failed to consent to a proposed amendment, waiver, discharge or termination which pursuant to the terms of Section 14.1 requires the consent of all of the Lenders affected and with respect to which the Required Lenders shall have granted their consent, then provided no Event of Default other than an Event of Default relating to the proposed amendment, waiver, discharge or termination at issue then exists, the Borrower shall have the right (unless such Non-Consenting Lender grants such consent) to replace such Non-Consenting Lender by deeming such Non-Consenting Lender to have assigned its Loans, and its Commitments hereunder to one or more assignees reasonably acceptable to the Administrative Agent; provided that: (a) all Obligations of the Borrower owing to such Non-Consenting Lender being replaced shall be paid in full to such Non-Consenting Lender concurrently with such assignment, and (b) the replacement Lender shall purchase the foregoing by paying to such Non-Consenting Lender a price equal to the principal amount thereof plus accrued and unpaid interest thereon. No action by or consent of the Non-Consenting Lender shall be necessary in connection with such assignment, which shall be immediately and automatically effective upon payment of such purchase price. In connection with any such assignment, the Borrower, Administrative Agent, such Non-Consenting Lender and the replacement Lender shall otherwise comply with Section 14.6.

#### 14.8. Adjustments; Set-off.

(a) If any Lender (a "benefited Lender") shall at any time receive any payment of all or part of its Loans, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off, pursuant to events or proceedings of the nature referred to in Section 12.5, or otherwise), in a greater proportion than any such payment to or collateral received by any other Lender, if any, in respect of such other Lender's Loans, or interest thereon, such benefited Lender shall purchase for cash from the other Lenders a participating interest in such portion of each such other Lender's Loan, or shall provide such other Lenders with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such benefited Lender to share the excess payment or benefits of such collateral or proceeds ratably with each of the Lenders; provided, however, that if all or any

portion of such excess payment or benefits is thereafter recovered from such benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest.

(b) After the occurrence and during the continuance of an Event of Default, in addition to any rights and remedies of the Lenders provided by law, each Lender shall have the right, without prior notice to the Borrower, any such notice being expressly waived by the Borrower to the extent permitted by applicable law, upon any amount becoming due and payable by the Borrower hereunder (whether at the stated maturity, by acceleration or otherwise) to set-off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final) (other than escrow, payroll, employee health and benefits, pension, fiduciary, 401(K), petty cash, trust and tax accounts), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Lender or any branch or agency thereof to or for the credit or the account of the Borrower. Each Lender agrees promptly to notify the Borrower and the Administrative Agent after any such set-off and application made by such Lender; provided that the failure to give such notice shall not affect the validity of such set-off and application. In the event that any Defaulting Lender shall exercise any such right of set-off, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 3.8 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the obligations owing to such Defaulting Lender as to which it exercised such right of set-off. The rights of each Lender and their respective Affiliates under this Section are in addition to other rights and remedies (including other rights of set-off) that such Lender or its Affiliates may have. Each Lender agrees to notify the Borrower and the Administrative Agent promptly after any such set-off and application; provided that the failure to give such notice shall not affect the validity of such setoff and application.

14.9. <u>Counterparts</u>. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by facsimile or other electronic transmission), and all of said counterparts taken together shall be deemed to constitute one and the same instrument. A set of the copies of this Agreement signed by all the parties shall be lodged with the Borrower and the Administrative Agent.

14.10. <u>Severability</u>. Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

14.11. <u>Integration</u>; <u>Order Controls</u>. This Agreement (together with the exhibits, annexes and schedules hereto) and the other Credit Documents represent the entire agreement of the Borrower, the Administrative Agent and the Lenders with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral or written, and there are no promises, undertakings, representations or warranties by the Administrative Agent or any Lender relative to subject matter hereof not expressly set forth or referred to herein or in the other Credit Documents. To the extent that any specific provision hereof is inconsistent with the Final DIP Order, the Final DIP Order shall control.

14.12. <u>GOVERNING LAW</u>. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES HEREUNDER SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK AND, TO THE EXTENT APPLICABLE, THE BANKRUPTCY CODE.

14.13. Submission to Jurisdiction; Consent to Service; Waivers.

(a) Each Credit Party hereby irrevocably and unconditionally:

(i) submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Credit Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the exclusive general jurisdiction of the Bankruptcy Court and, if the Bankruptcy Court does not have (or abstains from) jurisdiction, the courts of the State of New York, the courts of the United States of America for the Southern District of New York and appellate courts from any thereof;

(ii) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;

(iii) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Borrower at its addresses set forth in Section 14.2 or at such other address of which the Administrative Agent shall have been notified pursuant thereto;

(iv) agrees that nothing herein shall affect the right of the Agents or Lenders to effect service of process in any other manner permitted by law or shall limit the right of the Agents or Lenders to sue or enforce a judgment in any other jurisdiction; and

(v) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding against Agents or Lenders and their respective Affiliates, directors, employees, attorneys, agents or sub-agents, on any theory or liability for any special, indirect, exemplary, punitive or consequential damages (as opposed to direct or actual damages) (whether or not the claim therefor is based on contract, tort or duty imposed by any applicable legal requirement) arising out of, in connection with, as a result of, or in any way related to, this Agreement or any Credit Document or any agreement or instrument contemplated hereby or thereby or referred to herein or therein, the transactions contemplated hereby or thereby, any Loan or the use of the proceeds thereof or any act or omission or event occurring in connection therewith, and Borrower hereby waives, releases and agrees not to sue upon any such claim or any such damages, whether or not accrued and whether or not known or suspected to exist in its favor.

(b) By the execution and delivery of this Agreement, each Credit Party hereby irrevocably designates, appoints and empowers Intelsat US LLC (and Intelsat US LLC hereby irrevocably accepts such appointment) as its authorized agent upon which process may be served in any suit or proceeding arising out of or relating to this Agreement or the Credit Documents that may be instituted in any federal or state court in the State of New York.

(c) Each Credit Party, to the extent that it has or hereafter may acquire any immunity (sovereign or otherwise) from any legal action, suit or proceeding, from jurisdiction of any court or from setoff or any legal process (whether service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) with respect to itself or any of its property or assets, hereby waives and agrees not to plead or claim such immunity in respect of its obligations under this Agreement and the other Credit Documents (it being understood that the waivers contained in this paragraph (c) shall have the fullest extent permitted under the Foreign Sovereign Immunities Act of 1976, as amended, and are intended to be irrevocable and not subject to withdrawal for the purposes of such Act).

14.14. Acknowledgments. Each Credit Party hereby acknowledges that:

(a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Credit Documents;

(b) neither the Administrative Agent nor any Lender has any fiduciary relationship with or duty to any Credit Party arising out of or in connection with this Agreement or any of the other Credit Documents, and the relationship between Administrative Agent and Lenders, on one hand, and the Borrower, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and

(c) no joint venture is created hereby or by the other Credit Documents or otherwise exists by virtue of the transactions contemplated hereby among the Lenders or among any Credit Party and the Lenders.

## 14.15. <u>WAIVERS OF JURY TRIAL</u>. THE BORROWER, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVE TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN.

14.16. <u>Confidentiality</u>. Each of the Agents and the Lenders agree to maintain the confidentiality of the Information (as defined below), except that Confidential Information may be disclosed (a) to its Affiliates and to its and their respective Related Parties (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Confidential Information and instructed to keep such Confidential Information confidential) or any of their respective auditors; (b) to the extent required or requested by any regulatory authority or bank examiner purporting to have jurisdiction over such Person or its Related Parties (including any self-regulatory authority, such as the National Association of Insurance Commissioners); (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process; (d) to any other party hereto; (e) in connection with the exercise of any remedies hereunder or under any other Credit Document or any action or proceeding relating to this Agreement or any other Credit Document or the enforcement of rights hereunder or thereunder; (f) subject to an agreement containing provisions substantially the same as those of this Section 14.16, (or more restrictive) to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights and obligations under this Agreement, or (ii) any actual or prospective party (or its Related Parties) to any swap, derivative or other transaction under which payments are to be made by reference to the Borrower and its obligations, this Agreement or payments hereunder; (g) on a confidential basis to (i) any rating agency in connection with the issuance and monitoring of CUSIP numbers with respect to the DIP Facility; (h) with

the consent of the Borrower; or (i) to the extent such Confidential Information (x) becomes publicly available other than as a result of a breach of this Section 14.16, or (y) becomes available to any Agent, any Lender or any of their respective Affiliates on a nonconfidential basis from a source other than the Borrower who did not acquire such information as a result of a breach of this Section 14.16. In addition, the Agents and the Lenders may disclose the existence of this Agreement and information about this Agreement to market data collectors, similar service providers to the lending industry and service providers to the Agents or any Lender in connection with the administration of this Agreement, the other Credit Documents, and the Commitments. For purposes of this Section 14.16, "Confidential Information" means all information received from the Borrower or any of its Subsidiaries relating to the Borrower or any of its Subsidiaries or any Lender on a nonconfidential basis prior to disclosure by the Borrower or any of its Subsidiaries; provided that, in the case of information received from the Borrower or any of its Subsidiaries after the Closing Date, such information is clearly identified at the time of delivery as confidential.

14.17. Direct Website Communications.

(a) <u>Delivery</u>. (i) The Borrower may, at its option, provide to the Administrative Agent any information, documents and other materials that it is obligated to furnish to the Administrative Agent pursuant to the Credit Documents, including, without limitation, all notices, requests, financial statements, financial and other reports, certificates and other information materials, but excluding any such communication that (A) relates to a request for a new, or a conversion of an existing, borrowing or other extension of credit (including any election of an interest rate or interest period relating thereto), (B) relates to the payment of any principal or other amount due under this Agreement prior to the scheduled date therefor, (C) provides notice of any default or event of default under this Agreement or (D) is required to be delivered to satisfy any condition precedent to the effectiveness of this Agreement and/or any borrowing or other extension of credit thereunder (all such non-excluded communications being referred to herein collectively as "<u>Communications</u>"), by transmitting the Communications in an electronic/soft medium in a format reasonably acceptable to the Administrative Agent to an email address to be provided by the Administrative Agent. Nothing in this Section 14.17 shall prejudice the right of the Borrower, the Administrative Agent or any Lender to give any notice or other communication pursuant to any Credit Document in any other manner specified in such Credit Document.

(ii) Each Lender and each Agent agrees that the receipt of the Communications by such Agent at its e-mail address set forth above shall constitute effective delivery of the Communications to such Agent for purposes of the Credit Documents. Each Lender agrees that notice to it (as provided in the next sentence) specifying that the Communications have been posted to the Platform shall constitute effective delivery of the Communications to such Lender for purposes of the Credit Documents. Each Lender agrees (A) to notify the Administrative Agent in writing (including by electronic communication) from time to time of such Lender's e-mail address to which the foregoing notice may be sent by electronic transmission and (B) that the foregoing notice may be sent to such e-mail address.

(b) <u>Posting</u>. The Borrower further agrees that the Administrative Agent may make the Communications available to the Lenders by posting the Communications on SyndTrak or a substantially similar electronic transmission system (the "<u>Platform</u>"), so long as the access to such Platform is limited (i) to the Agents and the Lenders and (ii) remains subject the confidentiality requirements set forth in Section 14.16.

(c) <u>The Platform</u>. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE". NEITHER THE ADMINISTRATIVE AGENT NOR ANY OF ITS RELATED PARTIES WARRANTS THE ACCURACY OR COMPLETENESS OF THE COMMUNICATIONS OR THE ADEQUACY OF THE PLATFORM AND EACH EXPRESSLY DISCLAIMS LIABILITY FOR ERRORS OR OMISSIONS IN THE COMMUNICATIONS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NONINFRINGEMENT OF THIRD-PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS IS MADE BY THE ADMINISTRATIVE AGENT OR ANY OF ITS RELATED PARTIES IN CONNECTION WITH THE COMMUNICATIONS OR THE PLATFORM.

14.18. <u>USA PATRIOT Act</u>. Each Lender that is subject to the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "<u>Patriot Act</u>") and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies each Credit Party that pursuant to the requirements of the Patriot Act and the Beneficial Ownership Regulation, it is required to obtain, verify and record information that identifies each Credit Party, which information includes the name and address of each Credit Party and other information that will allow such Lender or the Administrative Agent, as applicable, to identify each Credit Party in accordance with the Patriot Act and the Beneficial Ownership Regulation.

14.19. Conversion of Currencies.

(a) If, for the purpose of obtaining judgment in any court, it is necessary to convert a sum owing hereunder in one currency into another currency, each party hereto agrees, to the fullest extent that it may effectively do so, that the rate of exchange used shall be that at which in accordance with normal banking procedures in the relevant jurisdiction the first currency could be purchased with such other currency on the Business Day immediately preceding the day on which final judgment is given.

(b) The obligations of the Borrower in respect of any sum due to any party hereto or any holder of the obligations owing hereunder (the "<u>Applicable Creditor</u>") shall, notwithstanding any judgment in a currency (the "<u>Judgment Currency</u>") other than the currency in which such sum is stated to be due hereunder (the "<u>Agreement Currency</u>"), be discharged only to the extent that, on the Business Day following receipt by the Applicable Creditor of any sum adjudged to be so due in the Judgment Currency, the Applicable Creditor may in accordance with normal banking procedures in the relevant jurisdiction purchase the Agreement Currency with the Judgment Currency; if the amount of the Agreement Currency so purchased is less than the sum originally due to the Applicable Creditor in the Agreement Currency, the Borrower agrees, as a separate obligation and notwithstanding any such judgment, to indemnify the Applicable Creditor against such loss. The obligations of the Borrower contained in this Section 14.19 shall survive the termination of this Agreement and the payment of all other amounts owing hereunder.

14.20. Acknowledgement and Consent to Bail-In of Affected Financial Institutions. Notwithstanding anything to the contrary in any Credit Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Affected Financial Institution arising under any Credit Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of the applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

(a) the application of any Write-Down and Conversion Powers by the applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an Affected Financial Institution; and

(b) the effects of any Bail-In Action on any such liability, including, if applicable:

(i) a reduction in full or in part or cancellation of any such liability;

(ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such Affected Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Credit Document; or

(iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of the applicable Resolution Authority.

#### SECTION 15. Guarantee.

#### 15.1. Guarantee

(a) In order to induce the Agents and the Lenders to enter into this Agreement and to extend credit hereunder, and in recognition of the substantial direct and indirect benefits to be received by each Guarantor from the proceeds of the Loans, which will be used in part to enable the Borrower to make valuable transfers to the Guarantors in connection with the operation of their respective businesses, each Guarantor, which is a Subsidiary of the Borrower, hereby agrees with the Secured Parties as follows: each of the Guarantors hereby, jointly and severally, unconditionally and irrevocably guarantees as primary obligor and not merely as surety, to the Administrative Agent, for the benefit of the Secured Parties, the full and prompt payment when due, whether upon maturity, acceleration or otherwise, of any and all of the Obligations. If any or all of the Obligations of the Credit Parties to the Secured Parties becomes due and payable hereunder, each Credit Party irrevocably and unconditionally promises to pay such indebtedness to the Secured Parties, or order, on demand, together with any and all expenses which may be incurred by the Secured Parties in collecting any of the Obligations. This Guarantee is a guaranty of payment and not of collection. If claim is ever made upon any Secured Party for repayment or recovery of any amount or amounts received in payment or on account of any of the Obligation over such payee or any of its property or (ii) any settlement or compromise of any such claimat (including any Credit Party), then and in such event each Credit Party agrees that any such judgment, decree, order, settlement or compromise shall be binding upon it, notwithstanding any revocation of this Guarantee or other instrument evidencing any liability of the Credit Parties, and each Credit Party shall be and remain liable to the aforesaid payees hereunder for the amount so repaid or recovered to the same extent as if such amount had never originally been received by any such payee. For

the avoidance of doubt, each Credit Party expressly accepts and confirms for the purposes of articles 1278 to 1281 of the Luxembourg civil code that, notwithstanding any assignment, transfer and/or novation made pursuant to this Agreement, the guarantee given by it guarantees all Obligations (including without limitation, all obligations with respect to all rights and/or obligations so assigned, transferred or novated) and that any security interest created under any Security Document to which it is a party shall be preserved for the benefit of any new Secured Party.

(b) Each Credit Party further agrees to pay any and all reasonable, document and invoiced out-of-pocket expenses (including all reasonable fees and disbursements of counsel) that may be paid or incurred by the Administrative Agent or any other Secured Party in enforcing, or obtaining advice of counsel in respect of, any rights with respect to, or collecting, any or all of the Obligations and/or enforcing any rights with respect to, or collecting against, such Guarantor under this Guarantee.

(c) Each Guarantor agrees that the Obligations may, subject to Section 15.12 (Luxembourg *Guarantee Limitations*) and Section 15.13 (*UK Guarantee Limitations*), at any time and from time to time exceed the amount of the liability of such Guarantor hereunder without impairing this Guarantee or affecting the rights and remedies of the Administrative Agent or any other Secured Party hereunder.

(d) No payment or payments made by the Borrower, any of the Guarantors, any other guarantor or any other Person or received or collected by the Administrative Agent or any other Secured Party from the Borrower, any of the Guarantors, any other guarantor or any other Person by virtue of any action or proceeding or any set-off or appropriation or application at any time or from time to time in reduction of or in payment of the Obligations shall be deemed to modify, reduce, release or otherwise affect the liability of any Guarantor hereunder, which shall, notwithstanding any such payment or payments other than payments made by such Guarantor in respect of the Obligations or payments received or collected from such Guarantor in respect of the Obligations, remain liable for the Obligations up to the maximum liability of such Guarantor hereunder until the Obligations are paid in full, the Commitments are terminated.

(e) Each Guarantor agrees that whenever, at any time, or from time to time, it shall make any payment to the Administrative Agent or any other Secured Party on account of its liability hereunder, it will notify the Administrative Agent in writing that such payment is made under this Guarantee for such purpose.

15.2. <u>Right of Contribution</u>. Each Guarantor hereby agrees that to the extent that a Guarantor shall have paid more than its proportionate share of any payment made hereunder, such Guarantor shall be entitled to seek and receive contribution from and against any other Guarantor hereunder who has not paid its proportionate share of such payment. Each Guarantor's right of contribution shall be subject to the terms and conditions of Section 15.4 hereof. The provisions of this Section 15.2 shall in no respect limit the obligations and liabilities of any Guarantor to the Administrative Agent and the other Secured Parties, and each Guarantor shall remain liable to the Administrative Agent and the other Secured Parties for the full amount guaranteed by such Guarantor hereunder.

15.3. <u>Right of Set-off</u>. In addition to any rights and remedies of the Secured Parties provided by law, each Guarantor hereby irrevocably authorizes each Secured Party at any time and from time to time following the occurrence and during the continuance of an Event of Default without notice to such Guarantor or any other Guarantor, any such notice being expressly waived by each Guarantor, upon any amount becoming due and payable by such Guarantor hereunder (whether at stated maturity, by acceleration or otherwise) to set off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Secured Party to or for the credit or the account of such Guarantor. Each Secured Party shall notify such Guarantor promptly of any such set-off and the appropriation and application made by such Secured Party, provided that the failure to give such notice shall not affect the validity of such set-off and application.

15.4. No Subrogation. Notwithstanding any payment or payments made by any of the Guarantors hereunder or any set-off or appropriation and application of funds of any of the Guarantors by the Administrative Agent or any other Secured Party, no Guarantor shall be entitled to be subrogated to any of the rights of the Administrative Agent or any other Secured Party against the Borrower or any other Guarantor or any collateral security or guarantee or right of offset held by the Administrative Agent or any other Secured Party for the payment of the Obligations, nor shall any Guarantor seek or be entitled to seek any contribution or reimbursement from the Borrower or any other Guarantor in respect of payments made by such Guarantor hereunder, until all amounts owing to the Administrative Agent and the other Secured Parties by the Credit Parties on account of the Obligations are Paid In Full and the Commitments are terminated. If any amount shall be paid to any Guarantor on account of such subrogation rights at any time when all the Obligations shall not have been Paid In Full, such amount shall be held by such Guarantor in trust for the Administrative Agent and the other Secured Parties, segregated from other funds of such Guarantor, and shall, forthwith upon receipt by such Guarantor, be turned over to the Administrative Agent

in the exact form received by such Guarantor (duly indorsed by such Guarantor to the Administrative Agent, if required), to be applied against the Obligations, whether due or to become due, in such order as the Administrative Agent may determine.

15.5. Amendments, etc. with Respect to the Obligations; Waiver of Rights. Each Guarantor shall remain obligated hereunder notwithstanding that, without any reservation of rights against any Guarantor and without notice to or further assent by any Guarantor, (a) any demand for payment of any of the Obligations made by the Administrative Agent or any other Secured Party may be rescinded by such party and any of the Obligations continued, (b) the Obligations, or the liability of any other party upon or for any part thereof, or any collateral security or guarantee therefor or right of offset with respect thereto, may, from time to time, in while or in part, be renewed, extended, amended, modified, accelerated, compromised, waived, surrendered or released by the Administrative Agent or any other Secured Party, (c) the Credit Agreement and any other documents executed and delivered in connection therewith, may be amended, modified, supplemented or terminated, in whole or in part, as the Administrative Agent (or the Required Lenders, as the case may be) may deem advisable from time to time, and (d) any collateral security, guarantee or right of offset at any time held by the Collateral Agent, the Administrative Agent or any other Secured Party for the payment of the Obligations may be sold, exchanged, waived, surrendered or released. Neither the Administrative Agent nor any other Secured Party shall have any obligation to protect, secure, perfect or insure any Lien at any time held by it as security for the Obligations or for this Guarantee or any property subject thereto. When making any demand hereunder against any Guarantor, the Administrative Agent or any other Secured Party may, but shall be under no obligation to, make a similar demand on any Credit Party or any Guarantor or guarantor, and any failure by the Administrative Agent. or any other Secured Party to make any such demand or to collect any payments from the Borrower or any Guarantor or guarantor or any release of the Borrower or any Guarantor or guarantor shall not relieve any Guarantor in respect of which a demand or collection is not made or any Guarantor not so released of its several obligations or liabilities hereunder, and shall not impair or affect the rights and remedies, express or implied, or as a matter of law, of the Administrative Agent or any other Secured Party against any Guarantor. For the purposes hereof, "demand" shall include the commencement and continuance of any legal proceedings

#### 15.6. Guarantee Absolute and Unconditional.

(a) Each Guarantor waives any and all notice of the creation, contraction, incurrence, renewal, extension, amendment, waiver or accrual of any of the Obligations, and notice of or proof of reliance by the Administrative Agent or any other Secured Party upon this Guarantee or acceptance of this Guarantee. The Obligations or any of them shall conclusively be deemed to have been created, contracted or incurred, or renewed, extended, amended, waived or accrued, in reliance upon this Guarantee, and all dealings between any of the Credit Parties, on the one hand, and the Administrative Agent and the other Secured Parties, on the other hand, likewise shall be conclusively presumed to have been had or consummated in reliance upon this Guarantee. Each Guarantor waives diligence, presentment, protest, demand for payment and notice of default or nonpayment to or upon any of the Credit Parties with respect to the Obligations. Each Guarantor understands and agrees that this Guarantee shall be construed as a continuing, absolute and unconditional guarantee of payment without regard to (a) the validity, regularity or enforceability of the Credit Agreement, any other Credit Document, any of the Obligations or any other collateral security therefor or guarantee or right of offset with respect thereto at any time or from time to time held by the Collateral Agent, the Administrative Agent or any other Secured Party, (b) any defense, set-off or counterclaim (other than a defense of payment or performance) that may at any time be available to or be asserted by any of the Credit Parties against the Administrative Agent or any other Secured Party or (c) any other circumstance whatsoever (with or without notice to or knowledge of such Credit Party) that constitutes, or might be constitute, an equitable or legal discharge of any of the Credit Parties for the Obligations, or of such Guarantor under this Guarantee, in bankruptcy or in any other instance. When pursuing its rights and remedies hereunder against any Guarantor, the Administrative Agent and any other Secured Party may, but shall be under no obligation to, pursue such rights and remedies as it may have against any of the Credit Parties or any other Person or against any collateral security or guarantee for the Obligations or any right of offset with respect thereto, and any failure by the Administrative Agent or any other Secured Party to pursue such other rights or remedies or to collect any payments from any of the Credit Parties or any such other Person or to realize upon any such collateral security or guarantee or to exercise any such right of offset, or any release of any of the Credit Parties or any such other Person or any such collateral security, guarantee or right of offset, shall not relieve such Guarantor of any liability hereunder, and shall not impair or affect the rights and remedies, whether express, implied or available as a matter of law, of the Administrative Agent and the other Secured Parties against such Guarantor.

(b) This Guarantee shall remain in full force and effect and be binding in accordance with and to the extent of its terms upon each Guarantor and the successors and assigns thereof and shall inure to the benefit of the

Administrative Agent and the other Secured Parties and their respective successors, indorsees, transferees and assigns until all the Obligations (other than any contingent indemnity obligations not then due) shall have been satisfied by payment in full, the Commitments thereunder shall be terminated and no Letters of Credit thereunder shall be outstanding, notwithstanding that from time to time during the term of the Credit Agreement the Credit Parties may be free from any Obligations.

15.7. This Guarantee shall continue to be effective, or be reinstated, as the case may be, if at any time payment, or any part thereof, of any of the Obligations is rescinded or must otherwise be restored or returned by the Administrative Agent or any other Secured Party upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of the Borrower or any Guarantor, or upon or as a result of the appointment of a receiver, intervenor or conservator of, or trustee or similar officer for, the Borrower or any Guarantor or any substantial part of its property, or otherwise, all as though such payments had not been made.

15.8. Subordination. Any of the indebtedness of the Credit Parties now or hereafter owing to any other Credit Party is hereby subordinated to the Obligations of the Credit Parties owing to the Secured Parties; and if the Administrative Agent so requests at a time when an Event of Default exists, all such indebtedness of such Credit Parties to another Credit Party shall be collected, enforced and received by such other Credit Party for the benefit of the Secured Parties and be paid over to the Administrative Agent on behalf of the Secured Parties on account of the Obligations of such Credit Parties to the Secured Parties, but without affecting or impairing in any manner the liability of any Credit Party under the other provisions of this Guarantee. Prior to the transfer by any Credit Party of any note or negotiable instrument evidencing any of the indebtedness of such Credit Parties to any other Credit Party, such Credit Party shall mark such note or negotiable instrument with a legend that the same is subject to this subordination. Without limiting the generality of the foregoing, each Credit Party hereby agrees with the Secured Parties that it will not exercise any right of subrogation which it may at any time otherwise have as a result of this Guarantee (whether contractual, under Section 509 of the Bankruptcy Code or otherwise) until all Obligations have been irrevocably paid in full in cash.

15.9. <u>Payments</u>. Each Guarantor hereby guarantees that payments hereunder will be paid to the Administrative Agent without set-off or counterclaim in Dollars at the Administrative Agent's Office.

15.10. Each Subsidiary of the Borrower that is required to become a party to this Guarantee pursuant to Section 9.11 shall become a Guarantor, with the same force and effect as if originally named as a Guarantor herein, for all purposes of this Guarantee upon execution and delivery by such Subsidiary of a Joinder Agreement. The execution and delivery of any instrument adding an additional Guarantor as a party to this Guarantee shall not require the consent of any other Guarantor hereunder. The rights and obligations of each Guarantor hereunder shall remain in full force and effect notwithstanding the addition of any new Guarantor as a party to this Guarantee.

### 15.11. [Reserved].

15.12. Luxembourg Guarantee Limitation. Notwithstanding any provision to the contrary in this Agreement, the liability of any Guarantor incorporated under the laws of Luxembourg (a "Luxembourg Guarantor") under this Section 15 for the obligations of any Guarantor in which the relevant Guarantor has no direct or indirect equity interest, shall be limited at any time to a maximum amount not exceeding ninety-five per cent. (95%) of the sum of such Guarantor's "*capitaux propres*" (as referred to in Annex I to the Grand-Ducal Regulation dated 18 December 2015 setting out the form and content of the presentation of the balance sheet and profit and loss account, enforcing the Law of 19 December 2002 on the register of commerce and companies and the accounting and annual accounts of undertakings, as amended) (the "<u>Own Funds</u>") and such Guarantor's debt which is subordinated in right of payment (whether generally or specifically) to any claim of any Credit Party under any of the Credit Documents (the "Lux Subordinated Debt"), as determined on the basis of the then latest available annual accounts of such Guarantor duly established in accordance with applicable accounting rules, as at the date on which the guarantee under this Section 15 is called.

Where, for the purpose of the above determinations, (i) no duly established annual accounts are available for the relevant reference period (which will include a situation where, in respect of the determinations to be made above, no final annual accounts have been established in due time in respect of the then most recently ended financial year) or (ii) the relevant annual accounts do not adequately reflect the status of the Lux Subordinated Debt or Own Funds as envisaged above, an independent auditor (*réviseur d'enterprises agréé*) of the Luxembourg Guarantor or, if no such auditor has been appointed, an independent reputable investment bank (acting in good faith) shall make the determination of the relevant Own Funds and Lux Subordinated Debt amounts based on such available elements and facts as deemed relevant by it at such time.

The above limitation shall not apply to:

(i) any amounts borrowed under any Credit Document and in each case made available, in any form whatsoever, to such Guarantor or any entity in which it has a direct or indirect equity interest; and

# (ii) for the avoidance of doubt, any Security Documents.

The obligations of each Luxembourg Guarantor under this Section 15 shall not extend to the guaranteeing or securing of any amount which would breach the prohibition on financial assistance as set out in the Luxembourg Law dated 10 August 1915 on commercial companies, as amended.

Notwithstanding anything herein to the contrary, the obligations and liabilities of any Luxembourg Guarantor under this Agreement shall not include any obligation or liability to the extent that, if so included, would constitute an abuse of assets as defined by article 1500-11 of the Luxembourg law dated 10 August 1915 on commercial companies, as amended.

# 15.13. UK Guarantee Limitations.

(A) Without limiting any specific exemptions set out in this Agreement, no UK Guarantor's obligations and liabilities under this Agreement and under any other guarantee or indemnity provision in a Credit Document (the "<u>UK Guarantee Obligations</u>") will extend to include any obligation or liability and no Collateral granted by a UK Guarantor will secure any UK Guarantee Obligation, if to the extent doing so would be unlawful financial assistance (notwithstanding any applicable exemptions and/or undertaking of any applicable prescribed whitewash or similar financial assistance procedures) in respect of the acquisition of shares in (a) itself (b) any other entity, company or corporation in respect of which it is a Subsidiary or (c) a member of the Group under the laws of its jurisdiction of incorporation.

(B) If, notwithstanding paragraph (A) of this Section 15.13, the giving of the guarantee in respect of the UK Guarantee Obligations or Collateral would be unlawful financial assistance, then, to the extent necessary to give effect to paragraph (A) above (and only to the extent legally effective in the relevant jurisdiction), the obligations of the UK Guarantor under the Credit Documents will be deemed to have been split into two tranches; *Tranche 1* comprising those obligations which can be secured by the UK Guarantee Obligations or Collateral without breaching or contravening relevant financial assistance laws and *Tranche 2* comprising the remainder of the obligations under the Credit Documents. The Tranche 2 obligations will be excluded from the relevant UK Guarantee Obligations.

IN WITNESS WHEREOF, each of the parties hereto has caused a counterpart of this Agreement to be duly executed and delivered as of the date first above written.

INTELSAT JACKSON HOLDINGS S.A. By: Name: Title: [SUBSIDIARY GUARANTOR] By: Name: Title:

[Signature Page to DIP Credit Agreement]

# CREDIT SUISSE AG, CAYMAN ISLANDS BRANCH, as Administrative Agent and Collateral

Agent

By: Name: Title:

By: Name: Title:

[Signature Page to DIP Credit Agreement]

[LENDERS] By: Name: Title:

[Signature Page to DIP Credit Agreement]

# <u>Exhibit B</u>

# Form of Order

[attached]

Edward O. Sassower, P.C. (admitted *pro hac vice* Steven N. Serajeddini, P.C. (admitted *pro hac vice* Anthony R. Grossi (admitted *pro hac vice* **KIRKLAND & ELLIS LLP KIRKLAND & ELLIS INTERNATIONAL LLP** 601 Lexington Avenue New York, New York 10022 Telephone: (212) 446-480 Facsimile: (212) 446-490 Michael A. Condyles (VA 27807) Peter J. Barrett (VA 46179) Jeremy S. Williams (VA 77469) Brian H. Richardson (VA 92477) **KUTAK ROCK LLP** 901 East Byrd Street, Suite 1000 Richmond, Virginia 23219-4071 Telephone: (804) 644-170 Facsimile: (804) 783-619

Co-Counsel to the Debtors and Debtors in Possession

# IN THE UNITED STATES BANKRUPTCY COURT FOR THE EASTERN DISTRICT OF VIRGINIA RICHMOND DIVISION

		)	
In re:		)	Chapter 11
		)	
INTELSAT S.A., et al.,[1]		)	Case No. 20-32299 (KLP)
		)	
	Debtors.	)	(Jointly Administered)
		)	

# ORDER (I) AUTHORIZING THE DEBTORS TO (A) CONSUMMATE A PROPOSED TRANSACTION AND (B) ENTER INTO AN AMENDMENT TO THE DIP CREDIT AGREEMENT AND (II) GRANTING RELATED RELIEF

Upon the motion (the "<u>Motion</u>")^[2] of the above-captioned debtors and debtors in possession (collectively, the "<u>Debtors</u>") for entry of an order (this "<u>Order</u>"), (i) authorizing the Debtors to (a) execute definitive agreements, obtain any necessary regulatory or other authorizations or consents, and consummate the Transaction without further order of the Court and (b) enter into the Investment Amendment and (ii) granting related relief; as set forth in the Motion; and upon the Supporting Declarations; and this Court having jurisdiction over this

^[1] Due to the large number of Debtors in these chapter 11 cases, for which joint administration has been granted, a complete list of the Debtor entities and the last four digits of their federal tax identification numbers is not provided herein. A complete list may be obtained on the website of the Debtors' claims and noticing agent at <u>https://cases.stretto.com/intelsat</u>. The location of the Debtors' service address is: 7900 Tysons One Place, McLean, VA 22102.

^[2] Capitalized terms used but not otherwise defined herein have the meanings ascribed to them in the Motion

² 

matter pursuant to 28 U.S.C. §§ 157 and 1334 and the *Standing Order of Reference from the United States District Court for the Eastern District of Virginia*, dated August 15, 1984; and this Court having found that it may enter a final order consistent with Article III of the United States Constitution; and this Court having found that venue of this proceeding and the Motion in this district is proper pursuant to 28 U.S.C. §§ 1408 and 1409; and this Court having found that the relief requested in the Motion is in the best interests of the Debtors' estates, their creditors, and other parties in interest; and this Court having found that no other notice need be provided; and this Court having reviewed the Motion, the Supporting Declarations, and having heard the statements in support of the relief requested therein at a hearing before this Court (the "Hearing"); and this Court having determined that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and upon all of the proceedings had before this Court; and after due deliberation and sufficient cause appearing therefor, it is hereby FOUND, CONCLUDED, AND DETERMINED THAT:

A. <u>Findings of Fact and Conclusions of Law</u>. To the extent that any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent that any of the following conclusions of law constitute findings of fact, they are adopted as such. The statutory predicates for the relief requested in the Motion are sections 105(a), 363, and 364 of the Bankruptcy Code and Bankruptcy Rule 6004.

B. <u>Jurisdiction</u>. This Court has jurisdiction over the Motion and over the property of the Debtors, including the cash to be used to fund the Transaction, pursuant to 28 U.S.C. §§ 157 and 1334. This is a core proceeding pursuant to 28 U.S.C. § 157(b)(2). Venue of these chapter 11 cases and the Motion in this district and Court is proper under 28 U.S.C. §§ 1408 and 1409.

C. <u>Final Order</u>. This Order constitutes a final order within the meaning of 28 U.S.C. § 158(a). Notwithstanding Bankruptcy Rule 6004(h), this Court finds that there is no just reason for delay in the implementation of this Order, and directs entry of judgment as set forth herein.

D. <u>Sale and Marketing Process</u>. The Sale Process has been competitive and was conducted in a fair and good faith manner. The Target retained a qualified investment bank and legal counsel to run the Sale Process and solicited proposals from multiple parties reasonably positioned to make investments in the businesses covered by the Transaction. Through multiple rounds, the Sale Process and related marketing process were designed to obtain the highest and best value for the Target.

E. <u>Sound Business Judgment</u>. The Debtors' entry into definitive documentation of the Transaction and the Investment Amendment and their consummation of the Transaction (a) constitute sound exercise of the Debtors' business judgment, made in good faith and for legitimate commercial reasons, and (b) are warranted and permissible under sections 105(a) and 363(b) of the Bankruptcy Code.

F. <u>Solvency and Reasonably Equivalent Value</u>. Based on the Seller's representations in the Share Purchase Agreement, the Seller is solvent after giving effect to the transactions contemplated under the Share Purchase Agreement and the transactions authorized by agreements ancillary to the Transaction. The consideration to be provided by the Debtors and received by the Seller in the Transaction is not less than the fair market value of the Target, and fair and reasonable to the Seller, and constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code and under the laws of the United States, any state, territory, possession, or the District of Columbia, and the terms and conditions of the Transaction, taken as a whole, are fair to the Seller. The Share Purchase Agreement has been proposed and negotiated in good faith and at arm's-length and the relief granted herein provides for negotiation of and entry into definitive documentation of the Transaction in good faith and at arm's-length.

G. <u>Best Interests of the Estates, Creditors, and Parties in Interest</u>. Given all of the circumstances of these chapter 11 cases and the Sale Process, the Transaction is in the best interests of the Debtors, their estates, their creditors, and other parties in interest, and should be approved.

# NOW, THEREFOR, IT IS HEREBY ORDERED THAT:

1. The Motion is granted as set forth herein.

2. The Debtors, and any new entity or entities that may be created to effectuate the Transaction, are authorized, but not required, to enter into the Share Purchase Agreement (substantially in the form attached hereto) and negotiate and execute definitive documentation of the Transaction with the Purchase Price and on other terms that are substantially consistent with the Share Purchase Agreement and are otherwise reasonable or appropriate in the Debtors' business judgment, without further order of the Court; provided, that, for the avoidance of doubt, notwithstanding any provision in this Order (including paragraph 4) to the contrary, (a) nothing in this Order shall authorize any party to take any action inconsistent with the DIP Order or the DIP Documents (as defined in the DIP Order); (b) any purported amendment or modification of any DIP Document in connection with the Transaction shall be ineffective unless such amendment or modification has been approved by the parties required to approve such amendment or modification as set forth in the applicable DIP Documents; (c) the Debtors shall consult with the advisors to the Jackson Ad Hoc Group and the Jackson Crossover Ad Hoc Group regarding the negotiation of, and the form and substance of, the Share Purchase Agreement and any other definitive documentation of the Transaction; and (d) any modification to the material terms of the Share Purchase Agreement attached hereto shall require (i) consent of the Jackson Ad Hoc Group and the Jackson Crossover Ad Hoc Group or (ii) approval of the Court after notice and a hearing. The Debtors shall provide copies of any material proposed waivers, consents, amendments, supplements, or modifications to any Designated Target Transaction Agreement (as defined in the Investment Amendment), or drafts of any material Designated Target Transaction Agreement not yet in definitive form, to the Jackson Ad Hoc Group and Jackson Crossover Ad Hoc Group as soon as reasonably practicable in advance of the execution and delivery of such document, and with reasonable time for review by counsel to the Jackson Ad Hoc Group and Jackson Crossover Ad Hoc Group.

3. The Debtors are authorized, but not required, to obtain any regulatory or other authorizations or consents, and take all actions reasonably necessary or appropriate to consummate the Transaction.

4. The Debtors are authorized, but not directed, to take other actions that are reasonably necessary or desirable in the Debtors' business judgment to effectuate the Transaction and otherwise permit the Debtors to operate their business following the Transaction, including, but not limited to: creating any new entity or entities, opening and funding bank accounts for such entities, entering into intercompany financing arrangements, and entering into ancillary agreements related to the Transaction and making payments to the Seller under such ancillary agreements; <u>provided</u>, <u>however</u>, for the avoidance of doubt, any new entity created in connection with the Transaction shall be a subsidiary of Intelsat Jackson Holdings S.A.

5. The Debtors, and any new entity or entities that may be created to effectuate the Transaction, are authorized to enter into the Investment Amendment and further conforming amendments as may be reasonably necessary or desirable in the Debtors' business judgment to effectuate the Transaction, pursuant to the requirements of the DIP Credit Agreement and the DIP Order.

6. The DIP Liens and the Adequate Protection Liens (each as defined in the DIP Order) shall attach to (i) the Designated Target Intercompany Loan and (ii) the Equity Interests in any Designated Target Subsidiary, in each case directly held by the Credit Parties, (each as defined in the Investment Amendment) in the same manner, and with the same priority, that the DIP Liens and Adequate Protection Liens attached to the DIP Collateral in the DIP Order, without the necessity of the execution by the Debtors (or recordation or other filing) of security agreements, control agreements, pledge agreements, financing statements, mortgages, or other similar documents.

7. The contents of the Motion satisfy the requirements of Bankruptcy Rule 6003(b).

8. Notice of the Motion as provided therein shall be deemed good and sufficient notice of such Motion and the requirements of Bankruptcy Rule 6004(a) and the Local Bankruptcy Rules are satisfied by such notice.

9. Notwithstanding Bankruptcy Rule 6004(h), the terms and conditions of this Order are immediately effective and enforceable upon its entry.

The Debtors are authorized to take all actions that are reasonably necessary or appropriate to effectuate the relief 10. granted in this Order in accordance with the Motion.

11. This Court retains exclusive jurisdiction with respect to all matters arising from or related to the implementation, interpretation, and enforcement of this Order. Dated: Richmond, Virginia United States Bankruptcy Judge

WE ASK FOR THIS:

<u>/s/ Jeremy Williams</u> Michael A. Condyles (VA 27807) Peter J. Barrett (VA 46179) Jeremy S. Williams (VA 77469) Brian H. Richardson (VA 92477) **KUTAK ROCK LLP** 901 East Byrd Street, Suite 1000 Richmond, Virginia 23219-4071 Telephone: (804) 644-170 Facsimile: (804) 783-619

- and -

Edward O. Sassower, P.C. (admitted *pro hac vice*) Steven N. Serajeddini, P.C. (admitted *pro hac vice*)

Anthony R. Grossi (admitted *pro hac vice*) **KIRKLAND & ELLIS LLP KIRKLAND & ELLIS INTERNATIONAL LLP** 601 Lexington Avenue New York, New York 10022 Telephone: (212) 446-480 Facsimile: (212) 446-490

Co-Counsel to the Debtors and Debtors in Possession

# **CERTIFICATION OF ENDORSEMENT UNDER LOCAL BANKRUPTCY RULE 9022-1(C)**

Pursuant to Local Bankruptcy Rule 9022-1(C), I hereby certify that the foregoing proposed order has been endorsed by or served upon all necessary parties.

/s/ Jeremy Williams

# <u>CERTIFICATION OF ENDORSEMENT</u> <u>UNDER LOCAL BANKRUPTCY RULE 9022-1(C)</u>

Pursuant to Local Bankruptcy Rule 9022-1(C), I hereby certify that the foregoing proposed order has been endorsed by or served upon all necessary parties.

/s/ Jeremy Williams

## CERTIFICATION

I, Stephen Spengler, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Intelsat S.A.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 5, 2020 /s/ Stephen Spengler Stephen Spengler Chief Executive Officer (Principal Executive Officer)

## CERTIFICATION

I, David Tolley, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Intelsat S.A.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 5, 2020 /<u>s/ David Tolley</u> David Tolley Chief Financial Officer (Principal Financial Officer)

# CERTIFICATION BY CHIEF EXECUTIVE OFFICER Pursuant To 18 U.S.C. Section 1350, As Adopted Pursuant To Section 906 of the Sarbanes-Oxley Act Of 2002

In connection with the Quarterly Report of Intelsat S.A. (the "Company") on Form 10-Q for the quarter ended September 30, 2020, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, Stephen Spengler, Chief Executive Officer of the Company, does hereby certify, pursuant to § 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. § 1350), that to his knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 5, 2020 /<u>s/ Stephen Spengler</u> Stephen Spengler Chief Executive Officer

The foregoing certification is made solely for the purposes of 18 U.S.C. § 1350 and is not being filed as part of the Report or as a separate disclosure document.

# CERTIFICATION BY CHIEF FINANCIAL OFFICER Pursuant To 18 U.S.C. Section 1350, As Adopted Pursuant To Section 906 of the Sarbanes-Oxley Act Of 2002

In connection with the Quarterly Report of Intelsat S.A. (the "Company") on Form 10-Q for the quarter ended September 30, 2020, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, David Tolley, Chief Financial Officer of the Company, does hereby certify, pursuant to § 906 of the Sarbanes-Oxley Act of 2002 (18 U.S.C. § 1350), that to his knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 5, 2020 /s/ David Tolley David Tolley Chief Financial Officer

The foregoing certification is made solely for the purposes of 18 U.S.C. § 1350 and is not being filed as part of the Report or as a separate disclosure document.